UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) **OF THE SECURITIES EXCHANGE ACT OF 1934**

Date of Report (Date of earliest event reported): January 25, 2023

ORCHESTRA BIOMED HOLDINGS, INC. (Exact name of registrant as specified in its charter)

Delaware	001-39421	92-2038755
(State or other jurisdiction	(Commission File Number)	(IRS Employer
of incorporation)		Identification No.)
	150 Union Square Drive	
	New Hope, Pennsylvania 18938	
(Ad	dress of principal executive offices, including zip coo	le)
Registra	nt's telephone number, including area code: (215) 86	52-5797
	Health Sciences Acquisitions Corporation 2	
	40 10th Avenue, Floor 7	
	New York, New York 10014	
(Form	ner name or former address, if changed since last rep	port)
e appropriate box below if the Form 8-	K filing is intended to simultaneously satisfy the fili	ng obligation of the registrant under a

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common stock, par value \$0.0001 per share	OBIO	The Nasdaq Stock Market LLC

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company \boxtimes

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

INTRODUCTORY NOTE

Unless the context otherwise requires, "New Orchestra" and the "Company" refer to Orchestra BioMed Holdings, Inc., a Delaware corporation (f/k/a Health Sciences Acquisitions Corporation 2), and its consolidated subsidiaries following the Closing (as defined below). Unless the context otherwise requires, references to "HSAC2" refer to Health Sciences Acquisitions Corporation 2, a Cayman Islands exempted company, prior to the Closing, references to "Domesticated HSAC" refer to Orchestra BioMed Holdings, Inc., a Delaware corporation (f/k/a Health Sciences Acquisitions Corporation 2) after the Domestication (as defined below) and prior to the Closing, and references to "Orchestra" refer to Orchestra BioMed, Inc. prior to the Closing. All references herein to the "Board" refer to the board of directors of New Orchestra.

Terms used in this Current Report on Form 8-K (this "Current Report") but not defined herein, or for which definitions are not otherwise incorporated by reference herein, shall have the meaning given to such terms in the Proxy Statement/Prospectus (as defined below) in the section titled "*Frequently Used Terms*" beginning on page 6 thereof, and such definitions are incorporated herein by reference.

Domestication and Business Combination Transaction

As previously announced, HSAC2 previously entered into an agreement and plan of merger, dated as of July 4, 2022 (as amended by Amendment No. 1 to Agreement and Plan of Merger, dated July 21, 2022, and Amendment No. 2 to Agreement and Plan of Merger, dated November 21, 2022, the "Merger Agreement"), by and among HSAC2, HSAC Olympus Merger Sub, Inc., a Delaware corporation and wholly owned subsidiary of HSAC2 ("Merger Sub"), and Orchestra.

On January 25, 2023, as contemplated by the Merger Agreement and described in the section titled "*Proposal 2—The Domestication Proposal*" of the final prospectus and definitive proxy statement, dated December 16, 2022 (the "Proxy Statement/Prospectus") and filed with the Securities and Exchange Commission (the "SEC") on December 16, 2022, HSAC2 filed a notice of deregistration with the Cayman Islands Registrar of Companies, together with the necessary accompanying documents, and filed a certificate of corporate domestication and a certificate of incorporation with the Secretary of State of Delaware, pursuant to which HSAC2 was domesticated and continued as a Delaware corporation, under the name of "Orchestra BioMed Holdings, Inc." (the "Domestication").

As a result of, and upon the effective time of the Domestication, among other things, (i) each issued and outstanding ordinary share, par value \$0.0001 per share, of HSAC2 (the "HSAC2 Ordinary Shares") converted automatically, on a one-for-one basis, into one share of HSAC2 common stock, par value \$0.0001 per share ("HSAC2 Common Stock"); and (ii) each issued and outstanding warrant to purchase HSAC2 Ordinary Shares converted automatically into a warrant to acquire one share of HSAC2 Common Stock.

As previously reported on the Current Report on Form 8-K filed with the SEC on January 24, 2023, HSAC2 held an extraordinary general meeting of its shareholders (the "Meeting"), at which the HSAC2 shareholders approved, among other matters, the Domestication and the Merger Agreement as described in the Proxy Statement/Prospectus.

On January 26, 2023 (the "Closing Date"), as contemplated by the Merger Agreement and described in the section of the Proxy Statement/Prospectus titled "*Proposal 1—The Business Combination Proposal*," New Orchestra consummated the merger transaction contemplated by the Merger Agreement (the "Closing"), whereby Merger Sub merged with and into Orchestra, the separate corporate existence of Merger Sub ceasing and Orchestra being the surviving corporation and a wholly owned subsidiary of New Orchestra (the "Merger" and, together with the Domestication, the "Business Combination"). We refer to HSAC2 Common Stock, after giving effect to the Business Combination, as "New Orchestra Common Stock."

Upon the Closing of the Merger, based on a ratio (the "Exchange Ratio") of 0.465 shares of HSAC2 Common Stock for each whole share of Orchestra common stock, par value \$0.0001 per share (the "Orchestra Common Stock"), 20,191,338 shares of New Orchestra Common Stock were issued to Orchestra stockholders (exclusive of the additional shares subject to earnout discussed below in this paragraph) and 5,523,834 shares of New Orchestra Common Stock were reserved for issuance pursuant to the Orchestra stock options and warrants converted into New Orchestra stock options and warrants in the Merger.

The foregoing description of the Business Combination does not purport to be complete and is qualified in its entirety by the full text of the Merger Agreement, which is attached hereto as Exhibits 2.1, 2.2 and 2.3, respectively, and are incorporated herein by reference.

HSAC2 Shareholder Support Agreement and Forfeiture

As described in the Proxy Statement/Prospectus, simultaneously with the execution of the Merger Agreement, HSAC2 and Orchestra entered into a parent support agreement with HSAC 2 Holdings, LLC (the "Sponsor") and certain other HSAC2 shareholders (as amended and restated on November 21, 2022, the "Parent Support Agreement") pursuant to which the Sponsor and such HSAC2 shareholders agreed, among other things, (a) to appear at any general meetings called to approve the Merger or any proposal to extend the period of time HSAC2 is afforded under its organizational documents and its prospectus to consummate an initial business combination, (b) not to redeem their shares or any other equity securities of HSAC2 now or in future acquired or beneficially owned, (c) to vote such shares and equity securities (i) in favor of the Domestication, the Merger and related transactions (except that any such additional equity securities acquired, including the 1,000,000 Forward Purchase Shares and any Backstop Purchases, would not be voted in favor of approving the Business Combination), (ii) in favor of any proposal to extend the period of time HSAC2 is afforded under its organizational documents and its prospectus to consummate an initial business combination, and (iii) against any change in the business, management or board of HSAC2 contrary to the Merger Agreement and against any other proposal reasonably expected to breach, prevent or impede the Merger, and (d) to waive anti-dilution and similar rights with respect to such shares, whether under the HSAC2 amended and restated memorandum and articles of association, applicable law, or a contract regarding the Merger and related transactions with HSAC2.

In addition, the Sponsor agreed that 25% or 1,000,000 shares of its New Orchestra Common Stock received in the Domestication will be forfeited to New Orchestra on the first business day following the fifth anniversary of the Closing unless, as to 500,000 shares, the volume-weighted average price of the New Orchestra Common Stock is greater than or equal to \$15.00 per share over any 20 trading days within any 30-trading day period (the "Initial Milestone Event"), and as to the remaining 500,000 shares, the volume-weighted average price of the New Orchestra Common Stock is greater than or equal to \$20.00 per share over any 20 trading days within any 30-trading day period (the "Final Milestone Event"). Further, the Sponsor and HSAC2's other initial shareholders prior to HSAC2's initial public offering (the "IPO") have agreed to subject (i) the 4,000,000 shares of New Orchestra Common Stock to be received in the Domestication in exchange for the 4,000,000 HSAC2 Ordinary Shares issued to HSAC2's initial shareholders prior to the IPO (the "Insider Shares") and (ii) the 450,000 shares of New Orchestra Common Stock to be received in the Domestication in exchange for the 4,000,000 warrants in HSAC2 purchased upon consummation of the IPO (the "Private Bares") at lock-up for up to 12 months following the Closing, and the Sponsor forfeited 50% of its 1,500,000 warrants in HSAC2 purchased upon consummation of the IPO (the "Private Warrants, for no consideration, immediately prior to the Closing (the "Sponsor Forfeiture"). Pursuant to the terms of the Merger Agreement, immediately following the Sponsor Forfeiture and prior to the Closing, HSAC2 issued 750,000 warrants to purchase New Orchestra Common Stock to eleven specified employees and directors of Orchestra. These new warrants have substantially similar terms to the forfeited Private Warrants, except that they will become exercisable between 24 and 36 months after the Closing.

The foregoing description of the Parent Support Agreement does not purport to be complete and is qualified in its entirety by the full text of the Parent Support Agreement, which is attached hereto as Exhibit 10.3 and is incorporated herein by reference.

Orchestra Stockholder Support Agreement

As described in the Proxy Statement/Prospectus, simultaneously with the execution of the Merger Agreement, HSAC2 and Orchestra entered into a support agreement with certain Orchestra stockholders, including Medtronic (the "Orchestra Support Agreement"), pursuant to which such stockholders have agreed (a) to appear at any stockholder meetings called to approve the Merger, (b) to vote such shares and equity securities (i) in favor of the Merger and related transactions, (ii) against any change in the business, management or board of Orchestra contrary to the Merger Agreement and (iii) against any other proposal reasonably expected to breach, prevent or impede the Merger.

The foregoing description of the Orchestra Support Agreement does not purport to be complete and is qualified in its entirety by the full text of the Orchestra Support Agreement, which is attached hereto as Exhibit 10.4 and is incorporated herein by reference.

Earnout

In connection with the Business Combination, existing Orchestra stockholders also had the opportunity to elect to participate in an earnout (the "Earnout") pursuant to which each such electing stockholder (an "Earnout Participant") may receive a portion of additional contingent consideration of up to 8,000,000 shares of New Orchestra Common Stock in the aggregate ("Earnout Consideration"). Approximately 91% of Orchestra stockholders elected to participate in the Earnout. Each Earnout Participant agreed to extend their applicable lock-up period from 6 months to 12 months, pursuant to an Earnout Election Agreement and such Earnout Participants will collectively be entitled to receive: (i) 4,000,000 shares of the Earnout Consideration, in the aggregate ("Initial Earnout Shares"), in the event that, from the time beginning immediately after the Closing until the fifth anniversary of the Closing Date (the "Earnout Period"), the Initial Milestone Event (as defined below) occurs; and (ii) an additional 4,000,000 shares of the Earnout Consideration, in the aggregate ("Final Earnout Shares" and, together with the Initial Earnout Shares, the "Earnout Shares"), in the event that, during the Earnout Period, the Final Milestone Event occurs.

The foregoing description of the Earnout does not purport to be complete and is qualified in its entirety by the full text of the Earnout Election Agreement, a form of which is attached hereto as Exhibit 10.5 and is incorporated herein by reference.

Forward Purchase Agreements

As described in the Proxy Statement/Prospectus, simultaneously with the execution of the Merger Agreement, HSAC2 and Orchestra entered into separate forward purchase agreements (each, as amended, a "Forward Purchase Agreement" and, together, the "Forward Purchase Agreements") with certain funds managed by RTW Investments, LP (the "RTW Funds") and Covidien Group S.à.r.l., an affiliate of Medtronic plc ("Medtronic" and the RTW Funds, each a "Purchasing Party"), pursuant to which each of the Purchasing Parties agreed to purchase \$10 million of HSAC2 Ordinary Shares from HSAC2 immediately prior to the Domestication (as defined below), less the dollar amount of HSAC2 Ordinary Shares holding redemption rights that the Purchasing Party acquires and holds until immediately prior to the Domestication (such HSAC2 Ordinary Shares either purchased from HSAC2 or acquired and held until immediately prior to the Domestication, the "Forward Purchase Shares").

The RTW Funds completed their purchases of HSAC2 Ordinary Shares under their Forward Purchase Agreement on or before July 22, 2022. Medtronic completed approximately \$9.9 million of purchases of HSAC2 Ordinary Shares under its Forward Purchase Agreement on or before January 20, 2023. Medtronic subsequently completed \$0.1 million in purchases of HSAC2 Ordinary Shares and/or New Orchestra Common Stock on or before January 30, 2023.

The foregoing description of the Forward Purchase Agreements does not purport to be complete and is qualified in its entirety by the full text of the Forward Purchase Agreements, which are attached hereto as Exhibits 10.23, 10.24 and 10.25, respectively, and are incorporated herein by reference.

Backstop Agreement

As described in the Proxy Statement/Prospectus, simultaneously with the execution of the Merger Agreement and Forward Purchase Agreements, HSAC2, Orchestra and the RTW Funds entered into a Backstop Agreement (the "Backstop Agreement"), pursuant to which the RTW Funds, jointly and severally, agreed to purchase such number of HSAC2 Ordinary Shares at a price of \$10.00 per share to the extent that the amount of cash remaining in HSAC2's working capital and trust account as of immediately prior to the closing of the Merger is less than \$60 million (the "Minimum Available Cash Condition") (which calculation excludes amounts received pursuant to Medtronic's Forward Purchase Agreement or are otherwise held in HSAC2's trust account established pursuant to our IPO (the "Trust Account") in respect of Medtronic's Forward Purchase Shares, but is inclusive of amounts received pursuant to the RTW Funds' Forward Purchase Agreement and otherwise held in the Trust Account in respect of the RTW Funds' Forward Purchase Shares (the "Sponsor Commitment")).

Pursuant to the Backstop Agreement, the RTW Funds purchased 1,808,512 HSAC2 Ordinary Shares on January 25, 2023, immediately prior to the Domestication.

The foregoing description of the Backstop Agreement does not purport to be complete and is qualified in its entirety by the full text of the Backstop Agreement, which is attached hereto as Exhibits 10.26 and 10.27 and is incorporated herein by reference.

Item 1.01. Entry into a Material Definitive Agreement.

Amended and Restated Registration Rights and Lock-Up Agreement

In connection with the Closing, on January 26, 2023, New Orchestra, the RTW Funds and certain existing shareholders of HSAC2 and stockholders of Orchestra entered into an amended and restated registration rights and lock-up agreement (the "Registration Rights Agreement") with respect to the resale of shares of New Orchestra held or acquired by such stockholders two business days prior to the filing of a registration statement with the SEC or a preeffective amendment thereto, and including any shares issuable on conversion of preferred stock, Earnout Consideration and shares acquired under the Forward Purchase Agreements and the Backstop Agreements.

The Registration Rights Agreement amends and restates the registration rights agreement that HSAC2 entered into as of August 3, 2020 in connection with its initial public offering. Subject to the Lock-Up described below, New Orchestra will file a registration statement to register the public resale of the shares as soon as reasonably practicable, but in any event within 120 calendar days following the Closing. In addition, subject to certain requirements and customary conditions, including with regard to the number of requests that may be made and when, such stockholders may request to sell all or any portion of their registrable securities in an underwritten offering so long as the total offering price is reasonably expected to exceed, in the aggregate, \$25 million. In addition, the stockholders party to the Registration Rights Agreement will have certain "piggy-back" registration rights that require New Orchestra to include such securities in registration statements that New Orchestra otherwise files, subject to certain requirements and customary conditions. The Registration Rights Agreement does not contain liquidated damages provisions or other cash settlement provisions resulting from delays in registration statements. The Registration Rights Agreement contains customary indemnification provisions.

Pursuant to the Registration Rights Agreement, the signatories thereto have agreed, subject to certain customary exceptions, not to (i) sell, assign, offer to sell, contract or agree to sell, grant any option to purchase or otherwise dispose of or agree to dispose of, directly or indirectly, any shares subject to lockup, (ii) establish or increase a put equivalent position or liquidation with respect to or decrease a call equivalent position within the meaning of Section 16 of the Exchange Act, with respect to any Lock-up Shares (as defined below), (iii) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any Lock-up Shares, or (iv) publicly announce an intention to effect any of the foregoing during the Lock-up Period (as defined below). The shares subject to lock-up are any shares of New Orchestra Common Stock or any security convertible into or exercisable or exchanged for New Orchestra Common Stock beneficially owned or owned of record by such Holder ("Lock-up Shares"), and the term "Lock-Up Period" means the period from the Closing until the earlier of: (1)(a) 12 months after the Closing with respect to the (i) 4,000,000 shares of New Orchestra Common Stock issued in the Domestication in exchange for 4,000,000 of HSAC2 Ordinary Shares that were issued to HSAC2's initial shareholders prior to its initial public offering, (ii) 450,000 shares of New Orchestra Common Stock issued in the Domestication in exchange for 450,000 of HSAC2 Ordinary Shares that were issued in a private placement simultaneously with HSAC2's initial public offering and (iii) any shares of New Orchestra Common Stock or any security convertible into or exchangeable for New Orchestra Common Stock beneficially owned or owned of record by RTW Investments, LP and its affiliates as of the Closing Date, and (b) six (6) months after the Closing with respect to all other Holders and New Orchestra Common Stock and (2) the date on which New Orchestra completes a liquidation, merger, stock exchange, reorganization or other similar transaction that results in all of the New Orchestra stockholders having the right to exchange their shares of New Orchestra Common Stock for cash, securities or other property.

The foregoing description of the Registration Rights Agreement does not purport to be complete and is qualified in its entirety by the terms and conditions of the Registration Rights Agreement, which is attached hereto as Exhibit 10.1 and is incorporated herein by reference.

Indemnification Agreements

On the Closing Date, New Orchestra entered into indemnification agreements with each of its directors and executive officers.

Each indemnification agreement provides for indemnification and advancements by New Orchestra of certain expenses and costs relating to claims, suits or proceedings arising from each director or executive officer's service to New Orchestra, or, at New Orchestra's request, service to other entities, as officers or directors to the maximum extent permitted by applicable law.

The foregoing description of the indemnification agreements does not purport to be complete and is qualified in its entirety by the terms and conditions of the indemnification agreements, a form of which is attached hereto as Exhibit 10.2 and is incorporated herein by reference.

Item 2.01. Completion of Acquisition or Disposition of Assets.

The disclosure set forth under "Introductory Note-Domestication and Business Combination Transaction" above is incorporated into this Item 2.01 by reference.

FORM 10 INFORMATION

Item 2.01(f) of Form 8-K states that if the predecessor registrant was a "shell company" (as such term is defined in Rule 12b-2 under the Securities Exchange Act of 1934, as amended (the "Exchange Act")), as HSAC2 was immediately before the Business Combination, then the registrant must disclose the information that would be required if the registrant were filing a general form for registration of securities on Form 10. Accordingly, the Company, as the successor registrant to HSAC2 following the consummation of the Business Combination, is providing the information below that would be included in a Form 10 if the Company were to file a Form 10. Please note that the information provided below relates to the Company as the combined company after the consummation of the Business Combination unless otherwise specifically indicated or the context otherwise requires.

Forward-Looking Statements.

Certain statements contained in this Current Report and in the documents incorporated herein by reference may constitute "forward-looking statements" for purposes of federal securities laws. Such statements can be identified by the fact that they do not relate strictly to historical or current facts. In addition, any statements that refer to projections, forecasts or other characterizations of future events or circumstances, including any underlying assumptions, are forward-looking statements. The words "anticipate," "believe," "contemplate," "continue," "could," "estimate," "expect," "intends," "may," "might," "plan," "possible," "potential," "project," "should," "will," "would" and similar expressions (including the negative of any of the foregoing) may identify forward-looking statements, but the absence of these words does not mean that a statement is not forward-looking. Forward-looking statements in this Current Report may include, for example, but are not limited to, statements about:

- New Orchestra's ability to raise financing in the future;
- New Orchestra's ability to realize the anticipated benefits of the Business Combination;
- the attraction and retention of qualified directors, officers, employees and key personnel of New Orchestra;

- New Orchestra's ability and/or the ability of third-party vendors and partners to manufacture its product candidates;
- New Orchestra's ability to source critical components or materials for the manufacture of its product candidates;
- New Orchestra's ability to achieve and sustain profitability;
- New Orchestra's ability to achieve its projected development and commercialization goals;
- the rate of progress, costs and results of New Orchestra's clinical studies and research and development activities;
- market acceptance of New Orchestra's product candidates, if approved;
- New Orchestra's ability to compete successfully with larger companies in a highly competitive industry;
- changes in New Orchestra's operating results which make future operations results difficult to predict;
- existing loan and security agreement covenants that may restrict business and financing activities;
- serious adverse events, undesirable side effects that could halt the clinical development, regulatory approval or certification, of New Orchestra's product candidates;
- New Orchestra's ability to manage growth or control costs related to growth;
- economic conditions that may adversely affect New Orchestra's business, financial condition and stock price;
- New Orchestra's reliance on third parties to drive successful marketing and sale of its initial product candidates;
- New Orchestra's reliance on third parties to manufacture and provide important materials and components for its products and product candidates;
- New Orchestra and its competitor's abilities to obtain necessary regulatory approvals and certifications for its product candidates in an uncomplicated and inexpensive manner;
- New Orchestra's ability to maintain compliance with regulatory and post-marketing requirements;
- adverse medical events, failure or malfunctions in connection with New Orchestra's product candidates and possible subjection to regulatory sanctions;
- healthcare costs containment pressures and legislative or administrative reforms which affect coverage and reimbursement practices of third-party payors;
- New Orchestra's ability to protect or enforce its intellectual property, unpatented trade secrets, know-how and other proprietary technology;
- New Orchestra's ability to obtain necessary intellectual property rights from third parties;
- New Orchestra's ability to protect its trademarks, trade names and build its names recognition;
- the ability to maintain the listing of the New Orchestra Common Stock on The Nasdaq Stock Market LLC ("Nasdaq"); and
- New Orchestra Common Stock's potential liquidity and trading.

These forward-looking statements are based on current expectations and beliefs concerning future developments and their potential effects. There can be no assurance that future developments affecting us will be those that we have anticipated. These forward-looking statements involve a number of risks, uncertainties (some of which are beyond our control) or other assumptions that may cause actual results or performance to be materially different from those expressed or implied by these forward-looking statements. These risks and uncertainties include, but are not limited to, those factors described under the heading "*Risk Factors*" in the Proxy Statement/Prospectus. Should one or more of these risks or uncertainties materialize, or should any of our assumptions prove incorrect, actual results may vary in material respects from those projected in these forward-looking statements. Some of these risks and uncertainties may in the future be amplified by the COVID-19 outbreak, and there may be additional risks that we consider immaterial or which are unknown. It is not possible to predict or identify all such risks. Readers are cautioned not to place undue reliance on forward-looking statements because of the risks and uncertainties related to them and to the risk factors. We do not undertake any obligation to update or revise any forward-looking statements, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws.

Business

The business of HSAC2 prior to the Business Combination is described in the Proxy Statement/Prospectus in the section titled "Business of HSAC2" and that information is incorporated herein by reference. The business of New Orchestra is described in the Proxy Statement/Prospectus in the section titled "Business of Orchestra" and that information is incorporated herein by reference.

Risk Factors

The risk factors related to New Orchestra's business and operations are set forth in the Proxy Statement/Prospectus in the section titled "*Risk Factors*" and that information is incorporated herein by reference.

Financial Information

The information set forth under Item 9.01 of this Current Report is incorporated herein by reference.

Management's Discussion and Analysis of Financial Condition and Results of Operations

Management's discussion and analysis of the financial condition and results of operations prior to the Business Combination is included in (a) the Company's Management's Discussion and Analysis of Financial Condition and Results of Operations beginning on page 310 of the Proxy Statement/Prospectus in the section titled "Orchestra's Management's Discussion and Analysis of Financial Condition and Results of Operations" are incorporated herein by reference and (b) HSAC2's Management's Discussion and Analysis of Financial Condition and Results of Operations beginning on page 213 of the Proxy Statement/Prospectus in the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations of HSAC2" are incorporated herein by reference.

Facilities

The disclosure contained in the Proxy Statement/Prospectus in the section titled "Business of Orchestra-Facilities" is incorporated herein by reference.

Security Ownership of Certain Beneficial Owners and Management

The following table sets forth information known to the Company regarding the beneficial ownership of New Orchestra Common Stock and voting power as of the Closing Date by:

- each person who is a named executive officer or director of New Orchestra;
- all executive officers and directors of New Orchestra as a group; and
- each person who is a beneficial owner of more than 5% of New Orchestra common stock.



Beneficial ownership is determined according to the rules of the SEC, which generally provide that a person has beneficial ownership of a security if he, she or it possesses sole or shared voting or investment power over that security, including options and warrants that are currently exercisable or exercisable within 60 days. Unless otherwise indicated, the Company believes that all persons named in the table below have sole voting and investment power with respect to the voting securities beneficially owned by them.

The beneficial ownership of New Orchestra Common Stock is based on 31,614,079 shares of such common stock outstanding as of January 26, 2023, immediately following the consummation of the Business Combination.

	Number of Shares Beneficially	
Name and Address of Beneficial Owner	Owned	% of Class
Directors and named executive officers of New Orchestra ^{(1)}		
David P. Hochman ⁽²⁾	734,407	2.3%
Darren R. Sherman ⁽³⁾	673,964	2.1%
Yuval Mika, Ph.D. ⁽⁴⁾	250,106	*
Jason Aryeh ⁽⁵⁾	73,825	*
Pamela Y. Connealy ⁽⁶⁾	15,758	*
Eric S. Fain, M.D. ⁽⁷⁾	47,058	*
Eric A. Rose, M.D. ⁽⁸⁾	27,900	*
Geoffrey W. Smith ⁽⁹⁾	1,713,297	5.4%
All directors and executive officers of New Orchestra following as a group (nine individuals)	3,611,533	10.8%
Five Percent Holders of New Orchestra:		
HSAC 2 Holdings, LLC ⁽¹⁰⁾	5,110,956	15.8%
Entities associated with RTW Investments, LP ⁽¹¹⁾	5,118,512	16.2%
Medtronic ⁽¹²⁾	4,992,588	15.8%
Perceptive Life Sciences Master Fund ⁽¹³⁾	1,850,976	5.8%

* Less than one percent.

(1) Unless otherwise indicated, the business address of each of the following individuals is 150 Union Square Drive, New Hope, PA 18938.

⁽²⁾ Consists of (i) 127,336 shares held directly by Mr. Hochman, of which 123,476 shares Mr. Hochman has the right to acquire dispositive power upon the settlement of restricted common stock awards as of or within 60 days, (ii) 422 shares over which Mr. Hochman has the right to acquire dispositive power upon exercise of warrants exercisable as of or within 60 days; (iii) 551,757 shares over which Mr. Hochman has the right to acquire dispositive power upon exercise of options exercisable as of or within 60 days; (iv) 51,342 shares held by the DPH 2008 Trust, over which Mr. Hochman has sole voting and dispositive power; (v) 3,140 shares held by the NSH 2008 Family Trust (the "NSH Trust"), over which Mr. Hochman has sole voting and dispositive power; and (vi) 410 shares over which the NSH Trust has the right to acquire dispositive power upon the exercise of warrants exercisable as of or within 60 days.

⁽³⁾ Consists of (i) 68,819 shares held directly by Mr. Sherman, of which 62,716 are restricted common stock awards that are fully vested; and (ii) 605,145 shares over which Mr. Sherman has the right to acquire dispositive power upon the exercise of options exercisable as of or within 60 days.

⁽⁴⁾ Consists of (i) 29,628 shares over which Mr. Mika has the right to acquire dispositive power upon the settlement of restricted common stock awards as of or within 60 days; and (ii) 220,478 shares over which Mr. Mika has the right to acquire dispositive power upon the exercise of options exercisable as of or within 60 days.

⁸

- (5) Consists of (i) 48,250 shares held directly by Mr. Aryeh; and (ii) 25,575 shares over which Mr. Aryeh has the right to acquire dispositive power upon the exercise of options exercisable as of or within 60 days.
- (6) Consists of 15,758 shares over which Ms. Connealy has the right to acquire dispositive power upon the exercise of options exercisable as of or within 60 days.
- (7) Consists of (i) 18,577 shares held by the Fain Living Trust (the "Fain Trust"), over which Mr. Fain has sole voting and dispositive power; (ii) 25,575 shares over which the Fain Trust has the right to acquire dispositive power upon the exercise of options exercisable as of or within 60 days; and (iii) 2,906 shares over which the Fain Trust has the right to acquire dispositive power upon the exercise of warrants exercisable as of or within 60 days.
- (8) Consists of (i) 25,575 shares over which Mr. Rose has the right to acquire dispositive power upon the exercise of options exercisable as of or within 60 days; and (ii) 2,325 shares over which Mr. Rose has the right to acquire dispositive power upon exercise of warrants exercisable as of or within 60 days.
- (9) Consists of (i) 25,575 shares over which Mr. Smith has the right to acquire dispositive power upon the exercise of options exercisable as of or within 60 days; (ii) 1,049,224 shares held by Ascent Biomedical Ventures II, L.P. ("ABV II"), (iii) 136,097 shares over which ABV II has the right to acquire dispositive power upon the exercise of warrants exercisable as of or within 60 days; (iv) 256,623 shares held by Ascent Biomedical Ventures Synecor, LP ("ABV Synecor"); (v) 519 shares over which ABV Synecor has the right to acquire dispositive power upon the exercise of warrants exercisable as of or within 60 days; (vi) 201,319 shares held by Ascent Biomedical Ventures II NY, LP ("ABV II NY"); and (vii) 43,940 shares over which ABV II NY has the right to acquire dispositive power upon the exercise of warrants exercise of warrants exercisable as of or within 60 days. ABV, LLC serves as general partner to ABV II, ABV Synecor and ABV II NY. Mr. Smith is a managing member of ABV, LLC. As such, Mr. Smith may be deemed to have voting and dispositive power over the shares held by ABV II, ABV Synecor and ABV II NY.
- (10) The Sponsor is governed by a board of directors consisting of three directors: Roderick Wong, Naveen Yalamanchi, and Alice Lee. Each director has one vote, and the approval of a majority of the directors is required to approve an action of the Sponsor. Under the so-called "rule of three," if voting and dispositive decisions regarding an entity's securities are made by three or more individuals, and a voting or dispositive decision requires the approval of a majority of those individuals, then none of the individuals is deemed a beneficial owner of the entity's securities. Based upon the foregoing analysis, no director of the Sponsor exercises voting or dispositive control over any of the shares held by the Sponsor, even those in which he or she directly holds a pecuniary interest. Accordingly, none of them will be deemed to have or share beneficial ownership of the shares held by the Sponsor.
- (11) Includes (i) 2,097,373 shares of New Orchestra Common Stock held by RTW Innovation Master Fund, Ltd.; (ii) 2,518,780 shares of New Orchestra Common Stock held by RTW Master Fund, Ltd.; and (iii) 502,359 shares of New Orchestra Common Stock held by RTW Venture Fund Limited. Roderick Wong, the former Chief Executive Officer of HSAC2, serves as the Managing Partner and Chief Investment Officer of RTW Investments, LP. Both he and RTW Investments, LP may be deemed the beneficial owner of the shares held by the RTW Funds and each disclaims beneficial ownership except to the extent of their pecuniary interest in the holders.
- (12) Consists of 4,992,588 shares held directly by Covidien Group S.à.r.l. The principal address of Covidien Group S.à.r.l. is c/o Medtronic, Inc., Operational Headquarters, 710 Medtronic Parkway, Minneapolis, MN 55432-5604.
- (13) Consists of (i) 1,781,226 shares held directly by Perceptive Life Sciences Master Fund, Ltd. ("Perceptive") and (ii) 69,750 shares over which Perceptive has the right to acquire dispositive power upon the exercise of warrants exercisable as of or within 60 days. Perceptive Advisors LLC ("Perceptive Advisors") serves as the investment manager to the Perceptive and may be deemed to beneficially own such shares. Joseph Edelman is the managing member of Perceptive Advisors and may be deemed to have voting and dispositive power over the shares held by Perceptive. The principal business address of the Perceptive is 51 Astor Place, 10th Floor, New York, NY 10003.

Directors and Executive Officers

The Company's directors and executive officers, composition of the committees of the Board and information with respect to the independence of the Board after the consummation of the Business Combination are described in the Proxy Statement/Prospectus in the section titled "Management after the Business Combination" and that information is incorporated herein by reference.

Executive Compensation

A description of the compensation of the named executive officers of HSAC2 before the consummation of the Business Combination, the named executive officers of Orchestra before the consummation of the Business Compensation and the named executive officers of New Orchestra after the consummation of the Business Combination is set forth in the Proxy Statement/Prospectus in the sections titled "*Executive Officers and Directors of HSAC2—Executive Compensation*," "*Executive and Director Compensation of Orchestra*," and "*Management After the Business Combination*," respectively, and that information is incorporated herein by reference.

Reference is made to the disclosure set forth below in Item 5.02 of this Current Report under the headings "Orchestra BioMed Holdings, Inc. 2023 Equity Incentive Plan" and "New Employment Agreements" which is incorporated herein by reference.

Director Compensation

A description of the compensation of the directors of HSAC2 before the consummation of the Business Combination is set forth in the Proxy Statement/Prospectus in the section titled "*Executive Officers and Directors of HSAC2—Executive Compensation*," and that information is incorporated herein by reference.

A description of the compensation of the directors of Orchestra before the consummation of the Business Combination is set forth in the Proxy Statement/Prospectus in the section titled "*Executive and Director Compensation of Orchestra*," and that information is incorporated herein by reference.

On January 26, 2023, the Compensation Committee of the Board adopted the Orchestra Biomed Holdings, Inc. Non-Employee Director Compensation Policy (the "New Orchestra Non-Employee Director Compensation Policy"), which is attached hereto as Exhibit and is incorporated herein by reference.

Certain Relationships and Related Party Transactions

Certain relationships and related party transactions of HSAC2 and the Company are described in the Proxy Statement/Prospectus in the section titled "Certain Relationships and Related Transactions" and that information is incorporated herein by reference.

Legal Proceedings

Information about legal proceedings is set forth in the section of the Proxy Statement/Prospectus "Business of Orchestra-Legal Proceedings" and such information is incorporated herein by reference.

Market Price of and Dividends on the Registrant's Common Equity and Related Stockholder Matters

Information about the ticker symbol, number of stockholders and dividends for HSAC2's securities is set forth in the Proxy Statement/Prospectus in the section titled "*Ticker Symbol, Market Price and Dividend Policy*" and such information is incorporated herein by reference.



Immediately following the Closing, there were 568 holders of record of New Orchestra Common Stock and 296 holders of record of New Orchestra Warrants to purchase New Orchestra Common Stock.

New Orchestra's common stock began trading on January 27, 2023 under the symbol "OBIO".

New Orchestra has not paid any cash dividends on shares of its common stock to date. The payment of cash dividends in the future will be dependent upon it revenues and earnings, if any, capital requirements and general financial condition. The payment of any dividends will be within the discretion of the Board. It is the present intention of the Board to retain all earnings, if any, for use in our business operations and, accordingly, the Board does not anticipate declaring any dividends in the foreseeable future. Further, if we incur any indebtedness, our ability to declare dividends may be limited by restrictive covenants we may agree to in connection therewith.

Reference is made to the disclosure described in the Proxy Statement/Prospectus in the section titled "*Proposal 8—The Equity Incentive Plan Proposal*," which is incorporated herein by reference. The Orchestra BioMed Holdings, Inc. 2023 Equity Incentive Plan (the "Equity Incentive Plan") was approved by HSAC2 shareholders at the Meeting.

Recent Sales of Unregistered Securities

Reference is made to the disclosure set forth below under Item 3.02 of this Current Report concerning the issuance and sale by the Company of certain unregistered securities, which is incorporated herein by reference.

Description of Registrant's Securities to Be Registered

The description of New Orchestra's securities is contained in the Proxy Statement/Prospectus in the section titled "Description of Securities after the Business Combination" and that information is incorporated herein by reference.

Immediately following the Closing, there were 31,614,079 shares of New Orchestra Common Stock issued and outstanding, held of record by 568 holders and 3,075,936 New Orchestra Warrants outstanding held of record by 296 holders. Such amounts do not include DTC participants or beneficial owners holding shares through nominee names.

Indemnification of Directors and Officers

Information about indemnification of the Company's directors and officers is set forth in the Proxy Statement/Prospectus in the section titled "Management after the Business Combination—Limitation of Liability and Indemnification of Directors and Officers" and that information is incorporated herein by reference. In connection with the Business Combination, New Orchestra entered into indemnification agreements with each of its directors and executive officers as of the Closing Date. The description of the indemnification agreements set forth above under Item 1.01 of this Current Report is incorporated herein by reference.

Financial Statements and Exhibits

The information set forth under Item 9.01 of this Current Report is incorporated herein by reference.

Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

The information set forth under Item 4.01 of this Current Report is incorporated herein by reference.

Item 3.02. Unregistered Sales of Equity Securities.

The disclosure set forth in the section titled "Introductory Note-Backstop Agreement" of this Current Report above is incorporated herein by reference.

The securities issued in connection with the Backstop Agreement were not registered under the Securities Act of 1933, as amended (the "Securities Act") in reliance on the exemption from registration provided by Section 4(a)(2) of the Securities Act.

Item 3.03. Material Modification to Rights of Security Holders.

In connection with the Domestication, HSAC2 filed a certificate of incorporation with the Secretary of State of the State of Delaware. The material terms of the certificate of incorporation and the general effect upon the rights of holders of HSAC2's capital stock are discussed in the Proxy Statement/Prospectus in the sections titled "*Proposal 2—The Domestication Proposal*" and "*Proposal 5—The Advisory Governance Proposals*," which are incorporated by reference herein.

As disclosed below in Item 8.01, in accordance with Rule 12g-3(a) under the Securities Exchange Act, New Orchestra is the successor issuer to HSAC2 and has succeeded to the attributes of HSAC2 as the registrant. In addition, the shares of New Orchestra Common Stock, as the successor to HSAC2, are deemed to be registered under Section 12(b) of the Exchange Act.

Certificate of Incorporation of New Orchestra and Bylaws of New Orchestra

Upon the effectiveness of the Domestication, HSAC2's memorandum and articles of association in effect immediately prior to the Domestication were replaced with a certificate of incorporation and bylaws of Domesticated HSAC2, which continued in effect through the Closing. The certificate of incorporation of Domesticated HSAC2 became the certificate of incorporation of New Orchestra (the "Charter"). The bylaws of Domesticated HSAC2 became the bylaws of New Orchestra (the "Bylaws"). Copies of the Charter and the Bylaws are attached hereto as Exhibit 3.1 and Exhibit 3.2 hereto, respectively, and are incorporated herein by reference.

The material terms of each and the Charter and the Bylaws and the general effect upon the rights of holders of New Orchestra's capital stock are included in the Proxy Statement/Prospectus under the sections titled "*Proposal 2—The Domestication Proposal*," "*Proposal 3—The Charter Approval Proposal*," "*Proposal 4—The Bylaws Proposal*" and "*Description of Securities after the Business Combination*," which are incorporated herein by reference.

Item 4.01. Changes in Registrant's Certifying Accountant.

Dismissal of independent registered public accounting firm

Effective upon the Closing, on January 26, 2023, the Board dismissed WithumSmith+Brown, PC ("Withum"), which served as HSAC2's independent registered public accounting firm prior to the Business Combination.

The report of Withum on the financial statements of HSAC2 as of December 31, 2022 and 2021 did not contain an adverse opinion or a disclaimer of opinion, and was not qualified or modified as to uncertainty, audit scope or accounting principles except for an explanatory paragraph in such report regarding substantial doubt about HSAC2's ability to continue as a going concern. During the fiscal years ended December 31, 2022 and 2021 and the subsequent interim period through January 26, 2023, there were no disagreements (as defined in Item 304(a)(1)(iv) of Regulation S-K) with Withum on any matter of accounting principles or practices, financial statement disclosure or auditing scope or procedures, which disagreements, if not resolved to the satisfaction of Withum, would have caused Withum to make reference to the subject matter of the disagreements in its reports covering such periods. In addition, no "reportable events," as defined in Item 304(a)(1)(v) of Regulation S-K, occurred within the period of Withum's engagement and the subsequent interim period preceding Withum's dismissal, other than the material weakness over financial reporting as a result of the Amended Form 10-K for the year ended December 31, 2021 filed with the SEC on March 31, 2022.

The Company provided Withum with a copy of the disclosures made pursuant to this Item 4.01 prior to the filing of this Current Report and requested that Withum furnish a letter addressed to the SEC dated January 31, 2023, which is filed as Exhibit 16.1 to this Current Report, stating whether it agrees with such disclosures, and, if not, stating the respects in which is does not agree.

Appointment of Ernst & Young LLP

Effective upon the Closing, the Board approved the engagement of Ernst & Young LLP ("EY") as the Company's independent registered public accounting firm for the fiscal year ending December 31, 2023, subject to the execution of an engagement letter. EY served as the independent registered public accounting firm for Orchestra prior to the Business Combination.

During the years ended December 31, 2021 and 2022 and the subsequent interim period through January 25, 2023, neither the Company nor anyone on its behalf consulted with EY regarding (i) the application of accounting principles to a specified transaction, either completed or proposed, or the type of audit opinion that might be rendered on the Company's financial statements, and no written report or oral advice was provided to the Company that EY concluded was an important factor considered by the Company in reaching a decision as to any accounting, auditing or financial reporting issue, or (ii) any matter that was the subject of a disagreement within the meaning of Item 304(a)(1)(iv) of Regulation S-K or any reportable event within the meaning of Item 304(a)(1)(v) of Regulation S-K.

Item 5.01. Changes in Control of Registrant.

Reference is made to the disclosure in the Proxy Statement/Prospectus in the section titled "*Proposal 1—The Business Combination Proposal*," which is incorporated herein by reference. Further reference is made to the information contained in Item 2.01 to this Current Report, which is incorporated herein by reference.

Item 5.02. Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

The information contained in the sections titled "Form 10 Information—Directors and Executive Officers," "Form 10 Information—Executive Compensation," "Form 10 Information—Director Compensation," "Form 10 Information—Certain Relationships and Related Transactions" and "Form 10 Information—Indemnification of Directors and Officers" in Item 2.01 to this Current Report is incorporated herein by reference.

Effective as of the Closing, the following people were appointed as directors of the Company:

- Class I directors: Eric A. Rose, M.D. and Jason Aryeh;
- Class II directors: Pamela Y. Connealy and Geoffrey W. Smith; and
- Class III directors: David P. Hochman, Darren R. Sherman and Eric S. Fain, M.D.

Effective as of the Closing, the executive officers of the Company are:

- David P. Hochman, Chief Executive Officer;
- Darren R. Sherman, President and Chief Operating Officer;
- Yuval Mika, Ph.D., General Manager and Chief Technology Officer, Bioelectronic Therapies; and
- Michael D. Kaswan, Chief Financial Officer.



Reference is made to the disclosure described in the Proxy Statement/Prospectus in the section titled "Management after the Business Combination" for biographical information about each of the directors and officers following the Business Combination, which is incorporated herein by reference.

Orchestra BioMed Holdings, Inc. 2023 Equity Incentive Plan

On January 26, 2023, the Equity Incentive Plan became effective. The Equity Incentive Plan is described in greater detail in the section of the Proxy Statement/Prospectus titled "*Proposal 8—The Equity Incentive Plan Proposal*," which is incorporated herein by reference.

The foregoing description of the Equity Incentive Plan, including the description in the Proxy Statement/Prospectus referenced above, does not purport to be complete and is subject to, and qualified in its entirety by, the full text of the Equity Incentive Plan, which is included herein as Exhibit 10.7 and is incorporated herein by reference.

New Employment Agreements

On January 26, 2023, Orchestra entered into employment agreements with each of David P. Hochman and Darren R. Sherman that governs the terms of their employment with Orchestra. The agreements provide that Mr. Hochman shall serve as Chief Executive Officer and Orchestra BioMed Founder and Mr. Sherman shall serve as President, Chief Operating Officer, and Orchestra BioMed Founder. Employment under the agreements is "at will" although Mr. Hochman and Mr. Sherman may be entitled to severance upon certain terminations as described below. Pursuant to the agreements, the executives receive annual base salary (\$595,000 for Mr. Hochman and \$495,000 for Mr. Sherman). In addition to this base compensation, the executives are each eligible to receive a discretionary annual bonus during each fiscal year of his employment with a target amount of 80% of base salary, with the performance metrics and goals required to receive such amount to be determined by the Orchestra Board in consultation with Orchestra's Chief Executive Officer. The agreements also provide for a future grant of equity compensation (the "Equity Grant") to each of the executives, to be made as soon as reasonably practicable following the beginning of Orchestra's first open trading window, so that the executive's long-term equity holdings in Orchestra are at least 4.6% of Orchestra's common stock in the case of Mr. Hochman and 4.3% in the case of Mr. Sherman. These percentages assume the future grant is in the form of options with a four-year vesting schedule commencing as of the Closing Date. However, the agreements contemplate that if instead the award is restricted stock units, the number of covered shares of Orchestra common stock will be reduced by a ratio based on two shares covered by restricted stock units being equal to three shares covered by stock options and that the award will vest over a to-be-agreed vesting schedule over a minimum of three years.

Board Membership Provisions under Messrs. Hochman and Sherman's Employment Agreements

The employment agreements provide that so long as each of Messrs. Hochman and Sherman remain in their current positions (Chief Executive Officer for Mr. Hochman and President and Chief Operating Officer for Mr. Sherman), Orchestra shall nominate each of them to serve as a member of the Orchestra Board. Messrs. Hochman and Sherman's service as a member of the Orchestra Board shall end upon certain specified events, including the termination of their employment and the expiration of their then current term if not reelected by Orchestra's stockholders.

Termination and Severance Provisions under Messrs. Hochman and Sherman's Employment Agreements

With respect to the each of Messrs. Hochman and Sherman, if Orchestra terminates his employment without Cause (as defined below), if he voluntarily resigns with Good Reason (as defined below), or in the event of his death, and subject to an effective release of claims, Orchestra will pay or provide him or his estate with continued base salary plus target bonus for 12 months. In addition, in the event of a termination without Cause or resignation for Good Reason, and subject to an effective of release of claims, Orchestra will pay or provide the executive with (i) if he validly elects to continue his healthcare coverage under applicable law, reimbursement of the applicable premium for up to 12 months, (ii) an extended stock option post-termination exercise period of up to 12 months, and (iii) all equity award held by the executive granted prior to the Closing Date will vest to the same extent such awards would have vested had the executive provided an additional 12 months of services. If the termination without Cause or resignation for Good Reason occurs within the period beginning three months prior to a change in control and ending 12 months following a change in control, then, subject to an effective release of claims, the continued base salary and target bonus component of severance specified under clause (i) shall be increased to 150% of annual base salary and target bonus, and (ii) all unvested equity awards held by the executive shall vest in full.

In addition, upon any termination of employment, each of Messrs. Hochman and Sherman shall be entitled to payment of accrued, but unpaid, base salary and reimbursement of previously and properly incurred business expenses.

Other Change in Control Provisions Not Part of Severance

The employment agreements provide that if a change in control occurs and the acquirer or surviving corporation refuses to assume or substitute for the executive's outstanding equity awards, the equity awards shall vest in full as of the date immediately preceding the change in control.

Definition of Cause under the Employment Agreements

"Cause" means each of:

- executive's breach of any of his obligations under the restrictive covenants section of his employment agreement;
- executive's breach of any of his obligations under his employment agreement (other than the restrictive covenants section of his agreement), which, to the extent curable, has not been cured within 30 days after the executive has been provided written notice of such breach;
- executive being convicted of, or pleading guilty or nolo contendere to, or being indicted for, any felony or any misdemeanor involving theft, fraud, dishonesty or moral turpitude; or
- fraud or embezzlement against Orchestra.

Definition of Good Reason under the Employment Agreements

"Good Reason" means each of:

• any change in the executive's position, title or reporting relationship with Orchestra that diminishes in any material respect the executive's authority, duties, or responsibilities; provided, however, that a change in authority, duties, or responsibilities due to Orchestra becoming a division, subsidiary, or other similar part of a larger organization shall not by itself constitute Good Reason;

- any material reduction in the executive's base compensation;
- the relocation of the executive's principal office or principal place of employment by more than fifty miles;
- or a material breach of the employment agreement by Orchestra;

provided, however, that the executive must provide written notice within 60 days of the occurrence of the potential Good Reason trigger; if curable, Orchestra must fail to cure the potential Good Reason trigger within 60 days of receipt of notice, and the executive must resign within 60 days following the expiration of the 60-day cure period.

Golden Parachute Provisions

The employment agreements contain customary golden parachute provisions providing that if, in connection with a change of control, amounts would constitute "parachute payments" under Section 280G of the Internal Revenue Code of 1986, as amended, and be subject to an excise tax under Section 4999 of the Internal Revenue Code of 1986, amounts will either be (i) reduced, or (ii) paid in full, whichever results in the greater after tax amount to the executive.

Item 5.03. Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

The disclosure set forth in Item 3.03 of this Current Report is incorporated in this Item 5.03 by reference.

Item 5.05. Amendments to the Registrant's Code of Ethics, or Waiver of a Provision of the Code of Ethics.

In connection with the Closing, on January 26, 2023, the Board approved and adopted a new Code of Business Conduct and Ethics applicable to all employees, officers and directors of the Company. A copy of the code is available in the Investors section of the Company's website at www.orchestrabiomed.com.

Item 5.06. Change in Shell Company Status.

As a result of the Business Combination, the Company ceased to be a shell company upon the closing of the Business Combination. The material terms of the Business Combination are described in the sections titled "*Proposal 1 – The Business Combination Proposal*" of the Proxy Statement/Prospectus, and are incorporated herein by reference.

Item 7.01. Regulation FD Disclosure.

On January 27, 2023, the Company issued a press release announcing the consummation of the Business Combination, and the listing of its common stock on Nasdaq. The press release is filed as Exhibit 99.1 to this Current Report and is incorporated herein by reference. The foregoing (including Exhibit 99.1) is being furnished pursuant to Item 7.01 and will not be deemed to be filed for purposes of Section 18 of the Exchange Act, or otherwise subject to the liabilities of that section, nor will it be deemed to be incorporated by reference in any filing under the Securities Act or the Exchange Act.

Item 8.01. Other Events.

By operation of Rule 12g-3(a) under the Exchange Act, the Company is the successor issuer to HSAC2 and has succeeded to the attributes of HSAC2 as the registrant, including HSAC2's SEC file number (001-39421) and CIK Code (0001814114). The Company's common stock is deemed to be registered under Section 12(b) of the Exchange Act, and the Company will hereafter file reports and other information with the SEC using HSAC2's SEC file number (001-39421).

The Company's common stock is listed for trading on Nasdaq under the symbol "OBIO," and the CUSIP number relating to the Company's common stock is 68572M 106.

Holders of HSAC2 shares who have filed reports under the Exchange Act with respect to those shares should indicate in their next filing, or any amendment to a prior filing, filed on or after the Closing Date that the Company is the successor to HSAC2.

Item 9.01. Financial Statements and Exhibits.

(a) Financial statements of businesses acquired.

The unaudited condensed consolidated financial statements of HSAC2 as of and for the three and nine months ended September 30, 2022 and 2021, and the related notes thereto beginning on page F-18 of the Proxy Statement/Prospectus, are incorporated herein by reference. These unaudited condensed consolidated financial statements should be read in conjunction with the historical audited financial statements of HSAC2 from May 25, 2020 (inception) through December 31, 2020 and the year ended December 31, 2021 and the related notes included in the Proxy Statement/Prospectus beginning on page F-2 thereof, which are incorporated herein by reference.

The unaudited consolidated financial statements of Orchestra as of and for the nine months ended September 30, 2022 and 2021 and the related notes included in the Proxy Statement/Prospectus beginning on page F-66 thereof are incorporated by reference herein. These unaudited consolidated financial statements should be read in conjunction with the historical audited financial statements of Orchestra for the years ended December 31, 2021 and 2020 and the related notes included in the Proxy Statement/Prospectus beginning on page F-36 thereof, which are incorporated herein by reference.

(b) Pro forma financial information.

The unaudited pro forma condensed combined financial information as of September 30, 2022 and for the nine months ended September 30, 2022 and the year ended December 31, 2021 is attached hereto as Exhibit 99.5 and is incorporated herein by reference.

(d) Exhibits.

Exhibit	
Number	Description
2.1#	Agreement and Plan of Merger dated as of July 4, 2022 by and among Health Sciences Acquisitions Corporation 2, HSAC Olympus
	Merger Sub, Inc., and Orchestra BioMed, Inc. (incorporated by reference to Annex A-1 of Amendment No. 4 of HSAC2's Form S-4 (File
	No. 333-266660), filed with the SEC on December 12, 2022).
2.2#	Amendment No. 1 to Agreement and Plan of Merger dated as of July 21, 2022 by and among Health Sciences Acquisitions Corporation 2,
	HSAC Olympus Merger Sub, Inc., and Orchestra BioMed, Inc. ((incorporated by reference to Annex A-2 of Amendment No. 4 of HSAC2's
2.3#	Form S-4 (File No. 333-266660), filed with the SEC on December 12, 2022). Amendment No. 2 to Agreement and Plan of Merger dated as of November 21, 2022 by and among Health Sciences Acquisitions
2.3#	<u>Amendment No. 2 to Agreement and Plan of Merger dated as of November 21, 2022 by and among Health Sciences Acquisitions</u> <u>Corporation 2, HSAC Olympus Merger Sub, Inc., and Orchestra BioMed, Inc. (incorporated by reference to Annex A-3 of Amendment No.</u>
	4 of HSAC2's Form S-4 (File No. 333-266660), filed with the SEC on December 12, 2022).
3.1+	<u>4 of HSAC2 s Form S-4 (File No. 353-200000), filed with the SEC on December 12, 2022).</u> Certificate of Incorporation of Orchestra BioMed Holdings, Inc.
3.1+ 3.2+	Bylaws of Orchestra BioMed Holdings, Inc.
3.2+ 4.1	Form of Common Stock Warrant, issued by Orchestra BioMed, Inc. in the Formation Mergers in exchange for Caliber, BackBeat and
4.1	FreeHold warrants (incorporated by reference to Exhibit 4.4 of HSAC2's Form S-4 (File No. 333-266660)).
4.2	Form of Amendment to Common Stock Warrant, issued by Orchestra BioMed, Inc. in the Formation Mergers in exchange for Caliber,
4.2	BackBeat and FreeHold warrants (incorporated by reference to Exhibit 4.5 of HSAC2's Form S-4 (File No. 333-266660)).
4.3	Form of Amended and Restated Common Stock Warrant, issued by Orchestra BioMed, Inc. to designees of Aegis Capital Corp.
4.5	(incorporated by reference to Exhibit 4.6 of HSAC2's Form S-4 (File No. 333-266660)).
4.4	Form of Special Advisory Common Stock Warrant, issued by Orchestra BioMed, Inc. to its strategic advisers, dated May 31, 2018
4.4	(incorporated by reference to Exhibit 4.7 of HSAC2's Form S-4 (File No. 333-266660))
4.5	Form of Amendment to Special Advisory Common Stock Warrant, issued by Orchestra BioMed, Inc. to its strategic advisers (incorporated
ч.5	by reference to Exhibit 4.8 of HSAC2's Form S-4 (File No. 333-266660)).
4.6	Form of Common Stock Warrant, issued by Orchestra BioMed, Inc. to SLD Capital Corp., dated August 13, 2018 (incorporated by
	reference to Exhibit 4.9 of HSAC2's Form S-4 (File No. 333-266660)).
4.7	Form of Amendment to Common Stock Warrant, issued by Orchestra BioMed, Inc. to SLD Capital Corp. (incorporated by reference to
	Exhibit 4.10 of HSAC2's Form S-4 (File No. 333-266660)).
4.8	Investors' Rights Agreement, by and among Orchestra BioMed, Inc. and the investors listed on Schedule A thereto, dated May 31, 2018
	(incorporated by reference to Exhibit 4.11 of HSAC2's Form S-4 (File No. 333-266660)).
4.9	Form of Subscription Agreement for shares of Orchestra BioMed, Inc. Series B-1 Preferred Stock (incorporated by reference to Exhibit
	<u>4.12 of HSAC2's Form S-4 (File No. 333-266660)).</u>
4.10	Common Stock Warrant, issued by Orchestra BioMed, Inc. to Silicon Valley Bank, dated December 10, 2019 (incorporated by reference to
	Exhibit 4.13 of HSAC2's Form S-4 (File No. 333-266660)).
4.11	Common Stock Warrant, issued by Orchestra BioMed, Inc. to Avenue Venture Opportunities Fund, L.P., dated June 3, 2022 (incorporated
	by reference to Exhibit 4.14 of HSAC2's Form S-4 (File No. 333-266660)).
4.12	Common Stock Warrant, issued by Orchestra BioMed, Inc. to Avenue Venture Opportunities Fund II, L.P., dated June 3, 2022 (incorporated
	by reference to Exhibit 4.15 of HSAC2's Form S-4 (File No. 333-266660)).
4.13	Form of New Orchestra Common Stock Warrant issued pursuant to the Parent Support Agreement (incorporated by reference to Exhibit
	<u>4.16 of HSAC2's Form S-4 (File No. 333-266660)).</u>
4.14+	Amended & Restated Warrant issued to HSAC 2 Holdings, LLC, dated January 26, 2023.

10.1+	Amended and Restated Registration Rights and Lock-Up Agreement, by and among Health Sciences Acquisitions Corporation 2,
	equityholders thereof and certain stockholders of Orchestra BioMed, Inc.
10.2*+	Form of Indemnification Agreement of Orchestra BioMed Holdings, Inc.
10.3	Amended and Restated Parent Support Agreement dated as of November 21, 2022 by and among Health Sciences Acquisitions Corporation
	2, Orchestra BioMed, Inc., HSAC 2 Holdings, LLC, Alice Lee, Stephanie A. Sirota, Pedro Granadillo, Stuart Peltz, Michael Brophy, and
	Carsten Boess (incorporated by reference to Exhibit 10.16 of HSAC2's Form S-4 (File No. 333-266660)).
10.4	Orchestra Support Agreement dated as of July 4, 2022 by and among Health Sciences Acquisitions Corporation 2, Orchestra BioMed, Inc.,
	and Covidien Group S.A.R.L. (incorporated by reference to Exhibit 10.5 to HSAC2's Current Report on Form 8-K filed with the SEC on
	<u>July 5, 2022).</u>
10.5	Form of Earnout Election Agreement, by and among Health Sciences Acquisitions Corporation 2, Orchestra BioMed, Inc. and the
	securityholders thereto (incorporated by reference to Exhibit 10.7 to HSAC2's Current Report on Form 8-K filed with the SEC on July 5,
	<u>2022).</u>
10.6*	Orchestra BioMed, Inc. 2018 Stock Incentive Plan (incorporated by reference to Exhibit 10.19 of HSAC2's Form S-4 (File No. 333-
	<u>266660).</u>
10.7*+	Orchestra BioMed Holdings, Inc. 2023 Equity Incentive Plan.
10.8*+	Form of Stock Option Grant Notice and Stock Option Agreement under the Orchestra BioMed Holdings, Inc. 2023 Equity Incentive Plan.
10.9*+	Form of Restricted Stock Unit Award Grant Notice and Restricted Stock Unit Award Agreement under the Orchestra BioMed Holdings,
	Inc. 2023 Equity Incentive Plan.
10.10^	Commercial Lease, by and between Caliber Therapeutics, Inc. and Union Square, L.P. for facilities at 150 and 140 Union Square Drive,
	New Hope, Pennsylvania, dated December 14, 2009 and amended June 22, 2010, February 1, 2011, September 18, 2012, January 15, 2015,
	January 20, 2017, August 8, 2017, and January 29, 2019 (incorporated by reference to Exhibit 10.23 of HSAC2's Form S-4 (File No. 333-
10.11	<u>266660)).</u>
10.11	Agreement of Lease, by and between Orchestra BioMed, Inc. and ESRT One Grand Central Place, L.L.C. for facilities at Room/Suite 2430,
	One Grand Central Place, 60 East 42nd Street, New York, New York, dated November 5, 2019 (incorporated by reference to Exhibit 10.24
10.12	of HSAC2's Form S-4 (File No. 333-266660)).
10.12	Irrevocable Standby Letter of Credit, issued by Silicon Valley Bank to Orchestra BioMed, Inc. for the benefit of ERST One Grand Central Place, L.L.C., dated October 28, 2019 and amended September 20, 2022 (incorporated by reference to Exhibit 10.25 of HSAC2's Form S-4
	Flace, L.L.C., dated October 28, 2019 and amended September 20, 2022 (incorporated by reference to Exhibit 10.25 of HSAC2's Form S-4 (File No. 333-266660)).
10.13	License Agreement, by and between MOTUS GI Holdings, Inc. and Orchestra BioMed, Inc. for facilities at Suite 310, 1301 East Broward
10.15	Boulevard, Fort Lauderdale, Florida, dated January 22, 2020 (incorporated by reference to Exhibit 10.26 of HSAC2's Form S-4 (File No.
	<u>333-266660)).</u>
10.14	Amendment to License Agreement, by and between MOTUS GI Holdings, Inc. and Orchestra BioMed, Inc., dated May 1, 2022
10.11	(incorporated by reference to Exhibit 10.27 of HSAC2's Form S-4 (File No. 333-266660)).
10.15^	Exclusive License and Collaboration Agreement, by and among Orchestra BioMed, Inc., BackBeat Medical, LLC, and Medtronic, Inc.
10.10	dated June 30, 2022 (incorporated by reference to Exhibit 10.28 of HSAC2's Form S-4 (File No. 333-266660)).
10.16^	Distribution Agreement, by and among Orchestra BioMed, Inc., Terumo Corporation and Terumo Medical Corporation, dated June 13,
	2019 (incorporated by reference to Exhibit 10.29 of HSAC2's Form S-4 (File No. 333-266660)).
	<u></u>

10.17	Loan and Security Agreement, by and among Orchestra BioMed, Inc., Avenue Venture Opportunities Fund II, L.P., and Avenue Venture
	Opportunities Fund II, L.P., dated June 3, 2022 (incorporated by reference to Exhibit 10.30 of HSAC2's Form S-4 (File No. 333-266660)).
10.18	Stock Purchase Agreement, by and among Orchestra BioMed, Inc., Orchestra Medical Ventures, LLC and Accelerated Technologies, Inc.,
	dated December 20, 2019 (incorporated by reference to Exhibit 10.31 of HSAC2's Form S-4 (File No. 333-266660)).
10.19*+#	Employment Agreement, by and between Orchestra BioMed Holdings, Inc. and David P. Hochman, dated January 26, 2023.
10.20*+#	Employment Agreement, by and between Orchestra BioMed Holdings, Inc. and Darren R. Sherman, dated January 26, 2023.
10.21*	Employment Agreement, by and between Orchestra BioMed, Inc. and Yuval Mika, dated May 31, 2018 (incorporated by reference to
	Exhibit 10.34 of HSAC2's Form S-4 (File No. 333-266660).
10.22*	Offer Letter, by and between Orchestra BioMed, Inc. and Michael D. Kaswan, dated January 9, 2022 (incorporated by reference to Exhibit
	<u>10.35 of HSAC2's Form S-4 (File No. 333-266660).</u>
10.23	Forward Purchase Agreement dated as of July 4, 2022, by and among Health Sciences Acquisitions Corporation 2, Orchestra BioMed, Inc.,
	and Covidien Group S.à.r.l. (incorporated by reference to Exhibit 10.1 to HSAC2's Current Report on Form 8-K filed with the SEC on July
	<u>5,2022).</u>
10.24	Forward Purchase Agreement dated as of July 4, 2022, by and among Health Sciences Acquisitions Corporation 2, Orchestra BioMed, Inc.,
	and RTW Master Fund, Ltd., RTW Innovation Master Fund, Ltd., and RTW Venture Fund Limited (incorporated by reference to Exhibit
	10.2 to HSAC2's Current Report on Form 8-K filed with the SEC on July 5, 2022).
10.25	Amendment No. 1 to Forward Purchase Agreement dated as of October 21, 2022, by and among Health Sciences Acquisitions Corporation
	2, Orchestra BioMed, Inc., and RTW Master Fund, Ltd., RTW Innovation Master Fund, Ltd., and RTW Venture Fund Limited (incorporated
	by reference to Exhibit 10.1 to HSAC2's Current Report on Form 8-K filed with the SEC on October 21, 2022).
10.26	Backstop Agreement dated as of July 4, 2022, by and among Health Sciences Acquisitions Corporation 2, Orchestra BioMed, Inc., RTW
	Master Fund, Ltd., RTW Innovation Master Fund, Ltd., and RTW Venture Fund Limited (incorporated by reference to Exhibit 10.3 to
	HSAC2's Current Report on Form 8-K filed with the SEC on July 5, 2022).
10.27	Amendment No. 1 to Backstop Agreement dated as of October 21, 2022, by and among Health Sciences Acquisitions Corporation 2,
	Orchestra BioMed, Inc., RTW Master Fund, Ltd., RTW Innovation Master Fund, Ltd., and RTW Venture Fund Limited (incorporated by
	reference to Exhibit 10.2 to HSAC2's Current Report on Form 8-K filed with the SEC on October 21, 2022).
16.1+	Letter from WithumSmith+Brown, PC to the SEC, dated January 31, 2023.
21.1+	List of Subsidiaries.
99.1+	Press Release, dated January 27, 2023.
99.2+	Unaudited pro forma condensed combined financial information and the accompanying notes as of and for the nine months ended
	September 30, 2022 and for the year December 31, 2021.
104 +	Cover Page Interactive Data File (formatted as Inline XBRL and contained in Exhibit 101).

+ Filed herewith.

* Indicates a management contract or compensatory plan.

- Certain identified information has been omitted pursuant to Item 601(b)(10) of Regulation S-K because such information is both (i) not material and (ii) information that the Registrant treats as private or confidential. The Registrant hereby undertakes to furnish supplemental copies of the unredacted exhibit upon request by the SEC.
- # Certain of the exhibits and schedules to this Exhibit have been omitted in accordance with Regulation S-K Item 601. The Registrant agrees to furnish a copy of all omitted exhibits and schedules to the SEC upon its request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

ORCHESTRA BIOMED HOLDINGS, INC.

By: /s/ David P. Hochman Name: David P. Hochman Title: Chief Executive Officer

Date: January 31, 2023

CERTIFICATE OF INCORPORATION OF ORCHESTRA BIOMED HOLDINGS, INC.

I.

The name of the corporation is Orchestra BioMed Holdings, Inc. (hereinafter called the "Corporation").

П.

The address of the registered office of the Corporation in the State of Delaware is 850 New Burton Road, Suite 201, the City of Dover, County of Kent 19904, and the name of the registered agent of the Corporation in the State of Delaware at such address is Cogency Global Inc.

III.

The nature of the business or purposes to be conducted or promoted by the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware ("DGCL").

The Corporation is being incorporated in connection with the domestication of Health Sciences Acquisitions Corporation 2, a Cayman Islands exempted company ("HSAC2"), to a Delaware corporation, which domestication is being effected in connection with the transactions contemplated by that certain Agreement and Plan of Merger entered into by HSAC2, HSAC Olympus Merger Sub, Inc., a Delaware corporation and wholly owned subsidiary of HSAC2, and Orchestra BioMed, Inc., a Delaware corporation, on July 4, 2022, as amended, and this Certificate of Incorporation (as amended and/or restated from time to time, including pursuant to any Preferred Stock Designation (as defined below), this "Certificate of Incorporation") is being filed simultaneously with a certificate of domestication effecting such domestication.

IV.

A. The Corporation is authorized to issue two classes of stock to be designated, respectively, "Common Stock," and "Preferred Stock." The total number of shares that the Corporation is authorized to issue is 350,000,000 shares, 340,000,000 shares of which shall be Common Stock (the "*Common Stock*"), and 10,000,000 shares of which shall be Preferred Stock (the "*Preferred Stock*"). The Common Stock shall have a par value of \$0.0001 per share, and the Preferred Stock shall have a par value of \$0.0001 per share.

B. The Preferred Stock may be issued from time to time in one or more series, the shares of each series to have such designations and powers, preferences, privileges and rights, and qualifications, limitations and restrictions thereof, as are stated and expressed herein and in the resolution or resolutions providing for the issue of such series adopted by the Board of Directors as hereafter prescribed (a "*Preferred Stock Designation*"). Subject to any limitation prescribed by law and the rights of any series of the Preferred Stock then outstanding, if any, authority is hereby expressly granted to and vested in the Board of Directors to authorize the issuance of all or any of the shares of the Preferred Stock in one or more series, and, with respect to each series of Preferred Stock, to fix the number of shares and state by the Preferred Stock Designation, the designations, powers, preferences, privileges and relative participating, optional, or other rights and such qualifications, limitations, or restrictions thereof, as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issuance of such shares and as may be permitted by the DGCL. The Board of Directors is also expressly authorized to increase (but not above the authorized number of shares of Preferred Stock) or decrease (but not below the number of shares of such series subsequent to the issuance of shares of that series.

C. The number of authorized shares of Preferred Stock, or Common Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the voting power of all of the outstanding shares of stock of the Corporation entitled to vote thereon, without a separate vote of the holders of the Preferred Stock, or of any series thereof, or Common Stock, irrespective of the provisions of Section 242(b)(2) of the DGCL, unless a vote of any such holders is required pursuant to the terms of any Preferred Stock Designation filed with respect to any series of Preferred Stock.

D. Except as provided above, the designations, powers, preferences, privileges and relative participating, optional, or other rights, and qualifications, limitations, or restrictions of the Common Stock are as follows:

1. Rights Relating to Dividends, Subdivisions and Combinations. Subject to the prior rights of holders of all classes and series of stock at the time outstanding having prior rights as to dividends, the holders of the Common Stock shall be entitled to receive, when, as and if declared by the Board of Directors, out of any assets of the Corporation legally available therefor, such dividends as may be declared from time to time by the Board of Directors. Any dividends paid to the holders of shares of Common Stock shall be paid pro rata, on an equal priority, pari passu basis, unless different treatment of the shares of each such class is approved by the affirmative vote of the holders of a majority of the outstanding shares of the applicable class of Common Stock treated adversely, voting separately as a class.

2. Voting Rights.

(a) Except as otherwise required by law or this Certificate of Incorporation (the "*Certificate of Incorporation*") (including any Preferred Stock Designation), at any annual or special meeting of the stockholders of the Corporation, holders of the Common Stock shall have the exclusive right to vote for the election of directors and on all other matters properly submitted to a vote of the stockholders.

(b) Except as otherwise required by law or the Certificate of Incorporation (including any Preferred Stock Designation), the holders of shares of Common Stock shall be entitled to one vote for each such share on each matter properly submitted to the stockholders on which the holders of the Common Stock are entitled to vote.

(c) Except as otherwise required by applicable law, holders of Common Stock, as such, shall not be entitled to vote on any amendment to the Certificate of Incorporation that relates solely to the terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to the Certificate of Incorporation or applicable law.

3. Liquidation Rights.

In the event of any voluntary or involuntary liquidation, dissolution or winding-up of the Corporation, upon the completion of the distributions required with respect to each series of Preferred Stock that may then be outstanding, the remaining assets of the Corporation legally available for distribution to stockholders shall be distributed on an equal priority, pro rata basis to the holders of Common Stock, unless different treatment of the shares of each such class is approved by the affirmative vote of the holders of a majority of the outstanding shares of Common Stock; provided, however, for the avoidance of doubt, compensation pursuant to any employment, consulting, severance or other compensatory arrangement to be paid to or received by a person who is also a holder of Common Stock does not constitute consideration or a "distribution to stockholders" in respect of the Common Stock.

V.

A. The liability of the directors or officers of the Corporation for monetary damages for breach of fiduciary duty as a director or officer shall be eliminated to the fullest extent authorized under applicable law, provided that this provision shall not eliminate or limit the liability of:

- 1. a director or officer for any breach of the director's or officer's duty of loyalty to the Corporation or its stockholders;
- 2. a director or officer for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law;
- 3. a director under Section 174 of the DGCL;
- 4. a director or officer for any transaction from which the director or officer derived an improper personal benefit; or
- 5. an officer in any action by or in the right of the Corporation.



Any amendment, modification or repeal of the foregoing sentence shall not adversely affect any right or protection of a director or officer of the Corporation hereunder in respect of any act or omission occurring prior to the time of such amendment, modification or repeal.

B. To the fullest extent permitted by applicable law, the Corporation is authorized to provide indemnification of (and advancement of expenses to) directors, officers and other agents of the Corporation (and any other persons to which applicable law permits the Corporation to provide indemnification) through Bylaw provisions, agreements with such agents or other persons, vote of stockholders or disinterested directors or otherwise in excess of the indemnification and advancement otherwise permitted by such applicable law.

C. If applicable law is amended after approval by the stockholders of this Article V to authorize corporate action further eliminating or limiting the personal liability of directors or officers, then the liability of a director or officer to the Corporation shall be eliminated or limited to the fullest extent permitted by applicable law as so amended. Any repeal or modification of this Article V shall only be prospective and shall not affect the rights under this Article V in effect at the time of the alleged occurrence of any action or omission to act giving rise to liability.

D. All references in this Article V to an "officer" shall mean a person who:

- 1. is or was the president, chief executive officer, chief operating officer, chief financial officer, chief legal officer, controller, treasurer or chief accounting officer of the Corporation at any time during the course of conduct alleged in the action or proceeding to be wrongful;
- 2. is or was identified in the Corporation's public filings with the United States Securities and Exchange Commission because such person is or was one of the most highly compensated executive officers of the Corporation at any time during the course of conduct alleged in the action or proceeding to be wrongful; or
- 3. has, by written agreement with the Corporation, consented to be identified as an officer for purposes of this Article V.

VI.

A. Unless the Corporation consents in writing to the selection of an alternative forum, (a) the Court of Chancery (the "*Chancery Court*") of the State of Delaware (or, in the event that the Chancery Court does not have jurisdiction, the federal district court for the District of Delaware or other state courts of the State of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for (i) any derivative action, suit or proceeding brought on behalf of the Corporation, (ii) any action, suit or proceeding asserting a claim of breach of a fiduciary duty owed by any director, officer or stockholder of the Corporation to the Corporation or to the Corporation's stockholders, (iii) any action, suit or proceeding asserting a claim against the Corporation or this Certificate of Incorporation (as either may be amended from time to time) or (iv) any action, suit or proceeding asserting a claim against the Corporation governed by the internal affairs doctrine; and (b) subject to the preceding provisions of this Article VI, the federal district courts of the United States of America shall be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act of 1933, as amended (the "*Securities Act*"). If any action the subject matter of which is within the scope of clause (a) of the immediately preceding sentence is filed in a court other than the courts in the State of Delaware (a "*Foreign Action*") in the name of any stockholder, such stockholder shall be deemed to have consented to (x) the personal jurisdiction of the state and federal courts in the State of Delaware in connection with any action brought in any such action by service upon such stockholder's coursel in the Foreign Action as agent for such stockholder.

B. Any person or entity purchasing or otherwise acquiring any interest in any security of the Corporation shall be deemed to have notice of and consented to this Article VI. Notwithstanding the foregoing, the provisions of this Article VI shall not apply to suits brought to enforce any liability or duty created by the Securities Exchange Act of 1934, as amended, or any other claim for which the federal courts of the United States have exclusive jurisdiction.

C. If any provision or provisions of this Article VI shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever, (a) the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Article VI (including, without limitation, each portion of any paragraph of this Article VI containing any such provision held to be invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and (b) the application of such provision to other persons or entities and circumstances shall not in any way be affected or impaired thereby.

VII.

For the management of the business and for the conduct of the affairs of the Corporation, and in further definition, limitation and regulation of the powers of the Corporation, of its directors and of its stockholders or any class thereof, as the case may be, it is further provided that:

A. Board of Directors.

1. Generally. The management of the business and the conduct of the affairs of the Corporation shall be vested in the Board of Directors. The authorized number of directors which shall constitute the Board of Directors shall be fixed by the Board of Directors in the manner provided in the Bylaws.

2. Election.

(a) Subject to the rights of the holders of any series of Preferred Stock to elect additional directors as specified in any Preferred Stock Designation, the directors shall be divided into three classes designated as Class I, Class II and Class III, respectively. Each class shall consist, as nearly as possible, of one-third of the total number of such directors. The Board of Directors is authorized to assign members of the Board of Directors already in office to such classes at the time the classification becomes effective. At the first annual meeting of stockholders following such initial classification of the Board of Directors, the initial term of office of the Class I directors shall expire and Class I directors, the initial term of three years. At the second annual meeting of stockholders following such initial classification of the Board of Directors, the initial term of office of the Class III term of three years. At the third annual meeting of stockholders following such initial classification of the Board of Directors, the initial term of office of the Class III term of three years. At the third annual meeting of stockholders following such initial classification of the Board of Directors, the initial term of office of the Class III directors shall be elected for a full term of three years. At the third annual meeting of stockholders following such initial classification of the Board of Directors, the initial term of office of the Class III directors shall expire and Class III directors shall be elected for a full term of three years. At each succeeding annual meeting of stockholders, directors shall be elected for a full term of three years. At each succeeding annual meeting of stockholders, directors shall be elected for a full term of three years to succeed the directors of the class whose terms expire at such annual meeting.

(b) At any time that applicable law prohibits a classified board as described in Section A.2.(a) of this Article VII, all directors shall be elected at each annual meeting of stockholders to hold office until the next annual meeting. The directors of the Corporation need not be elected by written ballot unless the Bylaws so provide.

(c) No stockholder entitled to vote at an election for directors may cumulate votes to which such stockholder is entitled unless required by applicable law at the time of such election. During such time or times that applicable law requires cumulative voting, every stockholder entitled to vote at an election for directors may cumulate such stockholder's votes and give one candidate a number of votes equal to the number of directors to be elected multiplied by the number of votes to which such stockholder's shares are otherwise entitled, or distribute the stockholder's votes on the same principle among as many candidates as such stockholder thinks fit. No stockholder, however, shall be entitled to so cumulate such stockholder's votes unless (i) the names of such candidate or candidates have been placed in nomination prior to the voting and (ii) the stockholder has given notice at the meeting, prior to the voting, of such stockholder's intention to cumulate such stockholder's votes. If any stockholder has given proper notice to cumulate votes, all stockholders may cumulate their votes for any candidates who have been properly placed in nomination. Under cumulative voting, the candidates receiving the highest number of votes, up to the number of directors to be elected.

(d) Notwithstanding the foregoing provisions of this section, each director shall serve until his or her successor is duly elected and qualified or until his or her earlier death, resignation or removal. No decrease in the number of directors constituting the Board of Directors shall shorten the term of any incumbent director.

3. Removal of Directors. Subject to any limitations imposed by applicable law, removal shall be as provided in Section 141(k) of the DGCL.

4. Vacancies. Subject to any limitations imposed by applicable law and subject to the rights of the holders of any series of Preferred Stock, any vacancies on the Board of Directors resulting from death, resignation, disqualification, removal or other causes and any newly created directorships resulting from any increase in the number of directors, shall, unless the Board of Directors determines by resolution that any such vacancies or newly created directorships shall be filled by the stockholders and except as otherwise provided by applicable law, be filled only by a majority of the directors then in office, although less than a quorum, or by the sole remaining director, and not by the stockholders. Any director elected in accordance with the preceding sentence shall hold office for the remainder of the full term of the director for which the vacancy was created or occurred and until such director's successor shall have been elected and qualified.

B. Stockholder Actions. No action shall be taken by the stockholders of the Corporation except at an annual or special meeting of stockholders called in accordance with the Bylaws, and no action shall be taken by the stockholders by written consent. Advance notice of stockholder nominations for the election of directors and of business to be brought by stockholders before any meeting of the stockholders of the Corporation shall be given in the manner provided in the Bylaws.

C. Bylaws. The Board of Directors is expressly empowered to adopt, amend or repeal the Bylaws. The stockholders shall also have the power to adopt, amend or repeal the Bylaws; provided, however, that, in addition to any vote of the holders of any class or series of stock of the Corporation required by law or by the Certificate of Incorporation, such action by stockholders shall require the affirmative vote of the holders of at least 66 2/3% of the voting power of all of the then-outstanding shares of the capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class.

VIII.

A. The Corporation reserves the right to amend, alter, change or repeal any provision contained in the Certificate of Incorporation, in the manner now or hereafter prescribed by the DGCL, except as provided in paragraph B. of this Article VIII, and all rights conferred upon the stockholders herein are granted subject to this reservation.

B. Notwithstanding any other provisions of the Certificate of Incorporation or any provision of law that might otherwise permit a lesser vote or no vote, but in addition to any affirmative vote of the holders of any particular class or series of the Corporation required by law or by the Certificate of Incorporation or any Preferred Stock Designation filed with respect to a series of Preferred Stock, the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) of the voting power of all of the then-outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required to alter, amend or repeal Articles V, VI, VII and VIII.

IN WITNESS WHEREOF, this Certificate of Incorporation has been executed on this 25th day of January, 2023.

By: /s/ Roderick Wong Name: Roderick Wong Title: Incorporator

BYLAWS

OF

ORCHESTRA BIOMED HOLDINGS, INC.

(A DELAWARE CORPORATION)

ARTICLE I

OFFICES

Section 1. Registered Office. The registered office of the corporation in the State of Delaware shall be as set forth in the certificate of incorporation of the corporation (the "Certificate of Incorporation").

Section 2. Other Offices. The corporation shall also have and maintain an office or principal place of business at such place as may be fixed by the Board of Directors of the corporation (the "*Board of Directors*"), and may also have offices at such other places, both within and without the State of Delaware, as the Board of Directors may from time to time determine or as may be necessary or convenient to the business of the corporation.

ARTICLE II

CORPORATE SEAL

Section 3. Corporate Seal. The Board of Directors may adopt a corporate seal. If adopted, the corporate seal shall consist of a die bearing the name of the corporation and the inscription, "Corporate Seal-Delaware." Said seal may be used by causing it or a facsimile thereof to be impressed or affixed or reproduced or otherwise.

ARTICLE III

STOCKHOLDERS' MEETINGS

Section 4. Place of Meetings. Meetings of the stockholders of the corporation may be held at such place (if any), either within or without the State of Delaware, as may be determined from time to time by the Board of Directors. The Board of Directors may, in its sole discretion, determine that the meeting shall not be held at any place, but may instead be held solely by means of remote communication as provided under the Delaware General Corporation Law (the "*DGCL*"). For the avoidance of doubt, the Board of Directors may, in its sole discretion, determine that a meeting of stockholders of the corporation may be held both in a place and by means of remote communication. For any meeting of stockholders to be held by remote communication, the corporation shall (i) implement reasonable measures to verify that each person deemed present and permitted to vote at the meeting by remote communication is a stockholder or proxy holder, (ii) implement reasonable measures to provide such stockholders and proxy holders a reasonable opportunity to participate in the meeting and to vote on matters submitted to the stockholders, including an opportunity to read or hear the proceedings of the meeting substantially concurrently with such proceedings, and (iii) if any stockholder or proxy holder votes or takes other action at the meeting by means of remote communication, a record of such vote or other action shall be maintained by the corporation.

Section 5. Annual Meeting.

(a) The annual meeting of the stockholders of the corporation, for the purpose of election of directors and for such other business as may properly come before it, shall be held at such place, if any, and on such date and at such time as shall be designated from time to time by the Board of Directors and stated in the corporation's notice of meeting of stockholders. Nominations of persons for election to the Board of Directors and proposals of business to be considered by the stockholders may be made at an annual meeting of stockholders: (i) pursuant to the corporation's notice of meeting of stockholders given by or at the direction of the Board of Directors; (ii) brought specifically by or at the direction of the Board of Directors or a duly authorized committee thereof; or (iii) by any stockholder of the corporation who was a stockholder of record at the time of giving the stockholder's notice provided for in Section 5(b) below, who is entitled to vote at the meeting and who complied with the notice procedures set forth in Section 5. For the avoidance of doubt, clause (iii) above shall be the exclusive means for a stockholder to make nominations and submit other business (other than matters properly included in the corporation's notice of meeting of stockholders and proxy statement under Rule 14a-8 under the Securities Exchange Act of 1934, as amended (the "1934 Act"), and the rules and regulations thereunder before an annual meeting of stockholders).

(b) At an annual meeting of the stockholders, only such business shall be conducted as is a proper matter for stockholder action under Delaware law and as shall have been properly brought before the meeting in accordance with Section 5(a) and the procedures below.

(i) For nominations for the election to the Board of Directors to be properly brought before an annual meeting by a stockholder pursuant to clause (iii) of Section 5(a), such stockholder must deliver written notice to the Secretary at the principal executive offices of the corporation on a timely basis as set forth in Section 5(b)(iii) and must update and supplement such written notice on a timely basis as set forth in Section 5(c). Such stockholder's notice shall set forth: (A) as to each nominee such stockholder proposes to nominate at the meeting: (1) the name, age, business address and residence address of such nominee; (2) the principal occupation or employment of such nominee; (3) the class or series and number of shares of each class or series of capital stock of the corporation that are owned beneficially and of record by such nominee; (4) the date or dates on which such shares were acquired and the investment intent of such acquisition; (5) a statement whether such nominee, if elected, intends to tender, promptly following such person's failure to receive the required vote for election or re-election at the next meeting at which such person would face election or re-election, an irrevocable resignation effective upon acceptance of such resignation by the Board of Directors; and (6) such other information concerning such nominee as would be required to be disclosed in a proxy statement soliciting proxies for the election of such nominee as a director in an election contest (even if an election contest is not involved), or that is otherwise required to be disclosed pursuant to Section 14 of the 1934 Act and the rules and regulations promulgated thereunder (including such person's written consent to being named in the corporation's proxy statement and associated proxy card as a nominee of the stockholder and to serving as a director if elected); and (B) the information required by Section 5(b)(iv). The corporation may require any proposed nominee to furnish such other information as it may reasonably require to determine the eligibility of such proposed nominee to serve (i) as an independent director (as such term is used in any applicable stock exchange listing requirements or applicable law) of the corporation or (ii) on any committee or sub-committee of the Board of Directors under any applicable stock exchange listing requirements or applicable law, and that could be material to a reasonable stockholder's understanding of the independence, or lack thereof, of such proposed nominee. The notice in this paragraph must also be accompanied by (X) a completed written questionnaire (in a form provided by the Corporation) with respect to the background, qualifications, stock ownership and independence of such proposed nominee, and such additional information with respect to such proposed nominee as would be required to be provided by the Company pursuant to Schedule 14A if such proposed nominee were a participant in the solicitation of proxies by the Company in connection with such annual or special meeting and (Y) a written representation and agreement (in form provided by the Corporation) that such nominee (i) if elected as director of the Corporation, intends to serve the entire term until the next meeting at which such nominee would face re-election and (ii) consents to being named as a nominee in the Corporation's proxy statement pursuant to Rule 14a-4(d) under the Exchange Act and any associated proxy card of the Corporation and agrees to serve if elected as a director.

(ii) Other than proposals sought to be included in the corporation's proxy materials pursuant to Rule 14a-8 under the 1934 Act, for business other than nominations for the election to the Board of Directors to be properly brought before an annual meeting by a stockholder pursuant to clause (iii) of Section 5(a), the stockholder must deliver written notice to the Secretary at the principal executive offices of the corporation on a timely basis as set forth in Section 5(b)(iii), and must update and supplement such written notice on a timely basis as set forth in Section 5(c). Such stockholder's notice shall set forth: (A) as to each matter such stockholder proposes to bring before the meeting, a brief description of the business desired to be brought before the meeting, the text of the proposal or business (including the text of any resolutions proposed for consideration and in the event that such business includes a proposal to amend the Bylaws of the corporation (the "*Bylaws*"), the language of the proposed amendment), the reasons for conducting such business at the meeting, and any material interest (including any anticipated benefit of such business to any Proponent (as defined below) other than solely as a result of its ownership of the corporation's capital stock, that is material to any Proponent individually, or to the Proponents in the aggregate) in such business of any Proponent; and (B) the information required by Section 5(b)(iv).

(iii) To be timely, the written notice required by Section 5(b)(i) or 5(b)(i) must be received by the Secretary at the principal executive offices of the corporation not later than the close of business on the ninetieth (90th) day nor earlier than the close of business on the one hundred twentieth (120th) day prior to the first anniversary of the preceding year's annual meeting; *provided, however*, that, subject to the last sentence of this Section 5(b)(ii), in the event that the date of the annual meeting is advanced more than thirty (30) days prior to or delayed by more than thirty (30) days after the anniversary of the preceding year's annual meeting and (B) not later than the close of business on the later of the ninetieth (90th) day prior to such annual meeting or, if later than the ninetieth (90th) day prior to such annual meeting for which notice has been given, or the public announcement thereof has been made, commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above.

(iv) The written notice required by Section 5(b)(i) or 5(b)(i) shall also set forth, as of the date of the notice and as to the stockholder giving the notice and the beneficial owner, if any, on whose behalf the nomination or proposal is made (each, a "*Proponent*" and collectively, the "*Proponents*"): (A) the name and address of each Proponent, as they appear on the corporation's books; (B) the class or series and number of shares of each class of capital stock of the corporation that are owned of record and beneficially by each Proponent; (C) a description of any agreement, arrangement or understanding (whether oral or in writing) with respect to such nomination or proposal between or among any Proponent and any of its affiliates or associates, and any others (including their names) acting in concert, or otherwise under the agreement, arrangement or understanding, with any of the foregoing; (D) a representation that the Proponents are holders of record or beneficial owners, as the case may be, of shares of the corporation entitled to vote at the meeting and intend to appear in person or by proxy at the meeting to nominate the person or persons specified in the notice (with respect to a notice under Section 5(b)(i)); (E) a representation as to whether the Proponents intend to deliver a proxy statement and form of proxy to holders of at least 67% of the corporation's voting power of shares entitled to vote on the election of directors (with respect to a notice under Section 5(b)(i)); (F) to the extent known by any Proponent, the name and address of any other stockholder's notice; and (G) a description of all Derivative Transactions (as defined below) by each Proponent during the previous twelve (12) month period, including the date of the transactions and the class, series and number of securities involved in, and the material economic terms of, such Derivative Transactions.

(c) A stockholder providing the written notice required by Section 5(b)(i) or 5(b)(i) shall update and supplement such notice in writing, if necessary, so that the information provided or required to be provided in such notice is true and correct in all material respects as of (i) the record date for the meeting and (ii) the date that is five (5) Business Days (as defined below) prior to the meeting and, in the event of any adjournment or postponement thereof, five (5) Business Days prior to such adjourned or postponed meeting. In the case of an update and supplement pursuant to clause (i) of this Section 5(c), such update and supplement shall be received by the Secretary at the principal executive offices of the corporation not later than five (5) Business Days after the record date for the meeting. In the case of an update and supplement pursuant to clause (ii) of this Section 5(c), such update and supplement shall be received by the Secretary at the principal executive offices of the corporation not later than five the meeting, and, in the event of any adjournment or postponement thereof, two (2) Business Days prior to such adjourned or postponement thereof, two (2) Business Days prior to such adjourned or postponed meeting.

(d) Notwithstanding anything herein to the contrary, in the event that the number of directors to be elected to the Board of Directors of the corporation at the annual meeting is increased effective after the time period for which nominations would otherwise be due under Section 5(b)(iii) and there is no public announcement by the corporation naming the nominees for the additional directorships at least one hundred (100) days prior to the first anniversary of the preceding year's annual meeting, a stockholder's notice required by this Section 5 shall also be considered timely, but only with respect to nominees for the additional directorships, if it shall be delivered to the Secretary at the principal executive offices of the corporation not later than the close of business on the tenth (10th) day following the day on which such public announcement is first made by the corporation.

(e) A person shall not be eligible for election or re-election as a director at the annual meeting unless the person is nominated either in accordance with clause (ii) or clause (iii) of Section 5(a). Except as otherwise required by law, the chairperson of the annual meeting shall have the power and duty to determine whether a nomination or any business proposed to be brought before the meeting was made, or proposed, as the case may be, in accordance with the procedures set forth in these Bylaws and, if any proposed nomination or business is not in compliance with these Bylaws, or the Proponent does not act in accordance with the representations in Sections 5(b)(iv)(D) and 5(b)(iv)(E), to declare that such proposal or nomination shall not be presented for stockholder action at the meeting and shall be disregarded, notwithstanding that proxies in respect of such nomination or such business may have been solicited or received.

(f) Notwithstanding the foregoing provisions of this Section 5, in order to include information with respect to a stockholder proposal in the proxy statement and form of proxy for a stockholders' meeting, a stockholder must also comply with all applicable requirements of the 1934 Act and the rules and regulations thereunder. Nothing in these Bylaws shall be deemed to affect any rights of stockholders to request inclusion of proposals in the corporation's proxy statement pursuant to Rule 14a-8 under the 1934 Act; *provided, however*, that any references in these Bylaws to the 1934 Act or the rules and regulations thereunder are not intended to and shall not limit the requirements applicable to proposals and/or nominations to be considered pursuant to Section 5(a).

(g) For purposes of Sections 5 and 6,

"1933 Act");

(1) "affiliates" and "associates" shall have the meanings set forth in Rule 405 under the Securities Act of 1933, as amended (the

(2) "Business Day" means any day other than Saturday, Sunday or a day on which banks are closed in New York City, New York.

(3) "Derivative Transaction" means any agreement, arrangement, interest or understanding entered into by, or on behalf or for the benefit of, any Proponent or any of its affiliates or associates, whether record or beneficial: (A) the value of which is derived in whole or in part from the value of any class or series of shares or other securities of the corporation; (B) that otherwise provides any direct or indirect opportunity to gain or share in any gain derived from a change in the value of securities of the corporation; (C) the effect or intent of which is to mitigate loss, manage risk or benefit of security value or price changes; or (D) that provides the right to vote or increase or decrease the voting power of, such Proponent, or any of its affiliates or associates, with respect to any securities of the corporation, which agreement, arrangement, interest or understanding may include, without limitation, any option, warrant, debt position, note, bond, convertible security, swap, stock appreciation right, short position, profit interest, hedge, right to dividends, voting agreement, performance-related fee or arrangement to borrow or lend shares (whether or not subject to payment, settlement, exercise or conversion in any such class or series), and any proportionate interest of such Proponent in the securities of the corporation held by any general or limited partnership, or any limited liability company, of which such Proponent is, directly or indirectly, a general partner or managing member; and

(4) "*public announcement*" shall mean disclosure in a press release reported by the Dow Jones Newswires, Associated Press or comparable national news service or in a document publicly filed by the corporation with the Securities and Exchange Commission pursuant to Section 13, 14 or 15(d) of the 1934 Act or by such other means reasonably designed to inform the public or security holders in general of such information.

Section 6. Special Meetings.

(a) Special meetings of the stockholders of the corporation may be called, for any purpose as is a proper matter for stockholder action under Delaware law, by (i) the Chairperson of the Board of Directors, (ii) the Chief Executive Officer, or (iii) the Board of Directors pursuant to a resolution adopted by the Board of Directors.

(b) For a special meeting called pursuant to Section 6(a), the person(s) calling the meeting shall determine the time and place, if any, of the meeting; provided, however, that only the Board of Directors or a duly authorized committee thereof may authorize a meeting solely by means of remote communication. Upon determination of the date, time and place, if any, of the meeting, the Secretary shall cause a notice of meeting to be given to the stockholders entitled to vote, in accordance with the provisions of Section 7. No business may be transacted at a special meeting otherwise than as specified in the notice of meeting.

(c) Nominations of persons for election to the Board of Directors may be made at a special meeting of stockholders at which directors are to be elected (i) by or at the direction of the Board of Directors or a duly authorized committee thereof or (ii) by any stockholder of the corporation who is a stockholder of record or beneficial owner at the time of giving notice provided for in this paragraph, who is entitled to vote at the meeting and who delivers written notice to the Secretary of the corporation setting forth the information required by Section 5(b)(i) and the information required by Section 5(b)(iv). In the event the corporation calls a special meeting of stockholders for the purpose of electing one or more directors to the Board of Directors, any such stockholder of record or beneficial owner may nominate a person or persons (as the case may be), for election to such position(s) as specified in the corporation's notice of meeting, if written notice setting forth the information required by Section 5(b)(i) and the information required by Section 5(b)(iv) shall be received by the Secretary at the principal executive offices of the corporation not later than the close of business on the later of the ninetieth (90th) day prior to such meeting or the tenth (10th) day following the day on which the corporation first makes a public announcement of the date of the special meeting at which directors are to be elected. The stockholder shall also update and supplement such information as required has been made, commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above.

(d) A person shall not be eligible for election or re-election as a director at the special meeting unless the person is nominated either in accordance with clause (i) or clause (ii) of Section 6(c). Except as otherwise required by law, the chairperson of the special meeting shall have the power and duty to determine whether a nomination was made in accordance with the procedures set forth in these Bylaws and, if any nomination or business is not in compliance with these Bylaws, to declare that such nomination shall not be presented for stockholder action at the meeting and shall be disregarded, notwithstanding that proxies in respect of such nomination may have been solicited or received.

(e) Notwithstanding the foregoing provisions of this Section 6, a stockholder must also comply with all applicable requirements of the 1934 Act and the rules and regulations thereunder with respect to matters set forth in this Section 6. Nothing in these Bylaws shall be deemed to affect any rights of stockholders to request inclusion of proposals in the corporation's proxy statement pursuant to Rule 14a-8 under the 1934 Act; *provided, however*; that any references in these Bylaws to the 1934 Act or the rules and regulations thereunder are not intended to and shall not limit the requirements applicable to nominations for the election to the Board of Directors or proposals of other businesses to be considered pursuant to Section 6(c).

Section 7. Notice of Meetings. Except as otherwise provided by law, notice, given in writing or by electronic transmission, of each meeting of stockholders shall be given not fewer than ten (10) nor more than sixty (60) days before the date of the meeting to each stockholder entitled to vote at such meeting, such notice to specify the place, if any, date and hour, in the case of special meetings, the purpose or purposes of the meeting, the record date for determining the stockholders entitled to notice of the meeting, and the means of remote communications, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at any such meeting. If mailed, notice is given when deposited in the United States mail, postage prepaid, directed to the stockholder at such stockholder's mailing address as it appears on the records of the corporation. If sent via electronic transmission, notice is given when directed to such stockholder's electronic mail address as it appears on the records of the corporation. If sent via electronic transmission, notice is given when directed to such stockholder's electronic transmission of an objection to receiving notice by electronic mail or such notice is prohibited by Section 232(e) of the DGCL. Notice of the time, place, if any, and purpose of any meeting of stockholders (to the extent required) may be waived in writing, signed by the person entitled to notice therea of no by electronic transmission by such person, either before or after such meeting, and will be waived by any stockholder by his or her attendance thereat in person, by remote communication, if applicable, or by proxy, except when the stockholder so waiving notice of such meeting, to the transaction of any business because the meeting attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any such meeting in all respects as if due notice thereof had been given.

Section 8. Quorum; Voting. At all meetings of stockholders, except where otherwise provided by statute or by the Certificate of Incorporation, or by these Bylaws, the presence, in person, by remote communication, if applicable, or by proxy duly authorized, of the holders of a majority of the voting power of the outstanding shares of stock entitled to vote at the meeting shall constitute a quorum for the transaction of business. In the absence of a quorum, any meeting of stockholders may be adjourned, from time to time, either by the chairperson of the meeting or by vote of the holders of a majority of the voting power of the shares represented thereat and entitled to vote thereon, but no other business shall be transacted at such meeting. The stockholders present at a duly called or convened meeting, at which a quorum is present, may continue to transact business until adjournment, notwithstanding the withdrawal of enough stockholders to leave less than a quorum. Except as otherwise provided by statute or by applicable stock exchange rules, or by the Certificate of Incorporation or these Bylaws, in all matters other than the election of directors, the affirmative vote of the holders of a majority of the voting power of the shares present in person, by remote communication, if applicable, or represented by proxy duly authorized at the meeting and entitled to vote generally on the subject matter shall be the act of the stockholders. Except as otherwise provided by statute, the Certificate of Incorporation or these Bylaws, directors shall be elected by a plurality of the votes of the shares present in person, by remote communication, if applicable, or represented by proxy duly authorized at the meeting and entitled to vote generally on the election of directors. Where a separate vote by a class or classes or series is required, except where otherwise provided by statute, by applicable stock exchange rules or by the Certificate of Incorporation or these Bylaws, a majority of the voting power of the outstanding shares of such class or classes or series, present in person, by remote communication, if applicable, or represented by proxy duly authorized, shall constitute a quorum entitled to take action with respect to that vote on that matter. Except where otherwise provided by statute, by applicable stock exchange rules or by the Certificate of Incorporation or these Bylaws, the affirmative vote of the holders of a majority (plurality, in the case of the election of directors) of voting power of such class or classes or series present in person, by remote communication, if applicable, or represented by proxy at the meeting shall be the act of such class or classes or series.

Section 9. Adjournment and Notice of Adjourned Meetings. Any meeting of stockholders, whether annual or special, may be adjourned from time to time either by the person(s) who called the meeting or the chairperson of the meeting, or by the vote of the holders of a majority of the voting power of the shares present in person, by remote communication, if applicable, or represented by proxy duly authorized at the meeting and entitled to vote thereon. When a meeting is adjourned to another time or place, if any, notice need not be given of the adjourned meeting if the time and place, if any, and means of remote communications, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken. At the adjourned meeting, the corporation may transact any business that might have been transacted at the original meeting. If the adjournment is for more than thirty (30) days, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjournment a new record date for determination of stockholders entitled to vote is fixed for the adjourned meeting, the Board of Directors shall fix as the record date for determining stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote at the adjourned meeting to each stockholder of record as of the record date so fixed for notice of such adjourned meeting, and shall give notice of the adjourned meeting to each stockholder of record as of the record date so fixed for notice of such adjourned meeting.

Section 10. Voting Rights. For the purpose of determining those stockholders entitled to vote at any meeting of the stockholders, except as otherwise provided by law, only persons in whose names shares stand on the stock records of the corporation on the record date shall be entitled to vote at any meeting of stockholders. Every person entitled to vote shall have the right to do so either in person, by remote communication, if applicable, or by an agent or agents authorized by a proxy granted in accordance with Delaware law. An agent so appointed need not be a stockholder. No proxy shall be voted or acted upon after three (3) years from its date of creation unless the proxy provides for a longer period. A proxy shall be irrevocable if it states that it is irrevocable and if, and only as long as, it is coupled with an interest sufficient in law to support an irrevocable power. A stockholder may revoke any proxy which is not irrevocable by attending the meeting and voting in person or by delivering to the Secretary of the corporation a revocation of the proxy or a new proxy bearing a later date. Voting at meetings of stockholders need not be by written ballot.

Section 11. Joint Owners of Stock. If shares or other securities having voting power stand of record in the names of two (2) or more persons, whether fiduciaries, members of a partnership, joint tenants, tenants in common, tenants by the entirety, or otherwise, or if two (2) or more persons have the same fiduciary relationship respecting the same shares, unless the Secretary is given written notice to the contrary and is furnished with a copy of the instrument or order appointing them or creating the relationship wherein it is so provided, their acts with respect to voting shall have the following effect: (a) if only one (1) votes, his or her act binds all; (b) if more than one (1) votes, the act of the majority so voting binds all; and (c) if more than one (1) votes, but the vote is evenly split on any particular matter, each faction may vote the securities in question proportionally, or any person voting the shares, or a beneficiary, may apply to the Delaware Court of Chancery for relief as provided in DGCL Section 217(b). If the instrument filed with the Secretary shows that any such tenancy is held in unequal interests, a majority or even-split for the purpose of subsection (c) shall be a majority or even-split in interest.

Section 12. List of Stockholders. The corporation shall prepare, at least ten (10) days before every meeting of stockholders, a complete list of the stockholders of record entitled to vote at said meeting, arranged in alphabetical order, showing the address of each stockholder and the number and class of shares registered in the name of each stockholder. The Corporation shall not be required to include electronic mail addresses or other electronic contact information on such list. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting for a period of at least ten (10) days prior to the meeting, or (b) during ordinary business hours, either at a place within the city where the meeting is to be held, which place shall be specified in the notice of the meeting, or, if not so specified, at the principal place of business of the corporation. In the event that the corporation determines to make the list available on an electronic network, the corporation may take reasonable steps to ensure that such information is available only to stockholders of the corporation. The list shall be open to examination of any stockholder during the time of the meeting as provided by law.

Section 13. Action without Meeting. Unless otherwise provided in the Certificate of Incorporation, no action shall be taken by the stockholders of the corporation except at an annual or a special meeting of the stockholders called in accordance with these Bylaws, and no action of the stockholders of the corporation may be taken by the stockholders by written consent or electronic transmission.

Section 14. Organization.

(a) At every meeting of stockholders, the Chairperson of the Board of Directors, or, if a chairperson has not been appointed, is absent or refuses to act, the Chief Executive Officer, or, if no Chief Executive Officer is then serving, is absent or refuses to act, the President, or, if the President is absent or refuses to act, a chairperson of the meeting designated by the Board of Directors, or, if the Board of Directors does not designate such chairperson, a chairperson chosen by a majority of the voting power of the stockholders entitled to vote, present in person or by proxy duly authorized, shall act as chairperson. The Chairperson of the Board may appoint the Chief Executive Officer as chairperson of the meeting. The Secretary, or, in his or her absence, an Assistant Secretary directed to do so by the chairperson of the meeting, shall act as secretary of the meeting.

(b) The Board of Directors of the corporation shall be entitled to make such rules or regulations for the conduct of meetings of stockholders as it shall deem necessary, appropriate or convenient. Subject to such rules and regulations of the Board of Directors, if any, the chairperson of the meeting shall have the right and authority to convene and (for any or no reason) to recess and/or adjourn the meeting, to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such chairperson, are necessary, appropriate or convenient for the proper conduct of the meeting, including, without limitation, establishing an agenda or order of business for the meeting, rules and procedures for maintaining order at the meeting and the safety of those present, limitations on participation in such meeting to stockholders of record of the corporation and their duly authorized and constituted proxies and such other persons as the chairperson shall permit, restrictions on entry to the meeting after the time fixed for the commencement thereof, limitations on the time allotted to questions or comments by participants and regulation of the opening and closing of the polls for balloting on matters which are to be voted on by ballot. The date and time of the opening and closing of the polls for each matter upon which the stockholders will vote at the meeting shall be announced at the meeting. Unless and to the extent determined by the Board of Directors or the chairperson of the meeting, meetings of stockholders shall not be required to be held in accordance with rules of parliamentary procedure.

(c) The corporation shall, in advance of any meeting of stockholders, appoint one (1) or more inspectors to act at the meeting and make a written report thereof. The corporation may designate one (1) or more persons as alternate inspectors to replace any inspector who fails to act. If no inspector or alternate is able to act at a meeting of stockholders, the chairperson of the meeting shall appoint one (1) or more inspectors to act at the meeting. Each inspector, before entering upon the discharge of the duties of inspector, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of such inspector's ability. The inspectors shall: (1) ascertain the number of shares outstanding and the voting power of each; (2) determine the shares represented at a meeting and the validity of proxies and ballots; (3) count all votes and ballots; (4) determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspectors; and (5) certify their determination of the number of shares represented at the meeting, and their count of all votes and ballots. The inspectors may appoint or retain other persons or entities to assist the inspectors in the performance of the duties of the inspectors. In determining the validity and counting of proxies and ballots, the inspectors shall be limited to an examination of the proxies, any envelopes submitted with those proxies, any information provided in accordance with Sections 211(e) or 212(c)(2) of the DGCL, or any information provided pursuant to Sections 211(a)(2)b.(i) or (iii) of the DGCL, ballots and the regular books and records of the corporation, except that the inspectors may consider other reliable information for the limited purpose of reconciling proxies and ballots submitted by or on behalf of banks, brokers, their nominees or similar persons which represent more votes than the holder of a proxy is authorized by the record owner to cast or more votes than the stockholder holds of record. If the inspectors consider other reliable information for the limited purpose permitted herein, the inspectors at the time they make their certification pursuant to Section 231(b)(5) of the DGCL shall specify the precise information considered by them including the person or persons from whom they obtained the information, when the information was obtained, the means by which the information was obtained and the basis for the inspectors' belief that such information is accurate and reliable.



ARTICLE IV

DIRECTORS

Section 15. Number and Term of Office. The authorized number of directors of the corporation shall be fixed exclusively from time to time by a resolution adopted by the majority of the Board of Directors. Directors need not be stockholders unless so required by the Certificate of Incorporation. If for any cause, the directors shall not have been elected at an annual meeting, they may be elected as soon thereafter as convenient at a special meeting of the stockholders called for that purpose in the manner provided in these Bylaws, or such vacancies may be filled in accordance with Section 18 herein.

Section 16. Powers. The business and affairs of the corporation shall be managed by or under the direction of the Board of Directors, except as may be otherwise provided by statute or by the Certificate of Incorporation.

Section 17. Classes of Directors. The directors shall be divided into classes as and to the extent provided in the Certificate of Incorporation, except as otherwise required by applicable law.

Section 18. Vacancies. Vacancies on the Board of Directors shall be filled as provided in the Certificate of Incorporation, except as otherwise required by applicable law.

Section 19. Resignation. Any director may resign at any time by delivering his or her notice in writing or by electronic transmission to the Secretary, such resignation to specify whether it will be effective at a particular time. If no such specification is made, the resignation shall be effective at the time of delivery of the resignation to the Secretary.

Section 20. Removal. Subject to the rights of holders of any series of Preferred Stock (as defined in the Certificate of Incorporation) to elect additional directors or remove such directors under specified circumstances, neither the Board of Directors nor any individual director may be removed except in the manner specified in Section 141(k) of the DGCL.

Section 21. Meetings.

(a) Regular Meetings. Unless otherwise restricted by the Certificate of Incorporation, regular meetings of the Board of Directors may be held at any time or date and at any place, if any, within or without the State of Delaware which has been designated by the Board of Directors and publicized among all directors, either orally or in writing, by telephone, including a voice-messaging system or other system designed to record and communicate messages, facsimile, telegraph or telex, or by electronic mail or other electronic means. No further notice shall be required for regular meetings of the Board of Directors.

(b) Special Meetings. Unless otherwise restricted by the Certificate of Incorporation, special meetings of the Board of Directors may be held at any date, time and place, if any, within or without the State of Delaware whenever called by the Chairperson of the Board, the Chief Executive Officer or the Board of Directors.

(c) Meetings by Electronic Communications Equipment. Any member of the Board of Directors, or of any committee thereof, may participate in a meeting by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and participation in a meeting by such means shall constitute presence in person at such meeting.

(d) Notice of Special Meetings. Notice of the time and place of all special meetings of the Board of Directors shall be given orally or in writing, by telephone, including a voice messaging system or other system or technology designed to record and communicate messages, facsimile, telegraph or telex, or by electronic mail or other electronic means, during normal business hours, at least twenty-four (24) hours before the date and time of the meeting. If notice is sent by U.S. mail, it shall be sent by first class mail, postage prepaid at least five (5) days before the date of the meeting. Notice of any special meeting may be waived in writing or by electronic transmission at any time before or after the meeting and will be waived by any director by attendance thereat, except when the director attends the meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened.

(e) Waiver of Notice. The transaction of all business at any meeting of the Board of Directors, or any committee thereof, however called or noticed, or wherever held, shall be as valid as though it had been transacted at a meeting duly held after regular call and notice, if a quorum be present and if, either before or after the meeting, each of the directors not present who did not receive notice shall sign a written waiver of notice or shall waive notice by electronic transmission. All such waivers shall be filed with the corporate records or made a part of the minutes of the meeting. Notice of any meeting will be waived by any director by attendance thereat, except when the director attends the meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened.

Section 22. Quorum and Voting.

(a) Unless the Certificate of Incorporation requires a greater number, a quorum of the Board of Directors shall consist of a majority of the directors currently serving on the Board of Directors (but in no event less than one-third of the total authorized number of directors); *provided, however,* at any meeting whether a quorum be present or otherwise, a majority of the directors present may adjourn from time to time until the time fixed for the next regular meeting of the Board of Directors, without notice other than by announcement at the meeting.

(b) At each meeting of the Board of Directors at which a quorum is present, all questions and business shall be determined by the affirmative vote of a majority of the directors present, unless a different vote be required by law, the Certificate of Incorporation or these Bylaws.

Section 23. Action Without Meeting. Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting, if all members of the Board of Directors or committee, as the case may be, consent thereto in writing or by electronic transmission. The consent or consents shall be filed with the minutes of proceedings of the Board of Directors or committee, in the same paper or electronic form as the minutes are maintained.

Section 24. Fees and Compensation. Directors shall be entitled to such compensation for their services as may be approved by the Board of Directors or a committee thereof to which the Board of Directors has delegated such responsibility and authority, including, if so approved, by resolution of the Board of Directors or a committee thereof to which the Board of Directors has delegated such responsibility and authority, a fixed sum and expenses of attendance, if any, for attendance at each regular or special meeting of the Board of Directors and at any meeting of a committee of the Board of Directors. Nothing herein contained shall be construed to preclude any director from serving the corporation in any other capacity as an officer, agent, employee, or otherwise and receiving compensation therefor.

Section 25. Committees.

(a) Executive Committee. The Board of Directors may appoint an Executive Committee to consist of one or more members of the Board of Directors. The Executive Committee, to the extent permitted by law and provided in the resolution of the Board of Directors, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the corporation, and may authorize the seal of the corporation to be affixed to all papers which may require it; but no such committee shall have the power or authority in reference to (i) approving or adopting, or recommending to the stockholders, any action or matter (other than the election or removal of directors) expressly required by the DGCL to be submitted to stockholders for approval, or (ii) adopting, amending or repealing any Bylaw of the corporation.

(b) Other Committees. The Board of Directors may, from time to time, appoint such other committees as may be permitted by law. Such other committees appointed by the Board of Directors shall consist of one (1) or more members of the Board of Directors and shall have such powers and perform such duties as may be prescribed by the resolution or resolutions creating such committees, but in no event shall any such committee have the powers denied to the Executive Committee in these Bylaws.

(c) Term. The Board of Directors, subject to any requirements of any outstanding series of Preferred Stock and the provisions of subsections (a) or (b) of this Section 25, may at any time increase or decrease the number of members of a committee or terminate the existence of a committee. The membership of a committee member shall terminate on the date of his or her death or voluntary resignation from the committee or from the Board of Directors may at any time for any reason remove any individual committee member and the Board of Directors may fill any committee vacancy created by death, resignation, removal or increase in the number of members of the committee. The Board of Directors may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee, and, in addition, in the absence or disqualification of any member of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not he, she or they constitute a quorum, may unanimously appoint another member of the Board of Directors to act at the meeting in the place of any such absent or disqualified member.

(d) Meetings. Unless the Board of Directors shall otherwise provide, regular meetings of the Executive Committee or any other committee appointed pursuant to this Section 25 shall be held at such times and places, if any, as are determined by the Board of Directors, or by any such committee, and when notice thereof has been given to each member of such committee, no further notice of such regular meetings need be given thereafter. Special meetings of any such committee may be held at any place, if any, which has been determined from time to time by such committee, and may be called by any director who is a member of such committee, upon notice to the members of such committee of the time and place of such special meeting given in the manner provided for the giving of notice to members of the Board of Directors of the time and place of special meetings of the Board of Directors. Notice of any regular or special meeting of any committee may be waived in writing or by electronic transmission at any time before or after the meeting and will be waived by any director by attendance thereat, except when the director attends such regular or special meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Unless otherwise provided by the Board of Directors in the resolutions authorizing the creation of the committee, a majority of the authorized number of members of any such committee shall constitute a quorum for the transaction of business, and the act of a majority of those members of the committee present at any meeting at which a quorum is present shall be the act of such committee.



Section 26. Duties of Chairperson of the Board of Directors. The Chairperson of the Board of Directors, if appointed and when present, shall preside at all meetings of the stockholders and the Board of Directors. The Chairperson of the Board of Directors shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers, as the Board of Directors shall designate from time to time.

Section 27. Organization. At every meeting of the directors, the Chairperson of the Board of Directors, or, if a Chairperson has not been appointed or is absent, the Chief Executive Officer (if a director), or, if a Chief Executive Officer is absent, the President (if a director), or if the President is absent, the most senior Vice President (if a director), or, in the absence of any such person, a chairperson of the meeting chosen by a majority of the directors present, shall preside over the meeting. The Secretary, or in his or her absence, any Assistant Secretary or other officer, director or other person directed to do so by the person presiding over the meeting, shall act as secretary of the meeting.

Section 28. Interested Directors. No contract or transaction between the corporation and one or more of its directors or officers, or between the corporation and any other corporation, partnership, association or other organization in which one or more of its directors or officers are directors or officers or have a financial interest, shall be void or voidable solely for this reason, or solely because the director or officer is present at or participates in the meeting of the Board of Directors or committee thereof which authorizes the contract or transaction, or solely because any such director's or officer's or officer's relationship or interest and as to the contract or transaction are disclosed or are known to the Board of Directors or the committee, and the Board of Directors or a quorum; or (ii) the material facts as to the director's or officer's relationship or interest directors be less than a quorum; or (ii) the material facts as to the contract or transaction are disclosed or are known to the stockholders relationship or interest and as to the contract or transaction are disclosed or are known to the stockholders or officer's relationship or interest and as to the contract or transaction are disclosed or are known to the stockholders or transaction is specifically approved in good faith by vote of the stockholders; or (iii) the contract or transaction is fair as to the Corporation as of the time it is authorized, approved or ratified by the Board of Directors or of a committee which authorizes the contract or transaction.

ARTICLE V

OFFICERS

Section 29. Officers Designated. The officers of the corporation shall include, if and when designated by the Board of Directors, the Chief Executive Officer, the President, one or more Vice Presidents, the Secretary, the Chief Financial Officer and the Treasurer. The Board of Directors may also appoint one or more Assistant Secretaries and Assistant Treasurers and such other officers and agents with such powers and duties as it shall deem necessary. The Board of Directors may assign such additional titles to one or more of the officers as it shall deem appropriate. Any one person may hold any number of offices of the corporation at any one time unless specifically prohibited therefrom by law. The salaries and other compensation of the officers of the corporation shall be fixed in the manner required by applicable law or stock exchange rules.

Section 30. Tenure and Duties of Officers.

(a) General. All officers shall be designated and hold office at the pleasure of the Board of Directors and until their successors shall have been duly elected and qualified, or until their earlier death, resignation, retirement, disqualification or removal from office. If the office of any officer becomes vacant for any reason, the vacancy may be filled by the Board of Directors.

(b) Duties of Chief Executive Officer. Unless an officer has been appointed Chief Executive Officer of the corporation, the President shall be the chief executive officer of the corporation and shall, subject to the control of the Board of Directors, have general supervision, direction and control of the business and officers of the corporation. To the extent that a Chief Executive Officer has been appointed and no President has been appointed, all references in these Bylaws to the President shall be deemed references to the Chief Executive Officer. The Chief Executive Officer shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers, as the Board of Directors shall designate from time to time.

(c) Duties of President. Unless another officer has been appointed Chief Executive Officer of the corporation, the President shall be the chief executive officer of the corporation and shall, subject to the control of the Board of Directors, have general supervision, direction and control of the business and officers of the corporation. The President shall perform other duties commonly incident to the officer and shall also perform such other duties and have such other powers, as the Board of Directors (or the Chief Executive Officer, if the Chief Executive Officer and President are not the same person and the Board of Directors has delegated the designation of the President's duties to the Chief Executive Officer) shall designate from time to time.

(d) Duties of Vice Presidents. A Vice President may assume and perform the duties of the President in the absence or disability of the President or whenever the office of President is vacant (unless the duties of the President are being filled by the Chief Executive Officer). A Vice President shall perform other duties commonly incident to their office and shall also perform such other duties and have such other powers as the Board of Directors or the Chief Executive Officer, or, if the Chief Executive Officer has not been appointed or is absent, the President shall designate from time to time.

(e) Duties of Secretary and Assistant Secretary. The Secretary shall attend all meetings of the stockholders and of the Board of Directors and shall record all acts and proceedings thereof in the minute book of the corporation. The Secretary shall give notice in conformity with these Bylaws of all meetings of the stockholders and of all meetings of the Board of Directors and any committee thereof requiring notice. The Secretary shall perform all other duties provided for in these Bylaws and other duties commonly incident to the office and shall also perform such other duties and have such other powers, as the Board of Directors shall designate from time to time. The Chief Executive Officer, or if no Chief Executive Officer is then serving, the President may direct any Assistant Secretary or other officer to assume and perform the duties of the Secretary in the absence or disability of the Secretary, and each Assistant Secretary shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers as the Board of Directors or the Chief Executive Officer, or if no Chief Executive Officer is then serving, the President shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers as the Board of Directors or the Chief Executive Officer, or if no Chief Executive Officer is then serving, the President shall designate from time to time.

(f) Duties of Chief Financial Officer. The Chief Financial Officer shall keep or cause to be kept the books of account of the corporation in a thorough and proper manner and shall render statements of the financial affairs of the corporation in such form and as often as required by the Board of Directors or the Chief Executive Officer, or if no Chief Executive Officer is then serving, the President. The Chief Financial Officer shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers as the Board of Directors or the Chief Financial Officer is then serving, the President shall designate from time to time. To the extent that a Chief Financial Officer has been appointed and no Treasurer has been appointed, all references in these Bylaws to the Treasurer shall be deemed references to the Chief Financial Officer, or if no Chief Executive Officer to assume and perform the duties of the Chief Financial Officer in the absence or disability of the Chief Financial Officer, and each Treasurer and Assistant Treasurer and each controller and assistant controller shall perform other duties of the Chief Financial Officer is then serving, the President to the officer in the absence or disability of the Chief Financial Officer, and each Treasurer and Assistant Treasurer and each controller and assistant controller shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers as the Board of Directors or the Chief Executive Officer, or if no Chief Executive Officer is then serving, the President may direct the Treasurer, if any, or any Assistant Treasurer, or the controller or any assistant controller to assume and perform the duties of the Chief Financial Officer in the absence or disability of the Chief Financial Officer and shall also perform such other duties and have such other powers as the Board of Directors or the Chief Executive Officer, or if no Chief Executive Officer is then serving, t

(g) Duties of Treasurer and Assistant Treasurer. Unless another officer has been appointed Chief Financial Officer of the corporation, the Treasurer shall be the chief financial officer of the corporation and shall keep or cause to be kept the books of account of the corporation in a thorough and proper manner and shall render statements of the financial affairs of the corporation in such form and as often as required by the Board of Directors or the Chief Executive Officer, or if no Chief Executive Officer is then serving, the President, and, subject to the order of the Board of Directors, shall have the custody of all funds and securities of the corporation. The Treasurer shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers as the Board of Directors or the Chief Executive Officer, or if no Chief Executive Officer is then serving, the President and Chief Financial Officer (if not Treasurer) shall designate from time to time. The Chief Executive Officer, or if no Chief Executive Officer is then serving, the President and Chief Financial Officer may direct any Assistant Treasurer or the controller or any assistant controller to assume and perform the duties of the Treasurer in the absence or disability of the Treasurer, and each Assistant Treasurer and each controller and assistant controller shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers as the Board of Directors or the Chief Executive Officer, or if no Chief Executive Officer is then serving, the President and Chief Financial Officer may direct any Assistant Treasurer or the controller or any assistant controller to assume and perform the duties commonly incident to the office and shall also perform such other duties and have such other powers as the Board of Directors or the Chief Executive Officer, or if no Chief Executive Officer is then serving, the President and Chief Financial controller and assistant c

Section 31. Delegation of Authority. The Board of Directors may from time to time delegate the powers or duties of any officer to any other officer or agent, notwithstanding any provision hereof.

Section 32. Resignations. Any officer may resign at any time by giving notice in writing or by electronic transmission to the Board of Directors or to the Chief Executive Officer, or if no Chief Executive Officer is then serving, to the President or to the Secretary. Any such resignation shall be effective when received by the person or persons to whom such notice is given, unless a later time is specified therein, in which event the resignation shall become effective at such later time. Unless otherwise specified in such notice, the acceptance of any such resignation shall not be necessary to make it effective. Any resignation shall be without prejudice to the rights, if any, of the corporation under any contract with the resigning officer.

Section 33. Removal. Any officer may be removed from office at any time, either with or without cause, by the Board of Directors, or by any committee or superior officer upon whom such power of removal may have been conferred by the Board of Directors.

ARTICLE VI

EXECUTION OF CORPORATE INSTRUMENTS AND VOTING OF SECURITIES OWNED BY THE CORPORATION

Section 34. Execution of Corporate Instruments. The Board of Directors may, in its discretion, determine the method and designate the signatory officer or officers, or other person or persons, to execute on behalf of the corporation any corporate instrument or document, or to sign on behalf of the corporation the corporate name without limitation, or to enter into contracts on behalf of the corporation, except where otherwise provided by applicable law or these Bylaws, and such execution or signature shall be binding upon the corporation. All checks and drafts drawn on banks or other depositaries on funds to the credit of the corporation or in special accounts of the corporation shall be signed by such person or persons as the Board of Directors shall authorize so to do. Unless (i) authorized or ratified by the Board of Directors or (ii) within the agency power of an officer or any designee of any such officer (each, an "*Authorized Employee*"), no officer, agent or employee other than an Authorized Employee shall have any power or authority to bind the corporation by any contract or engagement or to pledge its credit or to render it liable for any purpose or for any amount.

Section 35. Voting of Securities Owned by the Corporation. All stock and other securities and interests of other corporations and entities owned or held by the corporation for itself, or for other parties in any capacity, shall be voted, and all proxies with respect thereto shall be executed, by the person authorized so to do by resolution of the Board of Directors, or, in the absence of such authorization, by the Chairperson of the Board of Directors, the Chief Executive Officer, the President, or any Vice President.

ARTICLE VII

SHARES OF STOCK

Section 36. Form and Execution of Certificates.

(a) The shares of the corporation shall be represented by certificates, or shall be uncertificated if so provided by resolution or resolutions of the Board of Directors. Certificates, if any, for the shares of stock shall be in such form as is consistent with the Certificate of Incorporation and applicable law.

(b) Every holder of stock in the corporation represented by certificate shall be entitled to have a certificate signed by, or in the name of, the corporation by any two (2) authorized officers of the corporation, certifying the number of shares owned by such holder in the corporation. Any or all of the signatures on the certificate may be facsimiles. In case any officer, transfer agent, or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent, or registrar before such certificate is issued, it may be issued with the same effect as if he or she were such officer, transfer agent, or registrar at the date of issue.

Section 37. Lost Certificates. A new certificate or certificates shall be issued in place of any certificate or certificates theretofore issued by the corporation alleged to have been lost, stolen, or destroyed, upon the making of an affidavit of that fact by the person claiming the certificate of stock to be lost, stolen, or destroyed. The corporation may require, as a condition precedent to the issuance of a new certificate or certificates, the owner of such lost, stolen, or destroyed certificate or certificates, or the owner's legal representative, to agree to indemnify the corporation in such manner as it shall require or to give the corporation a surety bond in such form and amount as it may direct as indemnity against any claim that may be made against the corporation with respect to the certificate alleged to have been lost, stolen, or destroyed.

Section 38. Transfers.

(a) Transfers of record of shares of stock of the corporation shall be made only upon its books by the holders thereof, in person or by attorney duly authorized, and, in the case of stock represented by certificate, upon the surrender of a properly endorsed certificate or certificates for a like number of shares.

(b) The corporation shall have power to enter into and perform any agreement with any number of stockholders of any one or more classes or series of stock of the corporation to restrict the transfer of shares of stock of the corporation of any one or more classes owned by such stockholders in any manner not prohibited by the DGCL.

Section 39. Fixing Record Dates.

(a) In order that the corporation may determine the stockholders entitled to notice of any meeting of stockholders or any adjournment thereof, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall, subject to applicable law, not be more than sixty (60) nor fewer than ten (10) days before the date of such meeting. If the Board of Directors so fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board of Directors determines, at the time it fixes such record date, that a later date on or before the date of the meeting shall be the date for making such determination. If no record date is fixed by the Board of Directors, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which the meeting; *provided, however*, that the Board of Directors may fix a new record date for determination of stockholders entitled to vote at the adjourned meeting, and in such case shall also fix as the record date for stockholders entitled to notice of any the adjourned meeting.

(b) In order that the corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights or the stockholders entitled to exercise any rights in respect of any change, conversion or exchange of stock, or for the purpose of any other lawful action, the Board of Directors may fix, in advance, a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than sixty (60) days prior to such action. If no record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto.

Section 40. Registered Stockholders. The corporation shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends, and to vote as such owner, and shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of Delaware.

ARTICLE VIII

OTHER SECURITIES OF THE CORPORATION

Section 41. Execution of Other Securities. All bonds, debentures and other corporate securities of the corporation, other than stock certificates (covered in Section 35), may be signed by any executive officer (as defined in Article XI) or any other officer or person as may be authorized by the Board of Directors; *provided, however*; that where any such bond, debenture or other corporate security shall be authenticated by the manual signature, or where permissible facsimile signature, of a trustee under an indenture pursuant to which such bond, debenture or other corporate security may be the imprinted facsimile of the signatures of such persons. Interest coupons appertaining to any such bond, debenture or other corporate security, authenticated by a trustee as aforesaid, shall be signature of such person. In case any officer who shall have signed or attested any bond, debenture or other corporate security, or whose facsimile signature shall appear thereon or on any such interest coupon, shall have ceased to be such officer before the bond, debenture or other corporate security nevertheless may be adopted by the corporation and issued and delivered as though the person who signed the same or whose facsimile signature shall have been used thereon had not ceased to be such officer of the corporation.

ARTICLE IX

DIVIDENDS

Section 42. Declaration of Dividends. Dividends upon the outstanding capital stock of the corporation, subject to the provisions of the Certificate of Incorporation and applicable law, if any, may be declared by the Board of Directors. Dividends may be paid in cash, in property, or in shares of the corporation's capital stock, subject to the provisions of the Certificate of Incorporation and applicable law.

Section 43. Dividend Reserve. Before payment of any dividend, there may be set aside out of any funds of the corporation available for dividends such sum or sums as the Board of Directors from time to time, in its absolute discretion, thinks proper as a reserve or reserves to meet contingencies, or for equalizing dividends, or for repairing or maintaining any property of the corporation, or for such other purpose as the Board of Directors shall think conducive to the interests of the corporation, and the Board of Directors may modify or abolish any such reserve in the manner in which it was created.

ARTICLE X

FISCAL YEAR

Section 44. Fiscal Year. The fiscal year of the corporation shall end on December 31 or on such other date as may otherwise be fixed by resolution of the Board of Directors.

ARTICLE XI

INDEMNIFICATION

Section 45. Indemnification of Directors, Executive Officers, Employees and Other Agents.

(a) Directors and Executive Officers. The corporation shall indemnify and hold harmless, to the fullest extent permitted by the DGCL as it presently exists or may hereafter be amended, any director or officer of the Corporation who was or is made or is threatened to be made a party or is otherwise involved in any Proceeding (as defined below) by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director or officer of the corporation or, while serving as a director or officer of the corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, enterprise or non-profit entity, including service with respect to employee benefit plans (each a "covered person"), against all liability and loss suffered and expenses (including attorneys' fees, judgments, fines ERISA excise taxes or penalties and amounts paid in settlement) reasonably incurred by such person in connection with any such Proceeding. Notwithstanding the preceding sentence, except as otherwise provided in Section 45(d), the corporation shall be required to indemnify a person in connection with a Proceeding initiated by such person only if the Proceeding was authorized in the specific case by the Board.

(b) Other Officers, Employees and Other Agents. The corporation shall have the power to indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any employee or agent of the corporation who was or is made or is threatened to be made a party or is otherwise involved in any Proceeding by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was an employee or agent of the Corporation or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, enterprise or non-profit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses reasonably incurred by such person in connection with any such Proceeding.

(c) Expenses. The corporation shall advance to any person who was or is a party or is threatened to be made a party to any Proceeding by reason of the fact that he or she is or was a director or executive officer of the corporation, or is or was serving at the request of the corporation as a director or executive officer of another corporation, partnership, joint venture, trust or other enterprise, prior to the final disposition of the proceeding, promptly following request therefor, all expenses incurred by any director or executive officer in connection with such Proceeding provided, however, that if the DGCL requires, an advancement of expenses incurred by a director or executive officer in his or her capacity as a director or executive officer (and not in any other capacity in which service was or is rendered by such indemnitee, including, without limitation, service to an employee benefit plan) shall be made only upon delivery to the corporation of an undertaking (hereinafter an "*undertaking*"), by or on behalf of such indemnitee, to repay all amounts so advanced if it shall ultimately be determined by final judicial decision from which there is no further right to appeal (hereinafter a "*final adjudication*") that such indemnitee is not entitled to be indemnified for such expenses under this section or otherwise.

(d) Enforcement. Without the necessity of entering into an express contract, all rights to indemnification and advances to directors and executive officers under this Bylaw shall be deemed to be contractual rights and be effective to the same extent and as if provided for in a contract between the corporation and the director or executive officer. Any right to indemnification or advances granted by this section to a director or executive officer shall be enforceable by or on behalf of the person holding such right in the Court of Chancery of the State of Delaware if (i) the claim for indemnification or advances is denied by the Board of Directors, in whole or in part, or (ii) no disposition of such claim is made within ninety (90) days of request therefor. To the extent permitted by law, the claimant in such enforcement action, if successful in whole or in part, shall be entitled to be paid also the expense of prosecuting the claim to the fullest extent permitted by law. In connection with any claim for indemnification, the corporation shall be entitled to raise as a defense to any such action that the claimant has not met the standards of conduct that make it permissible under the DGCL or any other applicable law for the corporation to indemnify the claimant for the amount claimed. In connection with any claim by an executive officer of the corporation (except in any action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that such executive officer is or was a director of the corporation) for advances, the corporation shall be entitled to raise a defense as to any such action clear and convincing evidence that such person acted in bad faith or in a manner that such person did not reasonably believe to be in or not opposed to the best interests of the corporation, or with respect to any criminal action or proceeding that such person acted without reasonable cause to believe that his or her conduct was lawful. Neither the failure of the corporation (including its Board of Directors, independent legal counsel or its stockholders) to have made a determination prior to the commencement of such action that indemnification of the claimant is proper in the circumstances because he or she has met the applicable standard of conduct set forth in the DGCL or any other applicable law, nor an actual determination by the corporation (including its Board of Directors, independent legal counsel or its stockholders) that the claimant has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that claimant has not met the applicable standard of conduct. In any suit brought by a director or executive officer to enforce a right to indemnification or to an advancement of expenses hereunder, the burden of proving that the director or executive officer is not entitled to be indemnified, or to such advancement of expenses, under this section or otherwise shall be on the corporation.



(e) Non-Exclusivity of Rights. The rights conferred on any person by this Bylaw shall not be exclusive of any other right which such person may have or hereafter acquire under any applicable statute, provision of the Certificate of Incorporation, Bylaws, agreement, vote of stockholders or disinterested directors or otherwise, both as to action in his or her official capacity and as to action in another capacity while holding office. The corporation is specifically authorized to enter into individual contracts with any or all of its directors, officers, employees or agents respecting indemnification and advances, to the fullest extent not prohibited by the DGCL, or by any other applicable law.

(f) Survival of Rights. The rights conferred on any person by this Bylaw shall continue as to a person who has ceased to be a director or executive officer or officer, employee or other agent and shall inure to the benefit of the heirs, executors and administrators of such a person.

(g) Insurance. To the fullest extent permitted by the DGCL or any other applicable law, the corporation, upon approval by the Board of Directors, may purchase insurance on behalf of any person required or permitted to be indemnified pursuant to this section.

(h) Amendments. Any amendment, repeal or modification of this section shall only be prospective and shall not affect the rights under this Bylaw in effect at the time of the alleged occurrence of any action or omission to act that is the cause of any proceeding against any agent of the corporation.

(i) Saving Clause. If this Bylaw or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the corporation shall nevertheless indemnify each director and executive officer to the full extent not prohibited by any applicable portion of this section that shall not have been invalidated, or by any other applicable law. If this section shall be invalid due to the application of the indemnification provisions of another jurisdiction, then the corporation shall indemnify each director and executive officer to the full extent under any other applicable law.

(j) Certain Definitions. For the purposes of this Bylaw, the following definitions shall apply:

(i) The term "*Proceeding*" shall be broadly construed and shall include, without limitation, the investigation, preparation, prosecution, defense, settlement, arbitration and appeal of, and the giving of testimony in, any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative.

(ii) The term "*expenses*" shall be broadly construed and shall include, without limitation, court costs, attorneys' fees, witness fees, fines, amounts paid in settlement or judgment and any other costs and expenses of any nature or kind incurred in connection with any proceeding.

(iii) The term the "*corporation*" shall include, in addition to the resulting corporation, any constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, and employees or agents, so that any person who is or was a director, officer, employee or agent of such constituent corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, shall stand in the same position under the provisions of this section with respect to the resulting or surviving corporation as he or she would have with respect to such constituent corporation if its separate existence had continued.

(iv) References to a "*director*," "*executive officer*," "*officer*," "*employee*," or "*agent*" of the corporation shall include, without limitation, situations where such person is serving at the request of the corporation as, respectively, a director, executive officer, officer, employee, trustee or agent of another corporation, partnership, joint venture, trust or other enterprise.

(v) References to "other enterprises" shall include employee benefit plans; references to "fines" shall include any excise taxes assessed on a person with respect to an employee benefit plan; and references to "serving at the request of the corporation" shall include any service as a director, officer, employee or agent of the corporation which imposes duties on, or involves services by, such director, officer, employee, or agent with respect to an employee benefit plan, its participants, or beneficiaries; and a person who acted in good faith and in a manner such person reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interests of the corporation" as referred to in this section.

ARTICLE XII

NOTICES

Section 46. Notices.

(a) Notice to Stockholders. Notice to stockholders of stockholder meetings shall be given as provided in Section 7 herein. Without limiting the manner by which notice may otherwise be given effectively to stockholders, including under any agreement or contract with such stockholder, subject to Section 232(e) of the DGCL, any written notice to stockholders given by the corporation under any provision of the DGCL, the Certificate of Incorporation or the Bylaws shall be effective if given by a form of electronic transmission consented to by the stockholder to whom the notice is given. Any such consent shall be revocable by the stockholder by written notice or electronic transmission to the corporation. Notice shall be deemed given pursuant to this Section 45, (1) if by facsimile telecommunication, when directed to a number at which the stockholder has consented to receive notice; (2) if by a posting on an electronic network together with separate notice to the stockholder of such specific posting, upon the later of (a) such posting, and (b) the giving of such separate notice; and (3) if by any other form of electronic transmission, when directed to the stockholder. For purposes of these Bylaws, (1) "Electronic transmission" means any form of communication, not directly involving the physical transmission of paper, including the use of, or participation in, one or more electronic networks or databases (including one or more distributed electronic networks or databases), that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process; (2) "Electronic mail" means an electronic transmission directed to a unique electronic mail address (which electronic mail shall be deemed to include any files attached thereto and any information hyperlinked to a website if such electronic mail includes the contact information of an officer or agent of the corporation who is available to assist with accessing such files and information); and (3) "Electronic mail address" means a destination, commonly expressed as a string of characters, consisting of a unique user name or mailbox (commonly referred to as the "local part" of the address) and a reference to an internet domain (commonly referred to as the "domain part" of the address), whether or not displayed, to which electronic mail can be sent or delivered.

(b) Notice to Directors. Any notice required to be given to any director may be given by the method stated in subsection (a) or as otherwise provided in these Bylaws, with notice other than one which is delivered personally to be sent to such address as such director shall have filed in writing with the Secretary, or, in the absence of such filing, to the last known address of such director.

(c) Affidavit of Mailing. An affidavit of mailing, executed by a duly authorized and competent employee of the corporation or its transfer agent appointed with respect to the class of stock affected or other agent, specifying the name and address or the names and addresses of the stockholder or stockholders, or director or directors, to whom any such notice or notices was or were given, and the time and method of giving the same, shall in the absence of fraud, be prima facie evidence of the facts therein contained.

(d) Methods of Notice. It shall not be necessary that the same method of giving notice be employed in respect of all recipients of notice, but one permissible method may be employed in respect of any one or more, and any other permissible method or methods may be employed in respect of any other or others.

(e) Notice to Person with Whom Communication is Unlawful. Whenever notice is required to be given, under any provision of law or of the Certificate of Incorporation or Bylaws, to any person with whom communication is unlawful, the giving of such notice to such person shall not be required and there shall be no duty to apply to any governmental authority or agency for a license or permit to give such notice to such person. Any action or meeting which shall be taken or held without notice to any such person with whom communication is unlawful shall have the same force and effect as if such notice had been duly given. In the event that the action taken by the corporation is such as to require the filing of a certificate under any provision of the DGCL, the certificate shall state, if such is the fact and if notice is required, that notice was given to all persons entitled to receive notice except such persons with whom communication is unlawful.

(f) Notice to Stockholders Sharing an Address. Except as otherwise prohibited under the DGCL, any notice given under the provisions of the DGCL, the Certificate of Incorporation or the Bylaws shall be effective if given by a single written notice to stockholders who share an address if consented to by the stockholders at that address to whom such notice is given. Such consent shall have been deemed to have been given if such stockholder fails to object in writing to the corporation within sixty (60) days of having been given notice by the corporation of its intention to send the single notice. Any consent shall be revocable by the stockholder by written notice to the corporation.

ARTICLE XIII

AMENDMENTS

Section 47. Amendments. Subject to the limitations set forth in Section 45(h) of these Bylaws or the provisions of the Certificate of Incorporation, the Board of Directors is expressly empowered to adopt, amend or repeal the Bylaws. Any adoption, amendment or repeal of the Bylaws by the Board of Directors shall require the approval of a majority of the authorized number of directors. The stockholders also shall have power to adopt, amend or repeal the Bylaws; provided, however, that, in addition to any vote of the holders of any class or series of stock of the corporation required by law or by the Certificate of Incorporation, such action by stockholders shall require the affirmative vote of the holders of at least sixty- six and two-thirds percent (66 2/3%) of the voting power of all of the then-outstanding shares of the capital stock of the corporation entitled to vote generally in the election of directors, voting together as a single class.

ARTICLE XIV

LOANS TO OFFICERS

Section 48. Loans to Officers. Except as otherwise prohibited by applicable law, the corporation may lend money to, or guarantee any obligation of, or otherwise assist any officer or other employee of the corporation or of its subsidiaries, including any officer or employee who is a director of the corporation or its subsidiaries, whenever, in the judgment of the Board of Directors, such loan, guarantee or assistance may reasonably be expected to benefit the corporation. The loan, guarantee or other assistance may be with or without interest and may be unsecured, or secured in such manner as the Board of Directors shall approve, including, without limitation, a pledge of shares of stock of the corporation. Nothing in these Bylaws shall be deemed to deny, limit or restrict the powers of guaranty or warranty of the corporation at common law or under any statute.

ARTICLE XV

BOOKS AND RECORDS

Section 49. Books and Records. The books and records of the corporation may be kept within or outside the State of Delaware at such place or places as may from time to time be designated by the Board of Directors. Any books or records maintained by the corporation in the regular course of its business, including its stock ledger, books of account, and minute books, may be kept on, or by means of, or be in the form of, any information storage device or method; provided, however, that the books and records so kept can be converted into clearly legible paper form within a reasonable time. The corporation shall so convert any books or records so kept upon the request of any person entitled to inspect such records pursuant to the Certificate of Incorporation, these Bylaws or the DGCL.

THIS WARRANT AND THE COMMON STOCK ISSUABLE UPON EXERCISE OF THIS WARRANT HAVE BEEN ACQUIRED FOR INVESTMENT PURPOSES ONLY AND MAY NOT BE TRANSFERRED UNTIL (i) A REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "<u>1933 ACT</u>") SHALL HAVE BECOME EFFECTIVE WITH RESPECT THERETO OR (ii) RECEIPT BY THE COMPANY OF AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE COMPANY TO THE EFFECT THAT REGISTRATION UNDER THE 1933 ACT IS NOT REQUIRED IN CONNECTION WITH SUCH PROPOSED TRANSFER NOR IS SUCH TRANSFER IN VIOLATION OF ANY APPLICABLE STATE SECURITIES LAWS. THIS LEGEND SHALL BE ENDORSED UPON ANY WARRANT ISSUED IN EXCHANGE FOR THIS WARRANT OR ANY COMMON STOCK ISSUABLE UPON EXERCISE OF THIS WARRANT.

AMENDED AND RESTATED WARRANT TO PURCHASE

COMMON STOCK

OF

ORCHESTRA BIOMED HOLDINGS, INC.

January 25, 2023

W-08062020-1

This is to Certify That, FOR VALUE RECEIVED, HSAC 2 HOLDINGS, LLC, or its assigns ("Holder"), is entitled to purchase, subject to the provisions of this Amended and Restated Warrant to Purchase 1,500,000 (subject to Section (a) immediately below) fully paid, validly issued and nonassessable shares of Common Stock, par value of \$0.0001 per share ("Common Stock") of ORCHESTRA BIOMED HOLDINGS, INC., a Delaware Corporation (the "Company"), at a price of \$11.50 per share (this "Warrant") and supersedes and replaces the original Warrant, dated August 6, 2020, to purchase 1,500,000 fully paid, validly issued and nonassessable ordinary shares, par value \$0.0001 per share (the "Ordinary Shares") from the Company's predecessor, Health Sciences Acquisitions Corporation 2, a former Cayman Islands exempted company (the "Predecessor") at a price of \$11.50 per share. The number of shares of Common Stock to be received upon the exercise of this Warrant and the price to be paid for each share of Common Stock may be adjusted from time to time as hereinafter set forth. The Common Stock deliverable upon such exercise, as adjusted from time to time, are hereinafter sometimes referred to as "Warrant Shares," and the exercise price for an Ordinary Share in effect at any time, as adjusted from time to time, is hereinafter sometimes referred to as the "Exercise Price."

(a) EXERCISE OF WARRANT. This Warrant may be exercised in whole or in part at any time on the later of (i) 30 days after the consummation by the Predecessor of its initial merger, share exchange, asset acquisition, share purchase, reorganization or other similar business combination with one or more businesses or entities (the "<u>Business Combination</u>") (as described more fully in the Registration Statement on Form S-1 (No. 333-239922) as filed with by the Predecessor with the U.S. Securities and Exchange Commission or (ii) 12 months from the closing of the public offering of the Predecessor's Ordinary Shares, and terminating at 5:00 p.m., New York City time on five years from the consummation of the Business Combination (the "<u>Expiration Date</u>"); provided however, that immediately prior to the closing of the merger contemplated by that certain Agreement and Plan of Merger, dated as of July 4, 2022 (as amended, the "<u>Merger Agreement</u>") by and among the Predecessor, HSAC Olympus Merger Sub, Inc., a Delaware corporation and wholly owned subsidiary of Predecessor, and Orchestra BioMed, Inc., a Delaware corporation, and in accordance with that certain Amended and Restated Support Agreement, dated as of November 21, 2022, by and among the Predecessor, Orchestra and the holders set for the therein, without any further action by any other person, Holder shall automatically forfeit for no consideration 750,000 Warrant Shares subject to this Warrant, and this Warrant shall thereafter be exercisable for 750,000 Warrant Shares, as adjusted from time to time. Any portion of this Warrant not exercised on or before the Expiration Date shall become void, and all rights thereunder and all rights in respect thereof under this Agreement shall cease at the close of business on the Expiration Date The Company in its sole discretion may extend the duration of the Warrants by delaying the Expiration Date; provided, however, that the Company will provide at least twenty (20) days' prior written notice of any such extension to registe

(1) This Warrant may be exercised by presentation and surrender hereof to the Company at its principal office with the Purchase Form annexed hereto (the "<u>Purchase Form</u>") duly executed and accompanied by payment of the Exercise Price for the number of Warrant Shares specified in such Purchase Form (which may take the form of a "cashless exercise" pursuant to <u>Section (a)(2)</u> if so indicated in the Purchase Form).

(2) The Holder shall pay the Exercise Price in immediately available funds; provided, however, that the Holder may, in the Holder's sole discretion, satisfy its obligation to pay the Exercise Price through a "cashless exercise", in which event the Company shall issue to the Holder the number of Warrant Shares determined as follows:

X = Y(A-B)/A

where

X = the number of Warrant Shares to be issued to the Holder.

Y = the total number of Warrant Shares with respect to which this Warrant is being exercised.

A = the Fair Market Value (as defined below) of one share of Common Stock on the trading day immediately preceding the date on which Holder elects to exercise this Warrant by means of a "cashless exercise."

B = the Exercise Price then in effect for the applicable Warrant Shares at the time of such exercise.

For purposes of this Warrant, "Fair Market Value" means, for any security as of any date, the price determined by the first of the following clauses that applies: (a) if the Common Stock is then listed on a national securities exchange, the daily volume weighted average price of the Common Stock for such date (or the nearest preceding date) on the trading market on which the Common Stock are then listed as reported by Bloomberg L.P. (based on a Trading Day from 9:30 a.m. (New York City time) to 4:00 p.m. (New York City time)), (b) if the Common Stock are quoted on the OTC Bulletin Board or the OTC Market, the average closing bid price on such market for the five most recently completed trading days, (c) if paragraphs (a) or (b) are not applicable, if an appraiser hired by the Company has provided a report on the fair market value of a shares of Common Stock within the 12-month period preceding the date on which Holder elects to exercise this Warrant by means of a "cashless exercise," the fair market value of a share of Common Stock as determined by such appraiser, or (d) if none of the foregoing is applicable, the price determined by the Board of Directors of the Company in good faith.

(b) EFFECTIVE TIME OF EXERCISE. Each exercise of this Warrant shall be deemed to have been effected immediately prior to the close of business on the day on which the Purchase Form has been delivered to the Company (the "<u>Exercise Date</u>") as provided in <u>Section (a)</u>. At such time, the person or persons in whose name or names any certificates for Warrant Shares shall be issuable upon such exercise as provided in <u>Section (c)</u> below shall be deemed to have become the holder or holders of record of the Warrant Shares represented by such certificates. Notwithstanding the foregoing, if the Holder or any assignee does not enter into the Shareholders' Agreement within three (3) business days of being requested to do so by the Company, then the exercise of this Warrant will be deemed to not have been effective and void ab initio and the Holder or any assignee shall return any securities received by them pursuant to <u>Section (c)</u> to the Company.

(c) DELIVERY TO HOLDER.

(1) As soon as practicable after the exercise of this Warrant in whole or in part, and in any event within five (5) business days thereafter, the Company will cause to be issued in the name of, and delivered to, the Holder, or as such Holder (upon payment by such Holder of any applicable transfer taxes) may direct:

(A) a certificate or certificates for the number of Warrant Shares to which such Holder shall be entitled, and

(B) in case such exercise is in part only, a new warrant or warrants of like tenor, exercisable for in the aggregate the number of shares of Common Stock equal (giving effect to any adjustment therein) to the number of shares of Common Stock called for on the face of this Warrant minus the number of such shares purchased by the Holder upon such exercise.

(2) To the extent permitted by law and except as provided in this Warrant, the Company's obligations to issue and deliver Warrant Shares in accordance with the terms hereof are absolute and unconditional, irrespective of any action or inaction by the Holder to enforce the same, any waiver or consent with respect to any provision hereof, the recovery of any judgment against any person or entity or any action to enforce the same, or any setoff, counterclaim, recoupment, limitation or termination, or any breach or alleged breach by the Holder or any other person or entity of any obligation to the Company or any violation or alleged violation of law by the Holder or any other person or entity, and irrespective of any other circumstance that might otherwise limit such obligation of the Company to the Holder in connection with the issuance of Warrant Shares. Nothing herein shall limit the Holder's right to pursue any other remedies available to it hereunder, at law or in equity including, without limitation, a decree of specific performance and/or injunctive relief with respect to the Company's failure to timely deliver certificates representing Warrant Shares upon exercise of the Warrant as required pursuant to the terms hereof.

(d) RESERVATION OF SHARES. The Company shall at all times reserve for issuance and/or delivery upon exercise of this Warrant such number of shares of Common Stock (as adjusted pursuant to the terms hereof) as shall be required for issuance and delivery upon exercise of this Warrant. The Company covenants that all Warrant Shares so issuable and deliverable shall, upon issuance and the payment of the applicable Exercise Price in accordance with the terms hereof, be duly and validly authorized, issued and fully paid and nonassessable. The Company will take all such action as may be reasonably necessary to assure that such Warrant Shares may be issued as provided herein without violation of any applicable law or regulation, or of any requirements of any securities exchange or automated quotation system upon which the Common Stock may be listed.

(e) FRACTIONAL SHARES. No fractional shares or scrips representing fractional shares shall be issued upon the exercise of this Warrant. With respect to any fraction of a share called for upon any exercise hereof, the Company shall pay to the Holder an amount in cash equal to such fraction multiplied by the fair market value of a share of Common Stock.

(f) LOSS OR DESTRUCTION OF WARRANT. Upon receipt by the Company of evidence satisfactory to it of the loss, theft, destruction or mutilation of this Warrant, and (in the case of loss, theft or destruction) of reasonably satisfactory indemnification, and upon surrender and cancellation of this Warrant, if mutilated, the Company will execute and deliver a new Warrant of like tenor and date.

(g) RIGHTS OF THE HOLDER. The Holder shall not, by virtue hereof, be entitled to any rights of a shareholder in the Company, either at law or equity, and the rights of the Holder are limited to those expressed in this Warrant and are not enforceable against the Company except to the extent set forth herein.

(h) CERTAIN ADJUSTMENTS. The Exercise Price and number of Warrant Shares issuable upon exercise of this Warrant are subject to adjustment from time to time as set forth in this Section (h).

(1) Share Capitalizations and Sub-Divisions. If the Company, at any time while this Warrant is outstanding, (i) declares a share capitalization on its Common Stock or otherwise makes a distribution on any class of shares that is payable in Common Stock, (ii) subdivides its outstanding Common Stock into a larger number of shares, or (iii) combines its outstanding Common Stock into a smaller number of shares, then, in each such case, the Exercise Price shall be multiplied by a fraction, the numerator of which shall be the number Common Stock outstanding immediately before such event and the denominator of which shall be the number of Common Stock outstanding immediately after such event. Any adjustment made pursuant to this Section (h) (1) shall become effective immediately after the record date for the determination of shareholders entitled to receive such dividend or distribution, and any adjustment pursuant to clause (ii) or (iii) of this Section (h)(1) shall become effective immediately after the effective date of such subdivision or combination.

(2) Number of Warrant Shares. Simultaneously with any adjustment to the Exercise Price pursuant to $\underline{\text{Section (h)(1)}}$, the number of Warrant Shares that may be purchased upon exercise of this Warrant shall be increased or decreased proportionately, so that after such adjustment the aggregate Exercise Price payable hereunder for the increased or decreased number of Warrant Shares shall be the same as the aggregate Exercise Price in effect immediately prior to such adjustment.

(3) Fundamental Transactions. If, at any time while this Warrant is outstanding (i) the Company effects any merger or consolidation of the Company with or into another Person, (ii) the Company effects any sale of all or substantially all of its assets or a majority of its Common Stock is acquired by a third party, in each case, in one or a series of related transactions, (iii) any tender offer or exchange offer (whether by the Company or a third party that is conducting such an offer pursuant to an agreement or arrangement with the Company) is completed pursuant to which all or substantially all of the holders of Common Stock are permitted to tender or exchange their shares for other securities, cash or property, or (iv) the Company effects any reorganization or reclassification of the Common Stock or any compulsory share exchange pursuant to which the shares of Common Stock are effectively converted into or exchanged for other securities, cash or property (other than as a result of a subdivision or combination of Common Stock covered by Section (h)(1) above) (in any such case, a "Fundamental Transaction"), then the Holder shall have the right thereafter to receive, upon exercise of this Warrant, the same amount and kind of securities, cash or property as it would have been entitled to receive upon the occurrence of such Fundamental Transaction if it had been, immediately prior to such Fundamental Transaction, the holder of the number of Warrant Shares then issuable upon exercise in full of this Warrant without regard to any limitations on exercise contained herein (the "Alternate Consideration"). The Company shall not effect any such company shall not effect any such consumption or purchasing or otherwise acquiring such assets or other appropriate corporation or Person shall assume the obligation to deliver to the Holder, such Alternate Consideration as, in accordance with the foregoing provisions, the Holder may be entitled to receive, and the other obligations under this Warrant. The provisions of this <u>Sec</u>

(4) Notice of Adjustments. Upon the occurrence of each adjustment pursuant to this <u>Section (h)</u>, the Company at its expense will, at the written request of the Holder, promptly compute such adjustment, in good faith, in accordance with the terms of this Warrant and prepare a certificate setting forth such adjustment, including a statement of the adjusted Exercise Price and adjusted number or type of Warrant Shares or other property issuable upon exercise of this Warrant (as applicable), describing the transactions giving rise to such adjustments and showing in detail the facts upon which such adjustment is based. Upon written request, the Company will promptly deliver a copy of each such certificate to the Holder and to the Company's transfer agent.

(5) Notices To Warrant Holders. So long as this Warrant shall be outstanding, (i) if the Company shall pay any dividend or make any distribution of cash, securities or other property in respect of its Common Stock or (ii) if the Company shall offer to the holders of Common Stock for subscription or purchase by them any share of any class or any other rights or (iii) if any capital reorganization of the Company, reclassification of the shares of the Company consolidation or merger of the Company with or into another corporation, sale, lease or transfer of all or substantially all of the property and assets of the Company to another corporation, if the Company authorizes or approves, enters into any agreement contemplating or solicits shareholder approval for any Fundamental Transaction (each as defined below), or voluntary or involuntary dissolution, liquidation or winding up of the Company shall be effected, then in any such case, the Company shall cause to be mailed to the Holder, at least fifteen days prior the date specified in (x) or (y) below, as the case may be, a notice containing a brief description of the proposed action and stating the date on which (x) a record is to be taken for the purpose of such dividend, distribution or rights, or (y) such reclassification, reorganization, consolidation, merger, conveyance, lease, Fundamental Transaction, sales or issuances, dissolution, liquidation or winding up is to take place and the date, if any is to be fixed, as of which the holders of Common Stock or other securities shall receive cash or other property deliverable upon such reclassification, reorganization, consolidation, merger, conveyance, lease, Fundamental Transaction, sales or issuances, dissolution, liquidation or winding up.

(i) NOTICES. Any notice or request hereunder shall be in writing and may be given only by, and shall be deemed to have been received upon: (a) registered or certified mail, return receipt requested, on the date on which such notice or request is received as indicated in such return receipt; (b) delivery by a nationally recognized overnight courier, one business day after deposit with such courier; or (c) facsimile or other electronic transmission upon telephone or further electronic communication from the recipient acknowledging receipt (whether automatic or manual from recipient) of such facsimile or other electronic transmission. In the case of the Company, such notices and communications shall be addressed to Orchestra BioMed Holdings, Inc., 150 Union Square Drive, New Hope, PA 18938, Attention: Secretary, unless the Company shall notify the Holder that notices and communications should be sent to a different address (or facsimile number or electronic mail address), in which case such notices and communications shall be addressed to its address as set forth in the signature page hereto, unless the Holder shall notify the Company that notices and communications should be sent to a different address (or facsimile number or electronic mail address), in which case such notices and communications should be sent to a different address (or facsimile number or electronic mail address), in which case such notices and communications should be sent to a different address (or facsimile number or electronic mail address), in which case such notices and communications should be sent to a different address (or facsimile number or electronic mail address), in which case such notices and communications should be sent to a different address (or facsimile number or electronic mail address), in which case such notices and communications should be sent to a different address (or facsimile number or electronic mail address), in which case such notices and communications should be sent to a different address (or facsimile number o

(j) NO NET-CASH SETTLEMENT. Except as otherwise provided herein, in no event will the Holder be entitled to receive a net-cash settlement or other consideration in lieu of physical settlement in securities.

(k) MODIFICATION OF AGREEMENT. The provisions of this Warrant may from time to time be amended, modified or waived, by the Company and the holder of this Warrant.

(1) CHARGES, TAXES AND EXPENSES. Issuance and delivery of a reasonable number of certificates representing Common Stock upon exercise of this Warrant shall be made without charge to the Holder for any issue or transfer tax, transfer agent fee or other incidental tax or expense in respect of the issuance of such certificates, all of which taxes and expenses shall be paid by the Company; provided, however, that the Company shall not be required to pay any tax that may be payable in respect of any transfer involved in the registration of any certificates for Warrant Shares or the Warrants in a name other than that of the Holder or an affiliate thereof. The Holder shall be responsible for all other tax liability that may arise as a result of holding or transferring this Warrant or receiving Warrant Shares upon exercise hereof.

(m) SUCCESSORS AND ASSIGNS. This Warrant and the rights of the Holder hereunder may not be transferred and/or assigned by the Holder in any way whatsoever, and no transaction in respect thereof shall be made, either for consideration or for no consideration. This Warrant may not be assigned by the Company without the written consent of the Holder except to a successor in the event of a Fundamental Transaction. This Warrant shall be binding on and inure to the benefit of the parties hereto and the Company's successors. Subject to the preceding sentence, nothing in this Warrant shall be construed to give to any person or entity other than the Company and the Holder any legal or equitable right, remedy or cause of action under this Warrant. This Warrant may be amended only in writing signed by the Company and the Holder, or the Company's successors.

(n) GOVERNING LAW.

THIS WARRANT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

ANY LEGAL ACTION OR PROCEEDING WITH RESPECT TO THIS WARRANT SHALL BE BROUGHT IN THE COURTS OF THE STATE OF NEW YORK OR OF THE UNITED STATES OF AMERICA FOR THE SOUTHERN DISTRICT OF NEW YORK, AND, BY EXECUTION AND DELIVERY OF THIS WARRANT, EACH PARTY HEREBY ACCEPTS FOR ITSELF AND (TO THE EXTENT PERMITTED BY LAW) IN RESPECT OF ITS PROPERTY, GENERALLY AND UNCONDITIONALLY, THE JURISDICTION OF THE AFORESAID COURTS. EACH PARTY HEREBY IRREVOCABLY WAIVES ANY OBJECTION, INCLUDING, WITHOUT LIMITATION, ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS, WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY SUCH ACTION OR PROCEEDING IN SUCH RESPECTIVE JURISDICTIONS. THIS SUBMISSION TO JURISDICTION IS NON-EXCLUSIVE AND DOES NOT PRECLUDE A PARTY FROM OBTAINING JURISDICTION OVER ANOTHER PARTY IN ANY COURT OTHERWISE HAVING JURISDICTION.

EACH PARTY IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS OF ANY OF THE AFOREMENTIONED COURTS IN ANY SUCH ACTION OR PROCEEDING BY THE MAILING OF COPIES THEREOF BY REGISTERED OR CERTIFIED MAIL, POSTAGE PREPAID, TO IT AT THE ADDRESS SPECIFIED HEREIN OR SUCH OTHER ADDRESS AS IS SPECIFIED PURSUANT TO THE TERMS HEREOF (OR ITS ASSIGNMENT AND ASSUMPTION), SUCH SERVICE TO BECOME EFFECTIVE 30 DAYS AFTER SUCH MAILING. NOTHING HEREIN SHALL AFFECT THE RIGHT OF A PARTY OR ANY HOLDER TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR TO COMMENCE LEGAL PROCEEDINGS OR OTHERWISE PROCEED AGAINST ANOTHER PARTY IN ANY OTHER JURISDICTION.

[remainder of page intentionally left blank]

IN WITNESS WHEREOF, the Company has caused this Warrant to be duly executed as of the date of this Warrant.

ORCHESTRA BIOMED HOLDINGS, INC.

By:/s/ Roderick Wong, M.D.Name:Roderick Wong, M.D.Title:Chairman and Chief Executive Officer

Holder:

Accepted and Agreed:

HSAC 2 HOLDINGS, LLC

By: /s/ Alice Lee

Name: Alice Lee Title: Director

PURCHASE FORM

Dated

- (1) The undersigned hereby irrevocably elects to exercise the within Warrant to the extent of purchasing shares of Common Stock of Orchestra BioMed Holdings, Inc. (or such number of shares of Common Stock or other securities or property to which the undersigned is entitled in lieu thereof or in addition thereto under the provisions of the Warrant).
- (2) (a) The undersigned hereby elects to make payment with the enclosed bank draft, certified check or money order payable to the Company in payment of the exercise price determined under, and on the terms specified in, the Warrant, or
 - (b) The undersigned hereby elects to make payment on a cashless basis.
- (3) The undersigned hereby irrevocably directs that the said shares be issued and delivered as follows:

		Number of Shares of Common Stock (net	
		of any Common Stock	
		used to exercise on a	
Name(s) in Full	Address(es)	cashless basis	S.S. or IRS #

(4) If the Warrant was not exercised in full, please check the following:

The undersigned hereby irrevocably directs that any remaining portion of the warrant be issued and delivered as follows:

Name(s) in Full	Address(es)	Number of Shares	S.S. or IRS #
-----------------	-------------	------------------	---------------

Signature of Holder

Print Name

AMENDED AND RESTATED REGISTRATION RIGHTS AND LOCK-UP AGREEMENT

THIS AMENDED AND RESTATED REGISTRATION RIGHTS AND LOCK-UP AGREEMENT (this "*Agreement*"), dated as of January 26, 2022, is made and entered into by and among, (i) Health Sciences Acquisitions Corporation 2, a Delaware corporation (the "*Company*"), (ii) the equityholders designated as Sponsor Equityholders on <u>Schedule A</u> hereto (collectively, the "*Sponsor Equityholders*"); and (iii) certain stockholders of Orchestra BioMed, Inc. designated as Legacy Orchestra Equityholders on <u>Schedule B</u> hereto (the "*Legacy Orchestra Equityholders*" and, together with the Sponsor Equityholders and any person or entity who hereafter becomes a party to this Agreement pursuant to Section 6.2 of this Agreement, the "*Holders*" and each individually a "*Holder*"). Capitalized terms used but not otherwise defined in this Agreement shall have the meanings ascribed to such terms in the Merger Agreement (as defined below).

RECITALS

WHEREAS, the Company and the Sponsor Equityholders are parties to that certain Registration Rights Agreement, entered into as of August 3, 2020 (the "*Prior Agreement*");

WHEREAS, the Company, HSAC Olympus Merger Sub, Inc., a Delaware corporation ("*Merger Sub*"), and Orchestra BioMed, Inc., a Delaware corporation ("*Legacy Orchestra*"), are party to that certain Agreement and Plan of Merger, dated as of July 4, 2022 (as amended or restated from time to time, the "*Merger Agreement*"), pursuant to which the Company changed its jurisdiction of incorporation from the Cayman Islands to the State of Delaware (the "*Domestication*") and on the date hereof: (a) Merger Sub will merge (the "*Merger*") with and into Legacy Orchestra, with Legacy Orchestra surviving the Merger as a wholly owned subsidiary of the Company, and (b) the Company will change its name to "Orchestra BioMed, Inc.";

WHEREAS, the Legacy Orchestra Equityholders and the Sponsor Equityholders are receiving shares of Common Stock, par value \$0.0001 per share, of the Company (the "*Common Stock*") on or about the date hereof, pursuant to the Merger Agreement (the "*Merger Shares*");

WHEREAS, in connection with the consummation of the Merger, the parties to the Prior Agreement desire to amend and restate the Prior Agreement in its entirety as set forth herein, and the parties hereto desire to enter into this Agreement pursuant to which the Company shall grant the Holders certain registration rights with respect to the Registrable Securities (as defined below) on the terms and conditions set forth in this Agreement;

WHEREAS, pursuant to Section 6.7 of the Prior Agreement, no amendment, modification or termination of the Prior Agreement shall be binding upon any party unless executed in writing by such party; and

WHEREAS, all of the parties to the Prior Agreement desire to amend and restate the Prior Agreement in its entirety and enter into this Agreement, pursuant to which the Company shall grant the Holders certain registration rights with respect to certain securities of the Company, as set forth in this Agreement, effective as of the Closing.

NOW, THEREFORE, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

1.1 Definitions. The terms defined in this Article I shall, for all purposes of this Agreement, have the respective meanings set forth below:

"Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Chief Executive Officer or Chief Financial Officer of the Company, after consultation with counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, declared effective or used, as the case may be, and (iii) the Company has a bona fide business purpose for not making such information public.

"Backstop Agreement" shall mean that certain Backstop Agreement, dated as of July 4, 2022 by and among the Company, Orchestra BioMed, Inc. and the purchasing parties signatory thereto.

"Block Trade" shall mean an offering and/or sale of Registrable Securities by any Holder on a block trade or underwritten or other coordinated basis (whether firm commitment or otherwise) without substantial marketing efforts prior to pricing, including, without limitation, a same day trade, overnight trade or similar transaction.

"Board" shall mean the Board of Directors of the Company.

"Change in Control" shall mean any transfer (whether by tender offer, merger, stock purchase, consolidation or other similar transaction), in one transaction or a series of related transactions, to a person or group of affiliated persons of the Company's voting securities if, after such transfer, such person or group of affiliated persons would hold more than 50% of outstanding voting securities of the Company (or surviving entity) or would otherwise have the power to control the Board or to direct the operations of the Company.

"Commitment Shares" shall mean any shares of Common Stock that the Sponsor Equityholders or their respective designees receive in connection with the Domestication or the Closing, including as a result of purchases contemplated by the Sponsor Forward Purchase Agreement (as defined below), the Backstop Agreement, and any other Additional Agreement or any ancillary agreements with Orchestra BioMed, Inc. which designate such shares as Commitment Shares under this Agreement.

"Covidien Forward Purchase Shares" shall mean any shares of Common Stock that Covidien Group or its designees receive in connection with the Domestication as a result of purchases contemplated by the Covidien Forward Purchase Agreement.

"Commission" shall mean the Securities and Exchange Commission.

"Earnout Election Agreement" shall have the meaning given in the Merger Agreement.

"Earnout Shares" shall have the meaning given in the Merger Agreement.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.

"Forward Purchase Agreements" shall mean, collectively, that certain Forward Purchase Agreement by and among the Company, Orchestra BioMed, Inc., and certain funds managed by RTW Investments, LP as Purchasing Parties (the "Sponsor Forward Purchase Agreement"), and that certain Forward Purchase Agreement by and among the Company, Orchestra BioMed, Inc., and Covidien Group S.à.r.l. ("Covidien Group"), an affiliate of Medtronic plc (the "Covidien Forward Purchase Agreement").

"Founder Shares" shall mean the 4,000,000 shares of Common Stock of the Company issued to the Company's initial shareholders in connection with the Domestication upon the conversion of the 4,000,000 ordinary shares, par value \$0.0001 per share of the Company ("Ordinary Shares") issued to such initial shareholders prior to the Company's initial public offering.

"Holders" shall have the meaning given in the Preamble, for so long as such person or entity holds any Registrable Securities.

"Lock-up Period" shall have the meaning given in Section 4.1.1.

"Misstatement" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus, or necessary to make the statements in a Registration Statement or Prospectus (in the case of a Prospectus, in the light of the circumstances under which they were made) not misleading.

"*Permitted Transferees*" shall mean: (a) with respect to any Sponsor Equityholder, any person or entity to whom a Holder of Registrable Securities is permitted to transfer such Registrable Securities prior to the expiration of the Lock-up Period on approval of both the Company and the Sponsor, and (b) any person or entity to whom a Holder of Registrable Securities is permitted to transfer such Registrable Securities prior to the expiration of the Lock-up Period on approval of both the company and the Sponsor, and (b) any person or entity to whom a Holder of Registrable Securities is permitted to transfer such Registrable Securities prior to the expiration of the Lock-up Period under this Agreement and any other applicable agreement between such Holder and the Company, and to any transferee thereafter.

"Private Shares" shall mean the 450,000 shares of Common Stock issued by the Company in connection with the Domestication upon the conversion of Ordinary Shares that were privately purchased simultaneously with the consummation of the Company's initial public offering.

"*Private Warrants*" shall mean the 1,500,000 warrants to purchase Common Stock issued by the Company in connection with the Domestication upon the conversion of 1,500,000 warrants to purchase Ordinary Shares issued by the Company that were privately purchased simultaneously with the consummation of the Company's initial public offering.

"Prospectus" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.

"Registrable Security" shall mean (a) the Merger Shares (b) the Founder Shares, (c) the Private Shares, (c) the Private Warrants and the Common Stock issued or issuable upon the exercise of the Private Warrants, (d) the Working Capital Warrants and any shares of Common Stock issued or issuable upon the exercise of the Working Capital Warrants, (e) the Commitment Shares, (f) the Earnout Shares, (g) the Covidien Forward Purchase Shares, (h) any outstanding share of the Common Stock or any other equity security (including the shares of Common Stock issued or issuable upon the exercise of any other equity security) of the Company held by a Holder as of the date that is two (2) business days in advance of the filing with the Commission of a Registration Statement or a pre-effective amendment thereto, provided such Holder provides the Company with information regarding the amount of such holdings at least two (2) business days in advance of the filing with the Commission of a Registration Statement or a pre-effective amendment thereto, and (i) any other equity security of the Company issued or issuable with respect to any such share of the Common Stock by way of a stock dividend or stock split or in connection with a combination of shares, recapitalization, merger, consolidation, spin-off, reorganization or similar transaction; provided, however, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities upon the earliest occur of: (A) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (B) such securities shall have been otherwise transferred (other than to a Permitted Transferee), new certificates for such securities not bearing (or book entry positions not subject to) a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not require registration under the Securities Act; (C) such securities shall have ceased to be outstanding; (D) such securities may be sold without registration pursuant to Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission) (but with no volume or other restrictions or limitations); or (E) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.¹

"*Registration*" shall mean a registration effected by preparing and filing a Registration Statement, Prospectus or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.



"Registration Expenses" shall mean the out-of-pocket expenses of a Registration, including, without limitation, the following:

(A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Common Stock is then listed;

(B) fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);

(C) printing, messenger, telephone and delivery expenses;

(D) reasonable fees and disbursements of counsel for the Company;

(E) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration;

(F) in an Underwritten Offering, reasonable and documented fees and expenses of one (1) legal counsel selected by the majority-in-interest of the Demanding Holders; and

(G) the costs and expenses of the Company and any of its officers, directors, counsel or other representatives in connection with presentations or meetings undertaken in connection with the offering of the Registrable Securities, including, without limitation, expenses associated with the production of road show slides and graphics and the production and hosting of any electronic road shows, fees and expenses of any consultants engaged in connection with road show presentations, and travel, lodging, transportation, and other expenses of the officers, directors, counsel and other representatives of the Company incurred in connection with any such presentations or meetings.

"*Registration Statement*" shall mean any registration statement filed by the Company with the Commission that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.

"Securities Act" shall mean the Securities Act of 1933, as amended from time to time.

"Shelf" shall mean the Form S-1 Shelf, the Form S-3 Shelf or any subsequent Shelf Registration.

"Shelf Registration" shall mean a registration of securities pursuant to a Registration Statement filed with the Commission in accordance with and pursuant to Rule 415 promulgated under the Securities Act (or any successor rule then in effect).

"Sponsor" shall mean HSAC 2 Holdings, LLC.

"Sponsor Forward Purchase Agreement" shall have the meaning given in the definition of "Forward Purchase Agreements."

"*Transfer*" shall mean the (a) the sale or assignment of, offer to sell, contract or agreement to sell, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Exchange Act with respect to, any security, (b) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) public announcement of any intention to effect any transaction specified in clause (a) or (b).

"Underwriter" shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.

"Underwritten Registration" or "Underwritten Offering" shall mean a Registration in which securities of the Company are sold to an Underwritter in a firm commitment underwriting for distribution to the public.

"*Working Capital Warrants*" shall mean any warrants issued in payment for working capital loans from the Sponsor to the Company, including any warrants issued by the Company in connection with the Domestication upon the conversion of warrants issued in payment for working capital loans from the Sponsor.

ARTICLE II REGISTRATIONS

2.1 Shelf Registration.

2.1.1 Filing. The Company shall as soon as reasonably practicable, but in any event within one hundred and twenty (120) days after the Closing Date, file with the Commission a Registration Statement for a Shelf Registration on Form S-1 (the "Form S-1 Shelf") covering, subject to Section 3.4, the public resale of all of the Registrable Securities (determined as of two (2) business days prior to such filing) on a delayed or continuous basis and shall use its commercially reasonable efforts to cause such Form S-1 Shelf to be declared effective as soon as practicable after the filing thereof, but in no event later than the earlier of (i) the 30th calendar day (or the 90th calendar day if the Commission notifies the Company that it will "review" the Registration Statement) following the filing of the Registration Statement and (ii) the tenth (10th) business day after the date the Company is notified (orally or in writing, whichever is earlier) by the Commission that the Registration Statement will not be "reviewed" or will not be subject to further review. Such Form S-1 Shelf shall provide for the resale of the Registrable Securities included therein pursuant to any method or combination of methods legally available to, and requested by, any Holder named therein. The Company shall maintain a Shelf in accordance with the terms hereof and shall prepare and file with the Commission such amendments, including post-effective amendments, and supplements as may be necessary to keep a Shelf continuously effective, available for use to permit the Holders named therein to sell their Registrable Securities included therein and in compliance with the provisions of the Securities Act until such time as there are no longer any Registrable Securities. Following the filing of a Form S-1 Shelf, the Company shall use its commercially reasonable efforts to convert the Form S-1 Shelf (and any Subsequent Shelf Registration) to a Registration Statement on Form S-3 (the "Form S-3 Shelf") as soon as reasonably practicable after the Company is eligible to use Form S-3. As soon as reasonably practicable following the effective date of a Registration Statement filed pursuant to this Section 2.1.1, but in any event within one (1) business day of such date, the Company shall notify the Holders of the effectiveness of such Registration Statement. The Company's obligation under this Section 2.1.1 shall, for the avoidance of doubt be subject to Section 3.4 hereto.

2.1.2 Subsequent Shelf Registration. If any Shelf ceases to be effective under the Securities Act for any reason at any time while Registrable Securities are still outstanding, the Company shall, subject to Section 3.4, use its commercially reasonable efforts to as promptly as is reasonably practicable cause such Shelf to again become effective under the Securities Act (including using commercially reasonable efforts to obtain the prompt withdrawal of any order suspending the effectiveness of such Shelf), and shall use its commercially reasonable efforts to as promptly as is reasonably practicable amend such Shelf in a manner reasonably expected to result in the withdrawal of any order suspending the effectiveness of such Shelf or file an additional registration statement as a Shelf Registration (a "Subsequent Shelf Registration") registering the resale of all Registrable Securities (determined as of two (2) business days prior to such filing), and pursuant to any method or combination of methods legally available to, and requested by, any Holder named therein. If a Subsequent Shelf Registration is filed, the Company shall use its commercially reasonable efforts to (i) cause such Subsequent Shelf Registration to become effective under the Securities Act as promptly as is reasonably practicable after the filing thereof (it being agreed that the Subsequent Shelf Registration shall be an automatic shelf registration statement (as defined in Rule 405 promulgated under the Securities Act) if the Company is a well-known seasoned issuer (as defined in Rule 405 promulgated under the Securities Act) at the most recent applicable eligibility determination date) and (ii) keep such Subsequent Shelf Registration continuously effective, available for use to permit the Holders named therein to sell their Registrable Securities included therein and in compliance with the provisions of the Securities Act until such time as there are no longer any Registrable Securities. Any such Subsequent Shelf Registration shall be on Form S-3 to the extent that the Company is eligible to use such form. Otherwise, such Subsequent Shelf Registration shall be on another appropriate form. The Company's obligation under this Section 2.1.2 shall, for the avoidance of doubt be subject to Section 3.4 hereto.

2.1.3 <u>Requests for Underwritten Shelf Takedowns</u>. At any time and from time to time when an effective Shelf is on file with the Commission, any one or more Holder (any Holder being, in such case, a "*Demanding Holder*") may request to sell all or any portion of their Registrable Securities in an underwritten offering that is registered pursuant to the Shelf (each, an "*Underwritten Shelf Takedown*"); *provided* that the Company shall only be obligated to effect an Underwritten Offering if such offering: (i) shall include Registrable Securities proposed to be sold by the Demanding Holder(s), either individually or together with other Demanding Holders, with a total offering price reasonably expected to exceed, in the aggregate, \$25 million (the "*Minimum Takedown Threshold*"), or (ii) is comprised of all remaining Registrable Securities held by the Demanding Holder. All requests for Underwritten Shelf Takedowns shall be made by giving written notice to the Company, which shall specify the approximate number of Registrable Securities proposed to be sold in the Underwritten Offering. Subject to Section 2.3.4, a majority-in-interest of the Demanding Holders shall have the right to select the Underwriters for such offering (which shall consist of one or more reputable nationally recognized investment banks), subject to the Company's prior approval (which shall not be unreasonably withheld, conditioned or delayed). The Legacy Orchestra Equityholders, on the one hand, and the Sponsor Equityholders, on the other hand, may each demand not more than one (1) Underwritten Shelf Takedown during the six-month period; *provided, however*; that no Demanding Holder may request an Underwritten Shelf Takedown during the six-month period following the closing of a prior Underwritten Shelf Takedown. Notwithstanding anything to the contrary in this Agreement, the Company may effect any Underwritten Offering pursuant to any then effective Registration Statement, including a Form S-3, that is then available for such offering.

2.1.4 <u>Reduction of Underwritten Offering</u>. If the managing Underwriter or Underwriters in an Underwritten Shelf Takedown, in good faith, advises the Company, the Demanding Holders and Holders requesting piggyback rights pursuant to this Agreement with respect to such Underwritten Shelf Takedown (the "Requesting Holders") (if any) in writing that the dollar amount or number of Registrable Securities that the Demanding Holders and the Requesting Holders (if any) desire to sell, taken together with all other shares of Common Stock or other equity securities that the Company desires to sell and all other shares of Common Stock or other equity securities, if any, as to which a Registration has been requested pursuant to separate written contractual piggyback registration rights entered into after the date hereof held by any other stockholders who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without materially affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders and the Requesting Holders (if any) (pro rata based on the respective number of Registrable Securities that each Demanding Holder and Requesting Holder (if any) has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders and Requesting Holders have requested be included in such Underwritten Registration (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the shares of Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the shares of Common Stock or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements entered into after the date hereof with such persons and that can be sold without exceeding the Maximum Number of Securities.

2.1.5 Withdrawal. Prior to the pricing of an Underwritten Shelf Takedown, a majority-in-interest of the Demanding Holders initiating such Underwritten Offering shall have the right to withdraw from a Registration pursuant to such Underwritten Offering for any or no reason whatsoever upon written notification (a "Withdrawal Notice") to the Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Underwritten Offering; provided that any Legacy Orchestra Equityholder or Sponsor Equityholder may elect to have the Company continue an Underwritten Offering if the Minimum Takedown Threshold would still be satisfied by the Registrable Securities proposed to be sold in the Underwritten Offering by the Legacy Orchestra Equityholders and the Sponsor Equityholders. If withdrawn, a demand for an Underwritten Offering shall constitute a demand for an Underwritten Offering by the withdrawing Demanding Holder for purposes of Section 2.1.3, unless either (i) such Demanding Holder has not previously withdrawn any Underwritten Offering or (ii) such Demanding Holder reimburses the Company for all Registration Expenses with respect to such Underwritten Offering (or, if there is more than one Demanding Holder, a pro rata portion of such Registration Expenses based on the respective number of Registrable Securities that each Demanding Holder has requested be included in such Underwritten Offering); provided that, if a Legacy Orchestra Equityholder or a Sponsor Equityholder elects to continue an Underwritten Offering pursuant to the proviso in the immediately preceding sentence, such Underwritten Offering shall instead count as an Underwritten Offering demanded by such Legacy Orchestra Equityholder or the Sponsor Equityholders, as applicable, for purposes of Section 2.1.3. Following the receipt of any Withdrawal Notice, the Company shall promptly forward such Withdrawal Notice to any other Holders that had elected to participate in such Underwritten Shelf Takedown. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with an Underwritten Shelf Takedown prior to its withdrawal under this Section 2.1.5, other than if a Demanding Holder elects to pay such Registration Expenses pursuant to clause (ii) of the second sentence of this Section 2.1.5.

2.1.6 Notwithstanding the registration obligations set forth in this Section 2.1, in the event the Commission informs the Company that all of the Registrable Securities cannot, as a result of the application of Rule 415, be registered for resale as a secondary offering on a single registration statement, the Company agrees to promptly (i) inform each of the Holders and use its commercially reasonable efforts to file amendments to the Registration Statement as required by the Commission and/or (ii) withdraw the Registration Statement and file a new registration statement (a "New Registration Statement"), on Form S-3, or if Form S-3 is not then available to the Company for such registration statement, on such other form available to register for resale the Registrable Securities as a secondary offering; provided, however, that prior to filing such amendment or New Registration Statement, the Company shall use its commercially reasonable efforts to advocate with the Commission for the registration of all of the Registrable Securities in accordance with any publicly-available written or oral guidance, comments, requirements or requests of the Commission staff (the "SEC Guidance"), including without limitation, the Manual of Publicly Available Telephone Interpretations D.29. Notwithstanding any other provision of this Agreement, if any SEC Guidance sets forth a limitation of the number of Registrable Securities permitted to be registered on a particular Registration Statement as a secondary offering (and notwithstanding that the Company used commercially reasonable efforts to advocate with the Commission for the registration of all or a greater number of Registrable Securities), unless otherwise directed in writing by a Holder as to its Registrable Securities, the number of Registrable Securities to be registered on such Registration Statement will be reduced on a pro rata basis based on the total number of Registrable Securities held by the Holders, subject to a determination by the Commission that certain Holders must be reduced first based on the number of Registrable Securities held by such Holders. In the event the Company amends the Registration Statement or files a New Registration Statement, as the case may be, under clauses (i) or (ii) above, the Company will use its commercially reasonable efforts to file with the Commission, as promptly as allowed by Commission or SEC Guidance provided to the Company or to registrants of securities in general, one or more registration statements on Form S-3 or such other form available to register for resale those Registrable Securities that were not registered for resale on the Registration Statement, as amended, or the New Registration Statement.

2.1.7 Effective Registration. Except with respect to withdrawals covered by Section 2.1.5, a Registration shall not count as a Registration unless and until (i) the Registration Statement has been declared effective by the Commission and (ii) the Company has complied with all of its obligations under this Agreement with respect thereto; *provided, further*, that if, after such Registration Statement has been declared effective, an offering of Registration Statement with respect to such Registration shall be deemed not to have been declared effective, unless and until (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders initiating such Registration thereafter affirmatively elect to continue with such Registration and accordingly notify the Company in writing, but in no event later than five (5) days, of such election; *provided, further*, that the Company shall not be obligated or required to file another Registration Statement until the Registration Statement that has been previously filed with respect to a Registration pursuant to an Underwritten Demand Registration becomes effective or is subsequently terminated.

2.2 Piggyback Registration.

2.2.1 Piggyback Rights. If the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company including, without limitation, pursuant to Section 2.1 hereof), other than a Registration Statement (or any registered offering with respect thereto) (i) filed in connection with any employee stock option or other benefit plan, (ii) for an exchange offer or offering of securities solely to the Company's existing stockholders, (iii) for an offering of debt that is convertible into equity securities of the Company, (iv) pursuant to a Registration Statement on Form S-4 (or similar form that relates to a transaction subject to Rule 145 under the Securities Act or any successor rule thereto), (v) for a dividend reinvestment plan, or (vi) for a Block Trade, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than ten (10) days before the anticipated filing date of such Registration Statement or, in the case of an Underwritten Offering pursuant to a Shelf Registration, the applicable "red herring" prospectus or prospectus supplement used for marketing such offering, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (such Registration a "Piggyback Registration"). Subject to Section 2.2.2, the Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and, if applicable, shall use commercially reasonable efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering to permit the Registrable Securities requested by the Holders pursuant to this Section 2.2.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company included in such registered offering and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this Section 2.2.1 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.



2.2.2 <u>Reduction of Piggyback Registration</u>. If the managing Underwriter or Underwriters in an Underwritten Registration that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of shares of Common Stock or other equity securities that the Company desires to sell, taken together with (i) the shares of Common Stock or other equity securities of Registrable Securities hereunder, (ii) the Registrable Securities as to which registration has been requested pursuant to Section 2.2 hereof, and (iii) the shares of Common Stock or other equity securities, if any, as to which Registration or a registered offering has been requested pursuant to Section 2.2 hereof, and (iii) the shares of Common Stock or other equity securities, if any, as to which Registration or a registered offering has been requested pursuant to separate written contractual piggyback registration rights of stockholders of the Company other than the Holders of Registrable Securities, then:

(a) If the Registration or a registered offering is undertaken for the Company's account, the Company shall include in any such Registration or a registered offering (A) first, the shares of Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to Section 2.2.1 hereof, Pro Rata, based on the respective number of Registrable Securities that each Holder has so requested to be included in such Registration or such registered offering, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the shares of Common Stock or other equity securities, if any, as to which Registration or a registered offering has been requested pursuant to written contractual piggyback registration rights of stockholders of the Company entered into after the date hereof, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities;

(b) If the Registration or a registered offering is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration or a registered offering (A) first, the shares of Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities (B) second, to the extent that the Maximum Number of Securities pursuant to Section 2.2.1, Pro Rata, based on the respective number of Registrable Securities that each Holder has so requested to be included in such Registration or such registered offering, which can be sold without exceeding the Maximum Number of Securities; (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the shares of Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A), (B) and (C), the shares of Common Stock or other equity securities, if any, as to which Registration or a registered offering has been requested pursuant to separate written contractual piggyback registration rights entered into after the date hereof, of persons or entities other than the Holders of Registrable Securities; and

(c) If the Registration or registered offering is pursuant to a request by Holder(s) of Registrable Securities pursuant to Section 2.1 hereof, then the Company shall include in any such Registration or registered offering securities in the priority set forth in Section 2.1.4.

2.2.3 Piggyback Registration Withdrawal. Any Holder of Registrable Securities (other than a Demanding Holder, whose right to withdraw from an Underwritten Shelf Takedown, and related obligations, shall be governed by Section 2.1.5) shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration prior to the effectiveness of the Registration, the filing of the applicable "red herring" prospectus or prospectus supplement with respect to such Piggyback Registration used for marketing such transaction. Other than with respect to an Underwritten Shelf Takedown by a Demanding Holder pursuant to Section 2.1.3, the Company (whether on its own good faith determination or as the result of a request for withdrawal by persons or entities pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission at any time prior to the effectiveness of such Registration Statement or abandon the <u>Underwritten Shelf Takedown</u> in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement or abandon the <u>Underwritten Shelf Takedown</u> in connection with a Piggyback Registration at any time prior to the launch of such <u>Underwritten Shelf Takedown</u>. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this Section 2.2.3.

2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Piggyback Registration effected pursuant to Section 2.2 hereof shall not be counted as a Registration pursuant to an Underwritten Shelf Takedown effected under Section 2.1 hereof.

2.3 Block Trades.

2.3.1 Notwithstanding the foregoing, at any time and from time to time when an effective Shelf is on file with the Commission and effective, if a Demanding Holder wishes to engage in a Block Trade, (x) with a total offering price reasonably expected to exceed \$75 million in the aggregate or (y) with respect to all remaining Registrable Securities held by the Demanding Holder (the "*Minimum Block Threshold*"), then such Demanding Holder only needs to notify the Company of the Block Trade at least five (5) business days prior to the day such offering is to commence and the Company shall as expeditiously as possible use its commercially reasonable efforts to facilitate such Block Trade; *provided* that the Demanding Holders representing a majority of the Registrable Securities wishing to engage in the Block Trade shall use commercially reasonable efforts to work with the Company and any Underwriters prior to making such request in order to facilitate preparation of the registration statement, prospectus and other offering documentation related to the Block Trade.

2.3.2 Prior to the filing of the applicable "red herring" prospectus or prospectus supplement used in connection with a Block Trade, a majority-in-interest of the Demanding Holders initiating such Block Trade shall have the right to submit a Withdrawal Notice to the Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Block Trade, *provided* that any other Demanding Holder(s) may elect to have the Company continue a Block Trade if the Minimum Block Threshold would still be satisfied by the Registrable Securities proposed to be sold in the Block Trade by the remaining Demanding Holder(s). Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Block Trade prior to its withdrawal under this Section 2.3.2.

2.3.3 Notwithstanding anything to the contrary in this Agreement, Section 2.2 hereof shall not apply to a Block Trade initiated by a Demanding Holder pursuant to this Agreement.

2.3.4 The Demanding Holder in a Block Trade shall have the right to select the Underwriters for such Block Trade (which shall consist of one or more reputable nationally recognized investment banks).

2.3.5 Each Legacy Orchestra Equityholder and Sponsor Equityholder may demand no more than one (1) Block Trade pursuant to this Section 2.3 in any twelve (12) month period. For the avoidance of doubt, any Block Trade effected pursuant to this Section 2.3 shall not be counted as a demand for an Underwritten Shelf Takedown pursuant to Section 2.1.3 hereof.

ARTICLE III COMPANY PROCEDURES

3.1 <u>General Procedures</u>. If the Company is required to effect the Registration of Registrable Securities pursuant to this Agreement, the Company shall use its best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall, as expeditiously as possible:

3.1.1 prepare and file with the Commission within the time frame required by Section 2.1.1 (to the extent applicable) a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective, until all Registrable Securities covered by such Registration Statement have been sold or have ceased to be Registrable Securities;

3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be reasonably requested by any Holder with Registrable Securities registered on such Registration Statement or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus or have ceased to be Registrable Securities;

3.1.3 at least two (2) business days prior to filing a Registration Statement or Prospectus, or any amendment or supplement thereto (or such shorter period of time as may be (a) necessary in order to comply with the Securities Act, the Exchange Act and the rules and regulations promulgated thereunder or (b) advisable in order to reduce the number of days that sales are suspended pursuant to Section 3.4), furnish without charge to the Underwriters, if any, and each Holder of Registrable Securities included in such Registration, and each such Holder's legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and each Holder of Registrable Securities included in such Registration or the legal counsel for any such Holders may reasonably request in order to facilitate the disposition of the Registrable Securities owned by such Holders; *provided*, that the Company shall have no obligation to furnish any documents publicly filed or furnished with the Commission pursuant to the Electronic Data Gathering Analysis and Retrieval System ("*EDGAR*"); and provided further, *provided* that the Company shall provide each Holder and its legal counsel with a reasonable opportunity to review such documents and comment thereon, and the Company shall consider in good faith any comments provided by such Holder or their legal counsel;

3.1.4 prior to any public offering of Registrable Securities, use commercially reasonable efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as any Holder of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request (or provide evidence reasonably satisfactory to such Holders and their respective legal counsel that the Registrable Securities are exempt from such registration or qualification) and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; *provided, however*, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;



3.1.5 cause all Registrable Securities included in any Registration to be listed on such exchanges or otherwise designated for trading in the same manner as similar securities issued by the Company are then listed or designated;

3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;

3.1.7 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its commercially reasonable efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;

3.1.8 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in Section 3.4 hereof;

3.1.9 in the event of an Underwritten Offering, a Block Trade, or sale by a broker, placement agent or sales agent pursuant to such Registration, in each of the cases to the extent customary for a transaction of its type, permit a representative of the Holders (such representative to be selected by a majority of the participating Holders), the Underwriters or other financial institutions facilitating such Underwritten Offering, Block Trade or other sale pursuant to such Registration, if any, and any attorney, consultant or accountant retained by such Holders or Underwriters to participate, at each such person's or entity's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriters or financial institutions enter into a confidentiality agreement, in customary form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information and *provided, further*, that, except as required by Law, the Company may not include the name of any Holder or Underwriter or any information regarding any Holder or Underwriter in any Registration Statement or Prospectus, any amendment or supplement to such Registration Statement or Prospectus, any document that is to be incorporated by reference into such Registration Statement or Prospectus, or any response to any comment letter, without the prior written consent of such Holder or Underwriter a reasonable amount of time to review and comment on such applicable document, which comments the Company shall consider shall consider in good faith and, to the extent deemed appropriate by the Company in its sole discretion, include.

3.1.10 obtain a "comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Offering, a Block Trade or sale by a broker, placement agent or sales agent pursuant to such Registration in customary form and covering such matters of the type customarily covered by "comfort" letters for a transaction of its type as the managing Underwriter may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders;

3.1.11 in the event of an Underwritten Offering, a Block Trade or sale by a broker, placement agent or sales agent pursuant to such Registration, on the date the Registrable Securities are delivered for sale pursuant to such Registration, to the extent customary for a transaction of its type, obtain an opinion, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the participating Holders, the broker, placement agents or sales agent, if any, and the Underwriters, if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the participating Holders, broker, placement agents, sales agent, or Underwriter may reasonably request and as are customarily included in such opinions and negative assurance letters;

3.1.12 in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;

3.1.13 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule then in effect), and which requirement will be deemed to be satisfied if the Company timely files complete and accurate information on Forms 10-Q, 10-K and 8-K under the Exchange Act and otherwise complies with Rule 158 under the Securities Act;

3.1.14 if the Registration involves the Registration of Registrable Securities involving gross proceeds in excess of \$25 million, use its commercially reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter in such Underwritten Offering; and

3.1.15 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the participating Holders, consistent with the terms of this Agreement, in connection with such Registration.

Notwithstanding the foregoing, the Company shall not be required to provide any documents or information to an Underwriter, broker, sales agent or placement agent if such Underwriter, broker, sales agent or placement agent has not then been named with respect to the applicable Underwritten Offering or other offering involving a registration as an Underwriter, broker, sales agent or placement agent, as applicable.

3.2 <u>Registration Expenses</u>. Except as otherwise provided herein, the Registration Expenses of all Registrations, including a Registration that is postponed, suspended, delayed or withdrawn, and any failed Registrations that are withdrawn or abandoned by the Company for any reason shall be borne by the Company, unless a Demanding Holder elects to reimburse the Company pursuant to clause (ii) of the second sentence of Section 2.1.5. It is acknowledged by the Holders that each Holder shall bear, severally and not jointly, with respect to such Holder's Registrable Securities being sold, all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "Registration Expenses," all reasonable fees and expenses of any legal counsel representing the Holders.

3.3 <u>Requirements for Participation in Underwritten Offerings</u>. Notwithstanding anything in this Agreement to the contrary, if any Holder does not provide the Company with its requested Holder Information (as defined in Section 5.1.2), the Company may exclude such Holder's Registrable Securities from the applicable Registration Statement or Prospectus if the Company determines in good faith, based on the advice of counsel, that it is necessary or advisable to include such information in the applicable Registration Statement or Prospectus and such Holder continues thereafter to withhold such information. In addition, no person or entity may participate in any Underwritten Offering or other offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person or entity (i) agrees to sell such person's or entity's securities on the basis provided in any underwriting arrangements approved by the Company and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements. For the avoidance of doubt, the exclusion of a Holder's Registrable Securities as a result of this Section 3.3 shall not affect the registration of the other Registrable Securities to be included in such Registration.

3.4 Suspension of Sales; Adverse Disclosure.

3.4.1 Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until it has received copies of a supplemented or amended Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as reasonably practicable after the time of such notice), or until it is advised in writing by the Company that the use of the Prospectus may be resumed.



3.4.2 If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would (a) require the Company to make an Adverse Disclosure, (b) require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, or (c) in the good faith judgment of the majority of the Board such Registration, be seriously detrimental to the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness or continued use at such time, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time, but in no event no more than three (3) occasions or for more than sixty (60) consecutive days, or more than ninety (90) total calendar days, in each case, during any 12-month period, determined in good faith by the Company to be necessary for such purpose. In the event the Company exercises its rights under the preceding sentence, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities until such Holder receives written notice from the Company shall promptly notify the Holders of the expiration of any period during which it exercised its rights under this Section 3.4, and, upon the expiration of any such period, the Holders shall be entitled to resume the use of any such Prospectus in connection with any sale or offer to sell Registrable Securities.

3.5 <u>Reporting Obligations</u>. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such filings; *provided* that any documents publicly filed or furnished with the Commission pursuant to EDGAR shall be deemed to have been furnished or delivered to the Holders pursuant to this Section 3.5. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell shares of Common Stock held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Section 4(a)(1) of the Securities Act or Rule 144 promulgated under the Securities Act (or any successor rule then in effect), including providing any legal opinions. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

ARTICLE IV LOCK-UP

4.1 Lock-up.

4.1.1 Except as permitted by Section 4.2, each Legacy Orchestra Equityholder and Sponsor Equityholder (each, a "Lock-up Party") shall not Transfer (the "Lock-up") any shares of Common Stock or any security convertible into or exercisable or exchanged for Common Stock beneficially owned or owned of record by such Holder (the "Lock-up Shares") until the date that is the earlier of (i) six (6) months from the date hereof (or twelve (12) months with respect to the Founder Shares and the Private Shares and any shares of Common Stock or any security convertible into or exercisable or exchanged for Common Stock beneficially owned or owned of record by RTW and its Affiliates as of the date of this Agreement), or (ii) the date on which the Company completes a liquidation, merger, stock exchange, reorganization or other similar transaction that results in all of the Company's stockholders having the right to exchange their shares of Common Stock for cash, securities or other property (the "Lock-up Period"). Notwithstanding anything to the contrary in this Agreement, (A) to the extent any Holder has executed any other agreement with the Company or any of its Affiliates that provides for a longer lock-up Period than the Lock-up Period, the lock-up period in such other agreement shall control and (B) the Company may at any time or from time to time, in its sole discretion, elect to release from the Lock-up some or all of the Lock-up Shares purchased in connection with the Backstop Agreement by providing written notice of such election to the holders of such Lock-up Shares.

4.2 Exceptions. The provisions of Section 4.1 shall not apply to:

4.2.1 transactions relating to shares of Common Stock or warrants acquired in open market transactions;

4.2.2 Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock as a bona fide gift or charitable contribution;

4.2.3 Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock to a trust, family limited partnership or other entity formed for estate planning purposes for the primary benefit of the spouse, domestic partner, parent, sibling, child or grandchild of a Holder or any other person with whom a Holder has a relationship by blood, marriage or adoption not more remote than first cousin and Transfers to any such family member;

4.2.4 Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock by will or intestate succession or the laws of descent and distributions upon the death of a Holder (it being understood and agreed that the appointment of one or more executors, administrators or personal representatives of the estate of a Holder shall not be deemed a Transfer hereunder to the extent that such executors, administrators and/or personal representatives comply with the terms of this <u>Article IV</u> on behalf of such estate);

4.2.5 Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock pursuant to a qualified domestic order or in connection with a divorce settlement;

4.2.6 if a Holder is a corporation, partnership (whether general, limited or otherwise), limited liability company, trust or other business entity, (i) Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock to another corporation, partnership, limited liability company, trust or other business entity that controls, is controlled by or is under common control or management with a Holder (including, for the avoidance of doubt, where such Holder is a partnership, to its general partner or a successor partnership or fund, or any other funds managed by such partnership), or (ii) Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock as part of a dividend, distribution, transfer or other disposition of shares of Common Stock to partners, limited liability company members, direct or indirect stockholders or other equity holders of a Holder, including, for the avoidance of doubt, where such Holder is a partnerships, funds or investment vehicles controlled or managed by such partnership, to its general partner or a successor partnership, fund or investment vehicle, or any other partnerships, funds or investment vehicles controlled or managed by such partnership;

4.2.7 if the Holder is a trust, Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock to a trustor or beneficiary of such trust or to the estate of a beneficiary of such trust;

4.2.8 Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock to the Company's or the Holder's officers, directors, members, consultants or their affiliates;

4.2.9 pledges of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock as security or collateral in connection with any borrowing or the incurrence of any indebtedness by any Holder (provided such borrowing or incurrence of indebtedness is secured by a portfolio of assets or equity interests issued by multiple issuers);

4.2.10 Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock pursuant to a *bona fide* third-party tender offer, merger, asset acquisition, stock sale, recapitalization, consolidation, business combination or other transaction or series of related transactions involving a Change in Control of the Company, *provided* that in the event that such tender offer, merger, asset acquisition, stock sale, recapitalization is not completed, the securities subject to this Agreement shall remain subject to this Agreement;

4.2.11 Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock to the Company in connection with the liquidation or dissolution of the Company by virtue of the laws of the state of the Company's organization and the Company's organizational documents;

4.2.12 the establishment of a trading plan pursuant to Rule 10b5-1 promulgated under the Exchange Act, *provided* that such plan does not provide for the Transfer of any shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock during the Lock-Up Period; and

4.2.13 Transfers of shares of Common Stock or any security convertible into or exercisable or exchangeable for Common Stock to satisfy any U.S. federal, state, or local income tax obligations of the Lock-up Party (or its direct or indirect owners) arising from a change in the U.S. Internal Revenue Code of 1986, as amended (the "*Code*"), or the U.S. Treasury Regulations promulgated thereunder (the "*Regulations*") after the date on which the Merger Agreement was executed by the parties, and such change prevents the Merger from qualifying as a "reorganization" pursuant to Section 368 of the Code (and the Merger does not qualify for similar tax-free treatment pursuant to any successor or other provision of the Code or Regulations taking into account such changes), in each case solely and to the extent necessary to cover any tax liability as a direct result of the transaction;

PROVIDED, THAT IN THE CASE OF ANY TRANSFER OR DISTRIBUTION PURSUANT TO SECTIONS 4.2.2 THROUGH 4.2.9 AND 4.2.13, EACH DONEE, DISTRIBUTEE OR OTHER TRANSFEREE SHALL AGREE IN WRITING, IN FORM AND SUBSTANCE REASONABLY SATISFACTORY TO THE COMPANY, TO BE BOUND BY THE PROVISIONS OF THIS AGREEMENT.

4.3 <u>Null and Void</u>. If any Transfer of shares of Common Stock prior to the end of the Lock-up Period is made or attempted contrary to the provisions of this Agreement, such purported Transfer shall be null and void *ab initi*o, and the Company shall refuse to recognize any such purported transferee of the Common Stock as one of its equityholders for any purpose.

4.4 Legend. During the Lock-up Period, each certificate evidencing any Common Stock shall be stamped or otherwise imprinted with a legend in substantially the following form, in addition to any other applicable legends:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO RESTRICTIONS ON TRANSFER SET FORTH IN AN AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT, DATED AS OF JANUARY 26, 2023 (AS MAY BE AMENDED OR RESTATED FROM TIME TO TIME), A COPY OF WHICH IS ON FILE WITH THE SECRETARY OF THE COMPANY. NO TRANSFER, SALE, ASSIGNMENT, PLEDGE, HYPOTHECATION OR OTHER DISPOSITION OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE MAY BE MADE EXCEPT IN ACCORDANCE WITH THE PROVISIONS OF SUCH AGREEMENT."

Promptly upon the expiration of the Lock-up Period, the Company shall cause the removal of such legend and, if determined appropriate by the Company, any restrictive legend related to compliance with the federal securities laws from the certificates evidencing the Common Stock.

ARTICLE V INDEMNIFICATION AND CONTRIBUTION

5.1 Indemnification.

5.1.1 The Company agrees to indemnify and hold harmless, to the extent permitted by law, each Holder of Registrable Securities, its partners, members, managers, officers, directors and agents and each person or entity who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and out-of-pocket expenses (including without limitation actual, reasonable and documented attorneys' fees or other fees and expenses incurred thereby in connection with investigating or defending any such claim or proceeding), caused by any untrue or alleged untrue statement of material fact contained or incorporated by reference in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading (except insofar as the same are caused by or contained in any information or affidavit so furnished in writing to the Company by such Holder expressly for use therein). The Company shall indemnify the Underwriters, their officers and directors and each person or entity who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.

5.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish (or cause to be furnished) to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus (the "*Holder Information*") and, to the extent permitted by law, shall indemnify the Company, its directors, officers and agents and each person or entity who controls the Company (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and out-of-pocket expenses (including without limitation actual, reasonable and documented attorneys' fees) resulting from any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement is contained in (or not contained in, in the case of an omission) the Holder Information; *provided, however*, that the obligation to indemnify shall be several, not joint and several, among such Holders of Registrable Securities, and the liability of each such Holder of Registrable Securities shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement. The Holders of Registrable Securities shall indemnify the Underwriters, their officers, directors and each person or entity who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to indemnification of the Company.

5.1.3 Any person or entity entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (*provided*, that the failure to give prompt notice shall not impair any person's or entity's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one (1) counsel (plus one (1) local counsel) for all parties indemnified by such indemnified party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement includes a statement or admission of fault and culpability on the part of such indemnified party or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

5.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling person or entity of such indemnified party and shall survive the transfer of securities. The Company and each Holder of Registrable Securities participating in an offering also agrees to make such provisions as are reasonably requested by any indemnified party for contribution to such party in the event the Company's or such Holder's indemnification is unavailable for any reason.

5.1.5 If the indemnification provided under Section 5.1 hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and out-of-pocket expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and out-of-pocket expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by (or not made by, in the case of an omission), or relates to information supplied by (or not supplied by, in the case of an omission), such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action and the benefits received by such indemnified party or indemnifying party; provided, however, that the liability of any Holder under this Section 5.1.5 shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in Sections 5.1.1, 5.1.2 and 5.1.3 above, any legal or other fees, charges or out-of-pocket expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 5.1.5 were determined by pro rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this Section 5.1.5. No person or entity guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution pursuant to this Section 5.1.5 from any person or entity who was not guilty of such fraudulent misrepresentation.

ARTICLE VI MISCELLANEOUS

6.1 <u>Notices</u>. Any notice hereunder shall be sent in writing, addressed as specified below, and shall be deemed given: (a) if by hand or recognized courier service, by 4:00PM on a business day, addressee's day and time, on the date of delivery, and otherwise on the first business day after such delivery; (b) if by fax or email, on the date that transmission is confirmed electronically, if by 4:00PM on a business day, addressee's day and time, on the first business day, addressee's day and time, and otherwise on the first business day after the date of such confirmation; or (c) five days after mailing by certified or registered mail, return receipt requested. Notices shall be addressed to the respective parties as follows, or to such other address as a party shall specify to the others in accordance with this Section 6.1:

if to the Company, to:

Orchestra BioMed, Inc. 150 Union Square Drive New Hope PA 18938 Attn: David Hochman, Chairman & CEO E-mail: DHochman@orchestrabiomed.com,

with a copy to:

Paul Hastings LLP 200 Park Avenue New York, NY 10166 Attn: Samuel A. Waxman E-mail: samuelwaxman@paulhastings.com;

if to any Holder, at such Holder's address or contact information as set forth in the Company's books and records.

6.2 Assignment; No Third-Party Beneficiaries.

6.2.1 This Agreement and the rights, duties and obligations of the Company hereunder may not be assigned or delegated by the Company in whole or in part.

6.2.2 Subject to Section 6.2.4 and Section 6.2.5, this Agreement and the rights, duties and obligations of a Holder hereunder may be assigned in whole or in part to such Holder's Permitted Transferees; *provided* that, (a)(i) the Legacy Orchestra Equityholders shall be permitted to transfer their rights hereunder as a Legacy Orchestra Equityholder to one or more affiliates or any direct or indirect partners, members or equity holders of the Legacy Orchestra Equityholders (it being understood that no such transfer shall reduce any rights of the Legacy Orchestra Equityholders shall be permitted to transfer their rights hereunder as the Sponsor Equityholders to one or more of their respective affiliates or any direct or indirect partners, members or equity holders or such transfer shall reduce any rights of the Sponsor Equityholders to one or more of their shall reduce any rights of the Sponsor Equityholders or such transferees).

6.2.3 This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors and the permitted assigns of the Holders, which shall include Permitted Transferees.

6.2.4 This Agreement shall not confer any rights or benefits on any persons or entities that are not parties hereto, other than as expressly set forth in this Agreement and Section 6.2 hereof.

6.2.5 No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in Section 6.1 hereof and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any transfer or assignment made other than as provided in this Section 6.2 shall be null and void.

6.3 <u>Counterparts; Facsimile Signatures</u>. This Agreement may be executed in separate counterparts, each of which shall constitute an original, and all of which together shall constitute one and the same agreement binding on all the parties hereto. This Agreement shall become effective upon delivery to each party of an executed counterpart or the earlier delivery to each party of original, photocopied, or electronically transmitted signature pages that together (but need not individually) bear the signatures of all other parties. For the avoidance of doubt, each party agrees that an electronic copy of this Agreement shall be considered and treated like an original, and that an electronic or digital signature shall be as valid as a handwritten signature (including .pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000 (e.g., www.docusign.com)).

6.4 <u>Governing Law; Venue</u>. This Agreement shall be construed in accordance with and governed by the laws of the State of Delaware, without giving effect to the conflict of laws principles thereof. Any Action based upon, arising out of or related to this Agreement or the transactions contemplated hereby must be brought in the Court of Chancery of the State of Delaware (or, to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware), or, if it has or can acquire jurisdiction, in the United States District Court for the District of Delaware, and each of the parties irrevocably (i) submits to the exclusive jurisdiction of each such court in any such proceeding or Action, (ii) waives any objection it may now or hereafter have to personal jurisdiction, venue or to convenience of forum, (iii) agrees that all claims in respect of the proceeding or Action shall be heard and determined only in any such court, and (iv) agrees not to bring any proceeding or Action arising out of or relating to this Agreement or the transactions contemplated hereby in any other court. Nothing herein contained shall be deemed to affect the right of any party to serve process in any manner permitted by Law or to commence Proceedings or otherwise proceed against any other party in any other jurisdiction, in each case, to enforce judgments obtained in any Action brought pursuant to this Section 6.4.

6.5 <u>Waiver of Jury Trial</u>. THE PARTIES TO THIS AGREEMENT HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVE ANY RIGHT EACH SUCH PARTY MAY HAVE TO TRIAL BY JURY IN ANY ACTION OF ANY KIND OR NATURE, IN ANY COURT IN WHICH AN ACTION MAY BE COMMENCED, ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT OR ANY ADDITIONAL AGREEMENT, OR BY REASON OF ANY OTHER CAUSE OR DISPUTE WHATSOEVER BETWEEN OR AMONG ANY OF THE PARTIES TO THIS AGREEMENT OF ANY KIND OR NATURE. NO PARTY SHALL BE AWARDED PUNITIVE OR OTHER EXEMPLARY DAMAGES RESPECTING ANY DISPUTE ARISING UNDER THIS AGREEMENT OR ANY ADDITIONAL AGREEMENT.

6.6 <u>Amendments and Modifications</u>. Upon the written consent of the Company, the Holders of a majority-in-interest of the Registrable Securities at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; *provided*, *however*, that, notwithstanding the foregoing, any amendment hereto or waiver hereof that (i) adversely affects one Holder, solely in its capacity as a holder of the shares of capital stock of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder so affected, (ii) any amendment or waiver hereof that adversely affects the rights of Sponsor (or its successor) shall require the consent of the Sponsor (or its successor), and (iv) any amendment or waiver hereof that adversely affects the rights of the Legacy Orchestra Equityholders shall require the consent a majority-in-interest of the Registrable Securities held by the Legacy Orchestra Equityholders at the time in question. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereone by such aparty.

6.7 Other Registration Rights. The Company represents and warrants that no person or entity, other than a Holder of Registrable Securities, has any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities for its own account or for the account of any other person or entity. Further, the Company represents and warrants that this Agreement supersedes any other registration rights agreement or agreement with similar terms and conditions and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail. This Agreement supersedes, and amends and restates in its entirety, the Prior Agreement.

6.8 <u>Term</u>. This Agreement shall terminate upon the earlier of (i) the tenth (10th) anniversary of the date of this Agreement, (ii) the date as of which all of the Registrable Securities have been sold or disposed of, or (iii) with respect to any particular Holder, on the date such Holder no longer holds Registrable Securities. The provisions of Section 3.5 and Article V shall survive any termination.

6.9 <u>Holder Information</u>. Each Holder agrees, if requested in writing, to represent to the Company the total number of Registrable Securities held by such Holder in order for the Company to make determinations hereunder.

6.10 <u>Severability</u>. A determination by a court or other legal authority that any provision that is not of the essence of this Agreement is legally invalid shall not affect the validity or enforceability of any other provision hereof. The parties shall cooperate in good faith to substitute (or cause such court or other legal authority to substitute) for any provision so held to be invalid a valid provision, as alike in substance to such invalid provision as is lawful.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties have caused this Amended and Restated Registration Rights Agreement to be executed and delivered by their duly authorized representatives as of the date first written above.

COMPANY:

ORCHESTRA BIOMED, INC.

By: /s/ David Hochman

Name: David Hochman Title: Chief Executive Officer

SPONSOR EQUITYHOLDERS:

HSAC 2 HOLDINGS, LLC

By: /s/ Alice Lee Name: Alice Lee Title: Director

/s/ Pedro Granadillo

Pedro Granadillo

/s/ Stuart Peltz

Stuart Peltz

/s/ Michael Brophy

Michael Brophy

/s/ Carsten Boess

Carsten Boess

RTW Master Fund, Ltd.

By: /s/ Roderick Wong, M.D. Name: Roderick Wong, M.D. Title: Director

RTW Innovation Master Fund, Ltd.

By: /s/ Roderick Wong, M.D.

Name: Roderick Wong, M.D. Title: Director

RTW Venture Fund Limited

By: RTW Investments, LP, its Investment Manager

By: /s/ Roderick Wong, M.D.

Name: Roderick Wong, M.D.

Title: Managing Partner

[Signature Page to Registration Rights Agreement]

LEGACY ORCHESTRA EQUITYHOLDERS:

RTW Master Fund, Ltd.

	/s/ Roderick Wong, M.D.	
	Roderick Wong, M.D. Director	
RTW Innovation Master Fund, Ltd.		
By:	/s/ Roderick Wong, M.D.	
Name: Title:	Roderick Wong, M.D. Director	
RTW Venture Fund Limited		
By:	/s/ Roderick Wong, M.D.	
Name:	Roderick Wong, M.D.	
Title:		
By:	/s/ David Hochman	
Name:	David Hochman	
2	/s/ Darren Sherman	
Name:	Darren Sherman	
By:	/s/ Michael Kaswan	
Name:	Michael Kaswan	
By: Name:	/s/ Yuval Mika Yuval Mika	
Ivanie.	דעימו ויווגמ	
By:	/s/ Eric Fain	
Name:	Eric Fain	
By:	/s/ Eric Rose	
Name:	Eric Rose	
	/s/ Jason Aryeh Jason Aryeh	
ivanie.	-	
By: Name:	/s/ Pamela Connealy Pamela Connealy	
maine.		
By: Name:	/s/ Geoffrey W. Smith Geoffrey W. Smith	
INAME.		

Name: Geoffrey W. Smith

Covidien Group S.à.r.l.

By: /s/ Erik De Gres Name: Erik De Gres Title: General Manager

Perceptive Life Sciences Master Fund

By:	/s/ James Mannix
Name:	James Mannix
Title:	COO

Ascent Biomedical Ventures II NY, LP

By: /s/ Steve Hochberg

Name: Steve Hochberg Title: General Partner

Ascent Biomedical Ventures II, LP

By: /s/ Steve Hochberg

Name: Steve Hochberg Title: General Partner

The General Farmer

Ascent Biomedical Ventures Synecor, LP

By: /s/ Steve Hochberg

Name: Steve Hochberg

Title: General Partner

SCHEDULE A – SPONSOR EQUITYHOLDERS

Name and Address of Equityholder

HSAC 2 Holdings, LLC 40 10th Avenue, Floor 7 New York, New York 10014

Pedro Granadillo 40 10th Avenue, Floor 7 New York, New York 10014

Stuart Peltz 40 10th Avenue, Floor 7 New York, New York 10014

Michael Brophy 40 10th Avenue, Floor 7 New York, New York 10014

Carsten Boess 40 10th Avenue, Floor 7 New York, New York 10014

RTW Master Fund, Ltd. 40 10th Avenue, Floor 7 New York, New York 10014

RTW Innovation Master Fund, Ltd. 40 10th Avenue, Floor 7 New York, New York 10014

RTW Venture Fund Limited 40 10th Avenue, Floor 7 New York, New York 10014

A-1

SCHEDULE B – LEGACY ORCHESTRA EQUITYHOLDERS

Name and Address of Equityholder

RTW Master Fund, Ltd. 40 10th Avenue, Floor 7 New York, New York 10014

RTW Innovation Master Fund, Ltd. 40 10th Avenue, Floor 7 New York, New York 10014

RTW Venture Fund Limited 40 10th Avenue, Floor 7 New York, New York 10014

David Hochman Orchestra BioMed, Inc. 150 Union Square Drive New Hope, PA 18938

Darren Sherman Orchestra BioMed, Inc. 150 Union Square Drive New Hope, PA 18938

Michael Kaswan Orchestra BioMed, Inc. 150 Union Square Drive New Hope, PA 18938

Yuval Mika Orchestra BioMed, Inc. 150 Union Square Drive New Hope, PA 18938

Eric Fain Orchestra BioMed, Inc. 150 Union Square Drive New Hope, PA 18938

Eric Rose Orchestra BioMed, Inc. 150 Union Square Drive New Hope, PA 18938

Jason Aryeh Orchestra BioMed, Inc. 150 Union Square Drive New Hope, PA 18938 Pam Connealy Orchestra BioMed, Inc. 150 Union Square Drive New Hope, PA 18938

Geoffrey W Smith Orchestra BioMed, Inc. 150 Union Square Drive New Hope, PA 18938

Covidien Group S.à.r.l. c/o Medtronic, Inc., Operational Headquarters 710 Medtronic Parkway Minneapolis, MN 55432-5604

Perceptive Life Sciences Master Fund 51 Astor Place, 10th Floor New York, NY 10003

B-1

INDEMNIFICATION AGREEMENT

THIS INDEMNIFICATION AGREEMENT (the "<u>Agreement</u>") is made and entered into as of ______, 2023 between Orchestra BioMed Holdings, Inc., a Delaware corporation (the "<u>Company</u>"), and ______("<u>Indemnitee</u>").

WITNESSETH THAT:

WHEREAS, highly competent persons have become more reluctant to serve corporations as directors, officers or in other capacities unless they are provided with adequate protection through insurance or adequate indemnification against inordinate risks of claims and actions against them arising out of their service to and activities on behalf of the corporation;

WHEREAS, the Board of Directors of the Company (the "Board") has determined that, in order to attract and retain qualified individuals, the Company will attempt to maintain on an ongoing basis, at its sole expense, liability insurance to protect persons serving the Company and its subsidiaries from certain liabilities. Although the furnishing of such insurance has been a customary and widespread practice among United States-based corporations and other business enterprises, the Company believes that, given current market conditions and trends, such insurance may be available to it in the future only at higher premiums and with more exclusions. At the same time, directors, officers and other persons in service to corporations or business enterprises are being increasingly subjected to expensive and time-consuming litigation relating to, among other things, matters that traditionally would have been brought only against the Company or business enterprise itself. The Bylaws and Certificate of Incorporation of the Company, each as may be amended or restated from time to time (the "Bylaws" and the "Certificate of Incorporation", respectively) require indemnification of the officers and directors of the Company. Indemnitee may also be entitled to indemnification pursuant to the General Corporation Law of the State of Delaware ("DGCL"). The Bylaws and Certificate of Incorporation and the DGCL expressly provide that the indemnification provisions set forth therein are not exclusive, and thereby contemplate that agreements may be entered into between the Company and members of the Board, officers and other persons with respect to indemnification;

WHEREAS, the uncertainties relating to such insurance and to indemnification have increased the difficulty of attracting and retaining such persons;

WHEREAS, the Board has determined that the increased difficulty in attracting and retaining such persons is detrimental to the best interests of the Company's stockholders and that the Company should act to assure such persons that there will be increased certainty of such protection in the future;

WHEREAS, it is reasonable, prudent and necessary for the Company contractually to obligate itself to indemnify, and to advance expenses on behalf of, such persons to the fullest extent permitted by applicable law so that they will serve or continue to serve the Company free from undue concern that they will not be so indemnified;

WHEREAS, this Agreement is a supplement to and in furtherance of the Bylaws and Certificate of Incorporation of the Company and any resolutions adopted pursuant thereto, and shall not be deemed a substitute therefor, nor to diminish or abrogate any rights of Indemnitee thereunder; [and]

WHEREAS, Indemnitee does not regard the protection available under the Company's Bylaws and Certificate of Incorporation and insurance as adequate in the present circumstances, and may not be willing to serve as an officer or director without adequate protection, and the Company desires Indemnitee to serve in such capacity. Indemnitee is willing to serve, continue to serve and to take on additional service for or on behalf of the Company on the condition that he or she be so indemnified; [and]

[WHEREAS, Indemnitee has certain rights to indemnification and/or insurance provided by ______ ("<u>Fund</u>"), which Indemnitee and Fund intend to be secondary to the primary obligation of the Company to indemnify Indemnitee as provided herein, with the Company's acknowledgement and agreement to the foregoing being a material condition to Indemnitee's willingness to serve on the Board.]

NOW, THEREFORE, in consideration of Indemnitee's agreement to serve as an officer or director from and after the date hereof, the parties hereto agree as follows:

1. <u>Indemnity of Indemnitee</u>. The Company hereby agrees to hold harmless and indemnify Indemnitee to the fullest extent permitted by law, as such may be amended from time to time. In furtherance of the foregoing indemnification, and without limiting the generality thereof:

(a) <u>Proceedings Other Than Proceedings by or in the Right of the Company</u>. Indemnitee shall be entitled to the rights of indemnification provided in this <u>Section 1(a)</u> if, by reason of his or her Corporate Status (as hereinafter defined), the Indemnitee is, or is threatened to be made, a party to or participant in any Proceeding (as hereinafter defined) other than a Proceeding by or in the right of the Company. Pursuant to this <u>Section 1(a)</u>, Indemnitee shall be indemnified against all Expenses (as hereinafter defined), judgments, penalties, fines and amounts paid in settlement actually and reasonably incurred by him or her, or on his or her behalf, in connection with such Proceeding or any claim, issue or matter therein, if the Indemnitee acted in good faith and in a manner the Indemnitee reasonably believed to be in or not opposed to the best interests of the Company, and with respect to any criminal Proceeding, had no reasonable cause to believe the Indemnitee's conduct was unlawful.

(b) <u>Proceedings by or in the Right of the Company</u>. Indemnitee shall be entitled to the rights of indemnification provided in this <u>Section 1(b)</u> if, by reason of his or her Corporate Status, the Indemnitee is, or is threatened to be made, a party to or participant in any Proceeding brought by or in the right of the Company. Pursuant to this <u>Section 1(b)</u>, Indemnitee shall be indemnified against all Expenses actually and reasonably incurred by the Indemnitee, or on the Indemnitee's behalf, in connection with such Proceeding if the Indemnitee acted in good faith and in a manner the Indemnitee reasonably believed to be in or not opposed to the best interests of the Company; <u>provided</u>, <u>however</u>, if applicable law so provides, no indemnification against such Expenses shall be made in respect of any claim, issue or matter in such Proceeding as to which Indemnitee shall have been adjudged to be liable to the Company unless and to the extent that the Court of Chancery of the State of Delaware shall determine that such indemnification may be made.

(c) Indemnification for Expenses of a Party Who is Wholly or Partly Successful. Notwithstanding any other provision of this Agreement, to the extent that Indemnitee is, by reason of his or her Corporate Status, a party to and is successful, on the merits or otherwise, in any Proceeding, he or she shall be indemnified to the maximum extent permitted by law, as such may be amended from time to time, against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection therewith. If Indemnitee is not wholly successful in such Proceeding but is successful, on the merits or otherwise, as to one or more but less than all claims, issues or matters in such Proceeding, the Company shall indemnify Indemnitee against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection with each successfully resolved claim, issue or matter. For purposes of this Section 1 and without limitation, the termination of any claim, issue or matter in such a Proceeding by dismissal, with or without prejudice, shall be deemed to be a successful result as to such claim, issue or matter.

2. <u>Additional Indemnity</u>. In addition to, and without regard to any limitations on, the indemnification provided for in <u>Section 1</u> of this Agreement, the Company shall and hereby does indemnify and hold harmless Indemnitee against all Expenses, judgments, penalties, fines and amounts paid in settlement actually and reasonably incurred by him or her or on his or her behalf if, by reason of his or her Corporate Status, he or she is, or is threatened to be made, a party to or participant in any Proceeding (including a Proceeding by or in the right of the Company), including, without limitation, all liability arising out of the negligence or active or passive wrongdoing of Indemnitee. The only limitation that shall exist upon the Company's obligations pursuant to this <u>Agreement</u> shall be that the Company shall not be obligated to make any payment to Indemnitee that is finally determined (under the procedures, and subject to the presumptions, set forth in <u>Sections 6</u> and <u>7</u> hereof) to be unlawful.

3. Contribution.

(a) Whether or not the indemnification provided in Sections 1 and 2 hereof is available, in respect of any Proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such Proceeding), the Company shall pay, in the first instance, the entire amount of any judgment or settlement of such Proceeding without requiring Indemnitee to contribute to such payment and the Company hereby waives and relinquishes any right of contribution it may have against Indemnitee. The Company shall not enter into any settlement of any Proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such Proceeding) unless such settlement provides for a full and final release of all claims asserted against Indemnitee.

(b) Without diminishing or impairing the obligations of the Company set forth in the preceding subparagraph, if, for any reason, Indemnitee shall elect or be required to pay all or any portion of any judgment or settlement in any Proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such Proceeding), the Company shall contribute to the amount of Expenses, judgments, fines and amounts paid in settlement actually and reasonably incurred and paid or payable by Indemnitee in proportion to the relative benefits received by the Company and all officers, directors or employees of the Company, other than Indemnitee, who are jointly liable with Indemnitee (or would be if joined in such Proceeding), on the one hand, and Indemnitee, on the other hand, from the transaction or events from which such Proceeding arose; provided, however, that the proportion determined on the basis of relative benefit may, to the extent necessary to conform to law, be further adjusted by reference to the relative fault of the Company and all officers, directors or employees of the Company other than Indemnitee who are jointly liable with Indemnitee (or would be if joined in such Proceeding), on the one hand, and Indemnitee, on the other hand, in connection with the transaction or events that resulted in such Expenses, judgments, fines or settlement amounts, as well as any other equitable considerations that applicable law may require to be considered. The relative fault of the Company and all officers, directors or employees of the Company, other than Indemnitee, who are jointly liable with Indemnitee (or would be if joined in such Proceeding), on the one hand, and Indemnitee, on the other hand, in connection with the transaction or events that resulted in such Expenses, judgments, fines or settlement amounts, as well as any other equitable considerations that applicable law may require to be considered. The relative fault of the Company and all officers, directors or employees of the Company, other than I

(c) The Company hereby agrees to fully indemnify and hold Indemnitee harmless from any claims of contribution that may be brought by officers, directors, or employees of the Company, other than Indemnitee, who may be jointly liable with Indemnitee.

(d) To the fullest extent permissible under applicable law, if the indemnification provided for in this Agreement is unavailable to Indemnitee for any reason whatsoever, the Company, in lieu of indemnifying Indemnitee, shall contribute to the amount incurred by Indemnitee, whether for judgments, fines, penalties, excise taxes, amounts paid or to be paid in settlement and/or for Expenses, in connection with any claim relating to an indemnifiable event under this Agreement, in such proportion as is deemed fair and reasonable in light of all of the circumstances of such Proceeding in order to reflect (i) the relative benefits received by the Company and Indemnitee as a result of the event(s) and/or transaction(s) giving cause to such Proceeding and/or (ii) the relative fault of the Company (and its directors, officers, employees and agents) and Indemnitee in connection with such event(s) and/or transaction(s).

4. <u>Indemnification for Expenses of a Witness</u>. Notwithstanding any other provision of this Agreement, to the extent that Indemnitee is, by reason of his or her Corporate Status, a witness, or is made (or asked) to respond to discovery requests, in any Proceeding to which Indemnitee is not a party, he or she shall be indemnified against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection therewith.

5. <u>Advancement of Expenses</u>. Notwithstanding any other provision of this Agreement, the Company shall advance all Expenses incurred by or on behalf of Indemnitee in connection with any Proceeding by reason of Indemnitee's Corporate Status within thirty (30) days after the receipt by the Company of a statement or statements from Indemnitee requesting such advance or advances from time to time, whether prior to or after final disposition of such Proceeding. Such statement or statements shall reasonably evidence the Expenses incurred by Indemnitee and shall include or be preceded or accompanied by a written undertaking by or on behalf of Indemnitee to repay any Expenses advanced if it shall ultimately be determined by a court of competent jurisdiction in a final judgment, not subject to appeal, that Indemnitee is not entitled to be indemnified against such Expenses. Any advances and undertakings to repay pursuant to this Section 5 shall be unsecured and interest free.

6. <u>Procedures and Presumptions for Determination of Entitlement to Indemnification</u>. It is the intent of this Agreement to secure for Indemnitee rights of indemnity that are as favorable as may be permitted under the DGCL and public policy of the State of Delaware. Accordingly, the parties agree that the following procedures and presumptions shall apply in the event of any question as to whether Indemnitee is entitled to indemnification under this Agreement:

(a) To obtain indemnification under this Agreement, Indemnitee shall submit to the Company a written request, including therein or therewith such documentation and information as is reasonably available to Indemnitee and is reasonably necessary to determine whether and to what extent Indemnitee is entitled to indemnification. The Secretary of the Company shall, promptly upon receipt of such a request for indemnification, advise the Board in writing that Indemnitee has requested indemnification. Notwithstanding the foregoing, any failure of Indemnitee to provide such a request to the Company, or to provide such a request in a timely fashion, shall not relieve the Company of any liability that it may have to Indemnitee unless, and to the extent that, such failure actually and materially prejudices the interests of the Company.

(b) Upon written request by Indemnitee for indemnification pursuant to the first sentence of <u>Section 6(a)</u> hereof, a determination with respect to Indemnitee's entitlement thereto shall be made in the specific case by one of the following four methods, which shall be at the election of the Board (1) by a majority vote of the Disinterested Directors, even though less than a quorum, (2) by a committee of Disinterested Directors designated by a majority vote of the Disinterested Directors, even though less than a quorum, (3) if there are no Disinterested Directors or if the Disinterested Directors so direct, by independent legal counsel in a written opinion to the Board, a copy of which shall be delivered to the Indemnitee or (4) if so directed by the Board, by the stockholders of the Company.

(c) If the determination of entitlement to indemnification is to be made by Independent Counsel pursuant to Section 6(b) hereof, the Independent Counsel shall be selected as provided in this Section 6(c). The Independent Counsel shall be selected by the Board. Indemnitee may, within ten (10) days after such written notice of selection shall have been given, deliver to the Company a written objection to such selection; provided, however, that such objection may be asserted only on the ground that the Independent Counsel so selected does not meet the requirements of "Independent Counsel" as defined in Section 13 of this Agreement, and the objection shall set forth with particularity the factual basis of such assertion. Absent a proper and timely objection, the person so selected shall act as Independent Counsel. If a written objection is made and substantiated, the Independent Counsel selected may not serve as Independent Counsel unless and until such objection is withdrawn or a court has determined that such objection is without merit. If, within twenty (20) days after submission by Indemnitee of a written request for indemnification pursuant to Section 6(a) hereof, no Independent Counsel shall have been made by the Indemnite to the Company's selection of Independent Counsel and/or for the appointment as Independent Counsel of a person selected by the court or by such other person as the court shall designate, and the person with respect to whom all objections are so resolved or the person so appointed shall act as Independent Counsel in connection with acting pursuant to Section 6(b) hereof, many ball pay any and all reasonable fees and expenses incident to the procedures of this Section 6(b) hereof. The Company with respect to whom all objections are so resolved or the person so appointed by such Independent Counsel in connection with acting pursuant to Section 6(b) hereof, and the Company shall pay all reasonable fees and expenses incident to the procedures of this Section 6(c), regardless of the manner

(d) In making a determination with respect to entitlement to indemnification hereunder, the person or persons or entity making such determination shall presume that Indemnitee is entitled to indemnification under this Agreement. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence. Neither the failure of the Company (including by its directors or independent legal counsel) to have made a determination prior to the commencement of any action pursuant to this Agreement that indemnification is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor an actual determination by the Company (including by its directors or create a presumption that Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct.

(e) Indemnitee shall be deemed to have acted in good faith if Indemnitee's action is based on the records or books of account of the Enterprise (as hereinafter defined), including financial statements, or on information supplied to Indemnitee by the officers of the Enterprise in the course of their duties, or on the advice of legal counsel for the Enterprise or on information or records given or reports made to the Enterprise by an independent certified public accountant or by an appraiser or other expert selected with reasonable care by the Enterprise. In addition, the knowledge and/or actions, or failure to act, of any director, officer, agent or employee of the Enterprise shall not be imputed to Indemnitee for purposes of determining the right to indemnification under this Agreement. Whether or not the foregoing provisions of this Section 6(e) are satisfied, it shall in any event be presumed that Indemnitee has at all times acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Company. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence.

(f) If the person, persons or entity empowered or selected under Section 6 to determine whether Indemnitee is entitled to indemnification shall not have made a determination within sixty (60) days after receipt by the Company of the request therefor, the requisite determination of entitlement to indemnification shall be deemed to have been made and Indemnitee shall be entitled to such indemnification absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law; provided, however, that such sixty (60) day period may be extended for a reasonable time, not to exceed an additional thirty (30) days, if the person, persons or entity making such determination with respect to entitlement to indemnification in good faith requires such additional time to obtain or evaluate documentation and/or information relating thereto; and provided further, that the foregoing provisions of this Section 6(f) shall not apply if the determination of entitlement to indemnification is to be made by the stockholders pursuant to Section 6(b) of this Agreement and if (A) within fifteen (15) days after receipt by the Company of the request for such determination, the Board or the Disinterested Directors, if appropriate, resolve to submit such determination to the stockholders for their consideration at an annual meeting thereof to be held within seventy five (75) days after such receipt and such determination, and such meeting is held for such purpose within sixty (60) days after heaving been so called and such determination is made thereat.

(g) Indemnitee shall cooperate with the person, persons or entity making such determination with respect to Indemnitee's entitlement to indemnification, including providing to such person, persons or entity upon reasonable advance request any documentation or information that is not privileged or otherwise protected from disclosure and that is reasonably available to Indemnitee and reasonably necessary to such determination. Any Independent Counsel, member of the Board or stockholder of the Company shall act reasonably and in good faith in making a determination regarding the Indemnitee's entitlement to indemnification under this Agreement. Any costs or expenses (including attorneys' fees and disbursements) incurred by Indemnitee in so cooperating with the person, persons or entity making such determination shall be borne by the Company (irrespective of the determination as to Indemnitee's entitlement to indemnification) and the Company hereby indemnifies and agrees to hold Indemnitee harmless therefrom.

(h) The Company acknowledges that a settlement or other disposition short of final judgment may be successful if it permits a party to avoid expense, delay, distraction, disruption and uncertainty. In the event that any Proceeding to which Indemnitee is a party is resolved in any manner other than by adverse judgment against Indemnitee (including, without limitation, settlement of such Proceeding with or without payment of money or other consideration) it shall be presumed that Indemnitee has been successful on the merits or otherwise in such Proceeding. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence.

(i) The termination of any Proceeding or of any claim, issue or matter therein, by judgment, order, settlement or conviction, or upon a plea of nolo contendere or its equivalent, shall not (except as otherwise expressly provided in this Agreement) of itself adversely affect the right of Indemnitee to indemnification or create a presumption that Indemnitee did not act in good faith and in a manner that he or she reasonably believed to be in or not opposed to the best interests of the Company or, with respect to any criminal Proceeding, that Indemnitee had reasonable cause to believe that his or her conduct was unlawful.

7. Remedies of Indemnitee.

(a) In the event that (i) a determination is made pursuant to Section 6 of this Agreement that Indemnitee is not entitled to indemnification under this Agreement, (ii) advancement of Expenses is not timely made pursuant to Section 5 of this Agreement, (iii) no determination of entitlement to indemnification is made pursuant to Section 6(b) of this Agreement within ninety (90) days after receipt by the Company of the request for indemnification, (iv) payment of indemnification is not made pursuant to this Agreement within ten (10) days after receipt by the Company of a written request therefor or (v) payment of indemnification is not made pursuant to 10 days after a determination has been made that Indemnitee is entitled to an adjudication in an appropriate court of the State of Delaware, or in any other court of competent jurisdiction, of Indemnitee's entitlement to such indemnification. Indemnitee shall commence such proceeding pursuant to this Section 7(a). The Company shall not oppose Indemnitee's right to seek any such adjudication.

(b) In the event that a determination shall have been made pursuant to $\underline{\text{Section 6(b)}}$ of this Agreement that Indemnitee is not entitled to indemnification, any judicial proceeding commenced pursuant to this $\underline{\text{Section 7}}$ shall be conducted in all respects as a de novo trial on the merits, and Indemnitee shall not be prejudiced by reason of the adverse determination under $\underline{\text{Section 6(b)}}$.

(c) If a determination shall have been made pursuant to <u>Section 6(b)</u> of this Agreement that Indemnitee is entitled to indemnification, the Company shall be bound by such determination in any judicial proceeding commenced pursuant to this <u>Section 7</u>, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's misstatement not materially misleading in connection with the application for indemnification, or (ii) a prohibition of such indemnification under applicable law.

(d) In the event that Indemnitee, pursuant to this <u>Section 7</u>, seeks a judicial adjudication of his or her rights under, or to recover damages for breach of, this Agreement, or to recover under any directors' and officers' liability insurance policies maintained by the Company, the Company shall pay on his or her behalf, in advance, any and all Expenses actually and reasonably incurred by him or her in such judicial adjudication, regardless of whether Indemnitee ultimately is determined to be entitled to such indemnification, advancement of Expenses or insurance recovery.

(e) The Company shall be precluded from asserting in any judicial proceeding commenced pursuant to this <u>Section 7</u> that the procedures and presumptions of this Agreement are not valid, binding and enforceable and shall stipulate in any such court that the Company is bound by all the provisions of this Agreement. The Company shall indemnify Indemnitee against any and all Expenses and, if requested by Indemnitee, shall (within ten (10) days after receipt by the Company of a written request therefore) advance, to the extent not prohibited by law, such Expenses to Indemnitee, which are incurred by Indemnitee in connection with any action brought by Indemnitee for indemnification or advance of Expenses from the Company under this Agreement or under any directors' and officers' liability insurance policies maintained by the Company, regardless of whether Indemnitee ultimately is determined to be entitled to such indemnification, advancement of Expenses or insurance recovery, as the case may be.

(f) Notwithstanding anything in this Agreement to the contrary, no determination as to entitlement to indemnification under this Agreement shall be required to be made prior to the final disposition of the Proceeding; *provided, however*, that the foregoing shall not affect the obligations of the Company to advance Expenses prior to the final disposition of the Proceeding pursuant to this Agreement.

8. Non-Exclusivity; Survival of Rights; Insurance; Primacy of Indemnification; Subrogation.

(a) The rights of indemnification as provided by this Agreement shall not be deemed exclusive of any other rights to which Indemnitee may at any time be entitled under applicable law, the Certificate of Incorporation, the Bylaws, any agreement, a vote of stockholders, a resolution of directors of the Company, or otherwise. No amendment, alteration or repeal of this Agreement or of any provision hereof shall limit or restrict any right of Indemnitee under this Agreement in respect of any action taken or omitted by such Indemnitee in his or her Corporate Status prior to such amendment, alteration or repeal. To the extent that a change in the DGCL, whether by statute or judicial decision, permits greater indemnification than would be afforded currently under the Certificate of Incorporation, Bylaws and this Agreement, it is the intent of the parties hereto that Indemnitee shall enjoy by this Agreement the greater benefits so afforded by such change. No right or remedy herein conferred is intended to be exclusive of any other right or remedy, and every other right and remedy shall be cumulative and in addition to every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. The assertion or employment of any right or remedy hereunder, or otherwise, shall not prevent the concurrent assertion or employment of any other right or remedy.

(b) To the extent that the Company maintains an insurance policy or policies providing liability insurance for directors, officers, employees, or agents or fiduciaries of the Company or of any other Enterprise that such person serves at the request of the Company, Indemnitee shall be covered by such policy or policies in accordance with its or their terms to the maximum extent of the coverage available for any director, officer, employee, agent or fiduciary under such policy or policies. If, at the time of the receipt of a notice of a claim pursuant to the terms hereof, the Company has directors' and officers' liability insurance in effect, the Company shall give prompt notice of the commencement of such proceeding to the insurers in accordance with the procedures set forth in the respective policies. The Company shall thereafter take all necessary or desirable action to cause such insurers to pay, on behalf of the Indemnitee, all amounts payable as a result of such proceeding in accordance with the terms of such policies.

(c) [The Company hereby acknowledges that Indemnitee has certain rights to indemnification, advancement of expenses and/or insurance provided by the Fund and certain of its affiliates (collectively, the "Fund Indemnitors"). The Company hereby agrees (i) that it is the indemnitor of first resort (i.e., its obligations to Indemnitee are primary and any obligation of the Fund Indemnitors to advance expenses or to provide indemnification for the same expenses or liabilities incurred by Indemnitee are secondary), (ii) that it shall be required to advance the full amount of expenses incurred by Indemnitee and shall be liable for the full amount of all Expenses, judgments, penalties, fines and amounts paid in settlement to the extent legally permitted and as required by the terms of this Agreement and the Certificate of Incorporation or Bylaws of the Company (or any other agreement between the Company and Indemnitee), without regard to any rights Indemnitee may have against the Fund Indemnitors and (iii) that it irrevocably waives, relinquishes and releases the Fund Indemnitors from any and all claims against the Fund Indemnitors on behalf of Indemnitee with respect to any claim for which Indemnitee has sought indemnification from the Company shall affect the foregoing and the Fund Indemnitors shall have a right of contribution and/or be subrogated to the extent of such advancement or payment to all of the rights of recovery of Indemnitee against the Company. The Company and Indemnitors are express third party beneficiaries of the terms of this Section 8(c).]

(d) [Except as provided in paragraph (c) above, i]In the event of any payment under this Agreement, the Company shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee [(other than against the Fund Indemnitors)], who shall execute all papers required and take all action necessary to secure such rights, including execution of such documents as are necessary to enable the Company to bring suit to enforce such rights.

(e) [Except as provided in paragraph (c) above, t]The Company shall not be liable under this Agreement to make any payment of amounts otherwise indemnifiable hereunder if and to the extent that Indemnitee has otherwise actually received such payment under any insurance policy, contract, agreement or otherwise.

(f) [Except as provided in paragraph (c) above, t]The Company's obligation to indemnify or advance Expenses hereunder to Indemnitee who is or was serving at the request of the Company as a director, officer, employee or agent of any other Enterprise shall be reduced by any amount Indemnitee has actually received as indemnification or advancement of expenses from such other Enterprise.

9. Exception to Right of Indemnification. Notwithstanding any provision in this Agreement, the Company shall not be obligated under this Agreement to make any indemnity in connection with any claim made against Indemnitee:

(a) for which payment has actually been made to or on behalf of Indemnitee under any insurance policy or other indemnity provision, except with respect to any excess beyond the amount paid under any insurance policy or other indemnity provision[, provided, that the foregoing shall not affect the rights of Indemnitee or the Fund Indemnitors set forth in <u>Section 8(c)</u> above]; or

(b) for an accounting of profits made from the purchase and sale (or sale and purchase) by Indemnitee of securities of the Company within the meaning of Section 16(b) of the Securities Exchange Act of 1934, as amended, or similar provisions of state statutory law or common law; or

(c) in connection with any Proceeding (or any part of any Proceeding) initiated by Indemnitee, including any Proceeding (or any part of any Proceeding) initiated by Indemnitee against the Company or its directors, officers, employees or other indemnitees, unless (i) the Board authorized the Proceeding (or any part of any Proceeding) prior to its initiation, (ii) the Company provides the indemnification, in its sole discretion, pursuant to the powers vested in the Company under applicable law, (iii) otherwise authorized by <u>Section 7(d)</u> hereof, or (iv) otherwise required by applicable law.

10. Duration of Agreement. All agreements and obligations of the Company contained herein shall continue during the period Indemnitee is an officer or director of the Company (or is or was serving at the request of the Company as a director, officer, employee or agent of any other Enterprise) and shall continue until the latest of (i) three (3) years after the date that Indemnitee shall have ceased to serve as a director or agent of the Company (as indicated in the parenthetical above), (ii) the date of final termination of any Proceeding, including any appeal, then pending in respect of which Indemnitee is granted rights of indemnification or advancement hereunder (and of any proceeding, including any appeal, commenced pursuant to <u>Section 7</u> of hereof), or (iii) the expiration of all statutes of limitation applicable to possible Proceedings to which Indemnitee may be subject arising out of or relating to Indemnification can be provided under this Agreement. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective successors (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all of the business or assets of the Company), assigns, spouses, heirs, executors and personal and legal representatives. The Company shall require and cause any direct or indirect successor (whether by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of the Company), assigns, spouses, heirs, executors and personal and legal representatives. The Company shall require and cause any direct or indirect successor (whether by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of the Company statisfactory to Indemnitee, expressly to assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place.

11. <u>Security</u>. To the extent requested by Indemnitee and approved by the Board, the Company may at any time and from time to time provide security to Indemnitee for the Company's obligations hereunder through an irrevocable bank line of credit, funded trust or other collateral. Any such security, once provided to Indemnitee, may not be revoked or released without the prior written consent of the Indemnitee.

12. Enforcement.

(a) The Company expressly confirms and agrees that it has entered into this Agreement and assumes the obligations imposed on it hereby in order to induce Indemnitee to serve as an officer or director of the Company, and the Company acknowledges that Indemnitee is relying upon this Agreement in serving as an officer or director of the Company.

(b) Without limiting any of the rights of Indemnitee under the Bylaws or Certificate of Incorporation, this Agreement constitutes the entire agreement between the parties hereto with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral, written and implied, between the parties hereto with respect to the subject matter hereof.

(c) The Company shall not seek from a court, or agree to, a "<u>bar order</u>" that would have the effect of prohibiting or limiting the Indemnitee's rights to receive advancement of Expenses under this Agreement.

13. Definitions. For purposes of this Agreement:

(a) "<u>Corporate Status</u>" describes the status of a person who is or was a director, officer, employee, agent or fiduciary of the Company or of any other Enterprise that such person is or was serving at the express written request of the Company.

(b) "Disinterested Director" means a director of the Company who is not and was not a party to the Proceeding in respect of which indemnification is sought by Indemnitee.

(c) "<u>Enterprise</u>" shall mean the Company, any Subsidiary of the Company and any other corporation, partnership, limited liability company, joint venture, trust, employee benefit plan or other enterprise that Indemnitee is or was serving at the express written request of the Company as a director, officer, trustee, administrator, general partner, manager, managing member, fiduciary, board of directors' committee member, employee, agent or fiduciary. For purposes hereof, if Indemnitee is serving or has served as a director, officer, trustee, administrator, general partner, manager, managing member, fiduciary of a Subsidiary, Indemnitee shall be deemed to be serving, or to have been serving, in such capacity at the express written request of the Company.

(d) "Expenses" shall include all reasonable attorneys' fees, retainers, court costs, transcript costs, fees of experts, witness fees, travel expenses, duplicating costs, printing and binding costs, telephone charges, postage, delivery service fees and all other disbursements or expenses of the types customarily incurred in connection with prosecuting, defending, preparing to prosecute or defend, investigating, participating, being or preparing to be a witness in, or responding to, or objecting to, a request to provide discovery in, any Proceeding. Expenses also shall include Expenses incurred in connection with any appeal resulting from any Proceeding and any federal, state, local or foreign taxes imposed on the Indemnitee as a result of the actual or deemed receipt of any payments under this Agreement, including without limitation the premium, security for and other costs relating to any cost bond, supersede as bond or other appeal bond or its equivalent. Expenses, however, shall not include amounts paid in settlement by Indemnitee or the amount of judgments or fines against Indemnitee.

(e) "Independent Counsel" means a law firm, or a member of a law firm, that is experienced in matters of corporation law and neither presently is, nor in the past five years has been, retained to represent (i) the Company or Indemnitee in any matter material to either such party (other than with respect to matters concerning Indemnitee under this Agreement, or of other indemnitees under similar indemnification agreements), or (ii) any other party to the Proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement. The Company agrees to pay the reasonable fees of the Independent Counsel referred to above and to fully indemnify such counsel against any and all Expenses, claims, liabilities and damages arising out of or relating to this Agreement or its engagement pursuant hereto.

(f) "Proceeding" includes any threatened, pending or completed action, suit, arbitration, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or any other actual, threatened or completed proceeding, whether brought by or in the right of the Company or otherwise and whether civil, criminal, administrative or investigative, in which Indemnitee was, is or will be involved as a party or otherwise, by reason of his or her Corporate Status, by reason of any action taken by him or her or of any inaction on his or her part while acting in his or her Corporate Status; in each case whether or not he or she is acting or serving in any such capacity at the time any liability or expense is incurred for which indemnification can be provided under this Agreement; including one pending on or before the date of this Agreement.

(g) "*Subsidiary*" means any corporation, partnership, joint venture, limited liability company, association, or other entity in which the Company owns, directly or indirectly, fifty percent (50%) or more of the outstanding equity securities or interests, the holders of which are generally entitled to vote for the election of the board of directors or other governing body of such entity.

14. <u>Severability</u>. The invalidity or unenforceability of any provision hereof shall in no way affect the validity or enforceability of any other provision. Without limiting the generality of the foregoing, this Agreement is intended to confer upon Indemnitee indemnification rights to the fullest extent permitted by applicable laws. In the event any provision hereof conflicts with any applicable law, such provision shall be deemed modified, consistent with the aforementioned intent, to the extent necessary to resolve such conflict.

15. <u>Modification and Waiver</u>. No supplement, modification, termination or amendment of this Agreement shall be binding unless executed in writing by both of the parties hereto. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provisions hereof (whether or not similar) nor shall such waiver constitute a continuing waiver.

16. <u>Notice by Indemnitee</u>. Indemnitee agrees promptly to notify the Company in writing upon being served with or otherwise receiving any summons, citation, subpoena, complaint, indictment, information or other document relating to any Proceeding or matter that may be subject to indemnification covered hereunder. The failure to so notify the Company shall not relieve the Company of any obligation that it may have to Indemnitee under this Agreement or otherwise unless and only to the extent that such failure or delay materially prejudices the Company.

17. Notices. All notices and other communications given or made pursuant to this Agreement shall be in writing and shall be deemed effectively given (a) upon personal delivery to the party to be notified, (b) when sent by confirmed electronic mail or facsimile if sent during normal business hours of the recipient, and if not so confirmed, then on the next business day, (c) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid or (d) one (1) day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written verification of receipt. All communications shall be sent:

- (a) To Indemnitee at the address set forth below Indemnitee signature hereto.
- (b) To the Company at:

Orchestra BioMed Holdings, Inc. 150 Union Square Drive New Hope, PA 18938 Attention: Chief Executive Officer

or to such other address as may have been furnished to Indemnitee by the Company or to the Company by Indemnitee, as the case may be.

18. <u>Counterparts</u>. This Agreement may be executed in two (2) or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same Agreement. Counterparts may be delivered via facsimile, electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, e.g., www.docusign.com) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.

19. <u>Headings</u>. The headings of the paragraphs of this Agreement are inserted for convenience only and shall not be deemed to constitute part of this Agreement or to affect the construction thereof.

20. <u>Governing Law and Consent to Jurisdiction</u>. This Agreement and the legal relations among the parties shall be governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without regard to its conflict of laws rules. The Company and Indemnitee hereby irrevocably and unconditionally (i) agree that any action or proceeding arising out of or in connection with this Agreement shall be brought only in the Chancery Court of the State of Delaware (the "Delaware Court"), and not in any other state or federal court in the United States of America or any court in any other country, (ii) consent to submit to the exclusive jurisdiction of the Delaware Court for purposes of any action or proceeding arising out of or in connection with this Agreement, (iii) waive any objection to the laying of venue of any such action or proceeding in the Delaware Court and (iv) waive, and agree not to plead or to make, any claim that any such action or proceeding brought in the Delaware Court has been brought in an improper or inconvenient forum.

Signature Page To Follow



IN WITNESS WHEREOF, the parties hereto have executed this Indemnification Agreement on and as of the day and year first above written.

COMPANY

ORCHESTRA BIOMED HOLDINGS, INC.

By: Name: Title: Electronic Mail

INDEMNITEE

By: Name: Address: Electronic Mail:

SIGNATURE PAGE TO INDEMNIFICATION AGREEMENT

Orchestra BioMed Holdings, Inc. 2023 Equity Incentive Plan

Adopted by the Board of Directors: December 12, 2022 Approved by the Stockholders: January 26, 2023 Effective Date: January 26, 2023

1. GENERAL.

(a) Plan Purpose. The Company, by means of the Plan, seeks to secure and retain the services of Employees, Directors and Consultants, to provide incentives for such persons to exert maximum efforts for the success of the Company and any Affiliate and to provide a means by which such persons may be given an opportunity to benefit from increases in value of the Common Stock through the granting of Awards.

(b) Available Awards. The Plan provides for the grant of the following Awards: (i) Incentive Stock Options; (ii) Nonstatutory Stock Options; (iii) SARs; (iv) Restricted Stock Awards; (v) RSU Awards; and (vi) Other Awards.

(c) Adoption Date; Effective Date. The Plan will come into existence on the Adoption Date, but no Award may be granted prior to the Effective Date.

2. SHARES SUBJECT TO THE PLAN.

(a) Share Reserve. Subject to adjustment in accordance with Section 2(c) and any adjustments as necessary to implement any Capitalization Adjustments, the aggregate number of shares of Common Stock that may be issued pursuant to Awards will not exceed 3,455,303 shares plus a number of shares of Common Stock equal to the number of Returning Shares, if any, as such shares become available from time to time. In addition, subject to any adjustments as necessary to implement any Capitalization Adjustments, such aggregate number of shares of Common Stock will automatically increase on January 1 of each year for a period of ten years commencing on January 1, 2023 and ending on (and including) January 1, 2032, in an amount equal to the lesser of (i) 4.8% of the total number of shares of Common Stock outstanding on December 31 of the preceding year, (ii) 3,036,722 shares of Common Stock, and (iii) such number of shares of Common Stock determined by the Board or the Compensation Committee prior to January 1st of a given year.

(b) Aggregate Incentive Stock Option Limit. Notwithstanding anything to the contrary in Section 2(a) and subject to any adjustments as necessary to implement any Capitalization Adjustments, the aggregate maximum number of shares of Common Stock that may be issued pursuant to the exercise of Incentive Stock Options is 3,455,303 shares, which such amount shall be increased commencing on January 1, 2023 and ending on (and including) January 1, 2032, in an amount equal to the lesser of (i) 4.8% of the total number of shares of Common Stock outstanding on December 31 of the preceding year, (ii) 3,036,722 shares of Common Stock, and (iii) such number of shares of Common Stock determined by the Board or the Compensation Committee prior to January 1st of a given year.

(c) Share Reserve Operation.

(i) Limit Applies to Common Stock Issued Pursuant to Awards. For clarity, the Share Reserve is a limit on the number of shares of Common Stock that may be issued pursuant to Awards and does not limit the granting of Awards, except that the Company will keep available at all times the number of shares of Common Stock reasonably required to satisfy its obligations to issue shares pursuant to such Awards. Shares may be issued in connection with a merger or acquisition as permitted by, as applicable, Nasdaq Listing Rule 5635(c), NYSE Listed Company Manual Section 303A.08, NYSE American Company Guide Section 711 or other applicable rule, and such issuance will not reduce the number of shares available for issuance under the Plan.

(ii) Actions that Do Not Constitute Issuance of Common Stock and Do Not Reduce Share Reserve. The following actions do not result in an issuance of shares under the Plan and accordingly do not reduce the number of shares subject to the Share Reserve and available for issuance under the Plan: (1) the expiration or termination of any portion of an Award without the shares covered by such portion of the Award having been issued, (2) the settlement of any portion of an Award in cash (i.e., the Participant receives cash rather than Common Stock), (3) the withholding of shares that would otherwise be issued by the Company to satisfy the exercise, strike or purchase price of an Award; or (4) the withholding of shares that would otherwise be issued by the Company to satisfy a tax withholding obligation in connection with an Award. For the avoidance of doubt, with respect to a SAR, only shares of Common Stock which are issued upon settlement of the SAR shall count towards reducing the number of shares available for issuance under the Plan.

(iii) Reversion of Previously Issued Shares of Common Stock to Share Reserve. The following shares of Common Stock previously issued pursuant to an Award and accordingly initially deducted from the Share Reserve will be added back to the Share Reserve and again become available for issuance under the Plan: (1) any shares that are forfeited back to or repurchased by the Company because of a failure to meet a contingency or condition required for the vesting of such shares; (2) any shares that are reacquired by the Company to satisfy the exercise, strike or purchase price of an Award; and (3) any shares that are reacquired by the Company to satisfy a tax withholding obligation in connection with an Award.

3. ELIGIBILITY AND LIMITATIONS.

(a) Eligible Award Recipients. Subject to the terms of the Plan, Employees, Directors and Consultants are eligible to receive Awards.

(b) Specific Award Limitations.

(i) Limitations on Incentive Stock Option Recipients. Incentive Stock Options may be granted only to Employees of the Company or a "parent corporation" or "subsidiary corporation" thereof (as such terms are defined in Sections 424(e) and (f) of the Code).

(ii) Incentive Stock Option \$100,000 Limitation. To the extent that the aggregate Fair Market Value (determined at the time of grant) of Common Stock with respect to which Incentive Stock Options are exercisable for the first time by any Optionholder during any calendar year (under all plans of the Company and any Affiliates) exceeds \$100,000 (or such other limit established in the Code) or otherwise does not comply with the rules governing Incentive Stock Options, the Options or portions thereof that exceed such limit (according to the order in which they were granted) or otherwise do not comply with such rules will be treated as Nonstatutory Stock Options, notwithstanding any contrary provision of the applicable Option Agreement(s).



(iii) Limitations on Incentive Stock Options Granted to Ten Percent Stockholders. A Ten Percent Stockholder may not be granted an Incentive Stock Option unless (i) the exercise price of such Option is at least 110% of the Fair Market Value on the date of grant of such Option and (ii) the Option is not exercisable after the expiration of five years from the date of grant of such Option.

(iv) Limitations on Nonstatutory Stock Options and SARs. Nonstatutory Stock Options and SARs may not be granted to Employees, Directors and Consultants who are providing Continuous Service only to any "parent" of the Company (as such term is defined in Rule 405) unless the stock underlying such Awards is treated as "service recipient stock" under Section 409A because the Awards are granted pursuant to a corporate transaction (such as a spin off transaction) or unless such Awards otherwise comply with the distribution requirements of Section 409A.

(c) Aggregate Incentive Stock Option Limit. The aggregate maximum number of shares of Common Stock that may be issued pursuant to the exercise of Incentive Stock Options is the number of shares specified in Section 2(b).

(d) Non-Employee Director Compensation Limit. The aggregate value of all compensation granted or paid, as applicable, in each case following the Effective Date, to any individual for service as a Non-Employee Director with respect to any fiscal year, including Awards granted and cash fees paid by the Company to such Non-Employee Director for his or her service as a Non-Employee Director, will not exceed (i) \$750,000 in total value or (ii) in the event such Non-Employee Director is first appointed or elected to the Board during such fiscal year, \$1,000,000 in total value, in each case calculating the value of any equity awards based on the grant date fair value of such equity awards for financial reporting purposes.

4. OPTIONS AND STOCK APPRECIATION RIGHTS.

Each Option and SAR will have such terms and conditions as determined by the Board. Each Option will be designated in writing as an Incentive Stock Option or Nonstatutory Stock Option at the time of grant; provided, however, that if an Option is not so designated, then such Option will be a Nonstatutory Stock Option, and the shares purchased upon exercise of each type of Option will be separately accounted for. Each SAR will be denominated in shares of Common Stock equivalents. The terms and conditions of separate Options and SARs need not be identical; provided, however, that each Option Agreement and SAR Agreement will conform (through incorporation of provisions hereof by reference in the Award Agreement or otherwise) to the substance of each of the following provisions:

(a) Term. Subject to Section 3(b) regarding Ten Percent Stockholders, no Option or SAR will be exercisable after the expiration of ten years from the date of grant of such Award or such shorter period specified in the Award Agreement.

(b) Exercise or Strike Price. Subject to Section 3(b) regarding Ten Percent Stockholders, the exercise or strike price of each Option or SAR will not be less than 100% of the Fair Market Value on the date of grant of such Award. Notwithstanding the foregoing, an Option or SAR may be granted with an exercise or strike price lower than 100% of the Fair Market Value on the date of grant of such Award if such Award is granted pursuant to an assumption of or substitution for another option or stock appreciation right pursuant to a Corporate Transaction and in a manner consistent with the provisions of Sections 409A and, if applicable, 424(a) of the Code.

(c) Exercise Procedure and Payment of Exercise Price for Options. In order to exercise an Option, the Participant must provide notice of exercise to the Plan Administrator in accordance with the procedures specified in the Option Agreement or otherwise provided by the Company. The Board has the authority to grant Options that do not permit all of the following methods of payment (or otherwise restrict the ability to use certain methods) and to grant Options that require the consent of the Company to utilize a particular method of payment. The exercise price of an Option may be paid, to the extent permitted by Applicable Law and as determined by the Board, by one or more of the following methods of payment to the extent set forth in the Option Agreement:

(i) by cash or check, bank draft or money order payable to the Company;

(ii) pursuant to a "cashless exercise" program developed under Regulation T as promulgated by the U.S. Federal Reserve Board that, prior to the issuance of the Common Stock subject to the Option, results in either the receipt of cash (or check) by the Company or the receipt of irrevocable instructions to pay the exercise price to the Company from the sales proceeds;

(iii) by delivery to the Company (either by actual delivery or attestation) of shares of Common Stock that are already owned by the Participant free and clear of any liens, claims, encumbrances or security interests, with a Fair Market Value on the date of exercise that does not exceed the exercise price, provided that (1) at the time of exercise the Common Stock is publicly traded, (2) any remaining balance of the exercise price not satisfied by such delivery is paid by the Participant in cash or other permitted form of payment, (3) such delivery would not violate any Applicable Law or agreement restricting the redemption of the Common Stock, (4) any certificated shares are endorsed or accompanied by an executed assignment separate from certificate and (5) such shares have been held by the Participant for any minimum period necessary to avoid adverse accounting treatment as a result of such delivery;

(iv) if the Option is a Nonstatutory Stock Option, by a "net exercise" arrangement pursuant to which the Company will reduce the number of shares of Common Stock issuable upon exercise by the largest whole number of shares with a Fair Market Value on the date of exercise that does not exceed the exercise price, provided that (1) such shares used to pay the exercise price will not be exercisable thereafter and (2) any remaining balance of the exercise price not satisfied by such net exercise is paid by the Participant in cash or other permitted form of payment; or

(v) in any other form of consideration that may be acceptable to the Board and permissible under Applicable Law.

(d) Exercise Procedure and Payment of Appreciation Distribution for SARs. In order to exercise any SAR, the Participant must provide notice of exercise to the Plan Administrator in accordance with the SAR Agreement. The appreciation distribution payable to a Participant upon the exercise of a SAR will not be greater than an amount equal to the excess of (i) the aggregate Fair Market Value on the date of exercise of a number of shares of Common Stock equal to the number of Common Stock equivalents that are vested and being exercised under such SAR, over (ii) the strike price of such SAR. Such appreciation distribution may be paid to the Participant in the form of Common Stock or cash (or any combination of Common Stock and cash) or in any other form of payment, as determined by the Board and specified in the SAR Agreement.

(e) Transferability. Options and SARs may not be transferred to third-party financial institutions for value. The Board may impose such additional limitations on the transferability of an Option or SAR as it determines. In the absence of any such determination by the Board, the following restrictions on the transferability of Options and SARs will apply, provided that except as explicitly provided herein, neither an Option nor a SAR may be transferred for consideration and provided, further, that if an Option is an Incentive Stock Option, such Option may be deemed to be a Nonstatutory Stock Option as a result of such transfer:

(i) Restrictions on Transfer. An Option or SAR will not be transferable, except by will or by the laws of descent and distribution, and will be exercisable during the lifetime of the Participant only by the Participant; provided, however, that the Board may permit transfer of an Option or SAR in a manner that is not prohibited by applicable tax and securities laws upon the Participant's request, including to a trust if the Participant is considered to be the sole beneficial owner of such trust (as determined under Section 671 of the Code and applicable U.S. state law) while such Option or SAR is held in such trust, provided that the Participant and the trustee enter into a transfer and other agreements required by the Company.

(ii) Domestic Relations Orders. Notwithstanding the foregoing, subject to the execution of transfer documentation in a format acceptable to the Company and subject to the approval of the Board or a duly authorized Officer, an Option or SAR may be transferred pursuant to a domestic relations order.

(f) Vesting. The Board may impose such restrictions on or conditions to the vesting and/or exercisability of an Option or SAR as determined by the Board and vesting conditions may include achievement of one or more Performance Goals. Except as otherwise provided in the applicable Award Agreement or other written agreement between a Participant and the Company, vesting of Options and SARs will cease upon termination of the Participant's Continuous Service.

(g) Termination of Continuous Service for Cause. Except as explicitly otherwise provided in the Award Agreement or other written agreement between a Participant and the Company, if a Participant's Continuous Service is terminated for Cause, the Participant's Options and SARs will terminate and be forfeited immediately upon such termination of Continuous Service, and the Participant will be prohibited from exercising any portion (including any vested portion) of such Awards on and after the date of such termination of Continuous Service and the Participant will have no further right, title or interest in such forfeited Award, the shares of Common Stock subject to the forfeited Award, or any consideration in respect of the forfeited Award.

(h) Post-Termination Exercise Period Following Termination of Continuous Service for Reasons Other than Cause. Subject to Section 4(i), if a Participant's Continuous Service terminates for any reason other than for Cause, the Participant may exercise his or her Option or SAR to the extent vested, but only within the following period of time or, if applicable, such other period of time provided in the Award Agreement or other written agreement between a Participant and the Company; provided, however, that in no event may such Award be exercised after the expiration of its maximum term (as set forth in Section 4(a)):

(i) three months following the date of such termination if such termination is a termination without Cause (other than any termination due to the Participant's Disability or death);

(ii) 12 months following the date of such termination if such termination is due to the Participant's Disability;

(iii) 18 months following the date of such termination if such termination is due to the Participant's death; or

(iv) 18 months following the date of the Participant's death if such death occurs following the date of such termination but during the period such Award is otherwise exercisable (as provided in (i) or (ii) above).

Following the date of such termination, to the extent the Participant does not exercise such Award within the applicable Post-Termination Exercise Period (or, if earlier, prior to the expiration of the maximum term of such Award), such unexercised portion of the Award will terminate, and the Participant will have no further right, title or interest in terminated Award, the shares of Common Stock subject to the terminated Award, or any consideration in respect of the terminated Award.

(i) Restrictions on Exercise; Extension of Exercisability. A Participant may not exercise an Option or SAR at any time that the issuance of shares of Common Stock upon such exercise would violate Applicable Law. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company, if a Participant's Continuous Service terminates for any reason other than for Cause and, at any time during the last thirty days of the applicable Post-Termination Exercise Period: (i) the exercise of the Participant's Option or SAR would be prohibited solely because the issuance of shares of Common Stock upon such exercise would violate Applicable Law, or (ii) the immediate sale of any shares of Common Stock issued upon such exercise would violate the Company's Trading Policy, then the applicable Post-Termination Exercise Period will be extended to the last day of the calendar month that commences following the date the Award would otherwise expire, with an additional extension of the exercise period, generally without limitation as to the maximum permitted number of extensions; provided, however, that in no event may such Award be exercise after the expiration of its maximum term (as set forth in Section 4(a)).

(j) Non-Exempt Employees. No Option or SAR, whether or not vested, granted to an Employee who is a non-exempt employee for purposes of the Fair Labor Standards Act of 1938, as amended, will be first exercisable for any shares of Common Stock until at least six months following the date of grant of such Award. Notwithstanding the foregoing, in accordance with the provisions of the Worker Economic Opportunity Act, any vested portion of such Award may be exercised earlier than six months following the date of grant of such Award in the event of (i) such Participant's death or Disability, (ii) a Corporate Transaction in which such Award is not assumed, continued or substituted, (iii) a Change in Control, or (iv) such Participant's retirement (as such term may be defined in the Award Agreement or another applicable agreement or, in the absence of any such definition, in accordance with the Company's then current employment policies and guidelines). This Section 4(j) is intended to operate so that any income derived by a non-exempt employee in connection with the exercise or vesting of an Option or SAR will be exempt from his or her regular rate of pay.

(k) Whole Shares. Options and SARs may be exercised only with respect to whole shares of Common Stock or their equivalents.

5. AWARDS OTHER THAN OPTIONS AND STOCK APPRECIATION RIGHTS.

(a) Restricted Stock Awards and RSU Awards. Each Restricted Stock Award and RSU Award will have such terms and conditions as determined by the Board; provided, however, that each Restricted Stock Award Agreement and RSU Award Agreement will conform (through incorporation of the provisions hereof by reference in the Award Agreement or otherwise) to the substance of each of the following provisions:

(i) Form of Award.

(1) RSAs: To the extent consistent with the Company's Bylaws, at the Board's election, shares of Common Stock subject to a Restricted Stock Award may be (i) held in book entry form subject to the Company's instructions until such shares become vested or any other restrictions lapse, or (ii) evidenced by a certificate, which certificate will be held in such form and manner as determined by the Board. Unless otherwise determined by the Board, a Participant will have voting and other rights as a stockholder of the Company with respect to any shares subject to a Restricted Stock Award.

(2) RSUs: A RSU Award represents a Participant's right to be issued on a future date the number of shares of Common Stock that is equal to the number of restricted stock units subject to the RSU Award. As a holder of a RSU Award, a Participant is an unsecured creditor of the Company with respect to the Company's unfunded obligation, if any, to issue shares of Common Stock in settlement of such Award and nothing contained in the Plan or any RSU Agreement, and no action taken pursuant to its provisions, will create or be construed to create a trust of any kind or a fiduciary relationship between a Participant and the Company or an Affiliate or any other person. A Participant will not have voting or any other rights as a stockholder of the Company with respect to any RSU Award (unless and until shares are actually issued in settlement of a vested RSU Award).

(ii) Consideration.

(1) RSA: A Restricted Stock Award may be granted in consideration for (A) cash or check, bank draft or money order payable to the Company, (B) past services to the Company or an Affiliate, or (C) any other form of consideration as the Board may determine and permissible under Applicable Law.

(2) RSU: Unless otherwise determined by the Board at the time of grant, a RSU Award will be granted in consideration for the Participant's services to the Company or an Affiliate, such that the Participant will not be required to make any payment to the Company (other than such services) with respect to the grant or vesting of the RSU Award, or the issuance of any shares of Common Stock pursuant to the RSU Award. If, at the time of grant, the Board determines that any consideration must be paid by the Participant (in a form other than the Participant's services to the Company or an Affiliate) upon the issuance of any shares of Common Stock in settlement of the RSU Award, such consideration may be paid in any form of consideration as the Board may determine and permissible under Applicable Law.

(iii) Vesting. The Board may impose such restrictions on or conditions to the vesting of a Restricted Stock Award or RSU Award as determined by the Board, which may include achievement of one or more Performance Goals. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, vesting of Restricted Stock Awards and RSU Awards will cease upon termination of the Participant's Continuous Service.

(iv) Termination of Continuous Service. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company, if a Participant's Continuous Service terminates for any reason, (i) the Company may receive through a forfeiture condition or a repurchase right any or all of the shares of Common Stock held by the Participant under his or her Restricted Stock Award that have not vested as of the date of such termination as set forth in the Restricted Stock Award Agreement and (ii) any portion of his or her RSU Award that has not vested will be forfeited upon such termination and the Participant will have no further right, title or interest in the RSU Award, the shares of Common Stock issuable pursuant to the RSU Award, or any consideration in respect of the RSU Award.

(v) Dividends and Dividend Equivalents. Dividends or dividend equivalents may be paid or credited, as applicable, with respect to any shares of Common Stock subject to a Restricted Stock Award or RSU Award, as determined by the Board and specified in the Award Agreement).

(vi) Settlement of RSU Awards. A RSU Award may be settled by the issuance of shares of Common Stock or cash (or any combination thereof) or in any other form of payment, as determined by the Board and specified in the RSU Award Agreement. At the time of grant, the Board may determine to impose such restrictions or conditions that delay such delivery to a date following the vesting of the RSU Award.

(b) Other Awards. Other Awards may be granted either alone or in addition to Awards provided for under Section 4 and the preceding provisions of this Section 5. Subject to the provisions of the Plan, the Board will have sole and complete discretion to determine the persons to whom and the time or times at which such Other Awards will be granted, the number of shares of Common Stock (or the cash equivalent thereof) to be granted pursuant to such Other Awards and all other terms and conditions of such Other Awards.

6. Adjustments Upon Changes in Common Stock; Other Corporate Events.

(a) Capitalization Adjustments. In the event of a Capitalization Adjustment, the Board shall appropriately and proportionately adjust: (i) the class(es) and maximum number of shares of Common Stock subject to the Plan and the maximum number of shares by which the Share Reserve may annually increase pursuant to Section 2(a), (ii) the class(es) and maximum number of shares that may be issued pursuant to the exercise of Incentive Stock Options pursuant to Section 2(b), and (iii) the class(es) and number of securities and exercise price, strike price or purchase price of Common Stock subject to outstanding Awards. The Board shall make such adjustments, and its determination shall be final, binding and conclusive. Notwithstanding the foregoing, no fractional shares or rights for fractional shares of Common Stock shall be created in order to implement any Capitalization Adjustment. The Board shall determine an appropriate equivalent benefit, if any, for any fractional shares or rights to fractional shares that might be created by the adjustments referred to in the preceding provisions of this Section.

(b) Dissolution or Liquidation. Except as otherwise provided in the Award Agreement, in the event of a dissolution or liquidation of the Company, all outstanding Awards (other than Awards consisting of vested and outstanding shares of Common Stock not subject to a forfeiture condition or the Company's right of repurchase) will terminate immediately prior to the completion of such dissolution or liquidation, and the shares of Common Stock subject to the Company's repurchase rights or subject to a forfeiture condition may be repurchased or reacquired by the Company notwithstanding the fact that the holder of such Award is providing Continuous Service, provided, however, that the Board may determine to cause some or all Awards to become fully vested, exercisable and/or no longer subject to repurchase or forfeiture (to the extent such Awards have not previously expired or terminated) before the dissolution or liquidation is completed but contingent on its completion.

(c) Corporate Transaction. The following provisions will apply to Awards in the event of a Corporate Transaction except as set forth in Section 11, and unless otherwise provided in the instrument evidencing the Award or any other written agreement between the Company or any Affiliate and the Participant or unless otherwise expressly provided by the Board at the time of grant of an Award.

(i) Awards May Be Assumed. In the event of a Corporate Transaction, any surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company) may assume or continue any or all Awards outstanding under the Plan or may substitute similar awards for Awards outstanding under the Plan (including but not limited to, awards to acquire the same consideration paid to the stockholders of the Company pursuant to the Corporate Transaction), and any reacquisition or repurchase rights held by the Company in respect of Common Stock issued pursuant to Awards may be assigned by the Company to the successor of the Company (or the successor's parent company, if any), in connection with such Corporate Transaction. A surviving corporation or acquiring corporation (or its parent) may choose to assume or continue only a portion of an Award or substitute a similar award for only a portion of an Award, or may choose to assume or continue the Awards held by some, but not all Participants. The terms of any assumption, continuation or substitution will be set by the Board.

(ii) Awards Held by Current Participants. In the event of a Corporate Transaction in which the surviving corporation or acquiring corporation (or its parent company) does not assume or continue such outstanding Awards or substitute similar awards for such outstanding Awards, then with respect to Awards that have not been assumed, continued or substituted and that are held by Participants whose Continuous Service has not terminated prior to the effective time of the Corporate Transaction (referred to as the "Current Participants"), the vesting of such Awards (and, with respect to Options and Stock Appreciation Rights, the time when such Awards may be exercised) will be accelerated in full to a date prior to the effective time of such Corporate Transaction (contingent upon the effective time of the Corporate Transaction), and such Awards will terminate if not exercised (if applicable) at or prior to the effective time of the Corporate Transaction, and any reacquisition or repurchase rights held by the Company with respect to such Awards will lapse (contingent upon the effectiveness of the Corporate Transaction). With respect to Awards with performance-based vesting that will accelerate upon the occurrence of a Corporate Transaction pursuant to this subsection (ii) and that have multiple vesting levels depending on the level of performance, unless otherwise provided in the form of a cash payment, such cash payment will be made no later than 30 days following the occurrence of the Corporate Transaction.

(iii) Awards Held by Persons other than Current Participants. In the event of a Corporate Transaction in which the surviving corporation or acquiring corporation (or its parent company) does not assume or continue such outstanding Awards or substitute similar awards for such outstanding Awards, then with respect to Awards that have not been assumed, continued or substituted and that are held by persons other than Current Participants, such Awards will terminate if not exercised (if applicable) prior to the occurrence of the Corporate Transaction; provided, however, that any reacquisition or repurchase rights held by the Company with respect to such Awards will not terminate and may continue to be exercised notwithstanding the Corporate Transaction.

(iv) Payment for Awards in Lieu of Exercise. Notwithstanding the foregoing, in the event an Award will terminate if not exercised prior to the effective time of a Corporate Transaction, the Board may provide, in its sole discretion, that the holder of such Award may not exercise such Award but will receive a payment, in such form as may be determined by the Board, equal in value, at the effective time, to the excess, if any, of (1) the value of the property the Participant would have received upon the exercise of the Award (including, at the discretion of the Board, any unvested portion of such Award), over (2) any exercise price payable by such holder in connection with such exercise.

(d) Appointment of Stockholder Representative. As a condition to the receipt of an Award under this Plan, a Participant will be deemed to have agreed that the Award will be subject to the terms of any agreement governing a Corporate Transaction involving the Company, including, without limitation, a provision for the appointment of a stockholder representative that is authorized to act on the Participant's behalf with respect to any escrow, indemnities and any contingent consideration.

(e) No Restriction on Right to Undertake Transactions. The grant of any Award under the Plan and the issuance of shares pursuant to any Award does not affect or restrict in any way the right or power of the Company or the stockholders of the Company to make or authorize any adjustment, recapitalization, reorganization or other change in the Company's capital structure or its business, any merger or consolidation of the Company, any issue of stock or of options, rights or options to purchase stock or of bonds, debentures, preferred or prior preference stocks whose rights are superior to or affect the Common Stock or the rights thereof or which are convertible into or exchangeable for Common Stock, or the dissolution or liquidation of the Company, or any sale or transfer of all or any part of its assets or business, or any other corporate act or proceeding, whether of a similar character or otherwise.

7. Administration.

(a) Administration by Board. The Board will administer the Plan unless and until the Board delegates administration of the Plan to a Committee or Committees, as provided in subsection (c) below.

(b) Powers of Board. The Board will have the power, subject to, and within the limitations of, the express provisions of the Plan:

(i) To determine from time to time: (1) which of the persons eligible under the Plan will be granted Awards; (2) when and how each Award will be granted; (3) what type or combination of types of Award will be granted; (4) the provisions of each Award granted (which need not be identical), including the time or times when a person will be permitted to receive an issuance of Common Stock or other payment pursuant to an Award (and whether and to what degree any applicable Performance Goals have been attained); (5) the number of shares of Common Stock or cash equivalent with respect to which an Award will be granted to each such person; and (6) the Fair Market Value applicable to an Award.

(ii) To construe and interpret the Plan and Awards granted under it, and to establish, amend and revoke rules and regulations for its administration. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan or in any Award Agreement, in a manner and to the extent it deems necessary or expedient to make the Plan or Award fully effective.

(iii) To settle all controversies regarding the Plan and Awards granted under it.

(iv) To accelerate the time at which an Award may first be exercised or the time during which an Award or any part thereof will vest, notwithstanding the provisions in the Award Agreement stating the time at which it may first be exercised or the time during which it will vest.

(v) To prohibit the exercise of any Option, SAR or other exercisable Award during a period of up to 30 days prior to the consummation of any pending stock dividend, stock split, combination or exchange of shares, merger, consolidation or other distribution (other than normal cash dividends) of Company assets to stockholders, or any other change affecting the shares of Common Stock or the share price of the Common Stock including any Corporate Transaction, for reasons of administrative convenience.

(vi) To suspend or terminate the Plan at any time. Suspension or termination of the Plan will not Materially Impair rights and obligations under any Award granted while the Plan is in effect except with the written consent of the affected Participant.

(vii) To amend the Plan in any respect the Board deems necessary or advisable; provided, however, that stockholder approval will be required for any amendment to the extent required by Applicable Law. Except as provided above, rights under any Award granted before amendment of the Plan will not be Materially Impaired by any amendment of the Plan unless (1) the Company requests the consent of the affected Participant, and (2) such Participant consents in writing.

(viii) To submit any amendment to the Plan for stockholder approval.

(ix) To approve forms of Award Agreements for use under the Plan and to amend the terms of any one or more Awards, including, but not limited to, amendments to provide terms more favorable to the Participant than previously provided in the Award Agreement, subject to any specified limits in the Plan that are not subject to Board discretion; provided however, that, (1) the Board shall not, without stockholder approval, reduce the exercise or strike price of an Option or SAR (other than in connection with a Capitalization Adjustment) and, at any time when the exercise or strike price of an Option or SAR for a new Award with a lower (or no) purchase price or for cash, and (2) a Participant's rights under any Award will not be Materially Impaired by any such amendment unless (A) the Company requests the consent of the affected Participant, and (B) such Participant consents in writing.

(x) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company and that are not in conflict with the provisions of the Plan or Awards.

(xi) To adopt such procedures and sub-plans as are necessary or appropriate to permit and facilitate participation in the Plan by, or take advantage of specific tax treatment for Awards granted to, Employees, Directors or Consultants who are non-U.S. nationals or employed outside the United States (provided that Board approval will not be necessary for immaterial modifications to the Plan or any Award Agreement to ensure or facilitate compliance with the laws of the relevant non-U.S. jurisdiction).

(c) Delegation to Committee.

(i) General. The Board may delegate some or all of the administration of the Plan to a Committee or Committees. If administration of the Plan is delegated to a Committee, the Committee will have, in connection with the administration of the Plan, the powers theretofore possessed by the Board that have been delegated to the Committee, including the power to delegate to another Committee or a subcommittee of the Committee or subcommittee), subject, however, to such resolutions, not inconsistent with the provisions of the Plan, as may be adopted from time to time by the Board. Each Committee may retain the authority to concurrently administer the Plan with the Committee or subcommittee to which it has delegated its authority hereunder and may, at any time, revest in such Committee some or all of the powers previously delegated. The Board may retain the authority to concurrently administer the Plan with any Committee and may, at any time, revest in the Board some or all of the powers previously delegated.

(ii) Rule 16b-3 Compliance. To the extent an Award is intended to qualify for the exemption from Section 16(b) of the Exchange Act that is available under Rule 16b-3 of the Exchange Act, the Award will be granted by the Board or a Committee that consists solely of two or more Non-Employee Directors, as determined under Rule 16b-3(b)(3) of the Exchange Act, and, thereafter, any action establishing or modifying the terms of the Award will be approved by the Board or a Committee meeting such requirements to the extent necessary for such exemption to remain available.

(d) Effect of Board's Decision. All determinations, interpretations and constructions made by the Board or any Committee in good faith will not be subject to review by any person and will be final, binding and conclusive on all persons.

(e) Delegation to an Officer. The Board or any Committee may delegate to one or more Officers the authority to do one or both of the following (i) designate Employees who are not Officers to be recipients of Options and SARs (and, to the extent permitted by Applicable Law, other types of Awards) and, to the extent permitted by Applicable Law, the terms thereof, and (ii) determine the number of shares of Common Stock to be subject to such Awards granted to such Employees; provided, however, that the resolutions or charter adopted by the Board or any Committee evidencing such delegation will specify the total number of shares of Common Stock that may be subject to the Awards granted by such Officer and that such Officer may not grant an Award to himself or herself. Any such Awards will be granted on the applicable form of Award Agreement most recently approved for use by the Board or the Committee, unless otherwise provided in the resolutions approving the delegation authority. Notwithstanding anything to the contrary herein, neither the Board nor any Committee may delegate to an Officer who is acting solely in the capacity of an Officer (and not also as a Director) the authority to determine the Fair Market Value.

8. TAX WITHHOLDING

(a) Withholding Authorization. As a condition to acceptance of any Award under the Plan, a Participant authorizes withholding from payroll and any other amounts payable to such Participant, and otherwise agrees to make adequate provision for (including), any sums required to satisfy any U.S. and/or non-U.S. federal, state, or local tax or social insurance contribution withholding obligations of the Company or an Affiliate, if any, which arise in connection with the exercise, vesting or settlement of such Award, as applicable. Accordingly, a Participant may not be able to exercise an Award even though the Award is vested, and the Company shall have no obligation to issue shares of Common Stock subject to an Award, unless and until such obligations are satisfied.

(b) Satisfaction of Withholding Obligation. To the extent permitted by the terms of an Award Agreement, the Company may, in its sole discretion, satisfy any U.S. and/or non-U.S. federal, state or local tax or social insurance withholding obligation relating to an Award by any of the following means or by a combination of such means: (i) causing the Participant to tender a cash payment; (ii) withholding shares of Common Stock from the shares of Common Stock issued or otherwise issuable to the Participant in connection with the Award; (iii) withholding cash from an Award settled in cash; (iv) withholding payment from any amounts otherwise payable to the Participant; (v) by allowing a Participant to effectuate a "cashless exercise" pursuant to a program developed under Regulation T as promulgated by the U.S. Federal Reserve Board or (vi) by such other method as may be set forth in the Award Agreement.

(c) No Obligation to Notify or Minimize Taxes; No Liability to Claims. Except as required by Applicable Law, the Company has no duty or obligation to any Participant to advise such holder as to the time or manner of exercising such Award. Furthermore, the Company has no duty or obligation to warn or otherwise advise such holder of a pending termination or expiration of an Award or a possible period in which the Award may not be exercised. The Company has no duty or obligation to minimize the tax consequences of an Award to the holder of such Award and will not be liable to any holder of an Award for any adverse tax consequences to such holder in connection with an Award. As a condition to accepting an Award under the Plan, each Participant (i) agrees to not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates related to tax liabilities arising from such Award or other Company compensation and (ii) acknowledges that such Participant was advised to consult with his or her own personal tax, financial and other legal advisors regarding the tax consequences of the Award and has either done so or knowingly and voluntarily declined to do so. Additionally, each Participant acknowledges any Option or SAR granted under the Plan is exempt from Section 409A only if the exercise or strike price is at least equal to the "fair market value" of the Common Stock on the date of grant as determined by the Internal Revenue Service and there is no other impermissible deferral of compensation associated with the Award. Additionally, as a condition to accepting an Option or SAR granted under the Plan, each Participant agrees not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates in the event that the U.S. Internal Revenue Service asserts that such exercise price or strike price is less than the "fair market value" of the Common Stock on the date of grant as subsequently determined by the U.S. Internal Revenue Service.

(d) Withholding Indemnification. As a condition to accepting an Award under the Plan, in the event that the amount of the Company's and/or its Affiliate's withholding obligation in connection with such Award was greater than the amount actually withheld by the Company and/or its Affiliates, each Participant agrees to indemnify and hold the Company and/or its Affiliates harmless from any failure by the Company and/or its Affiliates to withhold the proper amount.

9. MISCELLANEOUS.

(a) Source of Shares. The stock issuable under the Plan will be shares of authorized but unissued or reacquired Common Stock, including shares repurchased by the Company on the open market or otherwise.

(b) Use of Proceeds from Sales of Common Stock. Proceeds from the sale of shares of Common Stock pursuant to Awards will constitute general funds of the Company.

(c) Corporate Action Constituting Grant of Awards. Corporate action constituting a grant by the Company of an Award to any Participant will be deemed completed as of the date of such corporate action, unless otherwise determined by the Board, regardless of when the instrument, certificate, or letter evidencing the Award is communicated to, or actually received or accepted by, the Participant. In the event that the corporate records (e.g., Board consents, resolutions or minutes) documenting the corporate action approving the grant contain terms (e.g., exercise price, vesting schedule or number of shares) that are inconsistent with those in the Award Agreement or related grant documents, the corporate records will control and the Participant will have no legally binding right to the incorrect term in the Award Agreement or related grant documents.

(d) Stockholder Rights. No Participant will be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares of Common Stock subject to such Award unless and until (i) such Participant has satisfied all requirements for exercise of the Award pursuant to its terms, if applicable, and (ii) the issuance of the Common Stock subject to such Award is reflected in the records of the Company.

(e) No Employment or Other Service Rights. Nothing in the Plan, any Award Agreement or any other instrument executed thereunder or in connection with any Award granted pursuant thereto will confer upon any Participant any right to continue to serve the Company or an Affiliate in the capacity in effect at the time the Award was granted or affect the right of the Company or an Affiliate to terminate at will and without regard to any future vesting opportunity that a Participant may have with respect to any Award (i) the employment of an Employee with or without notice and with or without cause, (ii) the service of a Consultant pursuant to the terms of such Consultant's agreement with the Company or an Affiliate or (iii) the service of a Director pursuant to the Bylaws of the Company or an Affiliate, and any applicable provisions of the corporate law of the U.S. state or non-U.S. jurisdiction in which the Company or the Affiliate is incorporated, as the case may be. Further, nothing in the Plan, any Award Agreement or any other instrument executed thereunder or in connection with any Award will constitute any promise or commitment by the Company or an Affiliate regarding the fact or nature of future positions, future work assignments, future compensation or any other term or condition of employment or service or confer any right or benefit under the Award or the Plan unless such right or benefit has specifically accrued under the terms of the Award Agreement and/or Plan.

(f) Change in Time Commitment. In the event a Participant's regular level of time commitment in the performance of his or her services for the Company and any Affiliates is reduced (for example, and without limitation, if the Participant is an Employee of the Company and the Employee has a change in status from a full-time Employee to a part-time Employee or takes an extended leave of absence) after the date of grant of any Award to the Participant, the Board may determine, to the extent permitted by Applicable Law, to (i) make a corresponding reduction in the number of shares or cash amount subject to any portion of such Award that is scheduled to vest or become payable after the date of such change in time commitment, and (ii) in lieu of or in combination with such a reduction, extend the vesting or payment schedule applicable to such Award. In the event of any such reduction, the Participant will have no right with respect to any portion of the Award that is so reduced or extended.

(g) Execution of Additional Documents. As a condition to accepting an Award under the Plan, the Participant agrees to execute any additional documents or instruments necessary or desirable, as determined in the Plan Administrator's sole discretion, to carry out the purposes or intent of the Award, or facilitate compliance with securities and/or other regulatory requirements, in each case at the Plan Administrator's request.

(h) Electronic Delivery and Participation. Any reference herein or in an Award Agreement to a "written" agreement or document will include any agreement or document delivered electronically, filed publicly at www.sec.gov (or any successor website thereto) or posted on the Company's intranet (or other shared electronic medium controlled by the Company to which the Participant has access). By accepting any Award the Participant consents to receive documents by electronic delivery and to participate in the Plan through any on-line electronic system established and maintained by the Plan Administrator or another third party selected by the Plan Administrator. The form of delivery of any Common Stock (e.g., a stock certificate or electronic entry evidencing such shares) shall be determined by the Company.

(i) Clawback/Recovery. All Awards granted under the Plan will be subject to recoupment in accordance with any clawback policy that the Company is required to adopt pursuant to the listing standards of any national securities exchange or association on which the Company's securities are listed or as is otherwise required by the Dodd-Frank Wall Street Reform and Consumer Protection Act or other Applicable Law and any clawback policy that the Company otherwise adopts, to the extent applicable and permissible under Applicable Law. In addition, the Board may impose such other clawback, recovery or recoupment provisions in an Award Agreement as the Board determines necessary or appropriate, including but not limited to a reacquisition right in respect of previously acquired shares of Common Stock or other cash or property upon the occurrence of Cause. No recovery of compensation under such a clawback policy will be an event giving rise to a Participant's right to voluntarily terminate employment upon a "resignation for good reason," or for a "constructive termination" or any similar term under any plan of or agreement with the Company.

(j) Securities Law Compliance. A Participant will not be issued any shares in respect of an Award unless either (i) the shares are registered under the Securities Act; or (ii) the Company has determined that such issuance would be exempt from the registration requirements of the Securities Act. Each Award also must comply with other Applicable Law governing the Award, and a Participant will not receive such shares if the Company determines that such receipt would not be in material compliance with Applicable Law.

(k) Transfer or Assignment of Awards; Issued Shares. Except as expressly provided in the Plan or the form of Award Agreement, Awards granted under the Plan may not be transferred or assigned by the Participant. After the vested shares subject to an Award have been issued, or in the case of Restricted Stock and similar awards, after the issued shares have vested, the holder of such shares is free to assign, hypothecate, donate, encumber or otherwise dispose of any interest in such shares provided that any such actions are in compliance with the provisions herein, the terms of the Trading Policy and Applicable Law.

(I) Effect on Other Employee Benefit Plans. The value of any Award granted under the Plan, as determined upon grant, vesting or settlement, shall not be included as compensation, earnings, salaries, or other similar terms used when calculating any Participant's benefits under any employee benefit plan sponsored by the Company or any Affiliate, except as such plan otherwise expressly provides. The Company expressly reserves its rights to amend, modify, or terminate any of the Company's or any Affiliate's employee benefit plans.

(m) Deferrals. To the extent permitted by Applicable Law, the Board, in its sole discretion, may determine that the delivery of Common Stock or the payment of cash, upon the exercise, vesting or settlement of all or a portion of any Award may be deferred and may establish programs and procedures for deferral elections to be made by Participants. Deferrals will be made in accordance with the requirements of Section 409A.

(n) Section 409A. Unless otherwise expressly provided for in an Award Agreement, the Plan and Award Agreements will be interpreted to the greatest extent possible in a manner that makes the Plan and the Awards granted hereunder exempt from Section 409A, and, to the extent not so exempt, in compliance with the requirements of Section 409A. If the Board determines that any Award granted hereunder is not exempt from and is therefore subject to Section 409A, the Award Agreement evidencing such Award will incorporate the terms and conditions necessary to avoid the consequences specified in Section 409A(a)(1) of the Code, and to the extent an Award Agreement is silent on terms necessary for compliance, such terms are hereby incorporated by reference into the Award Agreement. Notwithstanding anything to the contrary in this Plan (and unless the Award Agreement specifically provides otherwise), if the shares of Common Stock are publicly traded, and if a Participant holding an Award that constitutes "deferred compensation" under Section 409A is a "specified employee" for purposes of Section 409A, no distribution or payment of any amount that is due because of a "separation from service" (as defined in Section 409A without regard to alternative definitions thereunder) will be issued or paid before the date that is six months and one day following the date of such Participant's "separation from service" or, if earlier, the date of the Participant's death, unless such distribution or payment can be made in a manner that complies with Section 409A, and any amounts so deferred will be paid in a lump sum on the day after such six month period elapses, with the balance paid thereafter on the original schedule.

(o) Choice of Law. This Plan and any controversy arising out of or relating to this Plan shall be governed by, and construed in accordance with, the internal laws of the State of Delaware, without regard to conflict of law principles that would result in any application of any law other than the law of the State of Delaware.

10. COVENANTS OF THE COMPANY.

(a) Compliance with Law. The Company will seek to obtain from each regulatory commission or agency, as may be deemed necessary, having jurisdiction over the Plan such authority as may be required to grant Awards and to issue and sell shares of Common Stock upon exercise or vesting of the Awards; provided, however, that this undertaking will not require the Company to register under the Securities Act the Plan, any Award or any Common Stock issued or issuable pursuant to any such Award. If, after reasonable efforts and at a reasonable cost, the Company is unable to obtain from any such regulatory commission or agency the authority that counsel for the Company deems necessary or advisable for the lawful issuance and sale of Common Stock under the Plan, the Company will be relieved from any liability for failure to issue and sell Common Stock upon exercise or vesting of such Awards unless and until such authority is obtained. A Participant is not eligible for the grant of an Award or the subsequent issuance of Common Stock pursuant to the Award if such grant or issuance would be in violation of any Applicable Law.

11. Additional Rules for Awards Subject to Section 409A.

(a) Application. Unless the provisions of this Section of the Plan are expressly superseded by the provisions in the form of Award Agreement, the provisions of this Section shall apply and shall supersede anything to the contrary set forth in the Award Agreement for a Non-Exempt Award.

(b) Non-Exempt Awards Subject to Non-Exempt Severance Arrangements. To the extent a Non-Exempt Award is subject to Section 409A due to application of a Non-Exempt Severance Arrangement, the following provisions of this subsection (b) apply.

(i) If the Non-Exempt Award vests in the ordinary course during the Participant's Continuous Service in accordance with the vesting schedule set forth in the Award Agreement, and does not accelerate vesting under the terms of a Non-Exempt Severance Arrangement, in no event will the shares be issued in respect of such Non-Exempt Award any later than the later of: (i) December 31^{st} of the calendar year that includes the applicable vesting date, or (ii) the 60^{th} day that follows the applicable vesting date.

(ii) If vesting of the Non-Exempt Award accelerates under the terms of a Non-Exempt Severance Arrangement in connection with the Participant's Separation from Service, and such vesting acceleration provisions were in effect as of the date of grant of the Non-Exempt Award and, therefore, are part of the terms of such Non-Exempt Award as of the date of grant, then the shares will be earlier issued in settlement of such Non-Exempt Award upon the Participant's Separation from Service in accordance with the terms of the Non-Exempt Severance Arrangement, but in no event later than the 60th day that follows the date of the Participant's Separation from Service. However, if at the time the shares would otherwise be issued the Participant is subject to the distribution limitations contained in Section 409A applicable to "specified employees," as defined in Section 409A(a)(2)(B)(i) of the Code, such shares shall not be issued before the date that is six months following the date of such Participant's Separation from Service, or, if earlier, the date of the Participant's death that occurs within such six month period.

(iii) If vesting of a Non-Exempt Award accelerates under the terms of a Non-Exempt Severance Arrangement in connection with a Participant's Separation from Service, and such vesting acceleration provisions were not in effect as of the date of grant of the Non-Exempt Award and, therefore, are not a part of the terms of such Non-Exempt Award on the date of grant, then such acceleration of vesting of the Non-Exempt Award shall not accelerate the issuance date of the shares, but the shares shall instead be issued on the same schedule as set forth in the Grant Notice as if they had vested in the ordinary course during the Participant's Continuous Service, notwithstanding the vesting acceleration of the Non-Exempt Award. Such issuance schedule is intended to satisfy the requirements of payment on a specified date or pursuant to a fixed schedule, as provided under Treasury Regulations Section 1.409A-3(a)(4).

(c) Treatment of Non-Exempt Awards Upon a Corporate Transaction for Employees and Consultants. The provisions of this subsection (c) shall apply and shall supersede anything to the contrary set forth in the Plan with respect to the permitted treatment of any Non-Exempt Award in connection with a Corporate Transaction if the Participant was either an Employee or Consultant upon the applicable date of grant of the Non-Exempt Award.

(i) Vested Non-Exempt Awards. The following provisions shall apply to any Vested Non-Exempt Award in connection with a

Corporate Transaction:

(1) If the Corporate Transaction is also a Section 409A Change in Control then the Acquiring Entity may not assume, continue or substitute the Vested Non-Exempt Award. Upon the Section 409A Change in Control the settlement of the Vested Non-Exempt Award will automatically be accelerated and the shares will be immediately issued in respect of the Vested Non-Exempt Award. Alternatively, the Company may instead provide that the Participant will receive a cash settlement equal to the Fair Market Value of the shares that would otherwise be issued to the Participant upon the Section 409A Change in Control.

(2) If the Corporate Transaction is not also a Section 409A Change in Control, then the Acquiring Entity must either assume, continue or substitute each Vested Non-Exempt Award. The shares to be issued in respect of the Vested Non-Exempt Award shall be issued to the Participant by the Acquiring Entity on the same schedule that the shares would have been issued to the Participant if the Corporate Transaction had not occurred. In the Acquiring Entity's discretion, in lieu of an issuance of shares, the Acquiring Entity may instead substitute a cash payment on each applicable issuance date, equal to the Fair Market Value of the shares that would otherwise be issued to the Participant on such issuance dates, with the determination of the Fair Market Value of the shares made on the date of the Corporate Transaction.

(ii) Unvested Non-Exempt Awards. The following provisions shall apply to any Unvested Non-Exempt Award unless otherwise determined by the Board pursuant to subsection (e) of this Section.

(1) In the event of a Corporate Transaction, the Acquiring Entity shall assume, continue or substitute any Unvested Non-Exempt Award. Unless otherwise determined by the Board, any Unvested Non-Exempt Award will remain subject to the same vesting and forfeiture restrictions that were applicable to the Award prior to the Corporate Transaction. The shares to be issued in respect of any Unvested Non-Exempt Award shall be issued to the Participant by the Acquiring Entity on the same schedule that the shares would have been issued to the Participant if the Corporate Transaction had not occurred. In the Acquiring Entity's discretion, in lieu of an issuance of shares, the Acquiring Entity may instead substitute a cash payment on each applicable issuance date, equal to the Fair Market Value of the shares that would otherwise be issued to the Participant on such issuance dates, with the determination of Fair Market Value of the shares made on the date of the Corporate Transaction.

(2) If the Acquiring Entity will not assume, substitute or continue any Unvested Non-Exempt Award in connection with a Corporate Transaction, then such Award shall automatically terminate and be forfeited upon the Corporate Transaction with no consideration payable to any Participant in respect of such forfeited Unvested Non-Exempt Award. Notwithstanding the foregoing, to the extent permitted and in compliance with the requirements of Section 409A, the Board may in its discretion determine to elect to accelerate the vesting and settlement of the Unvested Non-Exempt Award upon the Corporate Transaction, or instead substitute a cash payment equal to the Fair Market Value of such shares that would otherwise be issued to the Participant, as further provided in subsection (e)(ii) below. In the absence of such discretionary election by the Board, any Unvested Non-Exempt Award shall be forfeited without payment of any consideration to the affected Participants if the Acquiring Entity will not assume, substitute or continue the Unvested Non-Exempt Awards in connection with the Corporate Transaction.

(3) The foregoing treatment shall apply with respect to all Unvested Non-Exempt Awards upon any Corporate Transaction, and regardless of whether or not such Corporate Transaction is also a Section 409A Change in Control.

(d) Treatment of Non-Exempt Awards Upon a Corporate Transaction for Non-Employee Directors. The following provisions of this subsection (d) shall apply and shall supersede anything to the contrary that may be set forth in the Plan with respect to the permitted treatment of a Non-Exempt Director Award in connection with a Corporate Transaction.

(i) If the Corporate Transaction is also a Section 409A Change in Control then the Acquiring Entity may not assume, continue or substitute the Non-Exempt Director Award. Upon the Section 409A Change in Control the vesting and settlement of any Non-Exempt Director Award will automatically be accelerated and the shares will be immediately issued to the Participant in respect of the Non-Exempt Director Award. Alternatively, the Company may provide that the Participant will instead receive a cash settlement equal to the Fair Market Value of the shares that would otherwise be issued to the Participant upon the Section 409A Change in Control pursuant to the preceding provision.

(ii) If the Corporate Transaction is not also a Section 409A Change in Control, then the Acquiring Entity must either assume, continue or substitute the Non-Exempt Director Award. Unless otherwise determined by the Board, the Non-Exempt Director Award will remain subject to the same vesting and forfeiture restrictions that were applicable to the Award prior to the Corporate Transaction. The shares to be issued in respect of the Non-Exempt Director Award shall be issued to the Participant by the Acquiring Entity on the same schedule that the shares would have been issued to the Participant if the Corporate Transaction had not occurred. In the Acquiring Entity's discretion, in lieu of an issuance of shares, the Acquiring Entity may instead substitute a cash payment on each applicable issuance date, equal to the Fair Market Value of the shares that would otherwise be issued to the Participant on such issuance dates, with the determination of Fair Market Value made on the date of the Corporate Transaction.

(e) If the RSU Award is a Non-Exempt Award, then the provisions in this Section 11(e) shall apply and supersede anything to the contrary that may be set forth in the Plan or the Award Agreement with respect to the permitted treatment of such Non-Exempt Award:

(i) Any exercise by the Board of discretion to accelerate the vesting of a Non-Exempt Award shall not result in any acceleration of the scheduled issuance dates for the shares in respect of the Non-Exempt Award unless earlier issuance of the shares upon the applicable vesting dates would be in compliance with the requirements of Section 409A.

(ii) The Company explicitly reserves the right to earlier settle any Non-Exempt Award to the extent permitted and in compliance with the requirements of Section 409A, including pursuant to any of the exemptions available in Treasury Regulations Section 1.409A-3(j)(4)(ix).

(iii) To the extent the terms of any Non-Exempt Award provide that it will be settled upon a Change in Control or Corporate Transaction, to the extent it is required for compliance with the requirements of Section 409A, the Change in Control or Corporate Transaction event triggering settlement must also constitute a Section 409A Change in Control. To the extent the terms of a Non-Exempt Award provide that it will be settled upon a termination of employment or termination of Continuous Service, to the extent it is required for compliance with the requirements of Section 409A, the termination event triggering settlement must also constitute a Separation From Service. However, if at the time the shares would otherwise be issued to a Participant in connection with a "separation from service" such Participant is subject to the distribution limitations contained in Section 409A (a)(2)(B)(i) of the Code, such shares shall not be issued before the date that is six months following the date of the Participant's Separation From Service, or, if earlier, the date of the Participant's death that occurs within such six month period.

(iv) The provisions in this subsection (e) for delivery of the shares in respect of the settlement of a RSU Award that is a Non-Exempt Award are intended to comply with the requirements of Section 409A so that the delivery of the shares to the Participant in respect of such Non-Exempt Award will not trigger the additional tax imposed under Section 409A, and any ambiguities herein will be so interpreted.

12. SEVERABILITY.

If all or any part of the Plan or any Award Agreement is declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity shall not invalidate any portion of the Plan or such Award Agreement not declared to be unlawful or invalid. Any Section of the Plan or any Award Agreement (or part of such a Section) so declared to be unlawful or invalid shall, if possible, be construed in a manner which will give effect to the terms of such Section or part of a Section to the fullest extent possible while remaining lawful and valid.

13. TERMINATION OF THE PLAN.

The Board may suspend or terminate the Plan at any time. No Incentive Stock Options may be granted after the tenth anniversary of the earlier of: (i) the Adoption Date, or (ii) the date the Plan is approved by the Company's stockholders. No Awards may be granted under the Plan while the Plan is suspended or after it is terminated.

14. DEFINITIONS.

As used in the Plan, the following definitions apply to the capitalized terms indicated below:

(a) "Acquiring Entity" means the surviving or acquiring corporation (or its parent company) in connection with a Corporate Transaction.

(b) "Adoption Date" means the date the Plan is first approved by the Board or Compensation Committee.

(c) "Affiliate" means, at the time of determination, any "parent" or "subsidiary" of the Company as such terms are defined in Rule 405 promulgated under the Securities Act. The Board may determine the time or times at which "parent" or "subsidiary" status is determined within the foregoing definition.

(d) "Applicable Law" means the Code and any applicable U.S. or non-U.S. securities, federal, state, material local or municipal or other law, statute, constitution, principle of common law, resolution, ordinance, code, edict, decree, rule, listing rule, regulation, judicial decision, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any applicable self-regulating organization such as the Nasdaq Stock Market, New York Stock Exchange or the Financial Industry Regulatory Authority).

(e) "Award" means any right to receive Common Stock, cash or other property granted under the Plan (including an Incentive Stock Option, a Nonstatutory Stock Option, a Restricted Stock Award, a RSU Award, a SAR, or any Other Award).

(f) "Award Agreement" means a written agreement between the Company and a Participant evidencing the terms and conditions of an Award. The Award Agreement generally consists of the Grant Notice and the agreement containing the written summary of the general terms and conditions applicable to the Award and which is provided to a Participant along with the Grant Notice.

(g) "Board" means the board of directors of the Company (or its designee). Any decision or determination made by the Board shall be a decision or determination that is made in the sole discretion of the Board (or its designee), and such decision or determination shall be final and binding on all Participants.

(h) "*Capitalization Adjustment*" means any change that is made in, or other events that occur with respect to, the Common Stock subject to the Plan or subject to any Award after the Effective Date without the receipt of consideration by the Company through merger, consolidation, reorganization, recapitalization, reincorporation, stock dividend, dividend in property other than cash, large nonrecurring cash dividend, stock split, reverse stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or any similar equity restructuring transaction, as that term is used in Statement of Financial Accounting Standards Board Accounting Standards Codification Topic 718 (or any successor thereto). Notwithstanding the foregoing, the conversion of any convertible securities of the Company will not be treated as a Capitalization Adjustment.

(i) "Cause" has the meaning ascribed to such term in any written agreement between the Participant and the Company defining such term and, in the absence of such agreement, such term means, with respect to a Participant, the occurrence of any of the following events: (i) the Participant's theft, dishonesty, willful misconduct, breach of fiduciary duty for personal profit, or intentional falsification of any Company or Affiliate documents or records; (ii) the Participant's material failure to abide by the Company's Code of Business Conduct and Ethics or other policies (including, without limitation, policies relating to confidentiality and reasonable workplace conduct and policies of any Affiliate, as applicable); (iii) the Participant's unauthorized use, misappropriation, destruction or diversion of any tangible or intangible asset or corporate opportunity of the Company or any of its Affiliates (including, without limitation, the Participant's improper use or disclosure of Company or Affiliate confidential or proprietary information); (iv) any intentional act by the Participant which has a material detrimental effect on the Company's or its Affiliate's reputation or business; (v) the Participant's repeated failure or inability to perform any reasonable assigned duties after written notice from the Company (or its Affiliate, as applicable) of, and a reasonable opportunity to cure, such failure or inability; (vi) any material breach by the Participant of any employment or service agreement between the Participant and the Company (or its Affiliate, as applicable), which breach is not cured pursuant to the terms of such agreement; or (vii) the Participant's conviction (including any plea of guilty or nolo contendere) of any criminal act involving fraud, dishonesty, misappropriation or moral turpitude, or which impairs the Participant's ability to perform his or her duties with the Company (or its Affiliate, as applicable). The determination that a termination of the Participant's Continuous Service is either for Cause or without Cause will be made by the Board with respect to Participants who are executive officers of the Company and by the Company's Chief Executive Officer or his or her designee with respect to Participants who are not executive officers of the Company. Any determination by the Company that the Continuous Service of a Participant was terminated with or without Cause for the purposes of outstanding Awards held by such Participant will have no effect upon any determination of the rights or obligations of the Company or such Participant for any other purpose.

(j) "*Change in Control*" or "*Change of Control*" means the occurrence, in a single transaction or in a series of related transactions, of any one or more of the following events; provided, however, to the extent necessary to avoid adverse personal income tax consequences to the Participant in connection with an Award, also constitutes a Section 409A Change in Control:

(i) any Exchange Act Person becomes the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities other than by virtue of a merger, consolidation or similar transaction. Notwithstanding the foregoing, a Change in Control shall not be deemed to occur (A) on account of the acquisition of securities of the Company directly from the Company, (B) on account of the acquisition of securities of the Company by an investor, any affiliate thereof or any other Exchange Act Person that acquires the Company's securities in a transaction or series of related transactions the primary purpose of which is to obtain financing for the Company through the issuance of equity securities, or (C) solely because the level of Ownership held by any Exchange Act Person (the "*Subject Person*") exceeds the designated percentage threshold of the outstanding voting securities as a result of a repurchase or other acquisition of voting securities by the Company, and after such share acquisition, the Subject Person becomes the Owner of any additional voting securities that, assuming the repurchase or other acquisition had not occurred, increases the percentage of the then outstanding voting securities Owned by the Subject Person over the designated percentage threshold, then a Change in Control shall be deemed to occur;

(ii) there is consummated a merger, consolidation or similar transaction involving (directly or indirectly) the Company and, immediately after the consummation of such merger, consolidation or similar transaction, the stockholders of the Company immediately prior thereto do not Own, directly or indirectly, either (A) outstanding voting securities representing more than 50% of the combined outstanding voting power of the surviving Entity in such merger, consolidation or similar transaction, in each case in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such transaction;

(iii) the stockholders of the Company approve or the Board approves a plan of complete dissolution or liquidation of the Company, or a complete dissolution or liquidation of the Company shall otherwise occur, except for a liquidation into a parent corporation;

(iv) there is consummated a sale, lease, exclusive license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries, other than a sale, lease, license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries to an Entity, more than 50% of the combined voting power of the voting securities of which are Owned by stockholders of the Company in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such sale, lease, license or other disposition; or

(v) individuals who, on the date the Plan is adopted by the Board, are members of the Board (the "*Incumbent Board*") cease for any reason to constitute at least a majority of the members of the Board; provided, however, that if the appointment or election (or nomination for election) of any new Board member was approved or recommended by a majority vote of the members of the Incumbent Board then still in office, such new member shall, for purposes of this Plan, be considered as a member of the Incumbent Board.

Notwithstanding the foregoing or any other provision of this Plan, (A) the term Change in Control shall not include a sale of assets, merger or other transaction effected exclusively for the purpose of changing the domicile of the Company, and (B) the definition of Change in Control (or any analogous term) in an individual written agreement between the Company or any Affiliate and the Participant shall supersede the foregoing definition with respect to Awards subject to such agreement; provided, however, that if no definition of Change in Control or any analogous term is set forth in such an individual written agreement, the foregoing definition shall apply.

(k) "Code" means the U.S. Internal Revenue Code of 1986, as amended, including any applicable regulations and guidance thereunder.

(I) "Committee" means the Compensation Committee and any other committee of one or more Directors to whom authority has been delegated by the Board or Compensation Committee in accordance with the Plan.

(m) "Common Stock" means the common stock of the Company.

(n) "Company" means Orchestra BioMed Holdings, Inc., a Delaware corporation, and any successor corporation thereto.

(o) "Compensation Committee" means the Compensation Committee of the Board.

(p) "Consultant" means any person, including an advisor, who is (i) engaged by the Company or an Affiliate to render consulting or advisory services and is compensated for such services, or (ii) serving as a member of the board of directors of an Affiliate and is compensated for such services. However, service solely as a Director, or payment of a fee for such service, will not cause a Director to be considered a "Consultant" for purposes of the Plan. Notwithstanding the foregoing, a person is treated as a Consultant under this Plan only if a Form S-8 Registration Statement under the Securities Act is available to register either the offer or the sale of the Company's securities to such person.

(q) "Continuous Service" means that the Participant's service with the Company or an Affiliate, whether as an Employee, Director or Consultant, is not interrupted or terminated. A change in the capacity in which the Participant renders service to the Company or an Affiliate as an Employee, Director or Consultant or a change in the Entity for which the Participant renders such service, provided that there is no interruption or termination of the Participant's service with the Company or an Affiliate, will not terminate a Participant's Continuous Service; provided, however, that if the Entity for which a Participant is rendering services ceases to qualify as an Affiliate, as determined by the Board, such Participant's Continuous Service will be considered to have terminated on the date such Entity ceases to qualify as an Affiliate. For example, a change in status from an Employee of the Company to a Consultant of an Affiliate or to a Director will not constitute an interruption of Continuous Service. To the extent permitted by law, the Board or the cheif executive officer of the Company, in that party's sole discretion, may determine whether Continuous Service will be considered interrupted in the case of (i) any leave of absence approved by the Board or cheif executive officer, including sick leave, military leave or any other personal leave, or (ii) transfers between the Company, an Affiliate, or their successors. Notwithstanding the foregoing, a leave of absence policy, in the written terms of any leave of absence agreement or policy applicable to the Participant, or as otherwise required by law. In addition, to the extent required for exemption from or compliance with Section 409A, the determination of "separation from service" as defined under U.S. Treasury Regulation Section 1.409A-1(h) (without regard to any alternative definition thereunder).

(r) "*Corporate Transaction*" means the consummation, in a single transaction or in a series of related transactions, of any one or more of the following events:

(i) a sale or other disposition of all or substantially all, as determined by the Board, of the consolidated assets of the Company and its Subsidiaries;

(ii) a sale or other disposition of at least 50% of the outstanding securities of the Company;

(iii) a merger, consolidation or similar transaction following which the Company is not the surviving corporation; or

(iv) a merger, consolidation or similar transaction following which the Company is the surviving corporation but the shares of Common Stock outstanding immediately preceding the merger, consolidation or similar transaction are converted or exchanged by virtue of the merger, consolidation or similar transaction into other property, whether in the form of securities, cash or otherwise.

(s) "Director" means a member of the Board.

(t) "determine" or "determined" means as determined by the Board or the Committee (or its designee) in its sole discretion.

(u) "Disability" means, with respect to a Participant, such Participant is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than 12 months, as provided in Section 22(e)(3) of the Code, and will be determined by the Board on the basis of such medical evidence as the Board deems warranted under the circumstances.

(v) "Effective Date" means the date of the closing of the transactions contemplated by the Merger Agreement.

(w) "*Employee*" means any person employed by the Company or an Affiliate. However, service solely as a Director, or payment of a fee for such services, will not cause a Director to be considered an "Employee" for purposes of the Plan.

(x) "Employer" means the Company or the Affiliate that employs the Participant.

(y) "Entity" means a corporation, partnership, limited liability company or other entity.

(z) "Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

(aa) "Exchange Act Person" means any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act), except that "Exchange Act Person" will not include (i) the Company or any Subsidiary of the Company, (ii) any employee benefit plan of the Company or any Subsidiary of the Company, (iii) an underwriter temporarily holding securities pursuant to a registered public offering of such securities, (iv) an Entity Owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their Ownership of stock of the Company; or (v) any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act) that, as of the Effective Date, is the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities.

(bb) "Fair Market Value" means, as of any date, unless otherwise determined by the Board, the value of the Common Stock (as determined on a per share or aggregate basis, as applicable) determined as follows: (i) if the Common Stock is listed on any established stock exchange or traded on any established market, the Fair Market Value will be the closing sales price for such stock as quoted on such exchange or market (or the exchange or market with the greatest volume of trading in the Common Stock) on the date of determination, as reported in a source the Board deems reliable; (ii) if there is no closing sales price for the Common Stock on the date of determination, then the Fair Market Value will be the closing selling price on the last preceding date for which such quotation exists; or (iii) in the absence of such markets for the Common Stock, or if otherwise determined by the Board, the Fair Market Value will be determined by the Board in good faith and in a manner that complies with Sections 409A and 422 of the Code.

(cc) "*Governmental Body*" means any: (a) nation, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) U.S. or non-U.S. federal, state, local, municipal, or other government; (c) governmental or regulatory body, or quasi-governmental body of any nature (including any governmental division, department, administrative agency or bureau, commission, authority, instrumentality, official, ministry, fund, foundation, center, organization, unit, body or Entity and any court or other tribunal, and for the avoidance of doubt, any tax authority) or other body exercising similar powers or authority; or (d) self-regulatory organization (including the Nasdaq Stock Market, New York Stock Exchange, and the Financial Industry Regulatory Authority).

(dd) "Grant Notice" means the notice provided to a Participant that he or she has been granted an Award under the Plan and which includes the name of the Participant, the type of Award, the date of grant of the Award, number of shares of Common Stock subject to the Award or potential cash payment right, (if any), the vesting schedule for the Award (if any) and other key terms applicable to the Award.

(ee) "*Incentive Stock Option*" means an option granted pursuant to Section 4 of the Plan that is intended to be, and qualifies as, an "incentive stock option" within the meaning of Section 422 of the Code.

(ff) "*Materially Impair*" means any amendment to the terms of the Award that materially adversely affects the Participant's rights under the Award. A Participant's rights under an Award will not be deemed to have been Materially Impaired by any such amendment if the Board, in its sole discretion, determines that the amendment, taken as a whole, does not materially impair the Participant's rights. For example, the following types of amendments to the terms of an Award do not Materially Impair the Participant's rights under the Award: (i) imposition of reasonable restrictions on the minimum number of shares subject to an Option that may be exercised, (ii) to maintain the qualified status of the Award as an Incentive Stock Option under Section 422 of the Code; (iii) to change the terms of an Incentive Stock Option in a manner that disqualifies, impairs or otherwise affects the qualified status of the Award as an Incentive Stock Option under Section 422 of the Code; (iv) to clarify the manner of exemption from, or to bring the Award into compliance with or qualify it for an exemption from, Section 409A; or (v) to comply with other Applicable Laws.

(gg) "Merger Agreement" means the Agreement and Plan of Merger, dated as of July 4, 2022, by and among the Company and the other parties thereto.

(hh) "*Non-Employee Director*" means a Director who either (i) is not a current employee or officer of the Company or an Affiliate, does not receive compensation, either directly or indirectly, from the Company or an Affiliate for services rendered as a consultant or in any capacity other than as a Director (except for an amount as to which disclosure would not be required under Item 404(a) of Regulation S-K promulgated pursuant to the Securities Act ("*Regulation S-K*")), does not possess an interest in any other transaction for which disclosure would be required under Item 404(a) of Regulation S-K, and is not engaged in a business relationship for which disclosure would be required pursuant to Item 404(b) of Regulation S-K; or (ii) is otherwise considered a "non-employee director" for purposes of Rule 16b-3.

(ii) "Non-Exempt Award" means any Award that is subject to, and not exempt from, Section 409A, including as the result of (i) a deferral of the issuance of the shares subject to the Award which is elected by the Participant or imposed by the Company or (ii) the terms of any Non-Exempt Severance Agreement.

(jj) "Non-Exempt Director Award" means a Non-Exempt Award granted to a Participant who was a Director but not an Employee on the applicable grant date.

(kk) "Non-Exempt Severance Arrangement" means a severance arrangement or other agreement between the Participant and the Company that provides for acceleration of vesting of an Award and issuance of the shares in respect of such Award upon the Participant's termination of employment or separation from service (as such term is defined in Section 409A(a)(2)(A)(i) of the Code (and without regard to any alternative definition thereunder)) ("Separation from Service") and such severance benefit does not satisfy the requirements for an exemption from application of Section 409A provided under Treasury Regulations Section 1.409A-1(b)(4), 1.409A-1(b)(9) or otherwise.

(II) "Nonstatutory Stock Option" means any option granted pursuant to Section 4 of the Plan that does not qualify as an Incentive Stock

Option.

(mm) "Officer" means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act.

the Plan.

(nn) "Option" means an Incentive Stock Option or a Nonstatutory Stock Option to purchase shares of Common Stock granted pursuant to

(oo) "*Option Agreement*" means a written agreement between the Company and the Optionholder evidencing the terms and conditions of the Option grant. The Option Agreement includes the Grant Notice for the Option and the agreement containing the written summary of the general terms and conditions applicable to the Option and which is provided to a Participant along with the Grant Notice. Each Option Agreement will be subject to the terms and conditions of the Plan.

(pp) "Optionholder" means a person to whom an Option is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Option.

(qq) "*Other Award*" means an award valued in whole or in part by reference to, or otherwise based on, Common Stock, including the appreciation in value thereof (e.g., options or stock rights with an exercise price or strike price less than 100% of the Fair Market Value at the time of grant) that is not an Incentive Stock Options, Nonstatutory Stock Option, SAR, Restricted Stock Award, or RSU Award.

(rr) "Other Award Agreement" means a written agreement between the Company and a holder of an Other Award evidencing the terms and conditions of an Other Award grant. Each Other Award Agreement will be subject to the terms and conditions of the Plan.

(ss) "Own," "Owned," "Owner," "Owner," "Ownership" means that a person or Entity will be deemed to "Own," to have "Owned," to be the "Owner" of or to have acquired "Ownership" of securities if such person or Entity, directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has or shares voting power, which includes the power to vote or to direct the voting, with respect to such securities.

(tt) "Participant" means an Employee, Director or Consultant to whom an Award is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Award.

(uu) "Performance Criteria" means the one or more criteria that the Board will select for purposes of establishing the Performance Goals for a Performance Period. The Performance Criteria that will be used to establish such Performance Goals may be based on any one of, or combination of, the following as determined by the Board: earnings (including earnings per share and net earnings); earnings before interest, taxes and depreciation; earnings before interest, taxes, depreciation and amortization; total stockholder return; return on equity or average stockholder's equity; return on assets, investment, or capital employed; stock price; margin (including gross margin); income (before or after taxes); operating income; operating income after taxes; pre-tax profit; operating cash flow; sales or revenue targets; increases in revenue or product revenue; expenses and cost reduction goals; improvement in or attainment of working capital levels; economic value added (or an equivalent metric); market share; cash flow; cash flow per share; share price performance; debt reduction; customer satisfaction; stockholders' equity; capital expenditures; debt levels; operating profit or net operating profit; workforce diversity; growth of net income or operating income; billings; pre-clinical development related compound goals; financing; regulatory milestones, including approval of a compound; stockholder liquidity; corporate governance and compliance; product commercialization; intellectual property; personnel matters; progress of internal research or clinical programs; progress of partnered programs; partner satisfaction; budget management; clinical achievements; completing phases of a clinical study (including the treatment phase); announcing or presenting preliminary or final data from clinical studies; in each case, whether on particular timelines or generally; timely completion of clinical trials; submission of INDs and NDAs and other regulatory achievements; partner or collaborator achievements; internal controls, including those related to the Sarbanes-Oxley Act of 2002; research progress, including the development of programs; investor relations, analysts and communication; manufacturing achievements (including obtaining particular yields from manufacturing runs and other measurable objectives related to process development activities); strategic partnerships or transactions (including in-licensing and out-licensing of intellectual property; establishing relationships with commercial entities with respect to the marketing, distribution and sale of the Company's products (including with group purchasing organizations, distributors and other vendors); supply chain achievements (including establishing relationships with manufacturers or suppliers of active pharmaceutical ingredients and other component materials and manufacturers of the Company's products); co-development, co-marketing, profit sharing, joint venture or other similar arrangements; individual performance goals; corporate development and planning goals; and other measures of performance selected by the Board or Committee.

(vv) "Performance Goals" means, for a Performance Period, the one or more goals established by the Board for the Performance Period based upon the Performance Criteria. Performance Goals may be based on a Company-wide basis, with respect to one or more business units, divisions, Affiliates, or business segments, and in either absolute terms or relative to the performance of one or more comparable companies or the performance of one or more relevant indices. Unless specified otherwise by the Board (i) in the Award Agreement at the time the Award is granted or (ii) in such other document setting forth the Performance Goals at the time the Performance Goals are established, the Board will appropriately make adjustments in the method of calculating the attainment of Performance Goals for a Performance Period as follows: (1) to exclude restructuring and/or other nonrecurring charges; (2) to exclude exchange rate effects; (3) to exclude the effects of changes to generally accepted accounting principles; (4) to exclude the effects of any statutory adjustments to corporate tax rates; (5) to exclude the effects of items that are "unusual" in nature or occur "infrequently" as determined under generally accepted accounting principles; (6) to exclude the dilutive effects of acquisitions or joint ventures; (7) to assume that any business divested by the Company achieved performance objectives at targeted levels during the balance of a Performance Period following such divestiture; (8) to exclude the effect of any change in the outstanding shares of Common Stock of the Company by reason of any stock dividend or split, stock repurchase, reorganization, recapitalization, merger, consolidation, spin-off, combination or exchange of shares or other similar corporate change, or any distributions to common stockholders other than regular cash dividends; (9) to exclude the effects of stock based compensation and the award of bonuses under the Company's bonus plans; (10) to exclude costs incurred in connection with potential acquisitions or divestitures that are required to expensed under generally accepted accounting principles; and (11) to exclude the goodwill and intangible asset impairment charges that are required to be recorded under generally accepted accounting principles. In addition, the Board retains the discretion to reduce or eliminate the compensation or economic benefit due upon attainment of Performance Goals and to define the manner of calculating the Performance Criteria it selects to use for such Performance Period. Partial achievement of the specified criteria may result in the payment or vesting corresponding to the degree of achievement as specified in the Award Agreement.

(ww) "*Performance Period*" means the period of time selected by the Board over which the attainment of one or more Performance Goals will be measured for the purpose of determining a Participant's right to vesting or exercise of an Award. Performance Periods may be of varying and overlapping duration, at the sole discretion of the Board.

(xx) "Plan" means this Orchestra BioMed Holdings, Inc. 2023 Equity Incentive Plan, as amended from time to time.

(yy) "*Plan Administrator*" means the person, persons, and/or third-party administrator designated by the Company to administer the day-to-day operations of the Plan and the Company's other equity incentive programs.

(zz) "Post-Termination Exercise Period" means the period following termination of a Participant's Continuous Service within which an Option or SAR is exercisable, as specified in Section 4(h).

(aaa) "Prior Plan" means the Orchestra BioMed, Inc. 2018 Stock Incentive Plan, as amended.

Section 5(a).

(bbb) "Restricted Stock Award" or "RSA" means an Award of shares of Common Stock granted pursuant to the terms and conditions of

(ccc) "Restricted Stock Award Agreement" means a written agreement between the Company and a holder of a Restricted Stock Award evidencing the terms and conditions of a Restricted Stock Award grant. The Restricted Stock Award Agreement includes the Grant Notice for the Restricted Stock Award and the agreement containing the written summary of the general terms and conditions applicable to the Restricted Stock Award and which is provided to a Participant along with the Grant Notice. Each Restricted Stock Award Agreement will be subject to the terms and conditions of the Plan.

(ddd) "*Returning Shares*" means shares subject to outstanding stock awards granted under the Prior Plan and that following the Effective Date: (A) are not issued because such stock award or any portion thereof expires or otherwise terminates without all of the shares covered by such stock award having been issued; (B) are not issued because such stock award or any portion thereof is settled in cash; (C) are forfeited back to or repurchased by the Company because of the failure to meet a contingency or condition required for the vesting of such shares; (D) are withheld or reacquired to satisfy the exercise, strike or purchase price; or (E) are withheld or reacquired to satisfy a tax withholding obligation.

(eee) "*RSU Award*" or "*RSU*" means an Award of restricted stock units representing the right to receive an issuance of shares of Common Stock which is granted pursuant to the terms and conditions of Section 5(a).

(fff) "RSU Award Agreement" means a written agreement between the Company and a holder of a RSU Award evidencing the terms and conditions of a RSU Award. The RSU Award Agreement includes the Grant Notice for the RSU Award and the agreement containing the written summary of the general terms and conditions applicable to the RSU Award and which is provided to a Participant along with the Grant Notice. Each RSU Award Agreement will be subject to the terms and conditions of the Plan.

time.

(ggg) "Rule 16b-3" means Rule 16b-3 promulgated under the Exchange Act or any successor to Rule 16b-3, as in effect from time to

(hhh) "Rule 405" means Rule 405 promulgated under the Securities Act.

(iii) "Section 409A" means Section 409A of the Code and the regulations and other guidance thereunder.

(jjj) "Section 409A Change in Control" means a change in the ownership or effective control of the Company, or in the ownership of a substantial portion of the Company's assets, as provided in Section 409A(a)(2)(A)(v) of the Code and Treasury Regulations Section 1.409A-3(i)(5) (without regard to any alternative definition thereunder).

(kkk) "Securities Act" means the U.S. Securities Act of 1933, as amended.

(III) "Share Reserve" means the number of shares available for issuance under the Plan as set forth in Section 2(a).

(mmm) "Stock Appreciation Right" or "SAR" means a right to receive the appreciation on Common Stock that is granted pursuant to the terms and conditions of Section 4.

(nnn) "SAR Agreement" means a written agreement between the Company and a holder of a SAR evidencing the terms and conditions of a SAR grant. The SAR Agreement includes the Grant Notice for the SAR and the agreement containing the written summary of the general terms and conditions applicable to the SAR and which is provided to a Participant along with the Grant Notice. Each SAR Agreement will be subject to the terms and conditions of the Plan.

(000) "Subsidiary" means, with respect to the Company, (i) any corporation of which more than 50% of the outstanding Common Stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether, at the time, stock of any other class or classes of such corporation will have or might have voting power by reason of the happening of any contingency) is at the time, directly or indirectly, Owned by the Company, and (ii) any partnership, limited liability company or other entity in which the Company has a direct or indirect interest (whether in the form of voting or participation in profits or capital contribution) of more than 50%.

(ppp) "Ten Percent Stockholder" means a person who Owns (or is deemed to Own pursuant to Section 424(d) of the Code) stock possessing more than 10% of the total combined voting power of all classes of stock of the Company or any Affiliate.

(qqq) "*Trading Policy*" means the Company's policy permitting certain individuals to sell Company shares only during certain "window" periods and/or otherwise restricts the ability of certain individuals to transfer or encumber Company shares, as in effect from time to time.

(rrr) "Unvested Non-Exempt Award" means the portion of any Non-Exempt Award that had not vested in accordance with its terms upon or prior to the date of any Corporate Transaction.

(sss) "Vested Non-Exempt Award" means the portion of any Non-Exempt Award that had vested in accordance with its terms upon or prior to the date of a Corporate Transaction.

ORCHESTRA BIOMED HOLDINGS, INC. STOCK OPTION GRANT NOTICE (2023 EQUITY INCENTIVE PLAN)

Orchestra BioMed Holdings, Inc. (the "Company"), pursuant to its 2023 Equity Incentive Plan (the "Plan"), has granted to you ("Optionholder") an option to purchase the number of shares of the Common Stock set forth below (the "Option"). Your Option is subject to all of the terms and conditions as set forth herein and in the Plan and the Stock Option Agreement, all of which are attached hereto and incorporated herein in their entirety. Capitalized terms not explicitly defined herein, but defined in the Plan or the Stock Option Agreement, shall have the meanings set forth in the Plan or the Stock Option Agreement, as applicable.

Optionholder:	
Date of Grant:	
Vesting Commencement Date:	
Number of Shares of Common Stock Subject to Option:	
Exercise Price (Per Share):	
Total Exercise Price:	
Expiration Date:	

Type of Grant: [Incentive Stock Option] OR [Nonstatutory Stock Option]

Exercise and

Vesting Schedule: Subject to the Optionholder's Continuous Service through each applicable vesting date, the Option will vest as follows: []. Notwithstanding the foregoing, vesting shall terminate upon the Optionholder's termination of Continuous Service.

Optionholder Acknowledgements: By your signature below, or by electronic acceptance or authentication in a form authorized by the Company, you understand and agree that:

- The Option is governed by this Stock Option Grant Notice, and the provisions of the Plan and the Stock Option Agreement, all of which are made a part of this document. Unless otherwise provided in the Plan, this Grant Notice and the Stock Option Agreement (together, the "Option Agreement") may not be modified, amended or revised, except in a writing signed by you and a duly authorized officer of the Company.
- If the Option is an Incentive Stock Option, it (plus other outstanding Incentive Stock Options granted to you) cannot be first exercisable for more than \$100,000 in value (measured by exercise price) in any calendar year. Any excess over \$100,000 is a Nonstatutory Stock Option.
- You consent to receive this Grant Notice, the Stock Option Agreement, the Plan, the prospectus prepared for the Plan (the "Prospectus") and any other Plan-related documents by electronic delivery and to participate in the Plan through an on-line or electronic system established and maintained by the Company or another third party designated by the Company.
- You have read and are familiar with the provisions of the Plan, the Stock Option Agreement, and the Prospectus. In the event of any conflict between the provisions in this Grant Notice, the Option Agreement, or the Prospectus and the terms of the Plan, the terms of the Plan shall control.
- The Option Agreement sets forth the entire understanding between you and the Company regarding the acquisition of Common Stock and supersedes all prior oral and written agreements, promises and/or representations on that subject, with the exception of other equity awards previously granted to you and any.
- written employment agreement, offer letter, severance agreement, written severance plan or policy, or other written agreement between the Company and you in each case that specifies the terms that should govern this Option.
- Counterparts may be delivered via facsimile, electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, Uniform Electronic Transactions Act or other applicable law) or other transmission method, and any counterpart so delivered, will be deemed to have been duly and validly delivered and be valid and effective for all purposes.

ORCHESTRA BIOMED HOLDINGS, INC.

OPTIONHOLDER:

Signature

Title:

By:

Signature

Date:

Date:

Orchestra BioMed Holdings, Inc. 2023 Equity Incentive Plan

STOCK OPTION AGREEMENT

As reflected by your Stock Option Grant Notice ("*Grant Notice*") Orchestra BioMed Holdings, Inc. (the "*Company*") has granted you an option under its 2023 Equity Incentive Plan (the "*Plan*") to purchase a number of shares of Common Stock at the exercise price indicated in your Grant Notice (the "*Option*"). Capitalized terms not explicitly defined in this Agreement, but defined in the Grant Notice or the Plan, shall have the meanings set forth in the Grant Notice or Plan, as applicable. The terms of your Option as specified in the Grant Notice and this Stock Option Agreement constitute your Option Agreement.

The general terms and conditions applicable to your Option are as follows:

1. GOVERNING PLAN DOCUMENT. Your Option is subject to all the provisions of the Plan. Your Option is further subject to all interpretations, amendments, rules and regulations, which may, from time to time, be promulgated and adopted pursuant to the Plan. In the event of any conflict between the Option Agreement and the provisions of the Plan, the provisions of the Plan shall control.

2. EXERCISE.

(a) You may generally exercise the vested portion of your Option for whole shares of Common Stock at any time during its term by delivery of payment of the exercise price and applicable withholding taxes and other required documentation to the Plan Administrator in accordance with the exercise procedures established by the Plan Administrator, which may include an electronic submission. Please review the Plan, which may restrict or prohibit your ability to exercise your Option during certain periods.

(b) To the extent permitted by Applicable Law, you may pay your Option exercise price as follows:

(i) cash, check, bank draft or money order;

(ii) subject to Company and/or Committee consent at the time of exercise, pursuant to a "cashless exercise" program as further described in the Plan, if at the time of exercise the Common Stock is publicly traded;

(iii) subject to Company and/or Committee consent at the time of exercise, by delivery of previously owned shares of Common Stock as further described in the Plan; or

(iv) subject to Company and/or Committee consent at the time of exercise, if the Option is a Nonstatutory Stock Option, by a "net exercise" arrangement as further described in the Plan.

3. TERM. You may not exercise your Option before the commencement of its term or after its term expires. The term of your option commences on the Date of Grant and expires upon the earliest of the following:

- (a) immediately upon the termination of your Continuous Service for Cause;
- (b) three months after the termination of your Continuous Service for any reason other than Cause, Disability or death;
- (c) 12 months after the termination of your Continuous Service due to your Disability;
- (d) 18 months after your death if you die during your Continuous Service;

(e) immediately upon a Corporate Transaction if the Board has determined that the Option will terminate in connection with a Corporate

Transaction,

- (f) the Expiration Date indicated in your Grant Notice; or
- (g) the day before the 10th anniversary of the Date of Grant.

Notwithstanding the foregoing, if you die during the period provided in Section 3(b) or 3(c) above, the term of your Option shall not expire until the earlier of (i) 18 months after your death, (ii) upon any termination of the Option in connection with a Corporate Transaction, (iii) the Expiration Date indicated in your Grant Notice, or (iv) the day before the 10th anniversary of the Date of Grant. Additionally, the Post-Termination Exercise Period of your Option may be extended as provided in the Plan.

To obtain the federal income tax advantages associated with an Incentive Stock Option, the Code requires that at all times beginning on the date of grant of your Option, and ending on the day three months before the date of your Option's exercise, you must be an employee of the Company or an Affiliate, except in the event of your death or Disability. If the Company provides for the extended exercisability of your Option under certain circumstances for your benefit, your Option will not necessarily be treated as an Incentive Stock Option if you exercise your Option more than three months after the date your employment terminates.

4. WITHHOLDING OBLIGATIONS.

(a) Regardless of any action taken by the Company or, if different, the Affiliate to which you provide Continuous Service (the "*Service Recipient*") with respect to any income tax, social insurance, payroll tax, fringe benefits tax, payment on account, or other tax-related items associated with the grant, vesting or exercise of the Option or sale of the underlying Common Stock or other tax-related items related to your participation in the Plan and legally applicable to you (the "*Tax Liability*"), you hereby acknowledge and agree that the Tax Liability is your ultimate responsibility and may exceed the amount, if any, actually withheld by the Company or the Service Recipient. You further acknowledge that the Company and the Service Recipient (i) make no representations or undertakings regarding any Tax Liability in connection with any aspect of this Option, including, but not limited to, the grant, vesting or exercise of the Option, the issuance of Common Stock pursuant to such exercise, the subsequent sale of shares of Common Stock, and the payment of any dividends on the shares; and (ii) do not commit to, and are under no obligation to structure the terms of the grant or any aspect of the Option to reduce or eliminate your Tax Liability or achieve a particular tax result. Further, if you are subject to Tax Liability in more than one jurisdiction, you acknowledge that the Company and/or the Service Recipient (or former service recipient, as applicable) may be required to withhold or account for Tax Liability in more than one jurisdiction.

(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or the Service Recipient to satisfy all Tax Liability. As further provided in Section 8 of the Plan, you hereby authorize the Company and any applicable Service Recipient to satisfy any applicable withholding obligations with regard to the Tax Liability by one or a combination of the following methods: (i) causing you to pay any portion of the Tax Liability in cash or cash equivalent in a form acceptable to the Company; (ii) withholding from any compensation otherwise payable to you by the Company or the Service Recipient; (iii) withholding from the proceeds of the sale of shares of Common Stock issued upon exercise of the Option (including by means of a "cashless exercise" pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board to the extent permitted by the Company, or by means of the Company acting as your agent to sell sufficient shares of Common Stock for the proceeds to settle such withholding requirements, on your behalf pursuant to this authorization without further consent); (iv) withholding shares of Common Stock otherwise issuable to you upon the exercise of the Option, provided that to the extent necessary to qualify for an exemption from application of Section 16(b) of the Exchange Act, if applicable, such share withholding procedure will be subject to the express prior approval of the Board or the Company's Compensation Committee; and/or (v) any other method determined by the Company to be in compliance with Applicable Law. Furthermore, you agree to pay the Company or the Service Recipient any amount the Company or the Service Recipient may be required to withhold, collect or pay as a result of your participation in the Plan or that cannot be satisfied by the means previously described. In the event it is determined that the amount of the Tax Liability was greater than the amount withheld by the Company or the Service Recipient, you agree to indemnify and hold the Company and/or the Service Recipient (as applicable) harmless from any failure by the Company or the applicable Service Recipient to withhold the proper amount.

(c) The Company may withhold or account for your Tax Liability by considering statutory withholding amounts or other withholding rates applicable in your jurisdiction(s), including (i) maximum applicable rates in your jurisdiction(s), in which case you may receive a refund of any overwithheld amount in cash (whether from applicable tax authorities or the Company) and you will have no entitlement to the equivalent amount in Common Stock or (ii) minimum or such other applicable rates in your jurisdiction(s), in which case, you may be solely responsible for paying any additional Tax Liability to the applicable tax authorities or to the Company and/or the Service Recipient. If the Tax Liability withholding obligation is satisfied by withholding shares of Common Stock, for tax purposes, you are deemed to have been issued the full number of shares of Common Stock subject to the exercised portion of the Option, notwithstanding that a number of the shares of Common Stock is held back solely for the purpose of paying such Tax Liability.

(d) You acknowledge that you may not be able to exercise your Option, even though the Option is vested, and that the Company shall have no obligation to issue shares of Common Stock, in each case, unless, and until you have fully satisfied any applicable Tax Liability, as determined by the Company. Unless any withholding obligation for the Tax Liability is satisfied, the Company shall have no obligation to deliver to you any Common Stock in respect of the Option.

5. INCENTIVE STOCK OPTION DISPOSITION REQUIREMENT. If your option is an Incentive Stock Option, you must notify the Company in writing within 15 days after the date of any disposition of any of the shares of the Common Stock issued upon exercise of your option that occurs within two years after the date of your option grant or within one year after such shares of Common Stock are transferred upon exercise of your option.

6. TRANSFERABILITY. Except as otherwise provided in the Plan, your Option is not transferable, except by will or by the applicable laws of descent and distribution, and is exercisable during your life only by you.

7. CORPORATE TRANSACTION. Your Option is subject to the terms of any agreement governing a Corporate Transaction involving the Company, including, without limitation, a provision for the appointment of a stockholder representative that is authorized to act on your behalf with respect to any escrow, indemnities and any contingent consideration.

8. No LIABILITY FOR TAXES. As a condition to accepting the Option, you hereby (a) agree to not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates related to tax liabilities arising from the Option or other Company compensation and (b) acknowledge that you were advised to consult with your own personal tax, financial and other legal advisors regarding the tax consequences of the Option, and have either done so, or knowingly and voluntarily declined to do so. Additionally, you acknowledge that the Option is exempt from Section 409A, only if the exercise price is at least equal to the "fair market value" of the Common Stock on the date of grant as determined by the Internal Revenue Service, and there is no other impermissible deferral of compensation associated with the Option. Additionally, as a condition to accepting the Option, you agree not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates in the event that the Internal Revenue Service asserts that such exercise is less than the "fair market value" of the Common Stock on the date of grant as subsequently determined by the Internal Revenue Service.

9. OBLIGATIONS; RECOUPMENT. You hereby acknowledge that the grant of your Option is additional consideration for any obligations (whether during or after employment) that you have to the Company not to compete, not to solicit its customers, clients or employees, not to disclose or misuse confidential information or similar obligations. Accordingly, if the Company reasonably determines that you breached such obligations, in addition to any other available remedy, the Company may, to the extent permitted by Applicable Law, recoup any income realized by you with respect to the exercise of your Option within two years of such breach. In addition, to the extent permitted by Applicable Law, this right to recoupment by the Company applies in the event that your employment is terminated for Cause, or if the Company reasonably determines that circumstances existed that it could have terminated your employment for Cause.

10. SEVERABILITY. If any part of this Option Agreement or the Plan is declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity will not invalidate any portion of this Option Agreement or the Plan not declared to be unlawful or invalid. Any Section of this Option Agreement (or part of such a Section) so declared to be unlawful or invalid, will, if possible, be construed in a manner which will give effect to the terms of such Section or part of a Section to the fullest extent possible, while remaining lawful and valid.

11. INDEBTEDNESS TO THE COMPANY. In the event that you have any loans, draws, advances or any other indebtedness owing to the Company at the time of exercise of all or a portion of the Option, the Company may deduct and not deliver that number of shares of Common Stock with a Fair Market Value subject to the Option equal to such indebtedness to satisfy all or a portion of such indebtedness, to the extent permitted by law and in a manner consistent with Section 409A of the Code, if applicable.

12. OTHER DOCUMENTS. You hereby acknowledge receipt of, or the right to receive a document providing the information required by Rule 428(b) (1) promulgated under the Securities Act, which includes the Prospectus. In addition, you acknowledge receipt of the Company's Trading Policy.

13. QUESTIONS. If you have questions regarding these or any other terms and conditions applicable to your Option, including a summary of the applicable federal income tax consequences, please see the Prospectus.

* * * *

Orchestra BioMed Holdings, Inc. RSU Award Grant Notice (2023 Equity Incentive Plan)

Orchestra BioMed Holdings, Inc. (the "*Company*") has awarded to you (the "*Participant*") the number of restricted stock units specified and on the terms set forth below in consideration of your services (the "*RSU Award*"). Your RSU Award is subject to all of the terms and conditions as set forth herein and in the Company's 2023 Equity Incentive Plan (the "*Plan*") and the Award Agreement (the "*Agreement*"), which are incorporated herein in their entirety. Capitalized terms not explicitly defined herein but defined in the Plan or the Agreement shall have the meanings set forth in the Plan or the Agreement.

Participant:

Date of Grant:

Vesting Commencement Date:

Number of Restricted Stock Units:

Vesting Schedule:

[_____]. Notwithstanding the foregoing, vesting shall terminate upon the Participant's termination of Continuous Service.

Issuance Schedule: One share of Common Stock will be issued for each restricted stock unit, which vests at the time set forth in Section 5 of the Agreement.

Participant Acknowledgements: By your signature below or by electronic acceptance or authentication in a form authorized by the Company, you understand and agree that:

- The RSU Award is governed by this RSU Award Grant Notice (the "*Grant Notice*"), and the provisions of the Plan and the Agreement, all of which are made a part of this document. Unless otherwise provided in the Plan, this Grant Notice and the Agreement (together, the "*RSU Award Agreement*") may not be modified, amended or revised except in a writing signed by you and a duly authorized officer of the Company.
- You have read and are familiar with the provisions of the Plan, the RSU Award Agreement and the prospectus prepared for the Plan (the *"Prospectus"*). In the event of any conflict between the provisions in the RSU Award Agreement, or the Prospectus and the terms of the Plan, the terms of the Plan shall control.
- The RSU Award Agreement sets forth the entire understanding between you and the Company regarding the acquisition of Common Stock and supersedes all prior oral and written agreements, promises and/or representations on that subject with the exception of: (i) other equity awards previously granted to you, and (ii) any written employment agreement, offer letter, severance agreement, written severance plan or policy, or other written agreement between the Company and you in each case that specifies the terms that should govern this RSU Award.

ORCHESTRA BIOMED HOLDINGS, INC.

PARTICIPANT:

By:

Title:

Date:

Signature

Date:

Signature

Orchestra BioMed Holdings, Inc. 2023 Equity Incentive Plan Award Agreement (RSU Award)

As reflected by your Restricted Stock Unit Grant Notice ("*Grant Notice*"), Orchestra BioMed Holdings, Inc. (the "*Company*") has granted you a RSU Award under its 2023 Equity Incentive Plan (the "*Plan*") for the number of restricted stock units as indicated in your Grant Notice (the "*RSU Award*"). The terms of your RSU Award as specified in this Award Agreement for your RSU Award (the "*Agreement*") and the Grant Notice constitute your "*RSU Award Agreement*". Defined terms not explicitly defined in this Agreement but defined in the Grant Notice or the Plan shall have the same definitions as in the Grant Notice or Plan, as applicable.

The general terms applicable to your RSU Award are as follows:

1. GOVERNING PLAN DOCUMENT. Your RSU Award is subject to all the provisions of the Plan. Your RSU Award is further subject to all interpretations, amendments, rules and regulations, which may, from time to time, be promulgated and adopted pursuant to the Plan. In the event of any conflict between the RSU Award Agreement and the provisions of the Plan, the provisions of the Plan shall control.

2. GRANT OF THE RSU AWARD. This RSU Award represents your right to be issued on a future date the number of shares of the Company's Common Stock that is equal to the number of restricted stock units indicated in the Grant Notice subject to your satisfaction of the vesting conditions set forth therein (the "*Restricted Stock Units*"). Any additional Restricted Stock Units that become subject to the RSU Award pursuant to Capitalization Adjustments as set forth in the Plan and the provisions of Section 3 below, if any, shall be subject, in a manner determined by the Board, to the same forfeiture restrictions, restrictions on transferability, and time and manner of delivery as applicable to the other Restricted Stock Units covered by your RSU Award.

3. DIVIDENDS. You shall receive no benefit or adjustment to your RSU Award with respect to any cash dividend, stock dividend or other distribution that does not result from a Capitalization Adjustment as provided in the Plan; provided, however, that this sentence shall not apply with respect to any shares of Common Stock that are delivered to you in connection with your RSU Award after such shares have been delivered to you.

4. WITHHOLDING OBLIGATIONS.

(a) Regardless of any action taken by the Company or, if different, the Affiliate to which you provide Continuous Service (the "*Service Recipient*") with respect to any income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related items associated with the grant or vesting of the RSU Award or sale of the underlying Common Stock or other tax-related items related to your participation in the Plan and legally applicable to you (the "*Tax Liability*"), you hereby acknowledge and agree that the Tax Liability is your ultimate responsibility and may exceed the amount, if any, actually withheld by the Company or the Service Recipient. You further acknowledge that the Company and the Service Recipient (i) make no representations or undertakings regarding any Tax Liability in connection with any aspect of this RSU Award, including, but not limited to, the grant or vesting of the RSU Award, the issuance of Common Stock pursuant to such vesting, the subsequent sale of shares of Common Stock, and the payment of any dividends on the Common Stock; and (ii) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the RSU Award to reduce or eliminate your Tax Liability or achieve a particular tax result. Further, if you are subject to Tax Liability in more than one jurisdiction, you acknowledge that the Company and/or the Service Recipient (or former service recipient, as applicable) may be required to withhold or account for Tax Liability in more than one jurisdiction.



(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or the Service Recipient to satisfy all Tax Liability. As further provided in Section 8 of the Plan, you hereby authorize the Company and any applicable Service Recipient to satisfy any applicable withholding obligations with regard to the Tax Liability by any of the following means or by a combination of such means: (i) causing you to pay any portion of the Tax Liability in cash or cash equivalent in a form acceptable to the Company; (ii) withholding from any compensation otherwise payable to you by the Company or the Service Recipient; (iii) withholding shares of Common Stock from the shares of Common Stock issued or otherwise issuable to you in connection with the Award; provided, however, that to the extent necessary to qualify for an exemption from application of Section 16(b) of the Exchange Act, if applicable, such share withholding procedure will be subject to the express prior approval of the Board or the Company's Compensation Committee; (iv) permitting or requiring you to enter into a "same day sale" commitment, if applicable, with a broker-dealer that is a member of the Financial Industry Regulatory Authority (a "FINRA Dealer"), pursuant to this authorization and without further consent, whereby you irrevocably elect to sell a portion of the shares of Common Stock to be delivered in connection with your Restricted Stock Units to satisfy the Tax Liability and whereby the FINRA Dealer irrevocably commits to forward the proceeds necessary to satisfy the Tax Liability directly to the Company or the Service Recipient; and/or (v) any other method determined by the Company to be in compliance with Applicable Law. Furthermore, you agree to pay the Company or the Service Recipient any amount the Company or the Service Recipient may be required to withhold, collect, or pay as a result of your participation in the Plan or that cannot be satisfied by the means previously described. In the event it is determined that the amount of the Tax Liability was greater than the amount withheld by the Company and/or the Service Recipient (as applicable), you agree to indemnify and hold the Company and/or the Service Recipient (as applicable) harmless from any failure by the Company or the applicable Service Recipient to withhold the proper amount.

(c) The Company may withhold or account for your Tax Liability by considering statutory withholding amounts or other withholding rates applicable in your jurisdiction(s), including (i) maximum applicable rates in your jurisdiction(s), in which case you may receive a refund of any over-withheld amount in cash (whether from applicable tax authorities or the Company) and you will have no entitlement to the equivalent amount in Common Stock or (ii) minimum or such other applicable rates in your jurisdiction(s), in which case you may be solely responsible for paying any additional Tax Liability to the applicable tax authorities or to the Company and/or the Service Recipient. If the Tax Liability withholding obligation is satisfied by withholding shares of Common Stock, for tax purposes, you are deemed to have been issued the full number of shares of Common Stock subject to the vested portion of the RSU Award, notwithstanding that a number of the shares of Common Stock is held back solely for the purpose of paying such Tax Liability.

(d) You acknowledge that you may not participate in the Plan and the Company shall have no obligation to deliver shares of Common Stock until you have fully satisfied the Tax Liability, as determined by the Company. Unless any withholding obligation for the Tax Liability is satisfied, the Company shall have no obligation to deliver to you any Common Stock in respect of the RSU Award.

5. DATE OF ISSUANCE. The issuance of shares in respect of the Restricted Stock Units is intended to comply with U.S. Treasury Regulations Section 1.409A-3(a) and will be construed and administered in such a manner. Subject to the satisfaction of the Tax Liability withholding obligation, if any, in the event one or more Restricted Stock Units vests, the Company shall issue to you one (1) share of Common Stock for each vested Restricted Stock Unit. Each issuance date determined by this paragraph is referred to as an "*Original Issuance Date*." The Original Issuance Date shall be the date the underlying Restricted Stock Unit vests except that if the Original Issuance Date otherwise would fall on a date that is not a business day, the Original Issuance Date shall instead occur on the next following business day. In addition, and notwithstanding the foregoing, if:

(a) the Original Issuance Date does not occur (1) during an "open window period" applicable to you, as determined by the Company in accordance with the Company's then-effective policy on trading in Company securities, or (2) on a date when you are otherwise permitted to sell shares of Common Stock on an established stock exchange or stock market (including but not limited to under a previously established written trading plan that meets the requirements of Rule 10b5-1 under the Exchange Act and was entered into in compliance with the Company's policies (a "10b5-1 Arrangement")); and

(b) either (1) a Tax Liability withholding obligation does not apply, or (2) the Company decides, prior to the Original Issuance Date, (A) not to satisfy the Tax Liability withholding obligation by withholding shares of Common Stock from the shares otherwise due, on the Original Issuance Date, to you under this Award, and (B) not to permit you to enter into a "same day sale" commitment with a broker-dealer (including, but not limited to, a commitment under a 10b5-1 Arrangement) and (C) not to permit you to pay your Tax Liability in cash, then the shares that would otherwise be issued to you on the Original Issuance Date will not be delivered on such Original Issuance Date and will instead be delivered on the first business day when you are not prohibited from selling shares of the Common Stock in the open public market, but in no event later than December 31 of the calendar year in which the Original Issuance Date occurs (that is, the last day of your taxable year in which the Original Issuance Date occurs), or, <u>if and only if</u> permitted in a manner that complies with U.S. Treasury Regulations Section 1.409A-1(b)(4), no later than the date that is the 15th day of the third calendar month of the applicable year following the year in which the shares of Common Stock under this Award are no longer subject to a "substantial risk of forfeiture" within the meaning of U.S. Treasury Regulations Section 1.409A-1(d).

6. TRANSFERABILITY. Except as otherwise provided in the Plan, your RSU Award is not transferable, except by will or by the applicable laws of descent and distribution

7. CORPORATE TRANSACTION. Your RSU Award is subject to the terms of any agreement governing a Corporate Transaction involving the Company, including, without limitation, a provision for the appointment of a stockholder representative that is authorized to act on your behalf with respect to any escrow, indemnities and any contingent consideration.

8. No LIABILITY FOR TAXES. As a condition to accepting the RSU Award, you hereby (a) agree to not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates related to tax liabilities arising from the RSU Award or other Company compensation and (b) acknowledge that you were advised to consult with your own personal tax, financial and other legal advisors regarding the tax consequences of the RSU Award and have either done so or knowingly and voluntarily declined to do so.

9. SEVERABILITY. If any part of this Agreement or the Plan is declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity will not invalidate any portion of this Agreement or the Plan not declared to be unlawful or invalid. Any Section of this Agreement (or part of such a Section) so declared to be unlawful or invalid will, if possible, be construed in a manner which will give effect to the terms of such Section or part of a Section to the fullest extent possible while remaining lawful and valid.

10. OTHER DOCUMENTS. You hereby acknowledge receipt of or the right to receive a document providing the information required by Rule 428(b) (1) promulgated under the Securities Act, which includes the Prospectus. In addition, you acknowledge receipt of the Company's Trading Policy.

11. QUESTIONS. If you have questions regarding these or any other terms and conditions applicable to your RSU Award, including a summary of the applicable federal income tax consequences please see the Prospectus.

Orchestra BioMed Holdings, Inc. Amended and Restated Employment Agreement

THIS AMENDED AND RESTATED EMPLOYMENT AGREEMENT (this "<u>Agreement</u>") is entered into as of January 26, 2023 (the "<u>Amendment Date</u>"), by and between Orchestra BioMed Holdings, Inc., a Delaware corporation (the "<u>Company</u>") and the parent company of Orchestra BioMed, Inc., and David Hochman ("<u>Executive</u>"). This Agreement amends and restates in its entirety that certain Employment Agreement (the "<u>Prior</u> <u>Agreement</u>") entered into by and between Orchestra BioMed, Inc. and Executive, dated May 31, 2018 (the "<u>Original Effective Date</u>").

WHEREAS, Executive was an employee of Orchestra BioMed, Inc. prior to its acquisition by the Company (the "Acquisition"), which was previously known as Health Sciences Acquisitions Corp 2;

WHEREAS, upon the closing of the Acquisition, Orchestra BioMed, Inc. became a subsidiary of the Company;

WHEREAS, the Company desires to retain Executive as an employee of the Company, and Executive desires to be employed by the Company, upon the terms and subject to the conditions set forth herein.

NOW, THEREFORE, in consideration of the employment of Executive by the Company and the mutual agreements hereinafter set forth, the parties hereto agree as follows:

1. <u>Employment</u>. During the Employment Period (as defined below), the Company hereby agrees to employ Executive as Chief Executive Officer and Orchestra BioMed Founder, and Executive hereby accepts such employment and agrees to serve as Chief Executive Officer and Orchestra BioMed Founder, all in accordance with the terms and conditions of this Agreement. Executive hereby represents and warrants that neither Executive's entry into this Agreement nor Executive's performance of Executive's obligations hereunder will conflict with or result in a breach of the terms, conditions, or provisions of any other agreement or obligation of any nature to which Executive is a party or by which Executive is bound, including, without limitation, any non-competition agreement, non-solicitation agreement or confidentiality agreement or policy entered into by Executive or to which Executive is subject.

2. <u>Term of Employment</u>. The term of Executive's employment under this Agreement will commence on the Amendment Date and will continue until terminated in accordance with <u>Section 8</u> hereof (such term, the "<u>Employment Period</u>"). For the avoidance of doubt, and notwithstanding any contrary provision herein, Executive's employment is "at will" although Executive may be eligible for specified payments and benefits in the event of certain terminations of employment as set forth in Sections 4(c) and 8.

3. <u>Position and Responsibilities</u>. During the Employment Period: (a) Executive shall serve in the position of Chief Executive Officer and Orchestra BioMed Founder; (b) Executive shall have the duties, responsibilities, functions and authority, including administrative, financial, executive and managerial, as shall be designated by the Company's board of directors (the "<u>Board</u>") from time to time; (c) Executive will use his best efforts to promote the interests, prospects, condition (financial and otherwise) and welfare of the Company and shall perform his duties and responsibilities to the best of his ability in a diligent, trustworthy, businesslike and efficient manner; and (d) Executive shall comply at all times with all policies and codes of conduct of the Company, as such policies and codes may change from time to time.

(a) <u>Board of Directors</u>. So long as he remains the Company's Chief Executive Officer, the Company shall nominate Executive to serve as a Director. Unless otherwise agreed upon by Executive and the Company, Executive's service as a Director shall continue until the first to occur of (i) the date that Executive is no longer employed by the Company and its affiliates, (ii) the death of the Director, (iii) the resignation by the Executive from the Board of Directors, and (iv) the expiration of the then current term as a Director if Executive is not re-elected by the Company's stockholders to serve as a Director during the term of this Agreement, so long as he is elected to serve as a Director by the Company's stockholders.

4. Compensation.

(a) <u>Salary</u>. During the Employment Period, the Company shall pay to Executive a base salary at the rate of \$595,000 per annum (the "<u>Base Salary</u>"). On an annual basis the Board or the compensation committee of the Board (the "Compensation Committee") shall review Executive's Base Salary and may increase such amount as it may deem advisable. The Base Salary shall be paid in accordance with the Company's standard payroll practices in effect from time to time.

(b) <u>Bonus</u>. Commencing on the Amendment Date, in addition to the Base Salary, Executive shall be eligible to receive a discretionary annual bonus during each fiscal year of the Employment Period, with a target bonus amount of eighty percent (80%) of the Base Salary. The performance metrics and goals required to receive such bonus amount, shall be determined by the Board in consultation with the Company's Chief Executive Officer within sixty (60) days following the commencement of each fiscal year or, if later, at the next regularly scheduled meeting of the Board that occurs at least five business days after the Chief Executive Officer has submitted recommended performance metrics to the Board. The actual bonus, if any, payable under this <u>Section 4(b)</u> shall be paid within thirty (30) days following receipt of the Company's audited financial statements for the applicable fiscal year, unless such payment is delayed due to an unforeseeable administrative impracticability, in which case, such bonus will be paid as soon as administratively practicable thereafter, but in no event later than the end of the fiscal year immediately following the applicable fiscal year; provided that Executive is an employee with the Company at such time of payment and Executive has not breached any provision of this Agreement.

(c) <u>Equity Interest</u>. Company shall award Executive stock options to acquire common stock of the Company (the "<u>Common Shares</u>") or restricted common stock units ("RSUs") (as agreed upon by Executive and the Board) as soon as reasonably practicable following the beginning of the first open trading window after the date of this Agreement under the Company's Insider Trading Policy, so that Executive's long-term incentive equity holdings represent, in the aggregate, the Equity Award Percentage (as defined below; such grant, the "<u>Equity Award Grant</u>"). Except as otherwise provided herein, the terms of the Equity Award Grant shall reflect the Company's standard terms and conditions for executive equity award grants.

(i) <u>Defined Terms</u>. For the purposes of this Agreement, the following defined terms shall be construed to have the meanings set forth or referenced below.

(1) "<u>Equity Award Percentage</u>" means four and six-tenths percent (4.6%) of the fully diluted Common Shares of the combined company as of immediately following the closing of the Acquisition; provided, however, that notwithstanding any contrary provision herein, the parties acknowledge that the 4.6% assumes an award solely in the form of options. To the extent the Equity Award is in restricted stock units, the number of shares covered by the Equity Award will be reduced reflecting a ratio of 2 shares covered by restricted stock units is equivalent to 3 shares covered by options (i.e., rounding down, a restricted stock unit award covering 666 shares of Company common stock is deemed equivalent to an option award covering 1,000 shares of Company common stock).

(2) "Equity Vesting Schedule" means that twenty-five percent (25%) of any options awarded to the Executive shall be fully vested on the first (1st) anniversary of the Amendment Date and the remaining seventy-five percent (75%) of any options awarded to the Executive shall vest on a quarterly basis thereafter over the following three years; and unless otherwise agreed, any RSUs awarded to the Executive shall vest on a schedule to be agreed upon between the Executive and the Board (provided such vesting schedule shall be over a minimum of three (3) years unless otherwise agreed by the Company and Executive and other than as provided herein). All vesting shall be subject to continued employment or service as a member of the Board through the applicable vesting date; provided, however, in the event of the termination of Executive's employment and the Board upon termination of employment), the parties agree to negotiate in good faith a reduction in the number of shares as to which the Equity Award Grant has not yet vested so that future vesting of the award more closely reflects grant sizes for non-employee directors.

additional terms:

(ii) Other Terms. The Equity Award Grant shall vest according to the Equity Vesting Schedule and contain the following

(1) Notwithstanding the terms of any document to the contrary, in the event that (i) a "Change of Control" (as defined in the Company's 2023 Equity Incentive Plan provided, however, that the Acquisition is not a Change of Control) occurs, and (ii) the surviving corporation or the acquiring corporation fails to either (A) continue or assume any or all of Executive's Company options or other stock awards or (B) substitute similar stock awards for Executive's Company options or other stock awards (it being understood that similar stock awards include, but are not limited to, awards to acquire the same consideration paid to the stockholders or the Company, as the case may be, pursuant to the Change of Control), then 100% of Executive's unvested Company options or other stock awards shall immediately vest as of the date immediately preceding the Change of Control.

(2) If the Executive is terminated without Cause (as defined below), or voluntarily terminates the Employment Period with Good Reason during the period commencing three (3) months prior to a Change of Control of and ending twelve (12) months following a Change of Control (the "Change of Control Period"), and subject to the Executive satisfying the Release Requirement (defined below), any unvested equity awards then held by the Executive shall become fully vested.

(3) If the Executive is terminated without Cause (as defined below), or voluntarily terminates the Employment Period with Good Reason outside of the Change of Control Period, and subject to the Executive satisfying the Release Requirement, the vesting of any equity held by the Executive as of the closing of the Acquisition shall be determined as if the Executive had remained employed for an additional twelve (12) months beyond his date of termination.

5. <u>Benefit Plans</u>. During the Employment Period, Executive shall be eligible to participate in and receive benefits under the employee benefit plans, equity incentive plans, practices, policies, perquisites and programs established and maintained by the Company from time to time, in the Company's sole discretion, for which executives of the Company are generally eligible, in each case, subject to the eligibility and participation requirements thereof.

6. <u>Business Expenses</u>. The Company, in accordance with policies and practices established by the Company from time to time, will pay or reimburse Executive for all expenses reasonably incurred by Executive during the Employment Period in connection with the performance of Executive's duties under this Agreement; <u>provided</u>, that Executive shall provide to the Company reasonable documentation or evidence of expenses for which Executive seeks reimbursement in accordance with the policies and procedures established by the Company from time to time. Executive's air travel shall be pursuant to Company policy for executives.

7. <u>Paid Time Off</u>. Executive shall be entitled, during the Employment Period, to four (4) weeks of paid vacation and sick time in each calendar year or pursuant to Company policy for executives, whichever is greater. Accrued vacation and sick time not taken in any calendar year may not be carried forward or be usable in any subsequent calendar year, nor shall Executive be entitled to compensation for unused vacation or sick days during the Employment Period or upon termination of employment. Paid holidays may be taken in accordance with the holiday policy and schedule of the Company as from time to time in effect.

8. Termination.

(a) <u>Termination For Cause</u>. The Company may terminate the Employment Period immediately upon the occurrence of any of the following events (each of which shall constitute "<u>Cause</u>"):

(i) Executive's breach of any of Executive's obligations under Section 9;

(ii) Executive's breach of any of Executive's obligations under this Agreement other than <u>Section 9</u>, which, to the extent curable, has not been cured within thirty (30) days after Executive has been provided written notice of such breach;

(iii) Executive being convicted of, or pleading guilty or nolo contendere to, or being indicted for, any felony or any misdemeanor involving theft, fraud, dishonesty or moral turpitude; or

(iv) Fraud or embezzlement against the Company.

If the Employment Period is terminated pursuant to this <u>Section 8(a)</u>, the Company shall have no further obligation to Executive except for (i) payment of Base Salary accrued but unpaid through the date of termination and (ii) subject to the provisions of <u>Section 6</u>, reimbursement of any expense properly incurred by Executive through the date of termination (collectively, the "<u>Accrued Obligations</u>"). The Accrued Obligations shall be paid in accordance with applicable state law and in no event later than thirty (30) days after the date of termination.

(b) <u>Termination upon Executive's Death</u>. The Employment Period shall terminate immediately upon the death of Executive. If the Employment Period is terminated pursuant to this <u>Section 8(b)</u>, the Company shall have the following obligations to Executive (or his estate) (i) the payment of the Accrued Obligations, which shall be paid in accordance with <u>Section 8(a)</u>, and (ii) the payment of Severance Obligations, which shall be paid in accordance with <u>Section 8(f)</u>.

(c) <u>Termination by the Company without Cause</u>. The Company may terminate the Employment Period at any time without Cause upon written notice to Executive. If the Employment Period is terminated pursuant to this <u>Section 8(c)</u>, the Company shall have the following obligations to Executive (or his estate): (i) Payment of all Accrued Obligations, which shall be paid in accordance with <u>Section 8(a)</u>; and (ii) Payment of the Severance Obligations as described in <u>Section 8(f)</u>.

(d) <u>Voluntary Resignation by Executive - Without Cause</u>. Executive may terminate the Employment Period for any reason upon not less than ten (10) days' prior written notice. If the Employment Period is terminated pursuant to this <u>Section 8(d)</u>, the Company shall have no further obligation to Executive except for Accrued Obligations, which shall be paid in accordance with <u>Section 8(a)</u>.

(e) <u>Voluntary Resignation by Executive - With Good Reason</u>. Executive may terminate the Employment Period, with Good Reason, upon not less than ten (10) days' prior written notice. If Executive terminates the Employment Period with Good Reason, the Company shall have the following obligations to Executive (or his estate):

(i) Payment of all Accrued Obligations, which shall be paid in accordance with Section 8(a); and

(ii) Payment of the Severance Obligations as described in Section 8(f).

(iii) For purposes of this <u>Section 8(e)</u>, "<u>Good Reason</u>" shall mean each of the following, without Executive's consent: (A) any change in the Executive's position, title or reporting relationship with the Company that diminishes in any material respect the Executive's authority, duties or responsibilities; provided, however, that a change in the Executive's authority, duties or responsibilities solely due to the Company becoming a division, subsidiary or other similar part of a larger organization, shall not by itself constitute Good Reason; (B) any material reduction in the Executive's base compensation; (C) the relocation of the Executive's principal office, or principal place of employment, to a location more than fifty (50) miles from the location of the Executive's principal office or principal place of business; or (D) a material breach of any provision hereof by the Company or any successor or assign. Notwithstanding the foregoing, "Good Reason" shall not be deemed to have occurred unless: (1) the Executive provides the Company with written notice that the Executive intends to terminate the Executive's employment hereunder for one of the grounds set forth in subsections (A), (B), (C) or (D) within sixty (60) days of such reason(s) occurring, (2) if such ground is capable of being cured, the Company has failed to cure such ground within a period of sixty (60) days from the date of such written notice, and (3) the Executive terminates the Executive's employment within sixty (60) days after the cure period has lapsed. For purposes of clarification, the above-listed conditions shall apply separately to each occurrence of Good Reason and failure to adhere to such conditions in the event of Good Reason shall not disqualify the Executive from asserting Good Reason for any subsequent occurrence of Good Reason.

(f) Severance Payments. In the event Executive's employment is terminated (i) by the Company without Cause or (ii) by Executive for Good Reason, the Company shall continue to pay Executive as severance, the aggregate sum of one hundred percent (100%) of Executive's then-effective Base Salary plus his target bonus for the year of termination proportionally for a period of twelve (12) months beginning on the date such termination becomes effective (such payments, the Company's "Severance Obligations"), and if the Executive validly elects to continue his healthcare coverage under COBRA, reimburse the Executive for the premiums he pays to continue healthcare coverage for the Executive and his eligible dependents for twelve (12) months or, if earlier, until the Executive becomes eligible for healthcare coverage from a subsequent employer; provided, as a condition to all amounts specified in this clause (f), that Executive executes a full and final release substantially in the form attached as Exhibit B hereto (with such changes as the Company may reasonably request to reflect changes in applicable law) and such release becomes irrevocably effective within 60 days following the termination of Executive's employment (such requirement for an irrevocably effective release, the "Release Requirement"). Any and all payments of Severance Obligations shall be less applicable withholding and deductions. Further, in the event Executive's employment is terminated (i) by the Company without Cause or (ii) by Executive for Good Reason, and subject to satisfaction of the Release Requirement, the stated post-termination exercise period of Executive's outstanding options shall be extended to twelve months following the termination of Executive's employment; provided, however, options shall expire earlier upon expiration of the maximum term or, subject to Section 4(c)(ii)(1), pursuant to a Change of Control if all Company options are terminated in connection with the Change of Control.

In the event of such termination without Cause or resignation for Good Reason during the Change of Control Period, the Severance Obligations will be one hundred and fifty percent (150%) of the sum of Executive's then-effective Base Salary plus his target bonus for the year of termination, in addition to the healthcare continuation and extended post-termination exercise period set forth above.

Payment of the Severance Obligations shall accrue until the Release Requirement is satisfied, with accrued amounts paid on the first regularly scheduled Company payroll date after the Release Requirement is satisfied; provided, however, if the 60-day period following the termination of Executive's employment spans two calendar years in no event will any of the Severance Obligations be paid prior to January 1 of the second calendar year (with amounts accruing until that date). Reimbursement of COBRA healthcare premiums shall be paid in accordance with the Company's standard expense reimbursement procedures subject to Executive submitting proof of payment within 60 days of paying the applicable premium.

9. Restrictive Covenants.

(a) Executive's Acknowledgment. Executive acknowledges that: (i) the Company is and will be engaged in its business during the Employment Period and thereafter; (ii) Executive is one of a limited number of persons who is and will be developing the business and will, during the Employment Period, have frequent and prolonged interactions with all customers and suppliers of the Company with respect to matters significantly affecting the business relationship between those customers and suppliers and the Company; (iii) Executive occupies a position of trust and confidence with the Company and is familiar with the Company's trade secrets and with other proprietary and confidential information concerning the Company and its business; and (iv) the agreements and covenants contained in this Section 9 are essential to protect the Company and the goodwill of the business and are a condition precedent to the Company entering into this Agreement. Executive acknowledges that he or she has carefully read this Agreement and has given careful consideration to the restraints imposed upon him by this Agreement, and is in full accord as to their necessity for the reasonable and proper protection of confidential and proprietary information and goodwill of the Company now existing or to be developed or created in the future.

(b) <u>Non-Solicitation</u>. Executive will not, during the Restricted Period, directly or indirectly (whether as an owner, partner, shareholder, member, agent, officer, director, manager, employee, independent contractor, consultant, or otherwise) with or through any individual or entity:

(i) solicit for employment or engagement, or identify for another person or entity to employ, engage or solicit for employment or engagement, any individual who is, or was at any time during the twelve (12) month period immediately preceding any such action, an employee of the Company, or otherwise seek to adversely influence or alter any such employee's relationship with the Company;

(ii) employ or engage any individual who is, or was at any time during the twelve (12) month period immediately preceding any such action, an employee of the Company; or

(iii) solicit or encourage any individual or entity that is, or was during the twenty four (24) month period immediately preceding any such action, a current or prospective customer, supplier or vendor of the Company (A) to terminate or otherwise adversely alter his, her or its relationship with the Company or (B) for any purposes which are competitive with Company's business.

(c) <u>Right to Conduct Activities</u>. During the term of this Agreement, Executive agrees to devote substantially all of his business time and effort to the performance of his services to the Company; provided however, Executive may serve as a member of the Board of Directors or other governing body of up to two (2) for-profit entities, as well as any entity in which the Company has an investment, so long as such service does not unreasonably interfere with Executive's obligations hereunder. In addition, the Company consents to Executive's continued service as a member of the Board of Directors or other governing body of any of the entities set forth on Exhibit A, or any other entity approved by the Company, so long as such service does not unreasonably interfere with Executive's obligations hereunder.

(d) <u>Scope/Severability</u>. The parties acknowledge that the Company's business is and will be conducted throughout the Restricted Territory and thus the covenants in this <u>Section 9</u> would be ineffective if the covenants were to be limited to a particular geographic area within the Restricted Territory. If any court of competent jurisdiction at any time deems the Restricted Period too lengthy, or the Restricted Territory too extensive, or any of the covenants set forth in this <u>Section 9</u> otherwise not fully enforceable, the other provisions of this <u>Section 9</u>, and this Agreement in general, will nevertheless stand and to the full extent consistent with law continue in full force and effect, and it is the intention and desire of the parties that the court treat any provisions of this Agreement which are not fully enforceable as having been modified to the extent deemed necessary by the court to render them reasonable and enforceable and that the court enforce them to such extent (for example, that Restricted Territory be deemed to comprise the largest territory permissible by law under the circumstances but not in excess of the territory provided for herein).

10. Equitable Remedies. Executive acknowledges and agrees that the agreements and covenants set forth in Section 9 of this Agreement are reasonable and necessary for the protection of the Company's business interests, that irreparable injury will result to the Company if Executive breaches any of the terms of said covenants, and that in the event of Executive's actual or threatened breach of any such covenants, the Company will have no adequate remedy at law. Executive accordingly agrees that, in the event of any actual or threatened breach by Executive of any of said covenants, the Company will be entitled to immediate injunctive and other equitable relief, without bond and without the necessity of showing actual monetary damages are an inadequate remedy. Nothing in this Section 10 will be construed as prohibiting the Company from pursuing any other remedies available to it for such breach or threatened breach, including the recovery of any damages that it is able to prove.

11. <u>Assistance</u>. From and after the date hereof (including after the termination of Executive's employment with the Company for any reason), Executive shall, upon reasonable notice, furnish the Company with such information as may be in Executive's possession or control, and cooperate with the Company, as the Company may reasonably request, in connection with any litigation, claim, or other dispute in which the Company is or may become a party, provided that following termination of Executive's employment with the Company, any such assistance shall not unreasonably interfere with Executive's personal or business affairs. The Company shall reimburse Executive for all reasonable out-of-pocket expenses incurred by Executive in fulfilling Executive's obligations under this <u>Section</u>

12. <u>Definitions</u>. The following terms used herein shall have the definitions set forth below:

"Restricted Period" means twelve (12) months following the date of termination.

"Restricted Territory" means the United States of America.

13. Effect of Prior Agreements. Unless otherwise provided herein, this Agreement contains the entire understanding between the Company and Executive relating to the subject matter hereof and, as of the Amendment Date, will supersede the Prior Agreement and any other agreement relating to the subject matter hereof between Executive and the Company or any of its subsidiaries (including, without limitation, Orchestra BioMed, Inc.). Executive acknowledges that he is not relying and has not relied on any oral or written statements, promises, representations or warranties (whether made by the Company or any equity holder, director, officer, employee, agent or other representative of the Company or otherwise) that are not expressly set forth in this Agreement and those documents expressly referred to herein. For the avoidance of doubt, the non-solicitation and other restrictive covenants contained in this Agreement shall be in addition to, and not in lieu of, and shall not in any way limit or be limited by, any restrictive covenant covering similar subject matter contained in any other agreement to which Executive is a party, including but not limited to, any asset purchase agreement, equity subscription agreement or stockholder or operating agreement with respect to any equity interest in the Company.

14. <u>Confidential Information and Intellectual Property Assignment Agreement</u>. Notwithstanding anything herein to the contrary, the terms of that certain Confidential Information and Intellectual Property Assignment Agreement, dated as of the date hereof, between Executive and the Company are hereby incorporated by reference.

15. <u>Withholding and Deductions</u>. All amounts payable to Executive (or his estate, as applicable) pursuant to this Agreement shall be subject to such withholding and deductions by the Company as required by law and the applicable benefit plans of the Company.

16. Code Section 409A. The Company and Executive agree that this Agreement and the rights granted to Executive hereunder are intended to meet the requirements of paragraphs (2), (3) and (4) of Section 409A(a)(1)(A) of the Internal Revenue Code of 1986, as amended, (the "Code"). Accordingly, the parties agree that they shall negotiate in good faith to revise any provisions of this Agreement that might otherwise fail to meet the requirements of paragraphs (2), (3) and (4) of Section 409A of Code. However, the Company does not guarantee any particular tax effect of payments under this Agreement, and in no event shall the Company have any obligation to "gross-up" or otherwise compensate Executive with respect to any tax effect of payments under this Agreement. With respect to any reimbursement of expenses of, or any provision of in-kind benefits to, Executive, as specified under this Agreement, such reimbursement of expenses or provision of in-kind benefits shall be subject to the following conditions: (1) the expenses eligible for reimbursement or the amount of in-kind benefits provided in one taxable year shall not affect the expenses eligible for reimbursement or the amount of in-kind benefits provided in any other taxable year, except for any medical reimbursement arrangement providing for the reimbursement of expenses referred to in Section 105(b) of the Code; (2) the reimbursement of an eligible expense shall be made no later than the end of the year after the year in which such expense was incurred; and (3) the right to reimbursement or in-kind benefits shall not be subject to liquidation or exchange for another benefit. "Termination of employment," or words of similar import, as used in this Agreement means, for purposes of any payments under this Agreement that are payments of deferred compensation subject to Section 409A, Executive's "separation from service" as defined in Section 409A. For purposes of Section 409A, the right to a series of installment payments under this Agreement shall be treated as a right to a series of separate payments. If Executive is deemed on the date of termination of employment to be a "specified employee" within the meaning of Section 409A of the Code, then with regard to the payment or the provision of any benefit that is considered nonqualified deferred compensation under Section 409A of the Code payable on account of a "separation from service," such payment or benefit shall not be made or provided until the date which is the earlier of (i) the expiration of the six (6) month period measured from the date of such separation from service and (ii) the date of Executive's death. Upon the expiration of the foregoing delay period, all payments and benefits so delayed shall be paid or provided or reimbursed in a lump sum. In the event of a conflict between the terms of this Section 16 and the terms of any other provision in this Agreement relating to payment timing, the terms of this Section 16 shall control.

17. <u>Code Section 280G</u>. If any payment or benefit the Executive would receive pursuant to a Change of Control from the Company or otherwise ("<u>Payment</u>") would (i) constitute a "parachute payment" within the meaning of Section 280G of the Code, and (ii) but for this sentence, be subject to the excise tax imposed by Section 4999 of the Code (the "<u>Excise Tax</u>"), then such Payment shall be equal to the Reduced Amount. The "<u>Reduced Amount</u>" shall be either (x) the largest portion of the Payment, whichever amount, after taking into account all applicable federal, state and local employment taxes, income taxes, and the Excise Tax (all computed at the highest applicable marginal rate), results in the Executive's receipt, on an after-tax basis, of the greater amount of the Payment notwithstanding that all or some portion of the Payment may be subject to the Excise Tax. Any reduction shall be made in the following manner: first a pro rata reduction of (i) cash payments subject to Section 409A of the Code as deferred compensation and (ii) cash payments or subject to Section 409A of the Code. Reduction in either cash payments or equity compensation benefits shall be made pro rata between and among benefits which are subject to Section 409A of the Code and benefits which are exempt from Section 409A of the Code.

The accounting firm engaged by the Company for general audit purposes as of the day prior to the effective date of the Change of Control shall perform the foregoing calculations. If the accounting firm so engaged by the Company is serving as accountant or auditor for the individual, entity or group effecting the Change of Control, the Company shall appoint a nationally recognized accounting firm to make the determinations required hereunder. The Company shall bear all expenses with respect to the determinations by such accounting firm to be made hereunder.

The accounting firm engaged to make the determinations hereunder shall provide its calculations, together with detailed supporting documentation, to the Company and the Executive within fifteen (15) calendar days after the date on which the Executive's right to a Payment is triggered (if requested at that time by the Company or the Executive) or such other time as requested by the Company or the Executive. If the accounting firm determines that no Excise Tax is payable with respect to a Payment, either before or after the application of the Reduced Amount, it shall furnish the Company and the Executive with an opinion reasonably acceptable to the Executive that no Excise Tax will be imposed with respect to such Payment. Any good faith determinations of the accounting firm made hereunder shall be final, binding and conclusive upon the Company and the Executive.

The Company agrees to cooperate with the Executive to minimize the amount of any payment to the Executive that is subject to Section 280G of the Code, including soliciting the consent of the Company's stockholders, as applicable, to such payment under Section 280G(b)(5) of the Code.

18. <u>Modification and Waiver</u>. This Agreement may not be modified or amended, nor may any provisions of this Agreement be waived, except by an instrument in writing signed by the parties hereto. No written waiver will be deemed to be a continuing waiver unless specifically stated therein, and each such waiver will operate only as to the specific term or condition waived and shall not constitute a waiver of such term or condition for the future or as to any act other than that specifically waived. No failure of any party hereto to exercise any right or remedy given to such party under this Agreement or otherwise available to such party hereto or to insist upon strict compliance by any other party hereto with its or his obligations hereunder, and no custom or practice of the parties in variance with the terms hereof, shall constitute a waiver of any party's right to demand exact compliance with the terms hereof.

19. <u>Indemnification</u>. The Company hereby indemnifies Executive to the maximum extent provided in the Company's By-Laws and organizational documents, as currently in effect. Executive shall be entitled to coverage under the directors' and officers' liability insurance on terms no less favorable to him in any respect than the coverage then being provided to any other current or former director or officer of the Company and which the Company shall maintain with minimum coverage of Five Million Dollars (\$5 million).

20. <u>Severability</u>. Without limiting <u>Section 9(c)</u>: (i) if, for any reason, any provision of this Agreement is held invalid, such invalidity will not affect any other provision of this Agreement, and each provision will, to the full extent consistent with law, continue in full force and effect and (ii) if any provision of this Agreement is held invalid in part, such invalidity will in no way affect the rest of such provision, and the rest of such provision, together with all other provisions of this Agreement, will, to the full extent consistent with law, continue in full force and effect.

21. <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party hereto to any other party hereto shall be in writing and shall be given by delivery in person, by facsimile transmission, by email, by overnight courier or by registered or certified mail, postage prepaid to the address set forth below or such other address as such party may give to the other parties by notice pursuant to this <u>Section 21</u>. Notice shall be deemed given on (a) the date such notice is personally delivered, (b) the date of scheduled delivery if sent by overnight courier, or (c) the date such notice is transmission is prior to 5:00 p.m. Eastern Time on a business day, or the next succeeding business day if such transmission is after 5:00 p.m. Eastern Time.

If to the Board or the Company:	Orchestra BioMed Holdings, Inc.
	150 Union Square Drive
	New Hope, PA 18938
	E-Mail: dhochman@orchestrabiomed.com
If to Executive:	David Hochman At such address or email address as is in the Company's personnel records

22. <u>Third Party Beneficiaries</u>. Nothing herein expressed or implied is intended or shall be construed to confer upon or give to any person or entity, other than the parties to this Agreement, their respective permitted successors and assigns, any rights or remedies under or by reason of this Agreement.

23. <u>Headings</u>. The headings and other captions in this Agreement are included solely for convenience of reference and will not control the meaning and interpretation of any provision of this Agreement.

24. <u>Governing Law</u>. The validity, interpretation, performance, and enforcement of this Agreement will be governed by the laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware. Executive and the Company hereby irrevocably submit to the exclusive jurisdiction of any state or federal court located in the State of Delaware, over any suit, action or proceeding, whether at law or in equity, arising out of or relating to or concerning this Agreement. Each party hereto agrees that it will not challenge the applicability of the laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction), in any suit or proceeding regarding the validity, interpretation, performance, or enforcement of any provision of this Agreement, regardless of who initiates such suit or proceeding.

25. <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE MATERS CONTEMPLATED HEREBY, PROVIDED, HOWEVER, THAT THE PARTIES HERETO AGREE THAT SUCH WAIVER SHALL NOT BE DEEMED TO CONSTITUTE A WAIVER OF ADJUDICATION BY A COURT HAVING APPROPRIATE JURISDICTION.

26. <u>Non-Assignability/Binding Effect</u>. This Agreement shall not be assignable by either party without the prior written consent of the other party. This Agreement will be binding upon and inure to the benefit of Executive, the Company and their respective successors and permitted assigns.

27. <u>Construction</u>. The parties hereto have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement.

28. <u>Survival</u>. Termination of the Employment Period in accordance with <u>Section 8</u> will not affect the provisions of <u>Section 3(a)</u> or <u>Sections 9</u> through <u>28</u> of this Agreement, each of which shall survive such termination.

29. <u>Counterparts</u>. This Agreement may be executed in separate counterparts and may be executed and delivered by facsimile or other electronic delivery, each of which is deemed to be an original and all of which, taken together, constitute one and the same agreement.

[Remainder of Page Intentionally Blank; Signature Page to Follow]



IN WITNESS WHEREOF, the Company has caused this Agreement to be executed by its duly authorized officer, and Executive has signed this Agreement, as of the date first above written.

ORCHESTRA BIOMED HOLDINGS, INC .:

By: /s/ Darren Sherman

Name:Darren ShermanTitle:President, Chief Operating Officer

EMPLOYEE:

/s/ David Hochman

David Hochman

Exhibit A

Boards of Directors

- Motus GI Holdings, Inc. (Chairman)
- Vivasure Medical Ltd. (Observer)
- Mollie Parnis Livingston Foundation (President of board)

Exhibit B

Form of Release

Orchestra BioMed Holdings, Inc. Amended and Restated Employment Agreement

THIS AMENDED AND RESTATED EMPLOYMENT AGREEMENT (this "<u>Agreement</u>") is entered into as of January 26, 2023 (the "<u>Amendment Date</u>"), by and between Orchestra BioMed Holdings, Inc., a Delaware corporation (the "<u>Company</u>") and the parent company of Orchestra BioMed, Inc., and Darren R. Sherman ("<u>Executive</u>"). This Agreement amends and restates in its entirety that certain Employment Agreement (the "<u>Prior</u> <u>Agreement</u>") entered into by and between Orchestra BioMed, Inc. and Executive, dated May 31, 2018 (the "<u>Original Effective Date</u>").

WHEREAS, Executive was an employee of Orchestra BioMed, Inc. prior to its acquisition by the Company (the "Acquisition"), which was previously known as Health Sciences Acquisitions Corp 2;

WHEREAS, upon the closing of the Acquisition, Orchestra BioMed, Inc. became a subsidiary of the Company;

WHEREAS, the Company desires to retain Executive as an employee of the Company, and Executive desires to be employed by the Company, upon the terms and subject to the conditions set forth herein.

NOW, THEREFORE, in consideration of the employment of Executive by the Company and the mutual agreements hereinafter set forth, the parties hereto agree as follows:

1. <u>Employment</u>. During the Employment Period (as defined below), the Company hereby agrees to employ Executive as President, Chief Operating Officer and Orchestra BioMed Founder, and Executive hereby accepts such employment and agrees to serve as President, Chief Operating Officer and Orchestra BioMed Founder, all in accordance with the terms and conditions of this Agreement. Executive hereby represents and warrants that neither Executive's entry into this Agreement nor Executive's performance of Executive's obligations hereunder will conflict with or result in a breach of the terms, conditions, or provisions of any other agreement or obligation of any nature to which Executive is a party or by which Executive is bound, including, without limitation, any non-competition agreement, non-solicitation agreement or confidentiality agreement or policy entered into by Executive or to which Executive is subject.

2. <u>Term of Employment</u>. The term of Executive's employment under this Agreement will commence on the Amendment Date and will continue until terminated in accordance with <u>Section 8</u> hereof (such term, the "<u>Employment Period</u>"). For the avoidance of doubt, and notwithstanding any contrary provision herein, Executive's employment is "at will" although Executive may be eligible for specified payments and benefits in the event of certain terminations of employment as set forth in Sections 4(c) and 8.

3. <u>Position and Responsibilities</u>. During the Employment Period: (a) Executive shall serve in the position of President, Chief Operating Officer and Orchestra BioMed Founder; (b) Executive shall have the duties, responsibilities, functions and authority, including administrative, financial, executive and managerial, as shall be designated by the Company's board of directors (the "<u>Board</u>") or Chief Executive Officer from time to time; (c) Executive will use his best efforts to promote the interests, prospects, condition (financial and otherwise) and welfare of the Company and shall perform his duties and responsibilities to the best of his ability in a diligent, trustworthy, businesslike and efficient manner; and (d) Executive shall comply at all times with all policies and codes of conduct of the Company, as such policies and codes may change from time to time.

(a) <u>Board of Directors</u>. So long as he remains the Company's President and Chief Operating Officer, the Company shall nominate Executive to serve as a Director. Unless otherwise agreed upon by Executive and the Company, Executive's service as a Director shall continue until the first to occur of (i) the date that Executive is no longer employed by the Company and its affiliates, (ii) the death of the Director, (iii) the resignation by the Executive from the Board of Directors, and (iv) the expiration of the then current term as a Director if Executive is not re-elected by the Company's stockholders to serve as a Director. Executive agrees to serve as a Director during the term of this Agreement, so long as he is elected to serve as a Director by the Company's stockholders.

4. Compensation.

(a) <u>Salary</u>. During the Employment Period, the Company shall pay to Executive a base salary at the rate of \$495,000 per annum (the "<u>Base Salary</u>"). On an annual basis the Board or the compensation committee of the Board (the "Compensation Committee") shall review Executive's Base Salary and may increase such amount as it may deem advisable. The Base Salary shall be paid in accordance with the Company's standard payroll practices in effect from time to time.

(b) <u>Bonus</u>. Commencing on the Amendment Date, in addition to the Base Salary, Executive shall be eligible to receive a discretionary annual bonus during each fiscal year of the Employment Period, with a target bonus amount of eighty percent (80%) of the Base Salary. The performance metrics and goals required to receive such bonus amount, shall be determined by the Board in consultation with the Company's Chief Executive Officer within sixty (60) days following the commencement of each fiscal year or, if later, at the next regularly scheduled meeting of the Board that occurs at least five business days after the Chief Executive Officer has submitted recommended performance metrics to the Board. The actual bonus, if any, payable under this <u>Section 4(b)</u> shall be paid within thirty (30) days following receipt of the Company's audited financial statements for the applicable fiscal year, unless such payment is delayed due to an unforeseeable administrative impracticability, in which case, such bonus will be paid as soon as administratively practicable thereafter, but in no event later than the end of the fiscal year immediately following the applicable fiscal year; provided that Executive is an employee with the Company at such time of payment and Executive has not breached any provision of this Agreement.

(c) <u>Equity Interest</u>. Company shall award Executive stock options to acquire common stock of the Company (the "<u>Common Shares</u>") or restricted common stock units ("RSUs") (as agreed upon by Executive and the Board) as soon as reasonably practicable following the beginning of the first open trading window after the date of this Agreement under the Company's Insider Trading Policy, so that Executive's long-term incentive equity holdings represent, in the aggregate, the Equity Award Percentage (as defined below; such grant, the "<u>Equity Award Grant</u>"). Except as otherwise provided herein, the terms of the Equity Award Grant shall reflect the Company's standard terms and conditions for executive equity award grants.

(i) <u>Defined Terms</u>. For the purposes of this Agreement, the following defined terms shall be construed to have the meanings set forth or referenced below.

(1) "<u>Equity Award Percentage</u>" means four and three-tenths percent (4.3%) of the fully diluted Common Shares of the combined company as of immediately following the closing of the Acquisition; provided, however, that notwithstanding any contrary provision herein, the parties acknowledge that the 4.3% assumes an award solely in the form of options. To the extent the Equity Award is in restricted stock units, the number of shares covered by the Equity Award will be reduced reflecting a ratio of 2 shares covered by restricted stock units is equivalent to 3 shares covered by options (i.e., rounding down, a restricted stock unit award covering 666 shares of Company common stock is deemed equivalent to an option award covering 1,000 shares of Company common stock).

(2) "Equity Vesting Schedule" means that twenty-five percent (25%) of any options awarded to the Executive shall be fully vested on the first (1st) anniversary of the Amendment Date and the remaining seventy-five percent (75%) of any options awarded to the Executive shall vest on a quarterly basis thereafter over the following three years; and unless otherwise agreed, any RSUs awarded to the Executive shall vest on a schedule to be agreed upon between the Executive and the Board (provided such vesting schedule shall be over a minimum of three (3) years unless otherwise agreed by the Company and Executive and other than as provided herein). All vesting shall be subject to continued employment or service as a member of the Board through the applicable vesting date; provided, however, in the event of the termination of Executive's employment and the Board upon termination of employment), the parties agree to negotiate in good faith a reduction in the number of shares as to which the Equity Award Grant has not yet vested so that future vesting of the award more closely reflects grant sizes for non-employee directors.

additional terms:

(ii) Other Terms. The Equity Award Grant shall vest according to the Equity Vesting Schedule and contain the following

(1) Notwithstanding the terms of any document to the contrary, in the event that (i) a "Change of Control" (as defined in the Company's 2023 Equity Incentive Plan provided, however, that the Acquisition is not a Change of Control) occurs, and (ii) the surviving corporation or the acquiring corporation fails to either (A) continue or assume any or all of Executive's Company options or other stock awards or (B) substitute similar stock awards for Executive's Company options or other stock awards (it being understood that similar stock awards include, but are not limited to, awards to acquire the same consideration paid to the stockholders or the Company, as the case may be, pursuant to the Change of Control), then 100% of Executive's unvested Company options or other stock awards shall immediately vest as of the date immediately preceding the Change of Control.

(2) If the Executive is terminated without Cause (as defined below), or voluntarily terminates the Employment Period with Good Reason during the period commencing three (3) months prior to a Change of Control of and ending twelve (12) months following a Change of Control (the "Change of Control Period"), and subject to the Executive satisfying the Release Requirement (defined below), any unvested equity awards then held by the Executive shall become fully vested.

(3) If the Executive is terminated without Cause (as defined below), or voluntarily terminates the Employment Period with Good Reason outside of the Change of Control Period, and subject to the Executive satisfying the Release Requirement, the vesting of any equity held by the Executive as of the closing of the Acquisition shall be determined as if the Executive had remained employed for an additional twelve (12) months beyond his date of termination.

5. <u>Benefit Plans</u>. During the Employment Period, Executive shall be eligible to participate in and receive benefits under the employee benefit plans, equity incentive plans, practices, policies, perquisites and programs established and maintained by the Company from time to time, in the Company's sole discretion, for which executives of the Company are generally eligible, in each case, subject to the eligibility and participation requirements thereof.

6. <u>Business Expenses</u>. The Company, in accordance with policies and practices established by the Company from time to time, will pay or reimburse Executive for all expenses reasonably incurred by Executive during the Employment Period in connection with the performance of Executive's duties under this Agreement; <u>provided</u>, that Executive shall provide to the Company reasonable documentation or evidence of expenses for which Executive seeks reimbursement in accordance with the policies and procedures established by the Company from time to time. Executive's air travel shall be pursuant to Company policy for executives.

7. <u>Paid Time Off</u>. Executive shall be entitled, during the Employment Period, to four (4) weeks of paid vacation and sick time in each calendar year or pursuant to Company policy for executives, whichever is greater. Accrued vacation and sick time not taken in any calendar year may not be carried forward or be usable in any subsequent calendar year, nor shall Executive be entitled to compensation for unused vacation or sick days during the Employment Period or upon termination of employment. Paid holidays may be taken in accordance with the holiday policy and schedule of the Company as from time to time in effect.

8. Termination.

(a) <u>Termination For Cause</u>. The Company may terminate the Employment Period immediately upon the occurrence of any of the following events (each of which shall constitute "<u>Cause</u>"):

(i) Executive's breach of any of Executive's obligations under Section 9;

(ii) Executive's breach of any of Executive's obligations under this Agreement other than <u>Section 9</u>, which, to the extent curable, has not been cured within thirty (30) days after Executive has been provided written notice of such breach;

(iii) Executive being convicted of, or pleading guilty or nolo contendere to, or being indicted for, any felony or any misdemeanor involving theft, fraud, dishonesty or moral turpitude; or

(iv) Fraud or embezzlement against the Company.

If the Employment Period is terminated pursuant to this <u>Section 8(a)</u>, the Company shall have no further obligation to Executive except for (i) payment of Base Salary accrued but unpaid through the date of termination and (ii) subject to the provisions of <u>Section 6</u>, reimbursement of any expense properly incurred by Executive through the date of termination (collectively, the "<u>Accrued Obligations</u>"). The Accrued Obligations shall be paid in accordance with applicable state law and in no event later than thirty (30) days after the date of termination.

(b) <u>Termination upon Executive's Death</u>. The Employment Period shall terminate immediately upon the death of Executive. If the Employment Period is terminated pursuant to this <u>Section 8(b)</u>, the Company shall have the following obligations to Executive (or his estate) (i) the payment of the Accrued Obligations, which shall be paid in accordance with <u>Section 8(a)</u>, and (ii) the payment of Severance Obligations, which shall be paid in accordance with <u>Section 8(f)</u>.

(c) <u>Termination by the Company without Cause</u>. The Company may terminate the Employment Period at any time without Cause upon written notice to Executive. If the Employment Period is terminated pursuant to this <u>Section 8(c)</u>, the Company shall have the following obligations to Executive (or his estate): (i) Payment of all Accrued Obligations, which shall be paid in accordance with <u>Section 8(a)</u>; and (ii) Payment of the Severance Obligations as described in <u>Section 8(f)</u>.

(d) <u>Voluntary Resignation by Executive - Without Cause</u>. Executive may terminate the Employment Period for any reason upon not less than ten (10) days' prior written notice. If the Employment Period is terminated pursuant to this <u>Section 8(d)</u>, the Company shall have no further obligation to Executive except for Accrued Obligations, which shall be paid in accordance with <u>Section 8(a)</u>.

(e) <u>Voluntary Resignation by Executive - With Good Reason</u>. Executive may terminate the Employment Period, with Good Reason, upon not less than ten (10) days' prior written notice. If Executive terminates the Employment Period with Good Reason, the Company shall have the following obligations to Executive (or his estate):

(i) Payment of all Accrued Obligations, which shall be paid in accordance with Section 8(a); and

(ii) Payment of the Severance Obligations as described in Section 8(f).

(iii) For purposes of this <u>Section 8(e)</u>, "<u>Good Reason</u>" shall mean each of the following, without Executive's consent: (A) any change in the Executive's position, title or reporting relationship with the Company that diminishes in any material respect the Executive's authority, duties or responsibilities; provided, however, that a change in the Executive's authority, duties or responsibilities solely due to the Company becoming a division, subsidiary or other similar part of a larger organization, shall not by itself constitute Good Reason; (B) any material reduction in the Executive's base compensation; (C) the relocation of the Executive's principal office, or principal place of employment, to a location more than fifty (50) miles from the location of the Executive's principal office or principal place of business; or (D) a material breach of any provision hereof by the Company or any successor or assign. Notwithstanding the foregoing, "Good Reason" shall not be deemed to have occurred unless: (1) the Executive provides the Company with written notice that the Executive intends to terminate the Executive's employment hereunder for one of the grounds set forth in subsections (A), (B), (C) or (D) within sixty (60) days of such reason(s) occurring, (2) if such ground is capable of being cured, the Company has failed to cure such ground within a period of sixty (60) days from the date of such written notice, and (3) the Executive terminates the Executive's employment within sixty (60) days after the cure period has lapsed. For purposes of clarification, the above-listed conditions shall apply separately to each occurrence of Good Reason and failure to adhere to such conditions in the event of Good Reason shall not disqualify the Executive from asserting Good Reason for any subsequent occurrence of Good Reason.

(f) Severance Payments. In the event Executive's employment is terminated (i) by the Company without Cause or (ii) by Executive for Good Reason, the Company shall continue to pay Executive as severance, the aggregate sum of one hundred percent (100%) of Executive's then-effective Base Salary plus his target bonus for the year of termination proportionally for a period of twelve (12) months beginning on the date such termination becomes effective (such payments, the Company's "Severance Obligations"), and if the Executive validly elects to continue his healthcare coverage under COBRA, reimburse the Executive for the premiums he pays to continue healthcare coverage for the Executive and his eligible dependents for twelve (12) months or, if earlier, until the Executive becomes eligible for healthcare coverage from a subsequent employer; provided, as a condition to all amounts specified in this clause (f), that Executive executes a full and final release substantially in the form attached as Exhibit B hereto (with such changes as the Company may reasonably request to reflect changes in applicable law) and such release becomes irrevocably effective within 60 days following the termination of Executive's employment (such requirement for an irrevocably effective release, the "Release Requirement"). Any and all payments of Severance Obligations shall be less applicable withholding and deductions. Further, in the event Executive's employment is termination exercise period of Executive's outstanding options shall be extended to twelve months following the termination of Executive's employment; provided, however, options shall expire earlier upon expiration of the maximum term or, subject to Section 4(c)(ii)(1), pursuant to a Change of Control if all Company options are terminated in connection with the Change of Control.

In the event of such termination without Cause or resignation for Good Reason during the Change of Control Period, the Severance Obligations will be one hundred and fifty percent (150%) of the sum of Executive's then-effective Base Salary plus his target bonus for the year of termination, in addition to the healthcare continuation and extended post-termination exercise period set forth above.

Payment of the Severance Obligations shall accrue until the Release Requirement is satisfied, with accrued amounts paid on the first regularly scheduled Company payroll date after the Release Requirement is satisfied; provided, however, if the 60-day period following the termination of Executive's employment spans two calendar years in no event will any of the Severance Obligations be paid prior to January 1 of the second calendar year (with amounts accruing until that date). Reimbursement of COBRA healthcare premiums shall be paid in accordance with the Company's standard expense reimbursement procedures subject to Executive submitting proof of payment within 60 days of paying the applicable premium.

9. Restrictive Covenants.

(a) Executive's Acknowledgment. Executive acknowledges that: (i) the Company is and will be engaged in its business during the Employment Period and thereafter; (ii) Executive is one of a limited number of persons who is and will be developing the business and will, during the Employment Period, have frequent and prolonged interactions with all customers and suppliers of the Company with respect to matters significantly affecting the business relationship between those customers and suppliers and the Company; (iii) Executive occupies a position of trust and confidence with the Company and is familiar with the Company's trade secrets and with other proprietary and confidential information concerning the Company and its business; and (iv) the agreements and covenants contained in this Section 9 are essential to protect the Company and the goodwill of the business and are a condition precedent to the Company entering into this Agreement. Executive acknowledges that he or she has carefully read this Agreement and has given careful consideration to the restraints imposed upon him by this Agreement, and is in full accord as to their necessity for the reasonable and proper protection of confidential and proprietary information and goodwill of the Company now existing or to be developed or created in the future.

(b) <u>Non-Solicitation</u>. Executive will not, during the Restricted Period, directly or indirectly (whether as an owner, partner, shareholder, member, agent, officer, director, manager, employee, independent contractor, consultant, or otherwise) with or through any individual or entity:

(i) solicit for employment or engagement, or identify for another person or entity to employ, engage or solicit for employment or engagement, any individual who is, or was at any time during the twelve (12) month period immediately preceding any such action, an employee of the Company, or otherwise seek to adversely influence or alter any such employee's relationship with the Company;

(ii) employ or engage any individual who is, or was at any time during the twelve (12) month period immediately preceding any such action, an employee of the Company; or

(iii) solicit or encourage any individual or entity that is, or was during the twenty four (24) month period immediately preceding any such action, a current or prospective customer, supplier or vendor of the Company (A) to terminate or otherwise adversely alter his, her or its relationship with the Company or (B) for any purposes which are competitive with Company's business.

(c) <u>Right to Conduct Activities</u>. During the term of this Agreement, Executive agrees to devote substantially all of his business time and effort to the performance of his services to the Company; provided however, Executive may serve as a member of the Board of Directors or other governing body of up to two (2) for-profit entities, as well as any entity in which the Company has an investment, so long as such service does not unreasonably interfere with Executive's obligations hereunder. In addition, the Company consents to Executive's continued service as a member of the Board of Directors or other governing body of any of the entities set forth on <u>Exhibit A</u>, or any other entity approved by the Company, so long as such service does not unreasonably interfere with Executive's obligations hereunder.

(d) <u>Scope/Severability</u>. The parties acknowledge that the Company's business is and will be conducted throughout the Restricted Territory and thus the covenants in this <u>Section 9</u> would be ineffective if the covenants were to be limited to a particular geographic area within the Restricted Territory. If any court of competent jurisdiction at any time deems the Restricted Period too lengthy, or the Restricted Territory too extensive, or any of the covenants set forth in this <u>Section 9</u> otherwise not fully enforceable, the other provisions of this <u>Section 9</u>, and this Agreement in general, will nevertheless stand and to the full extent consistent with law continue in full force and effect, and it is the intention and desire of the parties that the court treat any provisions of this Agreement which are not fully enforceable as having been modified to the extent deemed necessary by the court to render them reasonable and enforceable and that the court enforce them to such extent (for example, that Restricted Territory be deemed to comprise the largest territory permissible by law under the circumstances but not in excess of the territory provided for herein).

10. Equitable Remedies. Executive acknowledges and agrees that the agreements and covenants set forth in Section 9 of this Agreement are reasonable and necessary for the protection of the Company's business interests, that irreparable injury will result to the Company if Executive breaches any of the terms of said covenants, and that in the event of Executive's actual or threatened breach of any such covenants, the Company will have no adequate remedy at law. Executive accordingly agrees that, in the event of any actual or threatened breach by Executive of any of said covenants, the Company will be entitled to immediate injunctive and other equitable relief, without bond and without the necessity of showing actual monetary damages are an inadequate remedy. Nothing in this Section 10 will be construed as prohibiting the Company from pursuing any other remedies available to it for such breach or threatened breach, including the recovery of any damages that it is able to prove.

11. <u>Assistance</u>. From and after the date hereof (including after the termination of Executive's employment with the Company for any reason), Executive shall, upon reasonable notice, furnish the Company with such information as may be in Executive's possession or control, and cooperate with the Company, as the Company may reasonably request, in connection with any litigation, claim, or other dispute in which the Company is or may become a party, provided that following termination of Executive's employment with the Company, any such assistance shall not unreasonably interfere with Executive's personal or business affairs. The Company shall reimburse Executive for all reasonable out-of-pocket expenses incurred by Executive in fulfilling Executive's obligations under this <u>Section</u>

12. Definitions. The following terms used herein shall have the definitions set forth below:

"Restricted Period" means twelve (12) months following the date of termination.

"Restricted Territory" means the United States of America.

13. Effect of Prior Agreements. Unless otherwise provided herein, this Agreement contains the entire understanding between the Company and Executive relating to the subject matter hereof and, as of the Amendment Date, will supersede the Prior Agreement and any other agreement relating to the subject matter hereof between Executive and the Company or any of its subsidiaries (including, without limitation, Orchestra BioMed, Inc.). Executive acknowledges that he is not relying and has not relied on any oral or written statements, promises, representations or warranties (whether made by the Company or any equity holder, director, officer, employee, agent or other representative of the Company or otherwise) that are not expressly set forth in this Agreement and those documents expressly referred to herein. For the avoidance of doubt, the non-solicitation and other restrictive covenants contained in this Agreement shall be in addition to, and not in lieu of, and shall not in any way limit or be limited by, any restrictive covenant covering similar subject matter contained in any other agreement to which Executive is a party, including but not limited to, any asset purchase agreement, equity subscription agreement or stockholder or operating agreement with respect to any equity interest in the Company.

14. <u>Confidential Information and Intellectual Property Assignment Agreement</u>. Notwithstanding anything herein to the contrary, the terms of that certain Confidential Information and Intellectual Property Assignment Agreement, dated as of the date hereof, between Executive and the Company are hereby incorporated by reference.

15. <u>Withholding and Deductions</u>. All amounts payable to Executive (or his estate, as applicable) pursuant to this Agreement shall be subject to such withholding and deductions by the Company as required by law and the applicable benefit plans of the Company.

16. Code Section 409A. The Company and Executive agree that this Agreement and the rights granted to Executive hereunder are intended to meet the requirements of paragraphs (2), (3) and (4) of Section 409A(a)(1)(A) of the Internal Revenue Code of 1986, as amended, (the "Code"). Accordingly, the parties agree that they shall negotiate in good faith to revise any provisions of this Agreement that might otherwise fail to meet the requirements of paragraphs (2), (3) and (4) of Section 409A of Code. However, the Company does not guarantee any particular tax effect of payments under this Agreement, and in no event shall the Company have any obligation to "gross-up" or otherwise compensate Executive with respect to any tax effect of payments under this Agreement. With respect to any reimbursement of expenses of, or any provision of in-kind benefits to, Executive, as specified under this Agreement, such reimbursement of expenses or provision of in-kind benefits shall be subject to the following conditions: (1) the expenses eligible for reimbursement or the amount of in-kind benefits provided in one taxable year shall not affect the expenses eligible for reimbursement or the amount of in-kind benefits provided in any other taxable year, except for any medical reimbursement arrangement providing for the reimbursement of expenses referred to in Section 105(b) of the Code; (2) the reimbursement of an eligible expense shall be made no later than the end of the year after the year in which such expense was incurred; and (3) the right to reimbursement or in-kind benefits shall not be subject to liquidation or exchange for another benefit. "Termination of employment," or words of similar import, as used in this Agreement means, for purposes of any payments under this Agreement that are payments of deferred compensation subject to Section 409A, Executive's "separation from service" as defined in Section 409A. For purposes of Section 409A, the right to a series of installment payments under this Agreement shall be treated as a right to a series of separate payments. If Executive is deemed on the date of termination of employment to be a "specified employee" within the meaning of Section 409A of the Code, then with regard to the payment or the provision of any benefit that is considered nonqualified deferred compensation under Section 409A of the Code payable on account of a "separation from service," such payment or benefit shall not be made or provided until the date which is the earlier of (i) the expiration of the six (6) month period measured from the date of such separation from service and (ii) the date of Executive's death. Upon the expiration of the foregoing delay period, all payments and benefits so delayed shall be paid or provided or reimbursed in a lump sum. In the event of a conflict between the terms of this Section 16 and the terms of any other provision in this Agreement relating to payment timing, the terms of this Section 16 shall control.

17. <u>Code Section 280G</u>. If any payment or benefit the Executive would receive pursuant to a Change of Control from the Company or otherwise ("<u>Payment</u>") would (i) constitute a "parachute payment" within the meaning of Section 280G of the Code, and (ii) but for this sentence, be subject to the excise tax imposed by Section 4999 of the Code (the "<u>Excise Tax</u>"), then such Payment shall be equal to the Reduced Amount. The "<u>Reduced Amount</u>" shall be either (x) the largest portion of the Payment, whichever amount, after taking into account all applicable federal, state and local employment taxes, income taxes, and the Excise Tax (all computed at the highest applicable marginal rate), results in the Executive's receipt, on an after-tax basis, of the greater amount of the Payment notwithstanding that all or some portion of the Payment may be subject to the Excise Tax. Any reduction shall be made in the following manner: first a pro rata reduction of (i) cash payments subject to Section 409A of the Code as deferred compensation and (ii) cash payments or subject to Section 409A of the Code. Reduction in either cash payments or equity compensation benefits shall be made pro rata between and among benefits which are subject to Section 409A of the Code and benefits which are exempt from Section 409A of the Code.

The accounting firm engaged by the Company for general audit purposes as of the day prior to the effective date of the Change of Control shall perform the foregoing calculations. If the accounting firm so engaged by the Company is serving as accountant or auditor for the individual, entity or group effecting the Change of Control, the Company shall appoint a nationally recognized accounting firm to make the determinations required hereunder. The Company shall bear all expenses with respect to the determinations by such accounting firm to be made hereunder.

The accounting firm engaged to make the determinations hereunder shall provide its calculations, together with detailed supporting documentation, to the Company and the Executive within fifteen (15) calendar days after the date on which the Executive's right to a Payment is triggered (if requested at that time by the Company or the Executive) or such other time as requested by the Company or the Executive. If the accounting firm determines that no Excise Tax is payable with respect to a Payment, either before or after the application of the Reduced Amount, it shall furnish the Company and the Executive with an opinion reasonably acceptable to the Executive that no Excise Tax will be imposed with respect to such Payment. Any good faith determinations of the accounting firm made hereunder shall be final, binding and conclusive upon the Company and the Executive.

The Company agrees to cooperate with the Executive to minimize the amount of any payment to the Executive that is subject to Section 280G of the Code, including soliciting the consent of the Company's stockholders, as applicable, to such payment under Section 280G(b)(5) of the Code.

18. <u>Modification and Waiver</u>. This Agreement may not be modified or amended, nor may any provisions of this Agreement be waived, except by an instrument in writing signed by the parties hereto. No written waiver will be deemed to be a continuing waiver unless specifically stated therein, and each such waiver will operate only as to the specific term or condition waived and shall not constitute a waiver of such term or condition for the future or as to any act other than that specifically waived. No failure of any party hereto to exercise any right or remedy given to such party under this Agreement or otherwise available to such party hereto or to insist upon strict compliance by any other party hereto with its or his obligations hereunder, and no custom or practice of the parties in variance with the terms hereof, shall constitute a waiver of any party's right to demand exact compliance with the terms hereof.

19. <u>Indemnification</u>. The Company hereby indemnifies Executive to the maximum extent provided in the Company's By-Laws and organizational documents, as currently in effect. Executive shall be entitled to coverage under the directors' and officers' liability insurance on terms no less favorable to him in any respect than the coverage then being provided to any other current or former director or officer of the Company and which the Company shall maintain with minimum coverage of Five Million Dollars (\$5 million).

20. <u>Severability</u>. Without limiting <u>Section 9(c)</u>: (i) if, for any reason, any provision of this Agreement is held invalid, such invalidity will not affect any other provision of this Agreement, and each provision will, to the full extent consistent with law, continue in full force and effect and (ii) if any provision of this Agreement is held invalid in part, such invalidity will in no way affect the rest of such provision, and the rest of such provision, together with all other provisions of this Agreement, will, to the full extent consistent with law, continue in full force and effect.

21. <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party hereto to any other party hereto shall be in writing and shall be given by delivery in person, by facsimile transmission, by email, by overnight courier or by registered or certified mail, postage prepaid to the address set forth below or such other address as such party may give to the other parties by notice pursuant to this <u>Section 21</u>. Notice shall be deemed given on (a) the date such notice is personally delivered, (b) the date of scheduled delivery if sent by overnight courier, or (c) the date such notice is transmission is prior to 5:00 p.m. Eastern Time on a business day, or the next succeeding business day if such transmission is after 5:00 p.m. Eastern Time.

If to the Board or the Company:	Orchestra BioMed Holdings, Inc. 150 Union Square Drive
	New Hope, PA 18938 E-Mail: dhochman@orchestrabiomed.com
If to Executive:	Darren R. Sherman At such address or email address as is in the Company's personnel records

22. <u>Third Party Beneficiaries</u>. Nothing herein expressed or implied is intended or shall be construed to confer upon or give to any person or entity, other than the parties to this Agreement, their respective permitted successors and assigns, any rights or remedies under or by reason of this Agreement.

23. <u>Headings</u>. The headings and other captions in this Agreement are included solely for convenience of reference and will not control the meaning and interpretation of any provision of this Agreement.

24. <u>Governing Law</u>. The validity, interpretation, performance, and enforcement of this Agreement will be governed by the laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware. Executive and the Company hereby irrevocably submit to the exclusive jurisdiction of any state or federal court located in the State of Delaware, over any suit, action or proceeding, whether at law or in equity, arising out of or relating to or concerning this Agreement. Each party hereto agrees that it will not challenge the applicability of the laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction), in any suit or proceeding regarding the validity, interpretation, performance, or enforcement of any provision of this Agreement, regardless of who initiates such suit or proceeding.

25. <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE MATERS CONTEMPLATED HEREBY, PROVIDED, HOWEVER, THAT THE PARTIES HERETO AGREE THAT SUCH WAIVER SHALL NOT BE DEEMED TO CONSTITUTE A WAIVER OF ADJUDICATION BY A COURT HAVING APPROPRIATE JURISDICTION.

26. <u>Non-Assignability/Binding Effect</u>. This Agreement shall not be assignable by either party without the prior written consent of the other party. This Agreement will be binding upon and inure to the benefit of Executive, the Company and their respective successors and permitted assigns.

27. <u>Construction</u>. The parties hereto have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement.

28. <u>Survival</u>. Termination of the Employment Period in accordance with <u>Section 8</u> will not affect the provisions of <u>Section 3(a)</u> or <u>Sections 9</u> through <u>28</u> of this Agreement, each of which shall survive such termination.

29. <u>Counterparts</u>. This Agreement may be executed in separate counterparts and may be executed and delivered by facsimile or other electronic delivery, each of which is deemed to be an original and all of which, taken together, constitute one and the same agreement.

[Remainder of Page Intentionally Blank; Signature Page to Follow]

IN WITNESS WHEREOF, the Company has caused this Agreement to be executed by its duly authorized officer, and Executive has signed this Agreement, as of the date first above written.

ORCHESTRA BIOMED HOLDINGS, INC .:

By: /s/ David Hochman

Name: David Hochman Title: Chief Executive Officer

EMPLOYEE:

/s/ Darren Sherman

Darren R. Sherman

Exhibit A

Boards of Directors

· Motus GI Holdings, Inc.

Exhibit B

Form of Release



January 31, 2023

Office of the Chief Accountant Securities and Exchange Commission 100 F Street, NE Washington, DC 20549

Ladies and Gentlemen:

We have read Orchestra Biomed Holdings, Inc.'s (formerly known as Health Sciences Acquisitions Corporation 2) statements included under Item 4.01 of its Form 8-K dated January 25, 2023. We agree with the statements concerning our Firm under Item 4.01, in which we were informed of our dismissal on January 26, 2023. We are not in a position to agree or disagree with other statements contained therein.

Very truly yours,

/s/ WithumSmith + Brown, PC WithumSmith+Brown, PC

New York, New York

WithumSmith+Brown, PC 1411 Broadway, 9th Floor, New York, New York 10018-3496 T [212] 751 9100 F [212] 750 3262 withum.com

AN INDEPENDENT MEMBER OF HLB - THE GLOBAL ADVISORY AND ACCOUNTING NETWORK

Subsidiaries of Orchestra BioMed Holdings, Inc.

Name	State or Jurisdiction of Incorporation or Organization
Orchestra BioMed, Inc.	Delaware
BackBeat Medical, LLC	Delaware
Caliber Therapeutics, LLC	Delaware
FreeHold Surgical, LLC	Delaware
Accelerated Technologies Inc.	Delaware

Orchestra BioMed Debuts as Nasdaq-Traded Company with Lead Programs Targeting Hypertension and Artery Disease and Novel Partnership-Enabled Business Model

Business combination with Health Sciences Acquisitions Corporation 2, a special purpose acquisition company sponsored by an affiliate of RTW Investments, LP, completed on January 26, 2023

The combination was announced simultaneously with Orchestra's strategic collaboration with Medtronic to develop BackBeat Cardiac Neuromodulation Therapy™ as potential integrated hypertension treatment for cardiac pacemaker patients

Gross proceeds from transaction and previously completed financing led by RTW and Medtronic total approximately \$180 million and support cash runway into 2026

Orchestra BioMed Holdings, Inc. to commence trading on January 27 on Nasdaq Global Market under ticker symbol "OBIO"

NEW HOPE, Pa., Jan. 27, 2023 (GLOBE NEWSWIRE) – Orchestra BioMed, Inc. ("Orchestra BioMed"), a biomedical company accelerating high-impact technologies to patients through risk-reward sharing partnerships, today announced the closing of its previously announced business combination with Health Sciences Acquisitions Corporation 2 ("HSAC2") (Nasdaq: HSAQ). Common stock of the combined company (Nasdaq: OBIO or the "Company"), which is called "Orchestra BioMed Holdings, Inc.", will commence trading on January 27, 2023, on the Nasdaq Global Market under the ticker symbol "OBIO". Orchestra BioMed's management team, led by Chairman, Chief Executive Officer and Co-founder David Hochman, will lead the combined company.

The business combination, which had previously been approved by Orchestra BioMed stockholders, was approved by HSAC2 shareholders at a special general meeting held on January 24, 2023. Gross proceeds from the business combination were \$70 million including \$20 million in market purchases of HSAC2 stock by an affiliate of Medtronic plc ("Medtronic") and funds managed by RTW Investments, LP ("RTW"), a leading life sciences investment firm, as well as an additional investment by RTW concurrent with the closing. Together with proceeds from Orchestra BioMed's previously completed \$110 million Series D financing, total cash following the business combination is expected to provide the Company with a cash runway into 2026. The Series D financing included a \$40 million investment from Medtronic, a \$20 million investment from RTW, and investments from Perceptive Advisors, Terumo, SternAegis Ventures and others. The Company's cash resources are intended to support the execution of pivotal studies for BackBeat Cardiac Neuromodulation TherapyTM (CNTTM) for the treatment of hypertension in cardiac pacemaker patients and Virtue® Sirolimus AngioInfusionTM Balloon (SAB) in patients with coronary in-stent restenosis.

"Orchestra BioMed is pioneering an innovative business model designed to leverage partnerships with global medical device leaders like Medtronic and Terumo to accelerate the development and commercialization of high-impact device therapies like BackBeat CNT and Virtue SAB. With a strong balance sheet, a top-flight shareholder base, and world-class strategic collaborators, this transaction puts us in an excellent position to address the unmet needs of patients and physicians while seeking to generate differentiated growth for our stockholders," said Mr. Hochman. "We have several important milestones ahead of us, including the expected initiation of pivotal trials this year for both BackBeat CNT and Virtue SAB. These innovative technologies are supported by strong proof-of-concept clinical data, fueling our excitement for Orchestra BioMed as it enters the public markets and the next phase of its corporate evolution." Roderick Wong, M.D., Managing Partner and Chief Investment Officer of RTW, added, "BackBeat CNT and Virtue SAB are supported by promising data and novel partnerships with global leaders across medtech that we believe will help accelerate clinical development and position them for commercial success. We are proud to be supporting this significant milestone and look forward to our continued work with Orchestra BioMed's leadership team, which has an exceptional track record of bringing novel medical technologies to market."

Advisors

Jefferies LLC acted as lead financial advisor and a capital markets advisor to Orchestra BioMed. Piper Sandler & Co. acted as a capital markets advisor to Orchestra BioMed.

Chardan and Barclays acted as financial and capital markets advisors to HSAC2. Chardan acted as M&A advisor to HSAC2.

Paul Hastings LLP is served as legal counsel for Orchestra BioMed. Loeb & Loeb LLP served as legal counsel for HSAC2. Latham & Watkins LLP served as legal counsel for Jefferies LLC, Piper Sandler & Co., Chardan and Barclays.

About Orchestra BioMed

Orchestra BioMed is a biomedical innovation company accelerating high-impact technologies to patients through risk-reward sharing partnerships with leading medical device companies. Orchestra BioMed's partnership-enabled business model focuses on forging strategic collaborations with leading medical device companies to drive successful global commercialization of products it develops. Orchestra BioMed's flagship product candidates include BackBeat Cardiac Neuromodulation TherapyTM (CNTTM) for the treatment of hypertension, a significant risk factor for death worldwide, and Virtue[®] Sirolimus AngioInfusionTM Balloon (SAB) for the treatment of atherosclerotic artery disease, the leading cause of mortality worldwide. Orchestra BioMed has a strategic collaboration with Medtronic, one of the largest medical device companies in the world, for development and commercialization of BackBeat CNT for the treatment of hypertension in pacemaker-indicated patients, and a strategic partnership with Terumo Corporation, a global leader in medical technology, for development and commercialization of Virtue SAB for the treatment of artery disease. Orchestra BioMed has additional product candidates and plans to potentially expand its product pipeline through acquisitions, strategic collaborations, licensing, and organic development. For further information about Orchestra BioMed, please visit www.orchestrabiomed.com.

About BackBeat CNT and the Strategic Collaboration with Medtronic

BackBeat CNT is an investigational bioelectronic treatment designed to lower blood pressure. It is compatible with standard pacemakers as a firmware upgrade and has been evaluated in pilot studies in patients with hypertension who also are indicated for pacemakers. It is estimated that more than 70% of the approximately 1.1 million people globally who are implanted with cardiac pacemakers each year are also diagnosed with hypertension¹.

The recent peer-reviewed, double-blind, randomized pilot study, MODERATO II, showed that patients treated with the BackBeat CNT experienced net reductions of 8.1 mmHg in 24-hour ambulatory systolic blood pressure (aSBP) and 12.3 mmHg in office systolic blood pressure (oSBP) when compared to control patients at six months. Orchestra BioMed plans to conduct a global pivotal trial to further evaluate the safety and efficacy of the BackBeat CNT in lowering blood pressure in a similar target population of patients who have been indicated for, and recently received, a cardiac pacemaker implant. The strategic collaboration with Medtronic will provide Orchestra BioMed with development, clinical, and regulatory support for this planned multi-national study. Upon regulatory approval, Medtronic will have the global rights to commercialize BackBeat CNT-enabled pacing systems for this target population. Orchestra BioMed will share in the revenues generated from Medtronic sales of the BackBeat CNT-enabled pacing systems.

About Virtue SAB and the Strategic Collaboration with Terumo

Virtue SAB is a patented drug/device combination product candidate in development for the treatment of certain forms of artery disease that is designed to deliver a proprietary, investigational, extended-release formulation of sirolimus, SirolimusEFRTM, to the vessel wall during balloon angioplasty without any coating on the balloon surface or the need to leave a stent or other permanent implant in the artery. Virtue SAB demonstrated positive three-year clinical data in coronary in-stent restenosis (ISR) in the SABRE trial, a multi-center prospective, independent core lab-adjudicated clinical trial of 50 patients conducted in Europe. Virtue SAB has been granted Breakthrough Device designation by the U.S. Food and Drug Administration for specific indications relating to coronary ISR, coronary small vessel disease and peripheral artery disease below-the-knee.

Under the terms of their collaboration agreement, Orchestra BioMed and Terumo plan to execute a global clinical program in an effort to gain regulatory approval for commercial sale of Virtue SAB in multiple markets and indications. Terumo made an upfront payment of \$30 million to Orchestra BioMed and Terumo will potentially make additional future clinical and regulatory milestone payments. Orchestra BioMed will share meaningfully in future commercial revenues of Virtue SAB through royalties and per unit payments as the exclusive supplier of SirolimusEFR. Orchestra BioMed retains the rights to develop and license SirolimusEFR and other technologies used in Virtue SAB for clinical applications outside of coronary and peripheral vascular interventions.

About HSAC2

Health Sciences Acquisitions Corporation 2 was a blank check company formed for the purpose of effecting a merger, share exchange, asset acquisition, share purchase, reorganization or similar business combination with one or more businesses. The sponsor of HSAC2 was HSAC 2 Holdings, LLC, an affiliate of RTW Investments, LP.

About RTW Investments, LP

RTW Investments, LP is a New York-based, global, full life-cycle investment firm that focuses on identifying transformational and disruptive innovations across the biopharmaceutical and medical technologies sectors. As a leading partner of industry and academia, RTW combines deep scientific expertise with a solution-oriented investment approach to advance emerging medical therapies by building and supporting the companies and/or academics developing them. For further information about RTW, please visit www.RTWfunds.com.

References

1. Company estimates based on published sources, including National Inpatient Survey (NIS) and National Health and Nutrition Examination Survey (NHANES)



Forward-Looking Statements

Certain statements included in this document that are not historical facts are forward-looking statements for purposes of the safe harbor provisions under the United States Private Securities Litigation Reform Act of 1995. Forward-looking statements generally are accompanied by words such as "believe," "may," "will," "estimate," "continue," "anticipate," "intend," "expect," "should," "would," "plan," "predict," "potential," "seem," "seek," "future," "outlook" and similar expressions that predict or indicate future events or trends or that are not statements of historical matters. These forward-looking statements include, but are not limited to, statements relating to our expected cash runway, the potential efficacy of our BackBeat CNT product candidate, the timing of our planned pivotal trials, and the ability of our partnerships to accelerate clinical development. These statements are based on various assumptions, whether or not identified in this document, and on the current expectations of the Company's management and are not predictions of actual performance. These forward-looking statements are provided for illustrative purposes only and are not intended to serve as and must not be relied on as a guarantee, an assurance, a prediction, or a definitive statement of fact or probability. Actual events and circumstances are difficult or impossible to predict and may differ from assumptions. Many actual events and circumstances are beyond the control of the Company. These forward-looking statements are subject to a number of risks and uncertainties, including changes in domestic and foreign business, market, financial, political, and legal conditions; failure to realize the anticipated benefits of the business combination; risks related to regulatory approval of the Company's product candidates; the impact of COVID-19; and the risk factors identified under the heading "Risk Factors—Risks Related to Orchestra's Business and New Orchestra Following the Business Combination" in the proxy statement/prospectus filed

The Company operates in a very competitive and rapidly changing environment. New risks emerge from time to time. Given these risks and uncertainties, the Company cautions against placing undue reliance on these forward-looking statements, which only speak as of the date of this press release. The Company does not plan and, subject to applicable law, undertakes no obligation to update any of the forward-looking statements made herein, except as required by law.

Orchestra BioMed Investor Contact:

Bob Yedid LifeSci Advisors (646) 597-6989 Bob@lifesciadvisors.com

Orchestra BioMed Media Contact:

Bill Berry Berry & Company Public Relations LLC (212) 253-888 bberry@berrypr.com

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED COMBINED FINANCIAL INFORMATION

Defined terms included below have the same meaning as terms defined and included elsewhere in the Current Report on Form 8-K (the "Form 8-K") filed by Orchestra BioMed Holdings, Inc. with the Securities and Exchange Commission (the "SEC") on January 31, 2023 and, if not defined in the Form 8-K, the Proxy Statement. Unless the context otherwise requires, "Orchestra" refers to Orchestra BioMed, Inc. prior to the Closing, and "HSAC2" refers to Health Sciences Acquisition Corporation 2 prior to the Closing.

The following unaudited pro forma condensed consolidated combined financial information has been prepared in accordance with Article 11 of Regulation S-X as amended by the final rule, Release No. 33-10786 "Amendments to Financial Disclosures about Acquired and Disposed Businesses."

The historical financial information of HSAC2 was derived from the unaudited condensed financial statements of HSAC2 as of September 30, 2022 and for the nine months ended September 30, 2022 and the audited financial statements of HSAC2 as of and for the year ended December 31, 2021. The historical financial information of Orchestra was derived from the unaudited condensed consolidated financial statements of Orchestra as of September 30, 2022 and the audited consolidated financial statements of Orchestra as of September 30, 2022 and for the nine months ended September 30, 2022 and the audited consolidated financial statements of Orchestra as of and for the year ended December 31, 2021. Such unaudited pro forma financial information has been prepared on a basis consistent with the audited financial statements of HSAC2 and Orchestra, respectively, and should be read in conjunction with the audited historical financial statements and related notes. This information should be read together with HSAC2's and Orchestra's audited financial statements and related notes, the sections titled "Management's Discussion and Analysis of Results of Operations" in HSAC2's Proxy Statement/Prospectus filed with the SEC on December 16, 2022.

The unaudited pro forma condensed consolidated combined balance sheet as of September 30, 2022 combines the historical balance sheet of HSAC2 and the historical consolidated balance sheet of Orchestra on a pro forma basis as if the Business Combination and the related transactions contemplated by the Merger Agreement, summarized below, had been consummated on September 30, 2022. The unaudited pro forma condensed consolidated combined statements of operations for the nine months ended September 30, 2022 and for the year ended December 31, 2021, combines the historical statement of operations of HSAC2 and historical consolidated statement of operations of Orchestra for such periods on a pro forma basis as if the Business Combination and the transactions contemplated by the Merger Agreement, summarized below, had been consummated on January 1, 2021, the beginning of the earliest period presented. There were no pro forma adjustments required to eliminate activities between the companies.

The Business Combination will be accounted for as a reverse recapitalization, with no goodwill or other intangible assets recorded, in accordance with U.S. GAAP. Under this method of accounting, HSAC2 is treated as the "acquired" company for financial reporting purposes. Accordingly, the Business Combination will be treated as the equivalent of Orchestra issuing stock for the net assets of HSAC2, accompanied by a recapitalization. Orchestra has been determined to be the accounting acquirer because Orchestra, as a group, has retained a majority of the outstanding shares of the combined company as of the closing of the Business Combination, they have nominated all seven persons who will comprise the New Orchestra Board upon Closing, Orchestra's management will continue to manage the combined company and Orchestra's business will comprise the ongoing operations of the combined company.

These unaudited pro forma condensed consolidated combined financial statements are for informational purposes only. They do not purport to indicate the results that would have been obtained had the Business Combination and related transactions actually been completed on the assumed date or for the periods presented, or which may be realized in the future. The pro forma adjustments are based on the information currently available and the assumptions and estimates underlying the pro forma adjustments are described in the accompanying notes. Actual results may differ materially from the assumptions within the accompanying unaudited pro forma condensed consolidated combined financial information.

Description of the Business Combination

On July 4, 2022, HSAC2 entered into the Merger Agreement, pursuant to which the Business Combination between HSAC2 and Orchestra was effected in two steps. First, before the Closing, HSAC2 effected the Domestication by deregistering in the Cayman Islands and domesticating as a Delaware corporation in accordance with Section 388 of the Delaware General Corporation Law and the Companies Act. Second, at the Closing, the Merger was effected by Merger Sub merging with and into Orchestra, with Orchestra surviving such merger as the surviving entity. Upon consummation of the Business Combination, Orchestra became a wholly owned subsidiary of HSAC2. HSAC2 then changed its name to "Orchestra BioMed Holdings, Inc."

Simultaneously with the execution of the Merger Agreement, HSAC2 and Orchestra entered into the Forward Purchase Agreement with the RTW Funds and Medtronic, pursuant to which each of the Purchasing Parties agreed to purchase \$10.0 million of HSAC2 Ordinary Shares, for a total of \$20.0 million, less the dollar amount of HSAC2 Ordinary Shares holding redemption rights that the Purchasing Party acquires and holds until immediately prior to the Domestication.

Simultaneously with the execution of the Merger Agreement and Forward Purchase Agreements, HSAC2, Orchestra, and the RTW Funds entered into the Backstop Agreement pursuant to which the RTW Funds, jointly and severally, agreed to purchase such number of HSAC2 Ordinary Shares at a price of \$10.00 per share to the extent that the amount of Parent Closing Cash as of immediately prior to the closing of the Merger is less than \$60.0 million (which calculation excludes amounts received pursuant to the Medtronic Forward Purchase Agreement or are otherwise held in the Trust Account in respect of Medtronic's Forward Purchase Shares, but is inclusive of amounts received pursuant to the RTW Forward Purchase Agreement and otherwise held in the Trust Account in respect of the RTW Funds' Forward Purchase Shares).

On October 21, 2022, the parties amended the RTW Forward Purchase Agreement and the Backstop Agreement, pursuant to which the RTW Funds agreed that (1) the per share price under each of the RTW Forward Purchase Agreement and the Backstop Agreement will not exceed the redemption price available to HSAC2 shareholders exercising redemption rights at the Shareholder Meeting, (2) any shares purchased pursuant to the RTW Forward Purchase Agreement or the Backstop Agreement, or otherwise acquired by the RTW Funds outside of the existing redemption offer, will not be voted in favor of approving the Business Combination, and (3) the RTW Funds will waive redemption rights with respect to such purchases in the vote to approve the Business Combination.

The RTW Funds completed their purchases of HSAC2 Ordinary Shares under their Forward Purchase Agreement on or before July 22, 2022. Medtronic completed approximately \$9.9 million of purchases of HSAC2 Ordinary Shares under its Forward Purchase Agreement on or before January 20, 2023. Medtronic subsequently completed \$0.1 million in purchases of HSAC2 Ordinary Shares and/or New Orchestra Common Stock on or before January 30, 2023. Pursuant to the Backstop Agreement, the RTW Funds purchased 1,808,512 HSAC2 Ordinary Shares on January 25, 2023, immediately prior to the Domestication. The Sponsor and the Purchasing Parties have registration rights pursuant to the Amended and Restated Registration Rights Agreement with respect to the New Orchestra Common Stock.

In addition, the Sponsor has agreed that 25% or 1,000,000 shares of its New Orchestra Common Stock received in the Domestication will be forfeited to New Orchestra on the first business day following the fifth anniversary of the Closing unless, (i) as to 500,000 shares, the VWAP of the New Orchestra Common Stock is greater than or equal to \$15.00 per share over any 20 Trading Days within any 30-Trading Day period, and (ii) as to the remaining 500,000 shares, the VWAP of the New Orchestra Common Stock is greater than or equal to \$15.00 per share over any 20 Trading Days within any 30-Trading Day period, and (ii) as to the remaining 500,000 shares, the VWAP of the New Orchestra Common Stock is greater than or equal to \$20.00 per share over any 20-Trading Days within any 30-Trading Day period. Further, the Sponsor and HSAC2's other initial shareholders prior to its initial public offering have agreed to subject the 4,000,000 shares of New Orchestra Common Stock to be received in the Domestication in exchange for the 4,000,000 HSAC2 Ordinary Shares issued to HSAC2's initial shareholders prior to its initial public offering and 450,000 shares of New Orchestra Common Stock to be received in a private placement simultaneously with the HSAC2 initial public offering, to a lock-up for up to 12 months following the Closing and, subject to the Closing. Pursuant to the terms of the Merger Agreement, immediately following such forfeiture and prior to the Closing, HSAC2 issued 750,000 New Warrants to eleven specified employees and directors of Orchestra. These New Warrants have substantially similar terms to the forfeited Private Warrants, except that they will become exercisable between 24 and 36 months after the Closing.

On January 24, 2023, HSAC2 held the Meeting, at which time the HSAC2 shareholders approved, among other matters, the Domestication and the Merger Agreement.

The consideration paid at Closing by HSAC2 to Orchestra security holders was paid in shares of HSAC2 Common Stock subject to the Exchange Ratio.

Orchestra stockholders had the opportunity to elect to participate in an earnout pursuant to which each such Earnout Participant may receive Earnout Consideration of up to 8,000,000 shares of New Orchestra Common Stock in the aggregate. Each Earnout Participant agreed to extend their applicable Lock-up Period from 6 months to 12 months, pursuant to an Earnout Election Agreement and will be entitled to receive the Earnout Consideration as follows:

- Earnout Participants will collectively be entitled to receive 4,000,000 Initial Earnout Shares, in the event that, during the Earnout Period, over any 20 Trading Days within any 30-Trading Day period during the Earnout Period the VWAP of the New Orchestra Common Stock is greater than or equal to \$15.00 per share; and
- Earnout Participants will collectively be entitled to receive an additional 4,000,000 Final Earnout Shares, in the event that, during the Earnout Period, over any 20-Trading Days within any 30-Trading Day period during the Earnout Period the VWAP of the New Orchestra Common Stock is greater than or equal to \$20.00 per share.
- Upon the first change in control meeting certain conditions that occurs during the Earnout Period, if the corresponding valuation of New Orchestra is equal to or greater than \$15.00 per share (taking into consideration the Initial Earnout Shares in determining such calculation), the Initial Milestone Event will be deemed to have occurred and if equal to or greater than \$20.00 per share (taking into consideration the issuance of all Earnout Consideration in determining such calculation), the Final Milestone Event will be deemed to have occurred, in each case immediately prior to such change in control.

Orchestra accounts for the Earnout Shares as either equity-classified or liability-classified instruments based on an assessment of the Earnout Shares specific terms and applicable authoritative guidance in ASC 480, *Distinguishing Liabilities from Equity* ("ASC 480") and ASC 815, *Derivatives and Hedging* ("ASC 815"). Orchestra has preliminarily determined that the Earnout Shares are indexed to New Orchestra's stock and are therefore will be classified within stockholders' equity. The unaudited pro forma condensed combined financial information does not reflect pro forma adjustments related to the recognition of the Earnout Shares as the issuance of the shares would be represented by both an increase and offsetting decrease to additional paid-in capital. The unaudited pro forma condensed combined financial information does not reflect pro forma adjustments on a per share basis for the Earnout Shares because the earnout contingencies have not yet been met and because the earnout shares would be anti-dilutive.

The issuance of such Earnout Shares would dilute the value of all shares of New Orchestra Common Stock outstanding at the time of issuance. Assuming the current capitalization structure, the 4,000,000 Initial Earnout Shares that would become vested upon meeting the \$15.00 earnout threshold, would represent approximately 13% of total shares outstanding for the redemption scenarios set forth. Assuming the current capitalization structure, the total 8,000,000 shares representing the Initial Earnout Shares and Final Earnout Shares that would become vested upon meeting the \$20.00 earnout threshold, would represent approximately 26% of total shares outstanding for the redemption scenarios set forth.

Each share of Orchestra capital stock, if any, that is owned by HSAC2, Merger Sub, or Orchestra, or any of their subsidiaries (as treasury stock or otherwise) was canceled and extinguished without any conversion or consideration.

At the Effective Time, each issued and outstanding share of Orchestra Common Stock (other than any such shares of Orchestra Common Stock canceled as described above and any dissenting shares) was converted into the right to receive (1) a number of shares of HSAC2 Common Stock at the Exchange Ratio, and (2) shares of Earnout Consideration as, and subject to the contingencies, described above, including entry into an Earnout Election Agreement.

Each share of common stock, par value \$0.01 per share, of Merger Sub common stock issued and outstanding immediately prior to the Effective Time was converted into one newly issued share of Orchestra as the surviving corporation in the Merger.

At the Effective Time, each outstanding option to purchase shares of Orchestra Common Stock was converted into an option to purchase, subject to substantially the same terms and conditions as were applicable under such options prior to the Effective Time, shares of New Orchestra Common Stock equal to the number of shares subject to such option prior to the Effective Time multiplied by the Exchange Ratio, at an exercise price per share of New Orchestra Common Stock subject to such option divided by the Exchange Ratio.

Effective immediately prior to the Effective Time, each outstanding warrant to purchase shares of Orchestra capital stock was treated in accordance with the terms of the relevant agreements governing such warrants and converted into New Orchestra warrants. The outstanding warrants to purchase shares of Orchestra capital stock are expected to be equity classified upon the consummation of the Business Combination.

Extension Proposal

On July 1, 2022, HSAC2 filed a proxy statement seeking approval of the Extension Proposal from its shareholders to amend the Company's Existing Charter to: (a) extend from August 6, 2022 to November 6, 2022, the date by which, if the Company has not consummated a merger, amalgamation, share exchange, asset acquisition, share purchase, reorganization or similar business combination involving one or more businesses or entities, the Company must: (i) cease all operations except for the purpose of winding up; (ii) as promptly as reasonably possible but not more than ten business days thereafter, redeem the Public Shares; and (iii) as promptly as reasonably possible following such redemption liquidate and dissolve, subject in each case to its obligations under Cayman Islands law to provide for claims of creditors and in all cases subject to the other requirements of applicable law, and (b) allow the Company, without another shareholder vote, to elect to extend the date to consummate a business combination on a monthly basis for up to three times by an additional one month each time after November 6, 2022, upon five days' advance notice prior to the applicable deadlines, until February 6, 2023 or a total of up to six months after August 6, 2022, unless the closing of the Company's initial business combination shall have occurred.

On November 15, 2022, the Directors of the Company elected to extend the deadline until January 6, 2023 and on December 15, 2022, the Directors of the Company elected to further extend the deadline until February 6, 2023.

The submission of the Extension Proposal to amend HSAC2's Existing Charter entitled holders of Public Shares to redeem their shares for their pro rata portion of the funds held in the trust account established at the time of the HSAC2 initial public offering. In connection with the Extension Meeting, as of July 22, 2022, HSAC2 received requests for redemption from shareholders with respect to 9,237,883 HSAC2 Ordinary Shares.

HSAC2 held the Meeting on January 24, 2023 for the purposes of considering and voting upon, among other things, the business combination HSAC2 with Orchestra. The submission of the Business Combination to the shareholders entitles holders of Public Shares in the Company to redeem their shares for their pro rata portion of the funds held in the trust account established at the time of the HSAC2 initial public offering. In connection with the General Meeting, as of January 24, 2023, HSAC2 received requests for redemption from shareholders with respect to 1,597,888 HSAC2 Ordinary Shares.

The following summarizes the pro forma ownership of New Orchestra Common Stock following the Business Combination:

		Percentage of
	Number of Shares	Outstanding Shares
Orchestra et al halderr(])		
Orchestra stockholders ⁽¹⁾	13,881,338	45.3%
Medtronic ⁽²⁾	4,992,588	16.3%
HSAC2 Public Shareholders ⁽³⁾	3,141,641	10.3%
Sponsor and related parties ⁽⁴⁾⁽⁵⁾	8,598,512	28.1%
Pro forma Common Stock at September 30, 2022	30,614,079	100.0%

(1) Excludes (i) 8,000,000 Earnout Ordinary Shares as the earnout contingencies have not yet been met, (ii) shares issuable in connection with outstanding Orchestra options and warrants, (iii) shares available for issuance pursuant to the 2022 Equity Incentive Plan, (iv) 2,310,000 shares held by certain funds managed by RTW Investments, LP. and (v) shares held by Medtronic.

(2) Includes 4,000,000 shares of Orchestra and 992,588 HSAC2 Ordinary Shares.

(3) Reflects the redemption of 9,237,883 HSAC2 Ordinary Shares in connection with the Extension Amendment and 1,597,888 ordinary shares in connection with the Meeting. Excludes 1,992,588 Public Shares acquired by the RTW Funds and Medtronic in connection with the Forward Purchase Agreements and 30,000 Ordinary Shares held by officers of HSAC2.

(4) Excludes 1,000,000 HSAC2 Ordinary Shares subject to forfeiture upon the expiration of the Earnout Period, includes 1,000,000 HSAC2 Ordinary Shares purchased pursuant to the Forward Purchase Agreement with RTW Funds.

(5) Excludes 750,000 Private Warrants. If all potential sources of dilution were exercised and converted into HSAC2 Ordinary Shares the Sponsor and related parties would hold approximately 20.7%

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED COMBINED BALANCE SHEET AS OF SEPTEMBER 30, 2022 (in thousands, except share and per share data)

	HSAC2 (Historical)		Orchestra (Historical)		Transaction Accounting Adjustments		Pro Forma Combined		
ASSETS									
Current assets			*		*		_		
Cash and cash equivalents	\$	735	\$	96,995	\$	51,761	B	\$	152,486
						18,085	C		
						(2,372)			
						(7,118)			
						(5,600)	F		
Strategic investments				251					251
Accounts receivable, net				96					96
Inventory				330					330
Prepaid expenses and other current assets		137		828		—			965
Total current assets		872		98,500		54,756			154,128
Property and equipment, net				1,505					1,505
Right-of-use assets				2,339					2,339
Strategic investments, less current portion				2,495					2,495
Deposits and other assets				3,508		(2,937)	E		571
Investments held in Trust Account		67,777		5,500		(51,761)			
nivestinents nete in Trust Recount		07,777				(16,016)			
TT (1)							A		
Total assets	\$	68,649	\$	108,347	\$	(15,958)		\$	161,038
LIABILITIES, REDEEMABLE PREFERRED STOCK AND									
STOCKHOLDERS' EQUITY (DEFICIT)									
Current liabilities									
Accounts payable	\$	282	\$	7,686	\$	(137)	D	\$	5,571
recounts puyuote	Ψ	202	Ψ	7,000	Ψ	(2,260)		Ψ	5,571
Accrued expenses and other liabilities		1,159		4,746		(1,147)			4,758
		1,139		4,740		(1,147)	D		4,738
Operating lease liability - current		_				(1.972)	C		
Warrant liability		—		1,873		(1,873)	G		
Deferred revenue, current portion				5,143					5,143
Total current liabilities		1,441		20,132		(5,417)			16,156
Deferred revenue, less current portion				15,327		—			15,327
Loan payable, less current portion				9,453					9,453
Operating lease liability, less current portion				1,864					1,864
Other long-term liabilities				142					142
Deferred underwriting commissions		5,600				(5,600)	F		_
Total liabilities		7,041		46,918		(11,017)			42,942
Total habilities		7,041	_	40,918	-	(11,017)		-	42,942
						(1(01()			
Ordinary shares subject to possible redemption		67,676				(16,016)			
						(51,660)	Н		
Series A preferred stock		—		51,452		(51,452)			_
Series D-1 preferred stock				27,272		(27,272)			—
Series D-2 preferred stock				87,199		(87,199)	J		
Stockholders' equity (deficit)									
Preferred Stock									
Ordinary stock						1	Н		
						(1)	Ι		
New Orchestra Common Stock						1	Ι		3
						2	J		
Additional paid-in capital				85,490		18,085	Č		309,165
				00,190		(7,795)	E		505,100
						1,873	G		
						51,659	H		
						165,921			
							J V		
		(()(0)		(100.00.4)		(6,068)			(101.072)
Accumulated deficit		(6,068)		(189,984)		(1,088)	D		(191,072)
						6,068	K		
Total stockholders' equity (deficit)		(6,068)	_	(104,494)	_	228,658		_	118,096
Total liabilities, preferred stock and stockholders' equity (deficit)	\$	68,649	\$	108,347	\$	(15,958)		\$	161,038
	Ψ	00,017	φ	100,517	Ŷ	(15,750)		Ψ	101,050

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED COMBINED STATEMENT OF OPERATIONS FOR THE NINE MONTHS ENDED SEPTEMBER 30, 2022 (in thousands, except share and per share data)

Damana	HSAC2 (Historical)	Orchestra (Historical)	Transaction Accounting Adjustments	_	Pro Forma Combined
Revenue Darte and in management	¢	¢ 1.021	¢	¢	1.021
Partnership revenue Product revenue	\$ —	\$ 1,931	\$ —	\$,
		499			499
Total revenue		2,430		_	2,430
Expenses					
Cost of product revenues		158			158
Research and development		14,402			14,402
Selling, general and administrative	2,113	10,699	807	D 1	13,619
Administrative fee - related party	90	10,077	007	D.1	90
Total expenses	2,203	25,259	807	_	28,269
Operating loss	(2,203)	(22,829)	(807)	_	(25,839)
Other income (expense)					
Interest income from investments held in Trust Account	345	_	(345)	A.1	_
Interest income (expense), net	_	(419)	(2.12)		(419)
Loss on fair value adjustment of warrant liability	_	(1,124)	1,124	C.1	(,)
Loss on debt extinguishment		(682)	,		(682)
Gain on fair value of strategic investments	_	1,196	_		1,196
Total other income (expense)	345	(1,029)	779		95
Net loss	\$ (1,858)	\$ (23,858)	\$ (28)	\$	(25,744)
				-	
Net loss per share (Note 4)					
Weighted average shares outstanding	18,182,827	2,412,363			30,614,079
Basic and diluted net loss per share	\$ (0.10)	\$ (10.72)		\$	(0.84)
				=	

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED COMBINED STATEMENT OF OPERATIONS FOR THE YEAR ENDED DECEMBER 31, 2021 (in thousands, except share and per share data)

Damana	HSAC2 (Historical)	Orchestra (Historical)	Transaction Accounting Adjustments	Pro Forma Combined
Revenue Partnership revenue	\$	\$ (1,475)	¢	\$ (1,475)
Product revenue	s —	s (1,473) 693	۰ —	⁵ (1,473) 693
Total revenue				
Total levenue		(782)		(782)
Expenses				
Cost of product revenues	_	199	_	199
Research and development	—	12,890	—	12,890
Selling, general and administrative	275	7,928	1,088 B .	1 10,367
			1,076 D .	1
Administrative fee - related party	120	—	—	120
Total expenses	395	21,017	2,164	23,576
Loss from operations	(395)	(21,799)	(2,164)	(24,358)
Other income (expense)				
Interest income from investments held in Trust Account	16	_	(16) A.	1 —
Interest income (expense), net	—	(927)	—	(927)
Gain on fair value adjustment of warrant liability	—	699	(699) C.	1 —
Loss on fair value of strategic investments	—	(987)	—	(987)
Total other income (expense)	16	(1,215)	(715)	(1,914)
Net loss	\$ (379)	\$ (23,014)	\$ (2,879)	\$ (26,272)
Net loss per share (Note 4)				
Weighted average shares outstanding	20,450,000	2,111,161		30,614,079
Basic and diluted net loss per share	\$ (0.02)	\$ (10.90)		\$ (0.86)

NOTES TO UNAUDITED PRO FORMA CONDENSED CONSOLIDATED COMBINED FINANCIAL INFORMATION

Note 1. Basis of Presentation

The Business Combination has been accounted for as a reverse recapitalization, with no goodwill or other intangible assets recorded, in accordance with U.S. GAAP. Under this method of accounting, HSAC2 has been treated as the "accounting acquiree" and Orchestra as the "accounting acquirer" for financial reporting purposes. Accordingly, for accounting purposes, the Business Combination has been treated as the equivalent of Orchestra issuing shares for the net assets of HSAC2, followed by a recapitalization. The net assets of HSAC2 have been stated at historical cost. Operations prior to the Business Combination are those of Orchestra.

The unaudited pro forma condensed consolidated combined balance sheet as of September 30, 2022 gives effect to the Business Combination and related transactions as if they had been completed on September 30, 2022. The unaudited pro forma condensed consolidated combined statement of operations for the nine months ended September 30, 2022 and year ended December 31, 2021 gives effect to the Business Combination and related transactions as if they had been completed on January 1, 2021. These periods are presented on the basis that Orchestra is the acquirer for accounting purposes.

The pro forma adjustments reflecting the consummation of the Business Combination and the related transaction are based on certain currently available information and certain assumptions and methodologies that HSAC2 management believes are reasonable under the circumstances. The unaudited condensed consolidated combined pro forma adjustments, which are described in the accompanying notes, may be revised as additional information becomes available and is evaluated. Therefore, it is likely that the actual adjustments will differ from the pro forma adjustments, and it is possible that the difference may be material. HSAC2 management believes that its assumptions and methodologies provide a reasonable basis for presenting all of the significant effects of the Business Combination and the related transactions based on information available to management at this time and that the pro forma adjustments give appropriate effect to those assumptions and are properly applied in the unaudited pro forma condensed consolidated combined financial information.

The unaudited pro forma condensed consolidated combined financial information does not give effect to any anticipated synergies, operating efficiencies, tax savings, or cost savings that may be associated with the Business Combination. The unaudited pro forma condensed consolidated combined financial information is not necessarily indicative of the future consolidated results of operations or financial position of the post-combination company. They should be read in conjunction with the historical consolidated financial statements and notes thereto of HSAC2 and Orchestra.

Note 2. Accounting Policies and Reclassifications

Upon consummation of the Business Combination, management performed a comprehensive review of the two entities' accounting policies. As a result of the review, management did not identify any differences between the accounting policies of the two entities that would have a material impact on the unaudited pro forma condensed consolidated combined financial information. As a result, the unaudited pro forma condensed consolidated combined financial information.

As part of the preparation of these unaudited pro forma condensed consolidated combined financial statements, certain reclassifications were made to align HSAC2's financial statement presentation with that of Orchestra.

Preferred Stock Conversion

Immediately prior to the consummation of the Business Combination, each share of Orchestra's pre-merger preferred stock converted into Orchestra common stock. Upon the closing of the Business Combination (after giving effect to the conversion of Orchestra preferred stock into Orchestra common stock), all shares of Orchestra common stock outstanding have been converted into shares of New Orchestra common stock.



Accounting for Stock Option Conversion

The Company accounts for stock-based compensation arrangements with employees and non-employee consultants using a fair value method which requires the recognition of compensation expense for costs related to all stock-based payments, including stock options. As of the Effective Time, each Orchestra option that was outstanding prior to the business combination has been converted into an option to purchase shares of New Orchestra common stock upon substantially the same terms and conditions as were in effect with respect to such option immediately prior to the Effective Time, subject to specific terms and conditions. As there was no change in the terms of the options or material change in fair value, Orchestra did not recognize any incremental stock compensation expense.

Note 3. Adjustments to Unaudited Pro Forma Condensed Consolidated Combined Financial Information

The unaudited pro forma condensed consolidated combined financial information has been prepared to illustrate the effect of the Business Combination and related transactions and has been prepared for informational purposes only.

The following unaudited pro forma condensed consolidated combined financial information has been prepared in accordance with Article 11 of Regulation S-X as amended by the final rule, Release No. 33-10786 "Amendments to Financial Disclosures about Acquired and Disposed Businesses." Release No. 33-10786 replaces the existing pro forma adjustment criteria with simplified requirements to depict the accounting for the transaction ("Transaction Accounting Adjustments") and present the reasonably estimable synergies and other transaction effects that have occurred or are reasonably expected to occur ("Management's Adjustments"). The pro forma adjustments reflecting the consummation of the Business Combination and related transactions are based on certain currently available information and certain estimates, assumptions and methodologies that management believes are reasonable under the circumstances. The unaudited condensed consolidated combined pro forma adjustments, which are described in the accompanying notes, may be revised as additional information becomes available and is evaluated. New Orchestra has elected not to present Management's Adjustments and will only be presenting Transaction Accounting Adjustments in the unaudited pro forma condensed consolidated combined financial information. There were no pro forma adjustments required to eliminate activities between the companies.

The pro forma condensed combined financial information does not include an income tax adjustment. Upon closing of the Business Combination, it is likely that the combined company will record a valuation allowance against the total U.S. and state deferred tax assets as the recoverability of the tax assets is uncertain. The pro forma combined provision for income taxes does not necessarily reflect the amounts that would have resulted had the combined company filed consolidated income tax returns during the periods presented.

The pro forma basic and diluted earnings per share amounts presented in the unaudited pro forma condensed consolidated combined statements of operations are based upon the number of shares of New Orchestra Common Stock outstanding, assuming the Business Combination and related transactions occurred on the beginning of the earliest period presented. The pro forma basic and diluted earnings per share amounts exclude the impact of the Earnout Shares as the earnout contingencies have not yet been met and because the earnout shares would be anti-dilutive.

Adjustments to Unaudited Pro Forma Condensed Consolidated Combined Balance Sheet:

The adjustments included in the unaudited pro forma condensed consolidated combined balance sheet as of September 30, 2022 are as follows:

- A. Reflects the redemption payment totaling approximately \$16.0 million as a result of the redemption of 1,597,888 HSAC2 Ordinary Shares in connection with the Closing, inclusive of \$0.4 million of available interest at January 26, 2023.
- B. Reflects the reclassification of marketable securities of \$51.8 million held in the trust account to cash and cash equivalents.
- C. Reflects the proceeds of 1,808,512 HSAC2 Ordinary Shares purchased by certain funds managed by RTW Investments, LP pursuant to the Backstop Agreement.
- D. Represents HSAC2 transaction costs of \$2.4 million inclusive of advisory, banking, printing, legal, accounting fees and other professional fees that are expensed as a part of the Business Combination within accumulated deficit. Of the transaction costs, \$1.3 million has already been incurred and reflected in the historical financial statements of HSAC2, of which \$0.1 million had already been paid.
- E. Represents Orchestra transaction costs of \$7.8 million inclusive of advisory, banking, legal and other professional fees that were incurred as a part of the Business Combination within additional paid-in capital. Of the transaction costs, \$2.9 million had already been incurred pre-merger and reflected in the historical financial statements of Orchestra, of which \$0.7 million has already been paid.
- F. Reflects the settlement of \$5.6 million in deferred underwriting fee payable.
- G. Reflects the reclassification of \$1.9 million of Orchestra warrant liabilities to equity. The Orchestra warrants are expected to meet the fixed-forfixed indexation criteria to be equity classified following the consummation of the Business Combination.
- H. Reflects the reclassification of \$51.7 million of HSAC2 Ordinary Shares subject to possible redemption into permanent equity upon closing of the business combination.
- Reflects the conversion of 9,614,229 HSAC2 Ordinary Shares into 9,614,229 shares of New Orchestra Common Stock in the Domestication after reflecting the redemption of 10,835,771 HSAC2 Ordinary Shares in connection with the Extension Proposal and the Closing.
- J. Reflects the recapitalization of Orchestra's outstanding equity and temporary equity comprised of 35,694,179 shares of preferred stock and 2,518,359 shares of common stock, par value of \$0.0001 (aggregate value of \$165.9 million) reflected as an increase in additional paid-in capital.
- K. Reflects the reclassification of HSAC2's historical accumulated deficit.

Adjustments to Unaudited Pro Forma Condensed Consolidated Combined Statements of Operations

The pro forma adjustments included in the unaudited pro forma condensed consolidated combined statements of operations for the nine months ended September 30, 2022 and for the year ended December 31, 2021 are as follows:

- A.1 Reflects elimination of investment income on the trust account.
- B.1 Reflects transactions costs of \$1.1 million as if incurred on January 1, 2021, the date the Business Combination occurred for the purposes of the unaudited pro forma condensed consolidated combined statement of operations. The amount presented is comprised of transaction costs outlined in adjustment (C) that were not yet recognized and expensed in the historical statement of operations as part of the Business Combination.
- C.1 Reflects the reclassification of the Orchestra warrant liabilities to equity as of January 1, 2021 and the elimination of changes in the fair value of the warrant liabilities recorded in the statement of operations. The Orchestra warrants are expected to meet the fixed-for-fixed indexation criteria to be equity classified following the consummation of the Business Combination.
- D.1 Reflects additional stock compensation expense related to the grant of new warrants to purchase an aggregate of 750,000 shares of New Orchestra Common Stock to specified employees and directors of Orchestra.

Note 4. Net Loss per Share

Net loss per share was calculated using the historical weighted average shares outstanding, and the issuance of additional shares in connection with the Business Combination and the related transactions, assuming the shares were outstanding since January 1, 2021. As the Business Combination and the related transactions are being reflected as if they had occurred at the beginning of the period presented, the calculation of weighted average shares outstanding for basic and diluted net loss per share assumes that the shares issuable relating to the Business Combination and related have been outstanding for the entirety of all periods presented.

	Se	Vine Months Ended eptember 30, 2022 ⁽¹⁾ n thousands, ex		Year Ended ecember 31, 2021 ⁽¹⁾	
	(re data)	
Pro forma net loss	\$	(25,744)		(26,272)	
Weighted average shares outstanding - basic and diluted		30,614,079		30,614,079	
Net loss per share - basic and diluted	\$	(0.84)	\$	(0.86)	
Excluded securities: ⁽²⁾					
Private Warrants		750,000		750,000	
Orchestra Warrants		2,325,936		2,325,936	
Orchestra Options		3,929,298		3,929,298	
Sponsor and related parties shares subject to forfeiture on expiration of Earnout		1,000,000		1,000,000	
Earnout Shares		8,000,000		8,000,000	

⁽¹⁾ Pro forma net loss per share includes the related pro forma adjustments as referred to within the section "Adjustments to Unaudited Pro Forma Condensed Consolidated Combined Financial Information."

(2) The potentially dilutive outstanding securities were excluded from the computation of pro forma net loss per share, basic and diluted, because their effect would have been anti-dilutive and/or issuance or vesting of such shares is contingent upon the satisfaction of certain conditions which were not satisfied by the end of the periods presented.